Southampton

University of Southampton Research Repository

Copyright © and Moral Rights for this thesis and, where applicable, any accompanying data are retained by the author and/or other copyright owners. A copy can be downloaded for personal non-commercial research or study, without prior permission or charge. This thesis and the accompanying data cannot be reproduced or quoted extensively from without first obtaining permission in writing from the copyright holder/s. The content of the thesis and accompanying research data (where applicable) must not be changed in any way or sold commercially in any format or medium without the formal permission of the copyright holder/s.

When referring to this thesis and any accompanying data, full bibliographic details must be given, e.g.

Thesis: Author (Year of Submission) "Full thesis title", University of Southampton, name of the University Faculty or School or Department, PhD Thesis, pagination.

Data: Author (Year) Title. URI [dataset]

UNIVERSITY OF SOUTHAMPTON

Faculty of Social Sciences School of Mathematical Sciences

Dissipation and turbulence in general relativistic hydrodynamics

by

Thomas Celora ORCiD: 0000-0002-6515-3644

Thesis for the degree of Doctor of Philosophy

June 2023

University of Southampton

Abstract

Faculty of Social Sciences School of Mathematical Sciences

Doctor of Philosophy

Dissipation and turbulence in general relativistic hydrodynamics

by Thomas Celora

Hydrodynamics is one of the oldest research areas in physics, with applications across all macroscopic scales in the Universe. Despite the long history of successes, however, fluid modelling still presents severe conceptual and computational challenges. Not surprisingly, the hurdles become even more formidable for relativistic flows, and new issues come to the fore too. This work is concerned with advancing multi-fluid models in General Relativity, and in particular focuses on the modelling of dissipative fluids and turbulent flows. Such models are required for an accurate description of neutron star phenomenology, and binary neutron star mergers in particular. In fact, the advent of multi-messenger astronomy—started with the first detection of a binary neutron star coalescence in 2017—offers exciting prospects for exploring the extreme physics at play during such cosmic fireworks.

In this work we first focus on modelling dissipative fluids in relativity, and explore the arguably unique model that is ideally suited for describing dissipative multi-fluids in General Relativity. Modelling single fluids in relativity is already a hard task, but for neutron stars it is easy to argue that we need to understand even more complicated settings: the presence of superfluid/superconducting mixtures, for example, means that we need to go beyond single-fluid descriptions. We then consider turbulent flows and focus on how to perform "filtering" in a curved spacetime setting. We do so as most recent turbulent models in a Newtonian setting are based on the notion of spatial filtering. As the same strategy is beginning to be applied in numerical relativity, we focus on the foundational underpinnings and propose a novel scheme for carrying out filtering, ensuring consistency with the tenets of General Relativity. Finally, we discuss two applications of relevance for binary neutron star mergers. We focus on the modelling of $(\beta$ -)reactions in neutron star simulations, and provide a discussion of the magnetorotational instability that is suited to highly dynamical environments like mergers. We focus on these two problems as reactions are expected to source the dominant dissipative contribution to the overall dynamics, while the magneto-rotational instability is considered crucial for sustaining the development of turbulence in mergers.

Contents

Li	st of]	Figures		ix		
De	eclara	tion of	Authorship	xi		
Ac	cknov	vledge	ments	xiii		
No	Notation x					
1	Intr	oductio	on	1		
	1.1	Motiv	ation	1		
	1.2	Outlir	e	3		
	1.3	Relati	vistic perfect fluids	6		
Ι	Dis	sipativ	e (multi-)fluids in General Relativity	11		
2	Dise	sipative	e fluid models in General Relativity: an overview	12		
	2.1	Non-e	quilibrium thermodynamics	12		
		2.1.1	Linear irreversible thermodynamics	13		
		2.1.2	Causality and extended irreversible thermodynamics	14		
	2.2	Tradit	ional strategies for dissipative fluids	16		
		2.2.1	Eckart-Landau-Lifshitz models	17		
		2.2.2	Müller-Israel-Stewart models	19		
		2.2.3	Liu-Müller-Ruggeri and divergence-type theories	21		
	2.3 Variational models		ional models	23		
		2.3.1	Non dissipative multi-fluids models	23		
		2.3.2	Carter-like dissipative models	27		
		2.3.3	Andersson and Comer formalism	29		
	2.4	Hydro	odynamics as an effective field theory	34		
3	Linearizing an action-based formalism for dissipative (multi-)fluids 38					
	3.1	Main	assumptions of the model	38		
		3.1.1	Flux definition	39		
		3.1.2	Matter space volume forms	40		
	3.2	The no	on-dissipative limit	43		
		3.2.1	Laboratory vs. general relativistic set-up	43		
		3.2.2	Multiple equilibrium states	45		
		3.2.3	Global analysis of the non-dissipative limit	46		

		3.2.4 Local analysis of the non-dissipative limit	0	
		3.2.5 A final comment on equilibrium	4	
	3.3	Perturbations with respect to equilibrium 5	5	
	3.4	Energy density is stationary at equilibrium	0	
	3.5	The last piece of the puzzle	7	
	3.6	Model comparison	9	
		3.6.1 Equating the flux currents	0	
		3.6.2 Example: A viscous single fluid 7	'1	
	3.7	Cattaneo-type equations	6	
	3.8	Summary and outlook	9	
II	A	ovariant approach to large-eddy filtering 8	1	
4	Filte	ring relativistic hydrodynamics 8	2	
	4.1	A brief introduction to hydrodynamic turbulence	3	
	4.2	Averaging turbulent flows8	5	
	4.3	Averaging vs filtering	7	
	4.4	The spacetime view: Fermi Coordinates	9	
		4.4.1 On covariance and the Einstein equations	4	
	4.5	Averaging in the fluid frame	5	
		4.5.1 Baryon number conservation	7	
		4.5.2 Averaged matter dynamics	8	
		4.5.3 The equation of state	9	
	4.6	Fluid element filtering 10	1	
	4.7	Filtered Thermodynamics	2	
		4.7.1 The effective entropy	3	
		4.7.2 Energy cascade argument	5	
	4.8	An explicit closure model	6	
		4.8.1 Stability Analysis	7	
		48.2 Smagorinsky model 11	0	
		48.3 Fixing the Smagorinsky instability	3	
	4.9	Summary	7	
5	Filte	ring relativistic magneto-hydrodynamics 11	8	
-	5.1	A brief introduction to magneto-hydrodynamic turbulence	9	
	5.2	Magneto-hydrodynamics in the fibration: a shortcut	2	
	53	MHD covariant filtering: first steps	7	
	54	Outlook 12	9	
	0.1	Outlook	,	
II	l Bi	nary neutron-star merger applications 13	1	
6	Formulating bulk-viscosity for neutron star simulations			
	6.1	Simplifications must be made	2	
	6.2	The reactive system	4	
		 6.2.1 Thermodynamics of a reactive system	7	
			^	

	6.3	Appro	oximating the reactive system	140
		6.3.1	Multi-scale arguments and the reactive system	141
		6.3.2	Invariant manifold method with the electron fraction	143
		6.3.3	Partially resolved reactions and double counting	144
	6.4	Makir	ng contact with simulations	145
		6.4.1	What bulk viscous pressure approximation is suitable?	147
		6.4.2	How relevant is bulk viscosity in mergers?	149
		6.4.3	The impact of large-eddy filtering	150
	6.5	Sumn	nary and Outlook	154
7	Mag	gneto-r	otational instability in mergers: a local Newtonian analysis	157
	7.1	Backg	round gradients and plane-wave expansion	158
	7.2	The sl	owly evolving background	160
		7.2.1	Velocity gradient decomposition	161
	7.3	Non-i	nertial equations and the local frame	163
	7.4	Going	back to hydrodynamics	164
	7.5	Magn	eto-shear instability in the local frame	170
		7.5.1	Homogeneous background: a recap	171
		7.5.2	Sheared Background	172
		7.5.3	Background with vorticity	175
	7.6	Concl	uding remarks: The MRI in perspective	176
		7.6.1	The MRI vs the Rayleigh criterion	176
		7.6.2	The missing ingredient: Filtering	177
IV	Co	onclus	ions	179
Ar	openc	dix A	Transporting a tetrad and Fermi Coordinates	183
r	App	endix	A.1 Transporting a tetrad along a curve	183
	App	endix	A.2 Fermi coordinates	185
	r r			
Aŗ	openc		Multi-scale arguments and the invariant manifold method	189
	App	endix I	B.1 Invariant manifold approach	189
	App	endix I	B.2 Two timescale approach	190
	App	endix I	B.3 Linear fast dynamics	190
	Арр	endix	B.4 Constructing the fast terms	191
Aŗ	openc	dix C	Working with the CompOSE database	192
Aŗ	openc	dix D	Formulating the MRI in the local frame	196
	App	endix 1	D.1 Another look at the non-inertial MHD equations	198
	App	endix]	D.2 A closer look at the Rayleigh stability criterion	198
Aŗ	openc	dix E	The Routh-Hurwitz criterion	202
Re	ferer	nces		204

List of Figures

2.1	The pull-back from a point in the x^{th} matter space to the corresponding spacetime worldline. The points in matter space are labelled by X_x^A with $A = 1, 2, 3$. Figure taken from Andersson and Comer [21].	24
3.1	A depiction of the spacetime region \mathcal{M} , with one spatial axis suppressed. It has a characteristic spatial size ΔL and temporal size ΔT . Inside \mathcal{M} is a smaller region $\delta \mathcal{M}$ of characteristic spatial and temporal size δl and δt , respectively. The boundary $\partial \mathcal{M}$ consists of the initial and final time- slices $\partial \mathcal{M}_{-}$, $\partial \mathcal{M}_{+}$ and the timelike hypersurface $\partial \mathcal{M}_{L}$	47
3.2	vertical red lines, parameterized by τ , $\bar{\tau}$) and "Lagrangian displacements" which connect fluid elements (dashed horizontal blue lines, parameter- ized by λ).	56
4.1	Cartoon of the scales of the turbulent energy spectrum. Figure adapted from McDonough [151]	84
4.2	Left: shaded region resulting from combining the stability constraints obtained for the i) transverse modes (gapped, un-gapped, boosted, un-boosted), ii) longitudinal un-boosted modes (gapped and un-gapped), and iii) longitudinal boosted gapped modes. Right: sound speeds (solid) and damping rates (dashed) for the un-gapped longitudinal modes in the boosted frame. Colours match sound speeds with the corresponding damping rates. For illustrative purposes we used: $c_s = 0.16$, $v = 0.4$, $\eta = 1.2$, $\theta_1 = 0.5$ and a barotropic equation of state (both figures).	116
5.1	Cartoon of the perpendicular MHD turbulence spectrum. At scales larger than critical balance (i.e. $k < k_{CB}$) is shown the same scaling as in weak turbulence (i.e. $\propto k_{\perp}^{-2}$). At scales smaller than critical balance is shown the aligned cascade (i.e. $\propto k_{\perp}^{-3/2}$) periodically interrupted at k_1, k_2 and so on, while k_D represents the scale at which dissipative/resistive effects begin to prevail over inertial ones. Figure adapted from Schekochihin [192].	121
6.1	Illustrating the behaviour of $\beta(\omega)$ in two cases: the solution to the Catta- neo equation (i.e. the full extended irreversible thermodynamics (EIT) relaxation towards the Navier-Stokes limit, blue solid curve) and the parabolic case (the Navier-Stokes limit, orange dashed line). The shaded region indicates frequencies we assume we may not have "access" to numerically (this region moves towards higher frequencies when the nu- merical resolution is increased).	146

6.2	The maximum (for each point we assume that $\omega = A$) potential relative contribution of the bulk viscous approximation $\chi^{\text{max}}/p^{\text{eq}}$ at each point in phase space using the APR [210, 195] equation of state. We see that the bulk viscous pressure contribution can be large for most temperatures when $n \gtrsim 10^{-3}n_{\text{sat}}$. However, the bulk viscous approximation should only be used where the reaction rate cannot be resolved by the numerical simulation, which is where the grid frequency is greater than A . Also shown are contours at $A = \{10^3, 10^4, 10^7, 10^9\} \text{ s}^{-1}$ (solid, dashed, dotdash, dot). For current simulations, frequencies of $\sim 10^6 \text{ s}^{-1}$ are resolvable. This shows that the bulk viscous approximation should only be used for $T \gtrsim 10$ MeV, and as the grid resolution improves becomes less necessary.	151
7.1	Real and imaginary part of the solutions of eq. (7.59), with both the frequency and $ \mathbf{v}_A \cdot \mathbf{k} $ in units of $\sqrt{\text{Tr}(\sigma^2)}$. The solutions plotted correspond to the fastest growing modes evolving on top of an MHD sheared background. We see that the magnetic field has a stabilizing effect, as the growth rates are reduced with respect to those of the corresponding hydrodynamic modes. The stabilizing effect is all the more pronounced the more the wave-vector is aligned with the magnetic field lines, and is switched off for modes propagating in the directions perpendicular to the magnetic field lines. In particular, modes corresponding to sufficiently large values of $ \mathbf{v}_A \cdot \mathbf{k} $ are turned stable.	174
Арр	endix A.1 Left: Tetrad transported along the observer worldline. Right: Connecting geodesics starting from and perpendicular to the central curve. Figure adapted from Misner et al. [155].	186
Арр	endix C.1 Plot of \mathcal{A} for the APR equation of state used in [104]. The restoring term γ is calculated assuming the Fermi surface approximation remains valid. Contours are at $\mathcal{A} = \{10^3, 10^4, 10^7, 10^9\}s^{-1}$ (solid, dash, dot-dash, dot).	195

Declaration of Authorship

I declare that this thesis and the work presented in it is my own and has been generated by me as the result of my own original research.

I confirm that:

*

- 1. This work was done wholly or mainly while in candidature for a research degree at this University;
- 2. Where any part of this thesis has previously been submitted for a degree or any other qualification at this University or any other institution, this has been clearly stated;
- 3. Where I have consulted the published work of others, this is always clearly attributed;
- 4. Where I have quoted from the work of others, the source is always given. With the exception of such quotations, this thesis is entirely my own work;
- 5. I have acknowledged all main sources of help;
- 6. Where the thesis is based on work done by myself jointly with others, I have made clear exactly what was done by others and what I have contributed myself;
- 7. Parts of this work have been published as:

Signed:....

- Celora et al. [57],
- Celora et al. [58],
- Andersson et al. [26],
- Celora et al. [59].
- Celora et al. [60].

Date:....

Acknowledgements

Even though at times it feels like yesterday, almost four years have passed since I first set foot in Southampton. All the experiences I have had during this time have massively shaped the person I have become, and if I can proudly stand where I am today I owe it to the great people I shared parts of this journey with. Since the list is too long to acknowledge everyone personally, I will only thank here those that deserve special mentions and without whom this would not have been possible.

First of all, I am hugely indebted to my supervisors Nils Andersson and Ian Hakwe, and to Greg Comer—a.k.a. the fantastic trio. I have indeed been extremely lucky to have had not two, but three superb mentors. You have been an endless source of inspiration and guidance, and I did my best to learn as much as I could from you. Working with giants like you has been an immense pleasure and honour, and I hope our collaboration will continue for a long time.

I must also express my sincere thanks to the members of the Gravity Group, particularly those working on neutron star physics. I am honoured to have been part of such a group of brilliant scientists, and grateful for all I have learnt from you.

Next, I want to thank my mother Anna. You were the first one to show me how fascinating physics is, and I strongly believe you were the real trigger for everything that came later. I can never repay you for all the sacrifices you have made, which have allowed me to be here today. I must also thank my brother and sisters Agostino, Eleonora and Giuditta—rigorously in alphabetic order. I truly came to appreciate during my time here how much you all mean to me. Knowing that I can always count on you is priceless.

My friends also played a key role in this endeavour. Both the new ones I met in Southampton, and those who have known me for longer. Without you, I do not think I would have made it this far.

Last, but by no means least, I wish to thank my partner in crime Elisa. You are the one that has been closest to me throughout the emotional roller coaster of these years. I know I have been difficult way too often, and I cannot find the words to express how grateful I am for your love, support and faith in me. You should know that this achievement is also yours, and I look forward with indescribable excitement to the next steps of our journey.

Notation

General conventions and units.

In this thesis we work with geometric units, that is

$$\begin{split} G &= 6.674 \times 10^{-11} \, \text{m}^2 \, \text{kg}^{-1} \, \text{s}^{-2} = 1 \; , \\ c &= 2.997 \times 10^8 \, \text{m} \, \text{s}^{-1} = 1 \; , \end{split}$$

where G is Newton's gravitational constant and c is the speed of light in vacuum.

We use the "mostly-plus" signature for the metric, so that the flat line element in Minkowski coordinates $\{t, x, y, z\}$ takes the form

$$ds^2 = -dt^2 + dx^2 + dy^2 + dz^2 .$$

As such, time-like four vectors have negative norm.

Throughout this document we mainly work using explicit components notation for tensors. Space-time indices are denoted by "early" Latin characters a, b, c, ... and take values from 0 to 3, while spatial indices are denoted with "late" Latin characters i, j, k...and take values from 1 to 3. We reserve the Latin characters x, y, z to denote different chemical species/components. In a multi-component system made of neutrons (n) and protons (p) for example, the chemical index takes values x = n, p. Capital Latin characters A, B, C, ... will be used to indicate coordinates on the matter spaces only. Finally, we distinguish indices with respect to a coordinate basis from those relative to an orthonormal basis or tetrad. The latter are denoted with an additional "hat" symbol on top, such as $v^{\hat{a}}$.

We make use of the Einstein summation convention, where repeated indices (one contravariant or "up" and one covariant or "down") imply summation. For example

$$v_a v^a = \sum_{a=0}^3 v_a v^a$$
 ,

where v^a is an arbitrary four vector. Note that the Einstein summation convention does not apply to the chemical indices x, y, z. Similarly, we will not distinguish between "up or down" chemical indices.

Indices enclosed in round or square brackets denote, respectively, symmetrization and anti-symmetrization. If one (or more) of the indices within such brackets has "straight lines" to its left and right, the (anti-)symmetrization does not apply to it. For example, given a generic tensor A^{abc}

$$A^{a(bc)} = \frac{1}{2} \left(A^{abc} + A^{acb} \right) , \quad A^{a[bc]} = \frac{1}{2} \left(A^{abc} - A^{acb} \right) , \quad A^{(a|b|c)} = \frac{1}{2} \left(A^{abc} + A^{cba} \right) .$$

Derivatives and forms.

Throughout this document we make use of different notions of derivatives of a tensor. As usual, partial and covariant derivatives are denoted with ∂ and ∇ respectively. Working in a coordinate basis, the tangent space is spanned by the basis vectors

$$\partial_a = rac{\partial}{\partial x^a}$$
 .

and the covariant derivative of a generic tensor $T^{ab...}_{cd...}$ takes the form

$$\nabla_e T^{ab...}_{cd...} = \partial_e T^{ab...}_{cd...} + \Gamma^a_{fe} T^{fb...}_{cd...} + \Gamma^b_{ef} T^{af...}_{cd...} + \dots$$
$$- \Gamma^f_{ec} T^{ab...}_{fd...} - \Gamma^d_{ed} T^{ab...}_{cf...} - \dots$$

Clearly, this reduces to partial derivative for scalar quantities. The connection coefficients Γ^a_{bc} are taken as Christoffel symbols and are determined by the metric g_{ab} and its first derivatives as

$$\Gamma^a_{\ bc} = \frac{1}{2} g^{ad} \left(\partial_b g_{dc} + \partial_c g_{bd} - \partial_d g_{bc} \right) \ .$$

This means that, in particular, the connection is compatible with the metric $\nabla_a g_{bc} = 0$ and the connection is symmetric (i.e. torsion free) $\Gamma^a_{bc} = \Gamma^a_{(bc)}$.

An orthonormal basis $e_{\hat{a}}$ is linked to a coordinate one through the "matrix" $e_{\hat{b}}^{a}$ (or its inverse $e_{\hat{b}}^{\hat{a}}$) so that

$$\partial_a = e_a^{\hat{b}} e_{\hat{b}}$$

The covariant derivative of a tensor $T^{\hat{a}\hat{b}...}_{\hat{c}\hat{d}...}$ in a tetrad basis takes the form

$$\nabla_{e} T^{\hat{a}\hat{b}\dots}_{\hat{c}\hat{d}\dots} = \partial_{e} T^{\hat{a}\hat{b}\dots}_{\hat{c}\hat{d}\dots} + \omega^{\hat{a}}_{e\,\hat{f}} T^{\hat{f}\hat{b}\dots}_{\hat{c}\hat{d}\dots} + \omega^{\hat{b}}_{e\,\hat{f}} T^{\hat{a}\hat{f}\dots}_{\hat{c}\hat{d}\dots} + \dots \\ - \omega^{\hat{f}}_{e\,\hat{c}} T^{\hat{a}\hat{b}\dots}_{\hat{f}\hat{d}\dots} - \omega^{\hat{d}}_{e\,\hat{f}} T^{\hat{a}\hat{b}\dots}_{\hat{c}\hat{f}\dots} - \dots ,$$

where the spin coefficients $\omega_{a\,\hat{b}}^{\,\hat{c}}$ can be obtained from the connection coefficients via

$$\omega_{a\,\hat{b}}^{\ \hat{c}} = e_e^{\hat{c}} e_{\hat{b}}^d \Gamma^e_{ad} - e_{\hat{b}}^d \partial_a e_d^{\hat{c}} \,.$$

We also make use of the notion of Lie derivative. Given a tensor $T^{ab...}_{cd...}$ and a vector v^a , the Lie derivative of $T^{ab...}_{cd...}$ along v^a is given by

$$\mathcal{L}_{v}T^{ab...}_{cd...} = v^{d}\partial_{d}T^{ab...}_{cd...} - (\partial_{e}v^{a})T^{eb...}_{cd...} - (\partial_{e}v^{b})T^{ae...}_{cd...} - \dots$$
$$+ (\partial_{c}v^{e})T^{ab...}_{ed...} + (\partial_{d}v^{e})T^{ab...}_{ce...} + \dots$$

Whilst the Lie derivative does not require a metric or a connection to be defined, it is often convenient to rewrite the expression above in a form that is manifestly covariant. This can be obtained by substituting $\partial \rightarrow \nabla$ in the expression above, provided the connection is symmetric (the case of interest here as we work with Christoffel symbols).

We also make use of differential (p-)forms, namely covariant tensors that are totally anti-symmetric. Given a p-form *n* we write it in components notation using the "natural basis"

$$n=rac{1}{p!}n_{i_1\ldots i_p}dx^{i_1}\wedge\cdots\wedge dx^{i_p}$$

where \wedge is the exterior product and $n_{i_1 \dots i_p}$ is totally anti-symmetric in its indices. The exterior derivative *d* is an operation that takes a p-form as input and outputs a (p+1)-form. In components notation this is defined as

$$dn_{i_1\dots i_{p+1}} = (p+1)\partial_{[i_1}n_{i_2\dots i_{p+1}]}$$

In a similar fashion as for the Lie derivative, the exterior derivative of a p-form can be written in a manifestly covariant way by substituting $\partial \rightarrow \nabla$ in the expression above provided we work with a symmetric connection.

The Hodge-dual operation, denoted by *, takes a p-form as input and outputs a (n-p)-form, where n is the dimension of the manifold the p-form lives in. In component notation,

$$(*n)_{i_1\ldots i_q} = \varepsilon_{i_1\ldots i_q}^{j_1\ldots i_p} n_{j_1\ldots j_p}, \quad q=n-p$$

where ε is the spacetime volume form, defined as

$$\varepsilon_{abcd} = \sqrt{-g}[a \, b \, c \, d]$$
, $\varepsilon_{ab}^{\ cd} = \varepsilon_{abef} g^{ce} g^{df}$

with [*a b c d*] denoting the Levi-Civita symbol.

Riemann, Ricci and Einstein tensor.

The space-time curvature is encoded in the Riemann tensor, namely the space-time is flat provided the Riemann tensor vanishes. Given a generic co-vector (or 1-form) w_a , the Riemann tensor is defined as

$$[
abla_a,
abla_b] w_c = (
abla_a
abla_b -
abla_b
abla_a) w_c = -R^d_{\ cab} w_d \; .$$

In components notation,

$$R^a_{bcd} = \partial_c \Gamma^a_{bd} - \partial_d \Gamma^a_{bc} + \Gamma^e_{db} \Gamma^a_{ce} - \Gamma^e_{bc} \Gamma^a_{de}$$

The Ricci tensor, Ricci scalar and Einstein tensor are obtained from the Riemann tensor as

$$R_{ab} = R^{c}_{acb}$$
, $R = R^{a}_{a}$, $G_{ab} = R_{ab} - \frac{1}{2}Rg_{ab}$.

Projections and velocity gradients decomposition.

Throughout this document we often use parallel and perpendicular projection operators with respect to an observer. Given a time-like unit vector u^a , these are defined as

$$//_{b}^{a} = -u^{a}u_{b}$$
, $\perp_{b}^{a} = g_{b}^{a} + u^{a}u_{b}$.

We also make frequent use of the observer four velocity gradients decomposition. This is given by

$$abla_a u_b = -a_b u_a + \omega_{ab} + \sigma_{ab} + rac{1}{3} heta \perp_{ab}$$

where the acceleration and vorticity, shear and expansion rate are defined (respectively) as

$$a^b = u^a \nabla_a u^b$$
,
 $\omega_{ab} = \pm^c_{[a} \pm^d_{b]} \nabla_c u_d$,
 $\sigma_{ab} = \pm^c_{(a} \pm^d_{b)} \nabla_c u_d - \frac{1}{3} \theta \pm_{ab}$,
 $\theta = \nabla_a u^a = \pm^a_b \nabla_a u^b$.

Finally, the notion of Hodge-dual in the subspace orthogonal with respect to some observer u^a four velocity is frequently used in this work. This is defined in terms of the space-time volume form ε_{abcd} as

$$\varepsilon_{abc}^{\mathbf{u}} = \varepsilon_{dabc} u^d$$
.

Note that, for notational clarity, we will often drop the u-label and write the object as ε_{abc} instead. We do so whenever there is no risk of confusion.

Any variation from (or addition to) the notational conventions discussed here will be made explicit throughout the document.

Chapter 1

Introduction

1.1 Motivation

Fluid behaviour is relevant across all macroscopic scales in the Universe, from the interior of biological cells and the cardiovascular apparatus all the way up to planetary systems, stars, galaxies and beyond. This, together with the fact that fluids are vital to human survival, is arguably the reason why hydrodynamics—namely the study and modelling of fluid behaviour-is one of the oldest research areas in physics. And yet, despite fluids having attracted the attention of numerous scientists and engineers over the years, hydrodynamic modelling still presents severe conceptual and computational challenges. This is demonstrated, for example, by the "Navier-Stokes existence and smoothness problem" Millennium Prize, and/or the fact that turbulent flows-namely, the vast majority of real-life flows-are extremely costly to simulate. Such challenges are ever more formidable when we couple hydrodynamics to relativity, meaning that we consider fluids flowing at velocities commensurate to the speed of light and, at the same time, immersed in a strong gravitational field. The challenges added by the (successful) marriage of relativity and hydrodynamics are, not surprisingly, both conceptual and practical. To give one example, the most intuitive and natural extensions to relativity of the Navier-Stokes equations-further discussed in this thesis-are wellknown to give rise to problems [115]. On the practical side, simulations of relativistic fluids are intimidating by their complexity and computational cost.

As a small side-step before carrying on, let us state clearly that throughout this thesis we consider Einstein's General Relativity as the theory of gravity, even though we know that this cannot be the "ultimate theory". The indications of this are, in fact both theoretical and observational. On the one hand we know General Relativity breaks down on the quantum scales. On the other hand, we may view the need of including mysterious (and predominant) dark energy/matter contents in our cosmological models—required to match observational evidences such as the accelerated expansion of the Universe—as an indication that something is missing in our understanding. Nonetheless we will here take a pragmatic/conservative approach and consider General Relativity as the "correct" theory of gravity. This is well-motivated considering that General Relativity has passed, over the years, weak and strong-field tests with flying colours (see, for example, the recent review by Will [226]).

Given the tremendous challenges posed by the modelling of relativistic fluids, it is perfectly logical to wonder why we should care. The obvious reason is that there exist real systems whose description/modelling requires relativistic hydrodynamics. The most intriguing ones, possibly from a biased perspective, are neutron stars. Neutron stars form, quite dramatically, as result of a dying star that was born sufficiently massivenamely with mass $8M_{\odot} \leq M \leq 30M_{\odot}$, where M_{\odot} is the mass of our Sun [199, 186, 17]. Once it has emptied its nuclear fuel reservoir, such a massive star undergoes an extremely powerful explosion, known as "core-collapse supernova", that leaves behind a proto-neutron star. In essence, neutron stars are among the most exotic objects in the Universe: they contain roughly the same mass as the Sun squeezed into a 10 km radius sphere. Their extreme compactness, second only to black holes, means that General Relativity is an absolute must for accurately modelling neutron stars' phenomenology. At the same time, the extreme densities reached inside neutron stars (a few times that of an atomic nucleus) make them ideal laboratories to probe otherwise inaccessible physical regimes. In particular, realistic modelling of neutron star phenomenology-and its "validation" through simulations and observations-allow us to set constraints on the elusive equation of state encoding matter properties at such extreme densities, and test our understanding of gravity in the strong field regime at the same time.

When it comes to neutron star astrophysics, it is almost impossible not to mention the spectacular detection of 17th August 2017—hence dubbed *GW*170817. For the first time, the Laser Interferometer Gravitational-wave Observatory (LIGO) and Virgo teams captured the faint gravitational wave signal emitted during a binary neutron star inspiral [2]. This detection came roughly two years after the very first gravitational wave detection accomplished by the LIGO team on 14th September 2015 [1]. In this first remarkable event—known as *GW*150914—the gravitational waves were generated by two black holes inspiralling and merging, and travelled for about 1.3 billion years (at the speed of light) before reaching the detectors on Earth. Despite this earlier detection and the many more binary black hole merger detections that followed¹, *GW*170817 truly represents a milestone for astrophysics. The gravitational wave event was soon followed by

¹See https://www.ligo.org/detections.php for an up to date catalogue.

the detection of a post-merger electromagnetic counterpart [3] thus marking the beginning of a new "multi-messenger era"². Moreover, this event also confirmed the longstanding paradigm that much of the heavy elements in our Universe form precisely during such spectacular cosmic fireworks [123, 171].

Whilst it is undeniable that *GW*170817 represents an astonishing achievement, it has also demonstrated the range of exciting features we can explore with neutron star mergers [36, 185, 149, 189, 203]. Neutron star mergers provide an opportunity to explore many extremes of physics, from the state of matter beyond the nuclear saturation density (the elusive equation of state [164, 80, 218, 135]) to the formation of a long-lived merger remnant (likely a black hole [201, 45, 30]). Matter outflows and the associated rapid nuclear reactions determine the (hopefully) observable kilonova signature [152] and the twisting of the stars' magnetic field may help collimate an emerging jet and hence explain observed short gamma-ray bursts [167, 64].

Current gravitational-wave facilities, however, are only sensitive to the signal produced in the inspiral phase (during which the orbit shrinks due to energy lost in gravitational waves until the neutron stars touch and merge), even though much of the interesting dynamics happens at merger and during the post merger phase (at the end of which the system eventually settles down to either a black hole or neutron star). The violent merger dynamics is affected by the details of the transport properties of matter at such extreme densities and temperatures [12], and hence crucial to set tight constraints on the still-baffling equation of state. For this reason, new and more sensitive facilities are currently at the planning stage—such as the Einstein Telescope [146] and the Cosmic Explorer [182, 76] in the context of gravitational wave detectors—and expected to come online in the 2030s. With more sensitive instruments coming on-stream in the future, higher precision observations are anticipated and one may hope to extract more detailed information about the involved physics. These extremely exciting prospects constitute the motivation for the work presented in this thesis.

1.2 Outline

As present (and future) binary neutron star merger detections offer a wealth of opportunities to explore several extremes of physics, it is important to have an understanding of the different ingredients required to fully realize this potential—this will also provide further context for the results discussed within this thesis.

²In truth, it was the FERMI gamma ray telescope that sent out the trigger alert to LIGO and other telescopes. A large noise glitch in one of the LIGO detectors at the same time as the real signal, in fact prevented LIGO from sending out the alert first—as the gravitational wave signal arrived in the detectors roughly two seconds before the gamma ray burst detection. However, after having cleaned the data, LIGO confirmed the detection in gravitational waves of the first binary neutron star merger.

We obviously need sensitive detectors/telescopes (of all sorts) and, for the purpose of this thesis, it will suffice to say that efforts in this direction are under way. We obviously also need sophisticated tools to analyze the data, and extract useful physics information from a merger detection. Focusing on the gravitational aspects, for example, we need to contrast the gravitational wave signal to waveform templates. To construct the templates, we need accurate theoretical modelling of the entire merger process, from inspiral to ringdown [69]. Analytical models (based on the so-called Post-Newtonian expansion [172]) can accurately describe the inspiral phase, but they prove inadequate as the orbital separation of the two neutron stars (or black holes) approaches a few times their radii. To model the actual merger and post-merger phase we must rely on numerical simulations [184, 200]. This means that, as mergers are highly dynamical events, the full suite of Einstein equations have to be solved together with a suitable description of matter. Current simulations tend to involve an evolution of the ideal fluid equations, possibly augmented with a relatively simple scheme (or closure) to account for electromagnetic effects and (neutrino) radiation. Although these models already provide a formidable computational task, it is easy to argue that an accurate description of the physics involved in a merger would require even more complicated theoretical models. This is ever more true given the expected advancements in detector technology and sensitivity.

When it comes to neutron stars, it can be easily argued that a fluid model capable of accounting for all the relevant physics should involve at least four distinct components [21]. We expect, in fact, to find superfluid neutrons and superconducting protons as we go deep enough into the crust and core of a neutron star [103, 62], meaning that neutrons and protons flow independently from each other. We also expect a flow of electrons, coupled to the protons due to electromagnetic interactions, and finally a heat flux/entropy flow. This means that sophisticated multi-fluid models constitute an important part of the story, and a substantial part of this thesis is dedicated to discussing them. Moreover, as the violent dynamics of a merger is expected to push matter out of (local) thermodynamic equilibrium [104, 105], we also need to include dissipation in the models we would like to simulate. At the same time, however, while there is no doubt that multifluid models are crucial for modelling the phenomenology of neutron stars³, the question arises as to whether, say, superconductivity in the core will impact merger and post-merger dynamics. We may expect, for example, that the heating generated by the two neutron stars smashing against each other is going to melt the crust and push matter above the relevant critical temperature. In this respect, however, current simulations do not provide a definite answer, and some suggest that parts of the core could remain cold enough for superfluidity/superconductivity to play a role [104]. Obviously, the presence of superfluid/superconducting phases would change the underlying dissipative mechanisms. In essence, given the expected sensitivity of

³Pulsar glitches, for example, are considered smoking gun evidence for the presence of superfluid phases in the interiors of neutron star [107, 29].

future detectors, we are urged to check whether a more realistic modelling of neutron star matter can leave a measurable imprint in the detected signals.

In developing such complicated theoretical models, however, we need to bear in mind the extreme computational costs of simulating them, forcing us to consider simplifications whenever possible. The limitations associated with actual numerical implementations become ever more crucial if we consider that we expect turbulence to develop in mergers due to known fluid instabilities [138, 151]. In fact, the modelling of turbulence is an extremely complicated and subtle business already at the Newtonian level. In essence, we would like to make sure that the underlying physical features are faithfully represented by our theoretical models and, at the same time, make sure that we "get the physics right" in our numerical simulations. This is clearly a very challenging issue. As a final point, it is also crucial to keep in mind that we need to make sure that the physics we hope to explore is cleanly associated with particular observational features.

In this thesis we will discuss recent theoretical advancements in (electromagnetic) multifluid modelling in General Relativity, motivated by applications to neutron star astrophysics and neutron star mergers in particular. The introduction so far already suggests that there are a number of interconnected aspects to be kept in mind, from dissipation to turbulence. Given the complexity of the systems we aim to model, it is natural to consider each of these different aspects "one at a time". As such, the thesis is divided into three different parts, each of which is somewhat self-contained and could be read almost independently from the rest. Nonetheless, the discussions provided in the different parts complement each other to form a coherent story.

The first part of this thesis is devoted to the modelling of dissipative multi-fluids. Modelling dissipation in relativity is challenging already for single fluid models, and has baffled physicists for quite some time. Several ideas and prescriptions are currently on the market, and some of them are fairly recent. We will therefore start the first part by reviewing in chapter 2 the different modelling strategies. The discussion will highlight, in particular, that most of the present strategies do not seem to allow for an "easy" extension to multi-fluid models, which is presumably required for neutron stars. We will then continue in chapter 3 focusing on the (arguably) unique strategy that is clearly suited for this. In particular we will study the close to thermodynamic equilibrium regime of an action based variational model for dissipative multi-fluids.

The second part of this thesis is devoted to discussing some of the issues that arise when modelling turbulence in relativity. Turbulent models in Newtonian hydrodynamics often involve some notion of filtering/averaging, and such strategies are becoming fashionable also in the relativity community, with impressive successes [169, 5]. However, whilst there has been significant numerical effort going into extending the Newtonian logic to relativity, the foundational underpinnings for these strategies are not as well

explored as one may like. As such, after a brief introduction to hydrodynamic turbulence, we will focus in chapter 4 on discussing the foundational issues that arise in extending the Newtonian logic to the relativistic setting. This brings us to propose a novel scheme for carrying out filtering for relativistic fluids. Whilst the analysis in chapter 4 focuses on single-fluid models, in the following chapter 5 we will discuss the first steps towards extending such a scheme to multi-fluids. Even though at first sight this second part may seem relatively independent from the previous one, it is rather evident that the two are linked. A first indication of this comes from the well-known fact that energy and momentum transport are enhanced in turbulent flows. Furthermore, filtering an ideal fluid model—that is a model that does not account for dissipative effects—inevitably introduces additional terms in the equations that are akin to "effective dissipative terms", as we demonstrate in chapter 4.

While the first two parts of this thesis are rather theoretical—although the discussion in part II brings to the fore the key role played by computational limitations—we will continue in part III by considering two relevant binary neutron star merger applications. In particular, in chapter 6 we will focus on modelling reaction-sourced bulk viscosity for neutron star simulations—as reactions are expected to source the dominant dissipative mechanism in mergers. The topic is relatively well-explored from a "theoretical" perspective [11, 9, 10, 6, 8]—but still numerically challenging—so that our analysis aims to establish how the inevitable "limitations" of a numerical simulation (in terms of resolution) enter the discussion. In particular, we also explore the link to (or conflict with) strategies for dealing with turbulent flows. We then continue our journey focusing in chapter 7 on the magneto-rotational instability (MRI), which is considered a key mechanism for developing/sustaining turbulence in the outer envelopes of merger remnants. Crucially, the analysis is framed in a way suited for highly dynamical environments such as mergers. We provide our concluding remarks and comment on future works in part IV.

Supplemental material is provided in appendices. In appendix A we discuss the Fermi coordinates as these play a key role in chapter 4. Appendix B provides additional information about the multi-scale methods used in chapter 6, while in appendix C we make explicit contact with quantities that can be computed from standard equation of state tables as collected, for example, in the compOSE database [210]. In appendix D we link the analysis of chapter 7 to the usual MRI results/criteria. Finally, in appendix E we discuss the Routh-Hurwitz criterion, which is going to be used both in chapters 4 and 7.

1.3 Relativistic perfect fluids

As an appetizer before we begin the main part of our journey, let us cover some background material that is relevant to all the following parts. Quite naturally, dissipative fluid models build on the notion of perfect fluids, namely fluids where dissipative effects can be neglected. As for turbulence, we expect it to develop in fluid flows where inertial effects prevail over viscous/dissipative ones. As such, we need to understand how to model (relativistic) ideal fluids before we can meaningfully start talking about dissipation and turbulence. Excellent introductions to relativistic ideal fluids can be found in several textbooks/reviews—such as, for example, [134, 224, 155, 81, 184, 21]— and the material is often covered in introductions to General Relativity. Nonetheless, we take this as an opportunity to set the stage for the analysis presented in the following chapters.

Let us begin by writing down the Einstein field equations (in geometric units)

$$G^{ab} = 8\pi T^{ab} , \qquad (1.1)$$

where G^{ab} is the Einstein tensor associated with the spacetime metric, while T^{ab} is the stress-energy-momentum tensor associated with the matter content within the spacetime. In essence, these equations prescribe how the matter/energy content curves the spacetime and, in turn, the spacetime metric/curvature dictates how (test, freely-falling) particles move in spacetime. An important property of the Einstein tensor is that it is divergence free, meaning that

$$\nabla_a G^{ab} = 0 , \qquad (1.2)$$

which follows from its definition and the second Bianchi identity satisfied by the Riemann tensor. This means that, as a consequence of the Einstein field equations, we also have

$$\nabla_a T^{ab} = 0 , \qquad (1.3)$$

namely, energy and momentum are locally conserved. From a field-theory perspective, the conservation of the stress-energy-momentum tensor is associated with diffeomorphism invariance—and hence is analogous to conservation laws obeyed by the Noether currents whenever the field-theory has some underlying symmetry [144, 225]. Whilst the distinction between equations of motion and conserved Noether currents becomes important in the context of multi-fluid modelling—and we will come back to stress this issue—we here follow the "tradition" and refer to eq. (1.3) as the equations of motion of hydrodynamics.

It is immediately clear, however, that in order for eq. (1.3) to yield equations we can work with (or simulate), we need to specify the stress-energy-momentum tensor otherwise our fluid model is somewhat "empty". To do so, we first note that the stressenergy-momentum tensor can be decomposed—just like any other tensor—into parts that are, algebraically, parallel or orthogonal with respect to some observer $U^{a.4}$ In

⁴By observer we here mean a normalized time-like unit vector. The concept will be further developed in the following chapters.

particular, we can decompose the stress-energy-momentum tensor as

$$T^{ab} = \mathcal{E}U^a U^b + 2\mathcal{Q}^{(a} U^{b)} + \mathcal{S}^{ab}$$
(1.4)

where

 $\mathcal{E} = T^{ab} U_a U_b , \quad \mathcal{Q}^a = \perp^a_b T^{bc} U_c , \quad \mathcal{S}^{ab} = \perp^{ac} \perp^{bd} T_{cd} . \tag{1.5a}$

The different terms are then interpreted as follows: \mathcal{E} is the energy density measured by the observer, Q^a is the spatial momentum flux measured by the observer (and, by definition, satisfies $Q^a U_a = 0$) while S^{ab} encodes the spatial stresses (and is obviously a symmetric tensor). Whilst this decomposition is useful, it is just an algebraic decomposition. As such, we need to do some more work.

In order to arrive at the perfect fluid equations we now introduce the concept of locally co-moving observer, which we denote with u^a in order to distinguish it from the generic one introduced above. We assume that the particles constituting the fluid collide sufficiently often (using a classical mechanics language) that we can meaningfully talk about a mean velocity field associated with the particles' collective motion. The co-moving observer is then defined as the observer that moves with the mean flow. Next, we further assume that the frequent collisions are random in nature, so that an observer moving with the mean flow would observe an isotropic distribution of particles. This then brings us to consider the most generic stress-energy-momentum tensor that is consistent with such a notion of isotropy. In essence, we will require every quantity in the stress-energy-momentum tensor decomposition to be invariant with respect to spatial rotations (spatial with respect to the co-moving observer u^a).

We can rephrase the last assumption in a more formal way, by saying that we retain only those parts of T^{ab} —which belongs to a rank-two tensorial representation of the Lorentz group⁵ SO(3,1)—that transform as scalars under the rotation group SO(3). As the rotation group is a sub-group of the Lorentz one, the decomposition of different representations of the Lorentz group with respect to the rotation group are well known. In particular, a symmetric rank-two (Lorentz) tensor is made up of two scalars, one vector, and a spin-2 part with respect to the rotation group. The assumption of isotropy means that we take the vectorial and spin-2 part to vanish. However, let us here leave the formalities aside and proceed with a more intuitive discussion. We begin with the energy density, which is obviously a scalar under the Lorentz group. As such, the energy density is also a scalar with respect to rotations and we can keep it. Next, we observe that the momentum flux is a spatial vector living in the subspace orthogonal to the observer four-vector u^a . This means that a non vanishing momentum flux vector

⁵In fact, we can always introduce a non-coordinate basis/tetrad such that the metric is in Minkowski form at each point in spacetime. Using such a basis, any tensor becomes a Lorentz tensor. This has the additional advantage that it shows how in General Relativity we take the Poincaré group—namely the Lorentz group augmented with space-time translations—and require it to be a local/gauge symmetry of the theory. Invariance with respect to gauged space-time translations is nothing but invariance with respect to diffeomorphisms. See, for example, Carroll [50].

would identify a preferred direction in the subspace orthogonal to u^a . In order to not break isotropy, then, we need to assume that it vanishes. Finally, we focus on the spatial stresses and note these can be further decomposed as

$$S^{ab} = S \perp^{ab} + S^{\langle ab \rangle}$$
, where $S = S^a_a$, $S^{\langle ab \rangle} = S^{ab} - \frac{1}{3}g^{ab}S$. (1.6)

In short, S is the trace and $S^{\langle ab \rangle}$ is the (symmetric) trace-free part of the stress tensor. The trace is a Lorentz scalar so that the same logic as for the energy density applies. On the other hand, the trace-free part describes anisotropic stresses and hence must vanish in isotropic fluids. We conclude that the stress-energy-momentum tensor of an ideal fluid takes the form

$$T^{ab} = \varepsilon u^a u^b + p(g^{ab} + u^a u^b) .$$
(1.7)

Note that we have here re-labelled the energy density and isotropic stresses as ε and p (respectively), consistent with the standard notation. The isotropic stresses are then identified with the (equilibrium) thermodynamic pressure, usually given in the form of an equation of state (either in analytical form or provided as a table of data).

Since we now have a prescription for the stress-energy-momentum tensor, we can use it to make the energy-momentum conservation law more explicit. Working out the parallel and orthogonal projections of eq. (1.3) we obtain⁶

$$u^{a}\nabla_{a}\varepsilon + (\varepsilon + p)\nabla_{a}u^{a} = 0, \qquad (1.8a)$$

$$(p+\varepsilon)a^b + \perp^{bc} \nabla_c p = 0.$$
(1.8b)

The first relation is, intuitively, the familiar energy density balance law. It says, in fact, that the time derivative of the energy density is proportional to the expansion rate of the fluid (with a negative sign). If the fluid undergoes compression (negative expansion rate), the energy density will increase because i) the same energy is now stored in a smaller volume ii) of the "pdV"-work the fluid element does on its surroundings. The second equation is instead the relativistic Euler equation: it says that the fluid is accelerated in such a way to minimize pressure gradients, where the pre-factor in front of the four-acceleration a^b accounts for pressure contributions to the total mass-energy of the fluid.

We can think of the two equations in (1.8) as evolution equations for the energy density and the four velocity, respectively⁷. The system would then be closed provided we have an equation of state of the form $p(\varepsilon)$. Whilst this is a valid equation of state for, say, a fluid made of radiation, we would normally work with a two-parameter (or more) equation of state. For example, we can think of it as a function $p = p(n, \varepsilon)$, where

⁶It should be obvious by now that we here mean parallel and orthogonal with respect to the comoving observer.

⁷That we only need three equations to fully determine the four velocity is due to the fact that a fourth component can be obtained from the constraint $u^a u_a = -1$ provided the other three are known.

n is the particle number density⁸. This means that we need an additional evolution equation for n. As the particle number density is typically associated with baryons, we are then led to write a continuity equation

$$\nabla_a(nu^a) = 0 \Longrightarrow u^a \nabla_a n + n \nabla_a u^a = 0.$$
(1.9)

Note that the particle number density continuity equation plays the role of the mass continuity equation in non-relativistic fluids (see, e.g. [199]). Taken together, eqs. (1.8) and (1.9) form a closed system of equations⁹.

We conclude this introduction to relativistic perfect fluids showing why this model truly describes an ideal fluid. To do so, however, we need to talk a little bit about (equilibrium) thermodynamics. Let us first of all consider the first law of thermodynamics written in terms of quantities per unit volume (see, for example, [21])

$$d\varepsilon = Tds + \mu dn$$
, $T = \left(\frac{\partial\varepsilon}{\partial s}\right)_n$, $\mu = \left(\frac{\partial\varepsilon}{\partial n}\right)_s$ (1.10)

where μ , *T* are the chemical potential and temperature, while *s* in the entropy density and *n*, ε are the particle number and energy densities as before. Similarly, the "Euler relation" is

$$p + \varepsilon = Ts + \mu n . \tag{1.11}$$

In essence, what this shows is that we can think of the equation of state as some relation that allows us to express the various thermodynamic quantities in terms of our two favourite ones. Quite naturally from a theory perspective, we are then led to choose these as the energy and particle number densities. In particular, we can think of the entropy density as $s = s(n, \varepsilon)$. As such, we can derive an evolution equation for the entropy density starting from the relevant equations for the energy and particle densities. It is then easy to show that

$$T\nabla_a(su^a) = Tu^a \nabla_a s + Ts \nabla_a u^a = u^a \nabla_a \varepsilon - \mu u^a \nabla_a n + (p + \varepsilon - \mu n) \nabla_a u^a$$
$$= (u^a \nabla_a \varepsilon + (p + \varepsilon) \nabla_a u^a) - \mu (u^a \nabla_a n + n \nabla_a u^a) = 0, \quad (1.12)$$

where we have used the thermodynamic relations above as well as the continuity equations for the energy and particle number densities. In essence, this equation says that the entropy-current su^a satisfies a continuity equation as well. As thermodynamics dictates that dissipation is associated with an increase in entropy, the model truly represents an ideal fluid.

⁸In non-relativistic fluids, the pressure is often given as function of the mass density instead.

⁹We also note that more work is required to write the same equations in a form suitable for numerical implementations [184, 200, 100]. However, this would require us to cover additional background, which would be beyond the scope of this thesis anyway.

Part I

Dissipative (multi-)fluids in General Relativity

Chapter 2

Dissipative fluid models in General Relativity: an overview

The first part of this thesis focuses on modelling dissipative fluids in (general) relativity, which is a thorny issue that has kept physicists busy for quite some time. The covariant nature of General Relativity highlights the central role played by the reference frame used to describe a physical system. At the same time, the evolution of dissipative fluids must be consistent with thermodynamic principles and the arrow of time associated with the second law. As such, it is clear that the marriage of General Relativity and thermodynamics poses interesting foundational questions, so that it should not come a surprise that a number of authors have made significant contributions over the years¹.

Here we present an overview of the most important developments and results obtained over the past seventy years. The aim is not to draw a complete picture but only to introduce the main ideas and set the stage for what comes next. We will, in fact, continue in the following chapter 3 by focusing on the action-based dissipative framework of Andersson and Comer [20]. We do so as this is the only framework currently on the market that is suited for describing multi-fluid systems.

2.1 Non-equilibrium thermodynamics

As hydrodynamics is a macroscopic theory, fluid models must be complemented with some information about the microphysics of the system they aim to describe. For the ideal fluid case, for instance, such information may be the chemical potentials of the fluid constituents. These can be obtained within the equilibrium thermodynamics

¹Since the list of authors is indeed very long (and growing), we find it difficult to do justice to all of them here. Instead, we will acknowledge as we go along the most important contributions (for our purposes) made by the various authors, and point to [184, 21, 188, 42] for further references.

framework. However, if we move to the non-ideal case, new effects like viscosity or heat conductivity start to play a role in the macroscopic dynamics. To take these effects into account, one needs to provide hydrodynamic models with new information, typically in the form of transport coefficients. As these describe out of equilibrium processes, we start with a short discussion of non-equilibrium thermodynamics. Note that the analysis in this section is Newtonian. We will discuss how to re-phrase the same ideas in a relativistic setting later.

2.1.1 Linear irreversible thermodynamics

Classical equilibrium thermodynamics is a macroscopic theory that aims at describing the observed properties of a many-particle system—at thermodynamic equilibrium— in terms of a finite set of macroscopic variables (like energy E, volume V and particle's number N). It does not describe, however, the evolution of a system towards such equilibrium states, nor the damping of statistical fluctuations around it. These kind of phenomena are the realm of irreversible thermodynamics.

A first fundamental result in the study of irreversible processes is due to the pioneering work of Onsager [166], and it is now customary to refer to it as "Linear Irreversible Thermodynamics" (LIT). To sketch the main ideas of LIT, let us consider the entropy of a system out of (but close to) equilibrium, assuming that the entropy *S* is function of some state variables A_1 , A_2 , A_3 ... The key assumption of LIT is that the number of variables necessary to specify the out of equilibrium state of the system is the same as in equilibrium. Then we can write²

$$S = S_{\rm eq} - \frac{1}{2} \sum_{\beta,\gamma} G_{\beta\gamma} \alpha_{\beta} \alpha_{\gamma} , \qquad (2.1)$$

where α_{β} is the deviation from the equilibrium value of the corresponding variable A_{β} . If we then assume the *Onsager regression hypothesis*—which states that spontaneous fluctuations of the system decay with the same evolution law as the perturbations caused by external forces—we can write

$$\frac{d\alpha_{\beta}}{dt} = -\sum_{\gamma} M_{\beta\gamma} \alpha_{\gamma} , \qquad (2.2)$$

²In this section we will use the greek letters α , β ... to label the dissipative processes that lead to some entropy production. The Einstein summation convention does not apply to them.

where the matrix M describes the decay of spontaneous fluctuations. The entropy production in turn is³

$$\frac{dS}{dt} = -\sum_{\beta,\gamma} G_{\beta\gamma} \alpha_{\beta} \frac{d\alpha_{\gamma}}{dt} = \sum_{\gamma} J_{\gamma} X_{\gamma} , \qquad (2.3)$$

where we have introduced the *thermodynamic forces* $X_{\gamma} = -\sum_{\beta} G_{\beta\gamma} \alpha_{\beta}$ and *thermodynamic fluxes* $J_{\gamma} = d\alpha_{\gamma}/dt$. By means of eq. (2.2) we can rewrite the entropy production rate as

$$\Gamma_s = \frac{dS}{dt} = \sum_{\beta\gamma} L_{\beta\gamma} X_{\beta} X_{\gamma} , \qquad (2.4)$$

where the matrix $L = M \cdot G^{-1}$. The second law of thermodynamics then implies that the symmetric part of *L* must be a positive semi-definite matrix. While in his original discussion Onsager only considered thermodynamic variables that are even under time reversal, Casimir later extended the analysis to include also odd variables (e.g. the magnetic field) [56], demonstrating the so-called Onsager-Casimir relations:

$$L_{\beta\gamma} = \varepsilon_{\beta} \varepsilon_{\gamma} L_{\gamma\beta} , \qquad (2.5)$$

where ε_{β} is the parity of the variable α_{β} under time reversal.

2.1.2 Causality and extended irreversible thermodynamics

Despite LIT being successful and largely used in many different contexts, it has serious drawbacks that are disturbing already at the Newtonian level, and unacceptable at the relativistic one. For instance, the standard Fourier law for the heat-flux q (here given in 1+1 dimensions) $q = -\kappa \partial_x T$ can be deduced within the LIT program⁴. When the Fourier law is coupled to the balance law for the energy density ε of a system at rest $\partial_t \varepsilon = -\partial_x q$, it yields the heat equation

$$\frac{\partial T}{\partial t} - D \frac{\partial^2 T}{\partial x^2} = 0 , \qquad (2.6)$$

where we introduced the heat diffusivity D, defined as $D = \kappa/c_v$ where c_v is the heat capacity at constant volume. As a result, a temperature perturbation will propagate with infinite speed⁵. This is obviously against our intuition as we would expect thermal and viscous disturbances to propagate with a speed of the order of, say, the mean molecular speed. Whilst this problem may be ignored for non-relativistic applications where typical speeds are much smaller than the speed of light c, it comes back to bite

³Let us note for clarity that the thermodynamic quantities in this section are global and refer to the entire system under consideration. As a consequence, all equilibrium quantities are constant in time. This applies also to $G_{\beta\gamma}$ defined in eq. (2.1), which is evaluated at equilibrium.

⁴Here κ is the heat conductivity, while *T* denotes the temperature.

⁵To better understand this statement it is sufficient to calculate the kernel of the heat equation—that is, a Gaussian. As a result, an initial temperature perturbation localized at the origin will result after some finite time in a non-zero disturbance in the temperature field at all distances from the origin.

us in a special/general relativistic setting. The Extended Irreversible Thermodynamics (EIT) program is built in order to address the issues of LIT keeping the analysis at a thermodynamical level. We now sketch the main ideas of EIT and point to [121] for an exhaustive discussion.

The key assumption of EIT is that some (the majority) of the microscopic degrees of freedom rapidly decay towards their equilibrium values while others will do so on longer timescales. The causality issues of LIT are solved in the EIT paradigm by enlarging the set of variables that describe the out-of-equilibrium state of the system— through the inclusion of slowly decaying variables⁶. At the level of a phenomenolog-ical macroscopic theory, the additional variables can be chosen as the thermodynamic fluxes introduced above. The reason for this can be seen by going back to eq. (2.4). On the one hand, the entropy production rate—a central quantity in the Onsager LIT program—must be quadratic to retain consistency with the Second law of thermodynamics. On the other hand, the out-of-equilibrium entropy is in general a function of a larger set of variables—compared with the equilibrium one. If we then expand the entropy up to second order, like in eqs. (2.1) and (2.4), the dependence on these additional variables must be taken into account. This is exactly the "missing piece" in Onsager LIT, and the ultimate reason for its non-causal predictions. To better understand this, let us consider the simple example of a Newtonian fluid with heat-flux.

Let us start by assuming the (generalized) entropy density *s* of the system is a function of the energy density as usual and one additional non-equilibrium variable, the heat-flux *q*:

$$s = s(\varepsilon, q) . \tag{2.7}$$

As a result, its differential reads

$$ds = \theta^{-1}d\varepsilon + adq \Longrightarrow \frac{\partial s}{\partial t} = \theta^{-1}\frac{\partial \varepsilon}{\partial t} + a\frac{\partial q}{\partial t}.$$
 (2.8)

where both θ —which is a generalized temperature⁷—and *a* are functions of (ε, q) . Next, since in equilibrium there is no heat-flux and the entropy is maximized, we can expand the generalized entropy as $s(\varepsilon, q) = s_e(\varepsilon) - \frac{1}{2}s_2(\varepsilon)q^2$ and write

$$a = \left(\frac{\partial s}{\partial q}\right)_{\varepsilon} = -s_2 q , \qquad (2.9a)$$

$$\theta^{-1} = \left(\frac{\partial s}{\partial u}\right)_q = \frac{1}{T} - \frac{1}{2}\frac{\mathrm{d}s_2}{\mathrm{d}\varepsilon}q^2 \,.$$
(2.9b)

⁶This idea was first proposed by Müller [160].

⁷By generalized temperature we here mean a notion of temperature valid also out of equilibrium, noting that this notion is not unique [121].

The net result is that we can now compute $\partial_t s$ and write it in the form of a balance equation as

$$\frac{\partial s}{\partial t} = -\frac{\partial j_s}{\partial x} + \Gamma_s , \qquad (2.10)$$

where the entropy current j_s and the entropy production rate are

$$j_{\rm s} = \frac{1}{T}q \,, \tag{2.11a}$$

$$\Gamma_{\rm s} = q \left[\frac{\partial \left(T^{-1} \right)}{\partial x} - s_2 \frac{\partial q}{\partial t} \right] , \qquad (2.11b)$$

and we have used the energy conservation law as above. We can then proceed exactly as in the LIT program by writing the entropy production rate as the product of thermodynamic forces and fluxes; namely $\Gamma_s = qX_q$. Assuming a phenomenological law for the force $X_q = \mu(\varepsilon)q$ we obtain the following law for the heat-flux⁸

$$\tau \frac{\partial q}{\partial t} + q = -\kappa \frac{\partial T}{\partial x} . \tag{2.12}$$

In conclusion, by simply enlarging the set of state variables used to described an outof-equilibrium system—and then following the same strategy as in the Onsager LIT program—we obtained the Maxwell-Cattaneo law for the heat-flux (eq. (2.12)). We can then immediately use the last equation and the energy balance law to get an evolution equation for the temperature of the telegrapher type:

$$\tau \frac{\partial^2 T}{\partial t^2} + \frac{\partial T}{\partial t} - D \frac{\partial^2 T}{\partial x^2} = 0.$$
 (2.13)

This equation is hyperbolic and propagates temperature disturbances with a finite speed, its solution becoming indistinguishable from that of the heat-equation at late times.

2.2 Traditional strategies for dissipative fluids

Having discussed the basic ideas behind the EIT paradigm, we now proceed to describe the traditional strategies to model dissipative fluids in relativity. We will first focus on the so-called first-order theories—the Eckart-Landau-Lifshitz models—and then on the Müller-Israel-Stewart theory, following the presentation in [21]. We conclude with a brief discussion on the more abstract divergence-type theories.

⁸Where $\tau = s_2/\mu$ and $\kappa = (\mu T^2)^{-1}$, although this is not really crucial for the present discussion.
2.2.1 Eckart-Landau-Lifshitz models

With the term "first-order" theories one typically refers to the dissipative models introduced by Eckart [73] and Landau and Lifshitz [134]. They represent the most simple and natural generalization of the Newtonian Fourier-Navier-Stokes equations. These two models are essentially the same in terms of physics content—the difference being in the observer chosen to "measure" the various quantities—so we follow [21] and present them as one. Let us note that, however, the mathematical properties of the equations in the two models are not the same—and this impacts on their stability and causality properties.

In the Landau-Eckart models, the equations of motion are given by simple conservation laws⁹

$$\nabla_a n^a = 0 , \qquad (2.14a)$$

$$\nabla_a T^{ab} = 0 , \qquad (2.14b)$$

where n^a is the particle flux (or conserved baryon current) and T^{ab} is the total stressenergy-momentum tensor. In order to account for dissipative effects the stress-energymomentum tensor is decomposed as

$$T^{ab} = \underbrace{(\varepsilon + p)u^{a}u^{b} + pg^{ab}}_{\text{ideal terms}} + \underbrace{\chi \perp^{ab} + 2q^{(a}u^{b)} + \chi^{ab}}_{\text{dissipative terms}}, \qquad (2.15)$$

where we have separated the terms that would be present also in the ideal case from the rest. Making contact with the discussion in section 1.3, we have split $S = p + \chi$ and renamed $S^{\langle ab \rangle} = \chi^{ab}$. In practice, the stress-energy-momentum tensor of some reference equilibrium state is augmented by the dissipative fluxes introduced above. In Eckart-Landau-Lifshitz theories (as well as the Müller-Israel-Stewart model discussed later) the pressure and energy density of the fluid are <u>assumed</u> to be equal to the reference equilibrium values¹⁰. The terms q^a , χ , χ^{ab} represent dissipative fluxes, respectively the heat-flux, the bulk-viscous scalar and the shear-viscous tensor. These additional terms are required to satisfy the following algebraic constraints

$$u^a q_a = \chi^a_a = 0 , \qquad (2.16a)$$

$$u^a \chi_{ab} = 0 , \qquad (2.16b)$$

$$\chi_{[ab]} = 0. \tag{2.16c}$$

⁹These equations constitute the "matter-sector" of the theory, so that in General Relativity they must be coupled with Einstein equations.

¹⁰Or better, the reference equilibrium is defined as the one having the same energy and pressure, and vanishing thermodynamic fluxes.

At the end of the day, eq. (2.15) is just the same algebraic decomposition we encountered in section 1.3. In a similar fashion, the the particle flux n^a is linked to the observer 4-velocity as

$$n^a = nu^a + v^a , \qquad (2.17)$$

where the diffusion vector v^a is required to be orthogonal to the observer 4-velocity $u^a v_a = 0$. Hence it is proportional to the relative velocity between the observer and the particle flux. The Eckart or Landau-Lifschitz frame are obtained by choosing the observer to be respectively the matter frame or the momentum frame—practically this means setting either v^a or q^a to zero in the equations above.

Because of the additional quantities in the stress-energy-momentum tensor decomposition, eq. (2.14) are under-determined. In first-order theories, the system is closed by introducing an entropy flux s^a , which is assumed to be a linear combination of all the available vectors as

$$s^a = su^a + \beta q^a - \lambda v^a , \qquad (2.18)$$

with so-far unspecified coefficients. We also have

$$-u_a s^a = s , \qquad (2.19)$$

so that $s = s(\varepsilon, n)$ in eq. (2.18) is the entropy density (as measured by the chosen observer) and it is assumed to satisfy an equilibrium thermodynamic relation

$$n\nabla_a x_s = \frac{1}{T} \nabla_a \varepsilon - \frac{p+\varepsilon}{nT} \nabla_a n , \qquad (2.20)$$

where $x_s = s/n$ is the entropy per particle. By means of the last equation and the energy conservation law $u_b \nabla_a T^{ab} = 0$ we can write the entropy production rate as

$$\nabla_{a}s^{a} = q^{a}\left(\nabla_{a}\beta - \frac{1}{T}u^{b}\nabla_{b}u_{a}\right) + \left(\frac{1}{T} - \beta\right)\nabla_{a}q^{a} - \left(x_{s} + \lambda - \frac{p + \varepsilon}{nT}\right)\nabla_{a}\nu^{a} - \nu^{a}\nabla_{a}\lambda - \frac{\chi}{T}\nabla_{a}u^{a} - \frac{\chi^{ab}}{T}\nabla_{a}u_{b}.$$
(2.21)

To ensure consistency with the thermodynamic second law we first set

$$\beta = \frac{1}{T} , \qquad (2.22a)$$

$$\lambda = \frac{1}{nT} \left(p + \varepsilon - sT \right) = \frac{\mu}{T} .$$
 (2.22b)

In practice, we identify the parameters introduced in eq. (2.18) with the (inverse) temperature and the Gibbs free-energy per particle. Then, the second law is guaranteed to hold in the simplest possible way—often referred to as the "natural way"—by guessing the following constitutive relations for the dissipative fluxes

$$q^{a} = -\kappa T \perp^{ab} \left(\frac{1}{T} \nabla_{b} T + u^{c} \nabla_{c} u_{b} \right), \qquad (2.23a)$$

$$\nu^a = -\sigma T^2 \perp^{ab} \nabla_b \lambda , \qquad (2.23b)$$

$$\chi = -\zeta\theta , \qquad (2.23c)$$

$$\chi^{ab} = -\eta \sigma^{ab} , \qquad (2.23d)$$

where κ is the heat conductivity (as before) and we introduced the diffusion coefficient σ and the bulk-and shear viscosity coefficients ζ , η . Moreover, let us stress that θ here is the expansion rate of the observer (and not a generalized temperature), and similarly σ^{ab} is the shear tensor. We can now rewrite the entropy production rate as

$$\nabla_a s^a = \frac{q^a q_a}{\kappa T} + \frac{\chi^2}{\eta T} + \frac{\nu^a \nu_a}{\sigma T^2} + \frac{\chi^{ab} \chi_{ab}}{\eta T} \ge 0.$$
(2.24)

The system of equation is now determined and we just need to make sure that the transport coefficients—determined by microphysics— introduced above are all positive.

Let us conclude with a few comments on these first-order models. First of all, despite being quite natural and intuitive, these models have been shown to suffer from stability and causality issues. This means that if we set the system to deviate slightly from thermodynamic equilibrium, the deviations from it may grow rapidly in time (see [115]). Also, they give rise to non-causal behaviour. To better understand this it is sufficient to go back to eq. (2.23) and observe that, for instance, the heat equation is nothing more than a relativistic version of the usual Fourier law. The ultimate reason for this lies in the use of an equilibrium entropy (see eqs. (2.19) and (2.20)). In the parlance of the previous section this is an LIT description—hence, is not surprising that the resulting equations are non-causal.

More recent works (see [40, 133, 117] for instance) have shown that one can build firstorder theories that respects the stability/causality requirements. Because such models are motivated within a field-theory perspective, we will discuss these theories in section 2.4.

2.2.2 Müller-Israel-Stewart models

The take-home message from section 2.2.1 is that the most intuitive and simple way of including dissipative effects in (general) relativistic fluid models leads to problematic results. Inspired by the work of Müller [160] and supported by relativistic kinetic theory, Israel and Stewart proposed an extension of the first order theories to overcome their stability and causality flaws [118, 119, 120].

The logic of these "second-order theories" does not differ much from first-order ones. The main difference is in the ansatz for the entropy flux. In second-order theories the entropy flux is expanded up to second order in the dissipative quantities¹¹ (compare with eq. (2.18))

$$s^{a} = su^{a} + \frac{1}{T}q^{a} - \frac{1}{2T}\left(\beta_{0}\chi^{2} + \beta_{1}q^{b}q_{b} + \beta_{2}\chi_{bc}\chi^{bc}\right)u^{a} + \alpha_{0}\frac{\chi q^{a}}{T} + \alpha_{1}\frac{\chi^{ab}q_{b}}{T}.$$
 (2.25)

As a result, the number of unknown parameters—that have to be determined by microphysical calculations—is now larger. It is also interesting to note that the entropy measured in the Eckart frame is now

$$-u^{a}s_{a} = s - \frac{1}{2T} \left(\beta_{0}\chi^{2} + \beta_{1}q^{b}q_{b} + \beta_{2}\chi_{bc}\chi^{bc} \right).$$
(2.26)

Here *s* is, as before, the entropy density of the fluid at equilibrium—so that eq. (2.20) still holds. It is clear that the ansatz in eq. (2.25) is consistent with the EIT paradigm as second order combinations of the dissipative terms enter the formula for the out-of-equilibrium entropy of the system¹². Also, because the entropy is maximized at equilibrium, we get "for free" the following constraints on (some of) the additional parameters: β_0 , β_1 , $\beta_2 \ge 0$.

The strategy then follows the logic of first-order theories. Making use of the equation of motion $\nabla_a T^{ab} = 0$ and eq. (2.20) we arrive at

$$\nabla_{a}s^{a} = -\frac{1}{T}\chi\left[\theta + \beta_{0}u^{a}\nabla_{a}\chi - \alpha_{0}\nabla_{a}q^{a} - \gamma_{0}Tq^{a}\nabla_{a}\left(\frac{\alpha_{0}}{T}\right) + \frac{\chi T}{2}\nabla_{a}\left(\frac{\beta_{0}u^{a}}{T}\right)\right]$$
$$-\frac{1}{T}q^{a}\left[\frac{1}{T}\nabla_{a}T + a_{a} + \beta_{1}u^{b}\nabla_{b}q^{a} - \alpha_{0}\nabla_{a}\chi - \alpha_{1}\nabla_{b}\chi^{b}_{a} + \frac{T}{2}q_{a}\nabla_{b}\left(\frac{\beta_{1}u^{b}}{T}\right) - (1 - \gamma_{0})\chi T\nabla_{a}\left(\frac{\alpha_{0}}{T}\right) - (1 - \gamma_{1})T\chi^{b}_{a}\nabla_{b}\left(\frac{\alpha_{1}}{T}\right)\right]$$
$$-\frac{1}{T}\chi^{ab}\left[\nabla_{a}u_{b} + \beta_{2}u^{c}\nabla_{c}\chi_{ab} - \alpha_{1}\nabla_{a}q_{b} + \frac{T}{2}\chi_{ab}\nabla_{c}\left(\frac{\beta_{2}u^{c}}{T}\right) - \gamma_{1}Tq_{a}\nabla_{b}\left(\frac{\alpha_{1}}{T}\right)\right].$$
(2.27)

Let us note that, following [114] we included two additional parameters γ_0 , γ_1 because of the freedom we have in distributing the mixed quadratic terms. Consistency with the thermodynamic second law may then be enforced as in the previous subsection by assuming the entropy production rate is a sum of quadratic terms. This in turn yields the equation of motion for the dissipative fluxes and the system of equations is

¹¹It is quite common to write down the equations of the second-order theories in the Eckart frame, namely set the diffusion vector v^a to vanish. As this simplifies a little the expressions that follow, we will stick to this convention.

¹²Actually, it may be more correct to view/present the EIT program as an attempt to systematize the ideas behind Müller-Israel-Stewart theory.

now determined. Not surprisingly, they look like the general relativistic version of the Cattaneo laws

$$\tau_{\rm b}\dot{\chi} + \chi = -\zeta[\dots] , \qquad (2.28a)$$

$$\tau_{\rm s} \dot{\chi}_{ab} + \chi_{ab} = -2\eta [\dots] , \qquad (2.28b)$$

$$\tau_{\rm h}\dot{q}_a + q^a = -\kappa T \perp^{ab} [\dots]_b , \qquad (2.28c)$$

where we have introduced three different relaxation timescales τ_b , τ_s , τ_h —which can be related to the parameters introduced above—and the "dot" stands for the proper time derivative associated with the Eckart frame 4-velocity: $\dot{A} = u^a \nabla_a A$. We refer to, for example, [21, 184] for explicit expressions of the terms we omitted in square brackets. We nonetheless anticipate that they include, as one may expect, the first order forces namely the thermodynamic forces of the first order theories, such as the expansion rate for the bulk-viscous scalar—and additional terms quadratic in the fluxes themselves.

The Müller-Israel-Stewart (MIS) model has been proven to overcome the stability and causality issues of the first-order theories [114]: it possesses stable equilibrium states, and deviations from these states propagate causally¹³. However, a number of other issues remain to be addressed.

First, the Müller-Israel-Stewart model is based on an implicit expansion in deviations away from thermal equilibrium, and stability/causality are guaranteed only for the linearized system of equations. The non-linear behaviour is a completely different game, and not well-explored. One exception is the analysis of Hiscock and Lindblom [116], which explores the presence of non-linear pathologies (in an extremely simplified case), even though this relates to such an extreme regime that it may not be relevant for any physical or astrophysical application. When it comes to causality in the non-linear regime, whilst there have been recent progress for the bulk-viscous case [41], a definite answer is still missing. Second, from a field-theory perspective, the "second-order" expansion of the MIS model cannot be considered complete. Even though the dissipative terms are based on kinetic theory, the model contains only squares of first-order "thermodynamic fluxes" (as in the sense of Onsager [166]) in all possible combinations. Last, and maybe most importantly, the equations of motion are obtained from the conservation of the total stress-energy-momentum tensor of the system, and it is not clear how to extend the model to multi-fluid systems.

2.2.3 Liu-Müller-Ruggeri and divergence-type theories

As we have discussed above, the MIS model represents a significant improvement over the first order theories. Nonetheless, there is another open question regarding the MIS

¹³This can be intuitively understood from the Cattaneo-type form of the equations for the fluxes.

model we have not discussed, and we will briefly touch upon it now. Local wellposedness and strong hyperbolicity are not firmly established¹⁴. Motivated by this and the quest for a theory with more solid mathematical foundations, Müller and Ruggeri [161] have proposed a class of models (then slightly generalized by Geroch and Lindblom [91, 92]) known as divergence-type theories (see Rezzolla and Zanotti [184] and the review by Salazar and Zannias [190] for more details). This class of theories is based on three fundamental principles: i) Principle of Relativity ii) Maximum Entropy Principle iii) Hyperbolicity. In practice, the formal hydrodynamic equations are

$$\nabla_a n^a = 0 , \qquad (2.29a)$$

$$\nabla_a T^{ab} = 0 , \qquad (2.29b)$$

$$\nabla_a A^{abc} = I^{bc} , \qquad (2.29c)$$

where $A^{abc} = A^{a(bc)}$, $A^{ab}_{b} = 0$, $I^{bc} = I^{(bc)}$, $I^{a}_{a} = 0$ and the system is closed assuming A^{abc} , I^{bc} are functions of n^{a} , T^{ab} . The first two equations represent the conservation of particle flux and stress-energy-momentum tensor as before, while A^{abc} is supposed to be the third moment of the one particle distribution function of some underlying relativistic kinetic theory (see [141, 211, 61]). Similarly I^{bc} should represent an approximation to the second moment of the collisional integral of some underlying kinetic theory model. In addition, this class of theories must be completed by an entropy current s^{a} , again considered as function of n^{a} , T^{ab} .

The number of theories that can be formulated as in eq. (2.29) is obviously quite large, and one can see, for example, that the set contains both the Eckart-Landau-Lifschitz and the Müller-Israel-Stewart models. It is fair to say, however, that the number of acceptable theories can be somewhat reduced using constraints coming from the principles stated above—although to do this in practice one often assumes A^{abc} and I^{bc} are linear in the dissipative fluxes. While the final aim is that of constructing a framework in which discussing issues like stability, causality and hyperbolicity becomes relatively simple, this is (to the best of our knowledge) very much a work in progress.

Before we move on, it is worth mentioning an important result that was derived independently by Geroch [90] and Lindblom [140]. They argued that fluid states predicted by the causal divergence-type theories decay on very short timescales—of the order of the characteristic timescale of microscopic particle interactions—to ones that are welldescribed by the Eckart-Landau-Lifschitz model. In essence, the results suggest that while the first-order theories are non causal, unstable and not well-posed—and hence problematic for numerical applications—their physical predictions would be practically/experimentally indistinguishable from those of second order ones.

¹⁴The bulk-viscous MIS model has been shown to be weakly hyperbolic only recently in [41].

2.3 Variational models

In this section we review variational approaches to model dissipative multi-fluids in General Relativity. Most of these models are built on extending the action-based model for non dissipative multi-fluids first championed by Taub [209] and then developed by Carter and collaborators (see Carter [52], Comer and Langlois [67, 68], Carter and Langlois [54] and Andersson and Comer [21] for an up-to-date and pedagogical review). We start with a summary of the variational principle for non dissipative multi-fluids.

2.3.1 Non dissipative multi-fluids models

To model general relativistic multi-fluids systems we start from an action of the form

$$S = \int d^4x \sqrt{-g} (R + \Lambda) , \qquad (2.30)$$

where *R* is the Ricci scalar and the so-called "master function" Λ accounts for the matter content of the theory. To model a multi-fluid system with different chemical species (or constituents) labelled by x, y..., we take the master function Λ to depend on the particle fluxes n_x^a . Assuming the system to be isotropic, the master function is considered as a function of all the possible scalars that can be constructed from the fluxes n_x^a and the spacetime metric: $\Lambda = \Lambda(n_x^2, n_{xy}^2)$, where $n_x^2 = -n_x^a n_a^x$ and $n_{xy}^2 = -n_x^a n_a^y$. Then, performing the variation of the master function we obtain¹⁵

$$\delta\left(\sqrt{-g}\Lambda\right) = \sqrt{-g}\left[\sum_{\mathbf{x}}\mu_a^{\mathbf{x}}\delta n_{\mathbf{x}}^a + \frac{1}{2}\left(\Lambda g^{ab} + \sum_{\mathbf{x}}n_{\mathbf{x}}^a\mu_{\mathbf{x}}^b\right)\delta g_{ab}\right],\qquad(2.31)$$

where we have introduced the particle four-momenta

$$\mu_a^{\rm x} = \frac{\partial \Lambda}{\partial n_{\rm x}^a} = \mathcal{B}^{\rm x} n_a^{\rm x} + \sum_{\rm y \neq \rm x} \mathcal{A}^{\rm xy} n_a^{\rm y} , \qquad (2.32)$$

and

$$\mathcal{B}^{\mathrm{x}} = -2\frac{\partial\Lambda}{\partial n_{\mathrm{x}}^{2}},\qquad(2.33)$$

while the entrainment coefficients are defined as

$$\mathcal{A}^{\mathrm{xy}} = -\frac{\partial \Lambda}{\partial n_{\mathrm{xy}}^2} \,. \tag{2.34}$$

By inspecting eq. (2.31) we immediately learn two things. First, we see that the variational approach automatically accounts for the entrainment effect. Roughly speaking, entrainment is a non-dissipative interaction between the species and causes a species'

¹⁵Hereafter we ignore boundary terms unless they become relevant to the discussion.



FIGURE 2.1: The pull-back from a point in the x^{th} matter space to the corresponding spacetime worldline. The points in matter space are labelled by X_x^A with A = 1, 2, 3. Figure taken from Andersson and Comer [21].

four-momentum μ_a^x to be misaligned with its respective particle flux n_x^a . Entrainment was first recognized as an important dynamical effect in superfluid mixtures by Andreev and Bashkin [27] and, from a microphysical perspective, is akin to the notion of effective masses (gained by the electrons, say, when they move past a ion lattice). Second, we learn that an unconstrained variation on the fluxes gives rise to trivial equations of motion. In fact, the fluid equations that follow from eq. (2.31) are $\mu_a^x = 0$. This is a well-established result: to obtain non-trivial fluid equations of motion from a Lagrangian, the variation of the particle fluxes must be constrained [196, 52, 175].

A particularly elegant way of imposing the relevant constraint involves introducing the matter space, defined by identifying each currents worldline as a single point [55], see fig. 2.1 for an illustration of the idea. For each fluid, the matter space is a threedimensional manifold, so that when we introduce a set of coordinates X_x^A on, say, the x-fluid's matter space, we attach a "name", or label, to each fluid element. Because the entire worldline of each fluid element is mapped to a single matter space point, it is clear that the fluid element's label X_x^A , now considered as a collection of three scalars on spacetime, takes the same value at each point on the worldline. After assigning a label to each fluid element worldline, we can use the linear map Ψ_{xa}^A , defined as

$$\Psi^{A}_{\mathbf{x}a} \doteq \frac{\partial X^{A}_{\mathbf{x}}}{\partial x^{a}} , \qquad (2.35)$$

to push-forward (pull-back) vectors (co-vectors) between spacetime and the matter spaces. This is important because we can associate with each of the particle fluxes n_x^a a three-form n_{abc}^x by the standard Hodge-dual procedure:

$$n_{\rm x}^a = \frac{1}{3!} \varepsilon^{bcda} n_{bcd}^{\rm x} , \quad n_{abc}^{\rm x} = \varepsilon_{eabc} n_{\rm x}^e .$$
(2.36)

Now we can assume that the spacetime three-form n_{abc}^{x} is obtained by pulling back a corresponding matter space three-form, to be denoted n_{ABC}^{x} ; namely,

$$n_{abc}^{x} = \Psi_{x[a}^{A} \Psi_{xb}^{B} \Psi_{xc]}^{C} n_{ABC}^{x} , \qquad (2.37)$$

where, as usual, straight brackets indicate anti-symmetrization (and round ones symmetrization). Similarly, upon applying the Hodge-dual to the four-momentum μ_a^x , we can push-forward with the map and identify a matter space momentum "three-form" μ_x^{ABC} via

$$\mu_{\rm x}^{abc} = \varepsilon^{dabc} \,\mu_d^{\rm x} \,, \tag{2.38a}$$

$$\mu_{\rm x}^{ABC} = \Psi_{{\rm x}[a}^{A} \Psi_{{\rm x}b}^{B} \Psi_{{\rm x}c]}^{C} \, \mu_{\rm x}^{abc} \,. \tag{2.38b}$$

The main idea of the convective variational principle is to obtain the particle flux variation δn_x^a by first varying the matter-space three-form and then working backwards.

Generally speaking, there are two ways of tracking changes in a fluid system—Eulerian and Lagrangian. The first, to be denoted by a δ , measures changes in the fluid at fixed spacetime coordinates. The second, to be denoted Δ_x , measures changes following the motion of fluid elements. Locally, the two are related through the Lie derivative along some displacement vector field ξ_x^a as¹⁶

$$\Delta_{\rm x} = \delta + \mathcal{L}_{\xi_{\rm x}} , \qquad (2.39)$$

where \mathcal{L}_{ξ_x} is the Lie derivative with respect to ξ_x^a . Because the label X_x^A of a fluid element is fixed, we can assert

$$\Delta_{\mathbf{x}} X_{\mathbf{x}}^{A} = 0 \implies \delta X_{\mathbf{x}}^{A} = -\mathcal{L}_{\xi_{\mathbf{x}}} X_{\mathbf{x}}^{A} = -\Psi_{\mathbf{x}a}^{A} \xi_{\mathbf{x}}^{a} .$$
(2.40)

Now, it is easy to show that the particle flux variation δn_x^a is (see [21])

$$\delta n_{\mathrm{x}}^{a} = -\frac{1}{2} n_{\mathrm{x}}^{a} g^{bc} \delta g_{bc} - \frac{1}{3!} \varepsilon^{bcda} \mathcal{L}_{\xi_{\mathrm{x}}} n_{bcd}^{\mathrm{x}} \,. \tag{2.41}$$

As a result of eq. (2.41) we can write the *constrained* variation of the master function as

$$\delta\left(\sqrt{-g}\Lambda\right) = \sqrt{-g} \left[\frac{1}{2} \left(\Psi g^{ab} + \sum_{x} n_{x}^{a} \mu_{x}^{b}\right) \delta g_{ab} - \sum_{x} \left(f_{a}^{x} + \Gamma_{x} \mu_{a}^{x}\right) \xi_{x}^{a}\right], \quad (2.42)$$

¹⁶We note that this relation between Lagrangian and Eulerian variation works only to first order in the perturbation fields ξ_{xx}^a see Friedman and Schutz [83] for further details.

where we introduced the "generalized pressure" Ψ (not to be confused with the map introduced earlier)

$$\Psi = \Lambda - \sum_{\mathbf{x}} n_{\mathbf{x}}^{a} \mu_{a}^{\mathbf{x}} , \qquad (2.43)$$

while the force densities and creation rates (for each species)

$$f_a^{\mathbf{x}} = 2n_{\mathbf{x}}^b \nabla_{[b} \mu_{a]}^{\mathbf{x}} , \quad \Gamma_{\mathbf{x}} = \nabla_a n_{\mathbf{x}}^a .$$
(2.44)

Since the particle-flux three-forms n_{abc}^{\times} are pulled-back from the matter space, they are automatically closed—because n_{ABC}^{\times} is a three-form on a three dimensional matter space and the pull-back operation commutes with the exterior derivative. As a result the constrained variation gives zero creation rate for each particle-flux¹⁷:

$$\Gamma_{\rm x} = \nabla_a n_{\rm x}^a = \frac{1}{3!} \varepsilon^{bcda} \nabla_{[a} n_{bcd]}^{\rm x} = \frac{1}{4!} \varepsilon^{bcda} (dn)_{abcd} = 0 .$$
 (2.45)

Therefore, the term proportional to Γ_x actually drops out of eq. (2.42). Still, it is interesting to observe that such terms are formally present in the fluid equations.

We conclude this subsection by observing that the constrained variational principle also gives the total fluid stress-energy-momentum tensor as

$$T^{ab} = \Psi g^{ab} + \sum_{x} n^{a}_{x} \mu^{b}_{x} , \qquad (2.46)$$

and that it follows as an identity that

$$\nabla_a T^{ab} = \sum_{\mathbf{x}} f_a^{\mathbf{x}} = 0 , \qquad (2.47)$$

where the last equivalence is ultimately a consequence of the second Bianchi identity satisfied by the Riemann tensor. As a final point, let us go back to discuss an issue we briefly hinted at in section 1.3. Working with the multi-fluid framework, we can in fact appreciate the difference between equations of motion ($f_a^x = 0$) and energy-momentum conservation laws. In particular, the procedure is built in such a way that we automatically obtain as many equations as needed, whatever the number of particle fluxes is.

Before we move on to discuss extensions of the variational framework to the modelling of dissipative fluids, let us briefly comment on the case of massless particles, i.e. radiation. Our discussion of the variational model relies heavily on the notion of matter space, which we have introduced by "assigning a label" to each fluid element worldline. Next, by associating the fluid elements' worldlines to those of the particles in the system, we have used the matter space construction to automatically impose particles' number conservation (for each species separately), as appropriate for

¹⁷Here dn represents the exterior derivative of the differential form n.

a multi-fluid non-dissipative model. This logic obviously appears problematic if we consider, say, photons since their number is not conserved. At the same time, however, it is not clear whether modelling radiation as a fluid is the right thing to do in the first place. An acceptable description of radiation should in fact be able to model both the "trapped" regime—where the photons, say, interact sufficiently often that their meanfree-path is small enough and we can meaningfully introduce a notion of fluid element (see section 3.1.1)—and the "free-streaming" regime—where the photons do not interact often and are able to escape freely—as well as the transition between the two. As such, it appears to us that to model radiation it is best to start from a more fundamental approach, namely relativistic kinetic theory (see, e.g. [61, 184]). In particular, one can start from the relativistic Boltzmann equation and derive an infinite hierarchy of equations for the moments of the one-particle distribution function associated with the radiation field [212]. In practice, this hierarchy needs to be truncated at some level as we cannot work with or simulate an infinite set of equations. In particular, it is quite common to stop at second order (see, e.g. [183, 222, 178, 162]). This means that only the first two moments of the distribution function are evolved, and leads to the equations of radiation hydrodynamics (see, e.g. [153]). Whilst the end-result resembles a "fluid model" for radiation, the underlying theory is much more detailed, and hence better suited to describe radiation in the first place.

2.3.2 Carter-like dissipative models

We now describe two dissipative extensions of the variational model we just discussed. We start with a model for an heat-conducting medium and then move on to briefly discuss a variational model proposed by Carter.

Andersson-López-Monsalvo model for relativistic heat conduction

Here we will briefly describe of a model for an heat-conducting medium developed in Lopez-Monsalvo and Andersson [142] and Andersson and Lopez-Monsalvo [22]. The model linchpin is to include the heat flux through an entropy current s^a that can flow differently from matter n^a . As a result, the variational approach to multifluids is a natural starting point. This model is substantially a "correction" of a previous attempt by Carter (see Carter [51]). In his model Carter set to zero the entrainment between entropy and matter. This turns out to have a significant impact as the model without entrainment has been shown to violate causality (see Olson and Hiscock [165]). The take home message is that entrainment between entropy and matter is a fundamental ingredient in the description of an heat-conducting fluid—through entrainment the entropy current gains an effective mass and this results in an inertial heat response. The model starting point is the non-dissipative variational principle for a two-fluid system where the particle fluxes are n^a —that represents matter particles—and an entropy current s^a —which can be thought of as a gas of thermal excitations. Using the results of the previous subsection we can write the force densities as

$$f_a^{\mathbf{n}} = 2n^b \nabla_{[b} \mu_{a]} + \mu_a \nabla_b n^b , \qquad (2.48a)$$

$$f_a^{\rm s} = 2s^b \nabla_{[b} \theta_{a]} + \theta_a \nabla_b s^b , \qquad (2.48b)$$

where—following [142]—we named the matter particles and entropy 4-momenta (respectively) μ_a , θ_a . As we already pointed out earlier, even though the constrained variation is built in such a way that the particle fluxes are automatically conserved, the action-based force densities contain a term proportional to Γ_s or Γ_n . As a result we can build an hybrid model in the sense that the form of the force densities—and stressenergy-momentum-tensor—stems from the variational principle and, at the same time, we let the particle creation rates differ from zero. Because there is just one matter particle flux, we require $\Gamma_n = \nabla_b n^b = 0$ while $\Gamma_s = \nabla_b s^b \ge 0$ to retain consistency with thermodynamics. It is then convenient to work in the Eckart frame, that is we introduce the matter 4-velocity u^a such that $n^a = nu^a$ while for the entropy current we have

$$s^a = s^* (u^a + w^a) , (2.49)$$

where s^* is the entropy density measured by the matter particles. We also introduce a similar decomposition for the entropy 4-momentum

$$\theta_a = \left(\mathcal{B}^{\mathrm{s}}s^* + \mathcal{A}^{\mathrm{ns}}n\right)u_a + \mathcal{B}^{\mathrm{s}}s^*w_a \doteq \theta^*u_a + p_a , \qquad (2.50)$$

and then rewrite the heat-flux as

$$q_a = -\perp_{ab} u_c T^{bc} = s^* \theta^* w_a \doteq \theta^* \sigma_a .$$
(2.51)

Next, the energy density measured by matter particles is $\varepsilon = u_a u_b T^{ab} = -\Lambda + p_a \sigma^a$. Therefore, when the system is out of equilibrium and the heat flows relative to matter the energy density depends also on the heat-flux (encoded in the variables σ^a and p_a), i.e. we have an *extended Gibbs relation*

$$d\varepsilon = \mu dn + \theta^* ds^* + \sigma^a dp_a . \qquad (2.52)$$

We stress that this result arises automatically in the model and is consistent with the EIT picture—the key difference is that it is derived, not postulated. Finally, it is possible to rewrite the equation of motion for the entropy current as an equation for the heat-flux as

$$\tau \left(\dot{q}^a + q_c \nabla^a u^c \right) + q^a = \kappa \perp^{ab} \left(\nabla_b \theta^* + \theta^* a_b \right) , \qquad (2.53)$$

where the relaxation timescale τ can be rewritten in terms of the entrainment parameter

 A_{ns} and κ is the heat conductivity. The last equation is clearly a general relativistic generalization of the Cattaneo law (see eq. (2.12)) and has been shown to be consistent with the Israel-Stewart model in the linear regime (see [142]).

Carter variational principle for dissipative fluids

Another important step forward, at least from the formal point of view, is the variational model proposed by Carter. We now sketch the main ideas and assumptions of the model, while referring to the original work for the details [53]. We do so mainly to highlight the differences with what comes next.

The key quantity is again the master function Λ which now is assumed to depend also on a set of additional rank-2 (symmetric) tensors τ_{Σ}^{ab} . These new dynamical fields should be identified with viscous tensors and the label Σ is introduced to allow for different sources of viscosity separately—such as bulk and shear viscosity. The variation of the master function $\Lambda(n_x^a, \tau_{\Sigma}^{ab}, g_{ab})$ can then be written as

$$\delta\Lambda = \sum_{x} \mu_{a}^{x} \delta n_{x}^{a} + \frac{1}{2} \sum_{\Sigma} \frac{\partial \Lambda}{\partial \tau_{\Sigma}^{ab}} \delta \tau_{\Sigma}^{ab} + \frac{\partial \Lambda}{\partial g_{ab}} \delta g_{ab} .$$
(2.54)

In this approach, however, the action is only used to obtain the structure of the force terms, and the stress–energy–momentum conservation as a Noether identity. The final equations of motion are not obtained by setting to zero the variation of the action, but enforcing consistency with the second law of thermodynamics. Moreover, the identification of the new dynamical fields τ_{Σ}^{ab} with viscous tensors is so far just formal. To complete the identification with the usual thermodynamic fluxes a specific expansion in deviations away from thermal equilibrium had to be introduced, and the resulting model has been shown to belong to the same family as those of the MIS variety [174].

2.3.3 Andersson and Comer formalism

We now describe a recent action-based formalism for dissipative (general relativistic) multi-fluids proposed by Andersson and Comer [20], which is built by generalizing the non-dissipative model presented in section 2.3.1. As this is the starting point for the original results described in chapter 3, we will spend more time going into the details. We also note two important aspects of the model that distinguish it from the ones presented in the previous subsections. First, the approach is "fully variational" in the sense that the final equations of motion are obtained as Euler–Lagrange equations starting from an action—while the models discussed in the earlier section took the variational equations as starting point, and then modified them appropriately. This makes

the model well suited for describing dissipative multi-fluids¹⁸. Second, the model does not introduce any new dynamical field, focusing instead on the particle fluxes n_x^a .

In order to not get lost in the algebra, it is useful to start with two simple observations. First, at the microscopic level dissipation is the product of interaction between particles, which, at the fluid-dynamical level, intuitively translates into the idea of *interacting matter spaces*. Second, a central feature of dissipative fluids is having non-conserved fluxes, $\Gamma_x \neq 0$. As can be seen going back to eq. (2.45), flux conservation is—in the constrained variation—a direct consequence of having an associated closed three-form n_{abc}^x . As a result, if we want to keep working with the matter space construction and, at the same time, let the particle fluxes be non-conserved, we have to break the closure property of the three-forms.

Let us implement these two ideas by reviewing the constrained variation procedure. Clearly, we still have that the particle labels do not change if we follow them (in the Lagrangian sense) $\Delta_x X_x^A = 0$. Also, it is still true that

$$\Delta_{\mathbf{x}} \Psi_{\mathbf{x}a}^{A} = \Delta_{\mathbf{x}} \left(\frac{\partial X_{\mathbf{x}}^{A}}{\partial x^{a}} \right) = \frac{\partial}{\partial x^{a}} \left(\Delta_{\mathbf{x}} X_{\mathbf{x}}^{A} \right) = 0 .$$
 (2.55)

We can now work out (again) the particle flux variation δn_x^a without assuming the threeform n_{ABC}^x to be closed to get

$$\delta n_{\mathbf{x}}^{a} = -\frac{1}{2} n_{\mathbf{x}}^{a} g^{bc} \delta g_{bc} - \frac{1}{3!} \varepsilon^{bcda} \left(\mathcal{L}_{\xi_{\mathbf{x}}} n_{bcd}^{\mathbf{x}} - \Psi_{\mathbf{x}}^{B} [{}_{b} \Psi_{\mathbf{x}c}^{C} \Psi_{\mathbf{x}d}^{D}] \Delta_{\mathbf{x}} n_{BCD}^{\mathbf{x}} \right) \,. \tag{2.56}$$

But there is a deeper point to be made here. Formally, we can take n_{ABC}^{x} to be a particle measure form on the matter space, which "counts" the total number of species x particles in the system. If it is a tensor on matter space then it must be a function only of the matter space coordinates X_x^A . The fact that $n_{ABC}^x = n_{ABC}^x(X_x^A)$ implies $\Delta_x n_{ABC}^x = 0$, and the flux variation above reduces to the result for non-dissipative fluids. Therefore, to get the non-dissipative equations of motion one simply has to impose that the number of particles is conserved in the variation, or, equivalently, that the particle creation rates $\Gamma_x = \nabla_a n_x^a$ vanish. It then follows that a way to include dissipative processes (read: $\Gamma_x \neq 0$) at the level of the action principle is to break the tensorial nature of the matter space particle measure form n_{ABC}^x , and allow it to be a function of more than just the X_x^A . In other words, we are breaking the closure property of the n_{abc}^x .

Before we move on to discuss the model more in detail, let us briefly comment on the differences between dissipative fluids and radiation. Using the variational approach to model radiation seems problematic as the photon number is not conserved—although we argued in section 2.3.1 that using a fluid-scheme to model radiation appears too

¹⁸As opposed to simple fluids with, possibly, heat flow, which the standard approaches are designed for.

restrictive in the first place. As the matter space is introduced by labelling the worldlines, and we naively tend to associate the worldlines to the particles, one may be led to think that using the matter space construction to model a dissipative system is similarly problematic. We now consider a multi-fluid system undergoing reactions to argue that this is not quite the case. First of all, let us begin by recalling that all fluid models inevitably involve some notion of averaging (see also section 4.2). As such, it is probably more correct to think of the worldlines as associated with fluid elements-and we are going to discuss more in detail how these are defined in section 3.1.1—rather than the individual particles themselves. Moreover, while the system is undergoing reactions, it is still subject to, say, the conservation of baryons, leptons and so on. In essence, one could imagine some notion of a "global" matter space that, in the nondissipative limit factorizes as the direct sum of the individual matter spaces associated with each species. While the factorization in terms of matter spaces associated with the different species breaks down in the dissipative case, and one may be tempted to think of factorizing it in terms of baryons, leptons and so on, it still makes sense to consider fluid elements associated with the various species. The reason being that the dynamical behaviour of each of these is different—as they have, say, different charge and hence behave differently when immersed in a magnetic field. Admittedly, this discussion is somewhat hand-waving, and we are urged to make it more precise—which motivates the analysis and new results presented in chapter 3. Before we get there though, let us see where this idea brings us by considering two explicit examples.

Following the discussion in the original paper, we assume that the x-particle three-form depends also on the matter space coordinates of the y-matter spaces X_y^A . As a result we have

$$\Delta_{\mathbf{x}} n_{ABC}^{\mathbf{x}} = \sum_{\mathbf{y} \neq \mathbf{x}} \frac{\partial n_{ABC}^{\mathbf{x}}}{\partial X_{\mathbf{y}}^{D}} \Delta_{\mathbf{x}} X_{\mathbf{y}}^{D} = \sum_{\mathbf{y} \neq \mathbf{x}} \frac{\partial n_{ABC}^{\mathbf{x}}}{\partial X_{\mathbf{y}}^{D}} \left(\delta X_{\mathbf{y}}^{D} - \mathcal{L}_{\xi_{\mathbf{x}}} X_{\mathbf{y}}^{D} \right)$$
$$= \sum_{\mathbf{y} \neq \mathbf{x}} \frac{\partial n_{ABC}^{\mathbf{x}}}{\partial X_{\mathbf{y}}^{D}} \left(\xi_{\mathbf{x}}^{a} - \xi_{\mathbf{y}}^{a} \right) \partial_{a} X_{\mathbf{y}}^{D} .$$
(2.57)

Using the last equation in eq. (2.56) and defining a "resistivity coefficient" as

$$\mathbf{R}_{a}^{\mathrm{xy}} = \frac{1}{3!} \mu_{\mathrm{x}}^{ABC} \frac{\partial n_{ABC}^{\mathrm{x}}}{\partial X_{\mathrm{y}}^{D}} \, \Psi_{\mathrm{y}a}^{D} \,, \tag{2.58}$$

it is possible to write the fluid-part in the Lagrangian variation as

$$\mu_a^{\mathbf{x}} \delta n_{\mathbf{x}}^a = \text{``non-dissipative terms''} + \sum_{\mathbf{y} \neq \mathbf{x}} \mathbf{R}_a^{\mathbf{xy}} \left(\xi_{\mathbf{y}}^a - \xi_{\mathbf{x}}^a \right) \,. \tag{2.59}$$

The additional piece in the variation then changes the equations of motion to

$$f_a^{\rm x} + \Gamma_{\rm x} \mu_a^{\rm x} = \sum_{y \neq {\rm x}} \left({\rm R}_a^{\rm yx} - {\rm R}_a^{\rm xy} \right) ,$$
 (2.60)

where f_a^x is the same as in section 2.3.1. The take home message from this example is that by enlarging the set of quantities which the three-form can depend on we obtain additional terms that look like resistivity coefficients.

The natural follow-up question then is: can we perform a similar calculation and obtain, again from an action principle, additional terms in the equations that look like viscous tensors? The answer—demonstrated by Andersson and Comer [20]—is yes. Let us, in fact, consider a situation where the three form depends both on X_y^A and the projected metric

$$g_x^{AB} = \Psi_{xa}^A \Psi_{xb}^B g^{ab} .$$

We then have

$$\Delta_{\mathbf{x}} n_{ABC}^{\mathbf{x}} = \sum_{\mathbf{y} \neq \mathbf{x}} \frac{\partial n_{ABC}^{\mathbf{x}}}{\partial X_{\mathbf{y}}^{D}} \Delta_{\mathbf{x}} X_{\mathbf{y}}^{D} + \frac{\partial n_{ABC}^{\mathbf{x}}}{\partial g_{\mathbf{x}}^{DE}} \Delta_{\mathbf{x}} g_{\mathbf{x}}^{DE} .$$
(2.62)

The novel terms arising from $\Delta_{\mathbf{x}} g_{\mathbf{x}}^{DE}$ in the fluid-part of the Lagrangian variation then read

$$\mu_{a}^{x}\delta n_{x}^{a} = \dots + \frac{1}{3!}\mu_{x}^{ABC}\frac{\partial n_{ABC}^{x}}{\partial g_{x}^{DE}}\Delta_{x}g_{x}^{DE}$$

$$= \dots + \frac{1}{3!}\mu_{x}^{ABC}\frac{\partial n_{ABC}^{x}}{\partial g_{x}^{DE}}\Psi_{xa}^{D}\Psi_{xb}^{E}\left(\delta g^{ab} - 2\nabla^{(a}\xi_{x}^{b)}\right)$$

$$= \dots + \frac{1}{2}S_{x}^{ab}\delta g_{ab} - S_{ab}^{x}\nabla^{b}\xi_{x}^{a}, \qquad (2.63)$$

where we have defined the "viscosity tensor"

$$S_{ab}^{\mathbf{x}} = \frac{1}{3} \mu_{\mathbf{x}}^{ABC} \frac{\partial n_{ABC}^{\mathbf{x}}}{\partial g_{\mathbf{x}}^{DE}} \Psi_{\mathbf{x}a}^{D} \Psi_{\mathbf{x}b}^{E} , \qquad (2.64)$$

and the ... represent the non-dissipative terms plus the ones in eq. (2.59). From eq. (2.63) we the intuitively see that the additional dependence of the matter space three-form is going to affect both the force densities and the total stress-energy tensor of the system. Cranking through the algebra, the action variation can be written as

$$\delta\left(\sqrt{-g}\Lambda\right) = \sqrt{-g} \left[\frac{1}{2} \left(\Psi g^{ab} + \sum_{\mathbf{x}} n_{\mathbf{x}}^{a} \mu_{\mathbf{x}}^{b} + S_{\mathbf{x}}^{ab}\right) \delta g_{ab} - \sum_{\mathbf{x}} \left(f_{a}^{\mathbf{x}} + \Gamma_{\mathbf{x}} \mu_{a}^{\mathbf{x}} + \nabla^{b} S_{ba}^{\mathbf{x}} - \sum_{\mathbf{y} \neq \mathbf{x}} \left(\mathbf{R}_{a}^{\mathbf{yx}} - \mathbf{R}_{a}^{\mathbf{xy}}\right)\right) \xi_{\mathbf{x}}^{a}\right].$$
(2.65)

Therefore, the action-based model provides us with the total stress-energy-momentum tensor

$$T^{ab} = \Psi g^{ab} + \sum_{x} n^{a}_{x} \mu^{b}_{x} + S^{ab}_{x} , \qquad (2.66)$$

which contains the additional viscous stress-tensors S_x^{ab} . Also the resulting fluid equations of motion in this second case are

$$f_{a}^{x} + \Gamma_{x}\mu_{a}^{x} + \nabla^{b}S_{ba}^{x} = \sum_{y \neq x} \left(R_{a}^{yx} - R_{a}^{xy} \right) , \qquad (2.67)$$

and thus contain both a resistivity term and the four divergence of the new viscous tensor.

Having described these two examples, let us move on to the general formalism presented in Andersson and Comer [20]. The authors consider the case where the particle three-forms depend on

$$n_{ABC}^{x} = n_{ABC}^{x}(X_{x}^{A}, X_{y}^{A}, g_{x}^{AB}, g_{y}^{AB}, g_{xy}^{AB}), \qquad (2.68)$$

and the "mixed projected metrics" are defined as

$$g_{xy}^{AB} = \Psi_{xa}^{A} \Psi_{yb}^{B} g^{ab} .$$
 (2.69)

Performing the variations as in the examples above they arrive at the following equations of motion

$$f_a^{\mathbf{x}} + \Gamma_{\mathbf{x}} \mu_a^{\mathbf{x}} + \nabla^b D_{ba}^{\mathbf{x}} = R_a^{\mathbf{x}} , \qquad (2.70)$$

where

$$D_{ab}^{x} = S_{ab}^{x} + \sum_{y \neq x} s_{ab}^{yx} + \frac{1}{2} \left(S_{ba}^{xy} + S_{ab}^{yx} \right) , \qquad (2.71a)$$

$$R_a^{\mathbf{x}} = \sum_{\mathbf{y} \neq \mathbf{x}} \left(\mathbf{R}_a^{\mathbf{y}\mathbf{x}} - \mathbf{R}_a^{\mathbf{x}\mathbf{y}} \right) + \left(r_a^{\mathbf{y}\mathbf{x}} - r_a^{\mathbf{x}\mathbf{y}} \right) + \left(\mathcal{R}_a^{\mathbf{y}\mathbf{x}} - \mathcal{R}_a^{\mathbf{x}\mathbf{y}} \right).$$
(2.71b)

and

$$s_{ab}^{xy} = \frac{1}{3} \mu_x^{ABC} \frac{\partial n_{ABC}^x}{\partial g_y^{DE}} \Psi_{ya}^D \Psi_{yb}^E , \qquad (2.72a)$$

$$S_{ab}^{xy} = \frac{1}{3} \mu_x^{ABC} \frac{\partial n_{ABC}^x}{\partial g_{xy}^{DE}} \Psi_{xa}^D \Psi_{yb}^E , \qquad (2.72b)$$

$$r_a^{\rm xy} = \frac{1}{3!} \mu_{\rm x}^{ABC} \frac{\partial n_{ABC}^{\rm x}}{\partial g_{\rm y}^{DE}} \nabla_a \left(g^{bc} \Psi_{\rm yb}^D \Psi_{\rm yc}^E \right) , \qquad (2.72c)$$

$$\mathcal{R}_{a}^{\mathrm{xy}} = \frac{1}{3!} \mu_{\mathrm{x}}^{ABC} \frac{\partial n_{ABC}^{\mathrm{x}}}{\partial g_{\mathrm{xy}}^{DE}} g^{bc} \Psi_{\mathrm{x}b}^{D} \nabla_{a} \left(\Psi_{\mathrm{y}c}^{E} \right) , \qquad (2.72d)$$

while R_a^{xy} , S_{ab}^x are defined as in eqs. (2.58) and (2.64). Projecting the field equation along $u_x^a = n_x^a/n_x$, we see that

$$\left(-u_{\mathrm{x}}^{a}\mu_{a}^{\mathrm{x}}\right)\Gamma_{\mathrm{x}}=u_{\mathrm{x}}^{a}\nabla^{b}D_{ba}^{\mathrm{x}}-u_{\mathrm{x}}^{a}R_{a}^{\mathrm{x}},\qquad(2.73)$$

while the stress-energy-momentum tensor is

$$T^{ab} = \Psi g^{ab} + \sum_{x} n^{a}_{x} \mu^{b}_{x} + D^{ab} , \qquad (2.74)$$

and $D^{ab} = \sum_{x} D_x^{ab}$ is the sum of each species' total viscous tensor. Let us observe that it follows, as an identity, that $\sum_{x} R_a^x = 0$, and because of this we have automatically $\nabla_a T^{ab} = 0$. We also note that the "resistive terms" r_a^{xy} , \mathcal{R}_a^{xy} as well as the viscous tensors s_{ab}^{xy} , \mathcal{S}_{ab}^{xy} arise because we assume that n_{ABC}^x depends on g_y^{AB} and g_{xy}^{AB} , respectively. Finally, it is easy to see that, in general, the x-species total viscous tensor D_{ab}^x is not necessarily symmetric because \mathcal{S}_{ab}^{xy} is not. This property is, however, not inherited by the total viscous tensor of the system meaning $D_{ab} = \sum_{x} D_{ab}^x = D_{ba}$.

Let us conclude this section with some comments. First, the action-based dissipative formalism presented here is quite general and, as such usable—at least in principle—in a large number of astrophysical situations. For instance, it has already been used to build models beyond ideal magneto-hydrodynamics [24, 25, 23] due to the fact that it is intuitively clear how to couple the model to electromagnetism. Also, it is important to note that the action and the field equations are fully non-linear. The "variational" aspect of the approach is in the context of the action principle, and there is nothing in the variational process that says the field equations themselves have to be linear in the fields¹⁹. In fact, chapter 3 deals with a linearization of the present model in terms of deviations from—a self-consistently defined notion of—thermodynamic equilibrium.

2.4 Hydrodynamics as an effective field theory

After having discussed the "traditional" approaches and the recent variational efforts, we now turn our attention to some fairly recent work based on a field-theory take on the problem. As we will see, this work represent a significant change in perspective which is worth exploring. Moreover, we will adapt some of the ideas behind these strategies later on (see section 4.8.1), although in a different spirit.

Hydrodynamics can be viewed as the classical low energy limit of a more fundamental quantum (many body and thermal) field theory. Such a perspective is quite natural in the context of heavy-ion-collisions (see [188]). We also note that, as these theories have been developed having in mind heavy-ion-collision applications, they are mostly discussed in a special relativistic setting. An extension to General Relativity is not necessarily straightforward, and may require some careful thinking—as discussed in the next chapter.

¹⁹An obvious and familiar example illustrating this same feature comes to mind: the Einstein-Hilbert action yields the Einstein equations, which are notoriously non-linear in the metric.

We also mention that a similar view on hydrodynamics as a coarse-grained theory has been explored for non-dissipative fluid models as well. For instance, in Bhattacharya et al. [46] the authors proposed a variational non-dissipative hydrodynamic model based on a derivative expansion. Roughly speaking, the Lagrangian of the theory is built as the coarse-grained limit of some more fundamental one—the cutoff being the microscopic mean-free-path of the particles. Such an action is constructed summing all possible terms with a specific numbers of derivatives, each with its own coupling. In natural units, each derivative has the dimension of a mass. Since all terms in the action must have the same dimension, those with higher derivatives correspond to lower (mass) dimension couplings. These lower dimension couplings will take smaller values in the low energy limit, so that they are suppressed at the hydrodynamic level.

Moving on to dissipative fluids, two important lessons can be drawn from the traditional approaches and suggests we should investigate the effective field theory point of view. First, most of the dissipative theories—with the notable exception of the variational model of [20]—are intrinsically only valid in the "linear regime", i.e. close to some reference equilibrium state. Second, the relaxation effect we have described in section 2.2.3 suggests that while second-order theories may be required to overcome the problems of the Landau-Eckart models, their physical predictions are going to be practically indistinguishable form those of first-order ones. Furthermore, when fluctuations about equilibrium are included in the modelling, new problems arise. In particular, it turns out that corrections to the correlation functions coming from second order terms are smaller than those due to interactions between fluctuations. This is the so called breakdown of second order hydrodynamics. It is a relatively well known problem that is not specific to relativistic theories, and leads to stochastic hydrodynamics (see Landau and Lifshitz [134] and the recent review by Kovtun [132]). For all these reasons, it makes sense to ask whether one can fix the stability and causality issues directly at first order.

In the effective field-theory framework, dissipative hydrodynamics equations for single fluids can be constructed as follows. The equations of motion are given by the (special relativistic) conservation laws

$$\partial_a n^a = 0$$
 , (2.75a)

$$\partial_a T^{ab} = 0 , \qquad (2.75b)$$

where the conserved currents are decomposed as usual

$$n^a = nu^a + v^a , \qquad (2.76a)$$

$$T^{ab} = \varepsilon u^{a} u^{b} + (p + \chi) \perp^{ab} + 2q^{(a} u^{b)} + \chi^{ab} .$$
(2.76b)

To close the system, we need to provide explicit expressions for the dissipative fluxes via some constitutive equations. These are obtained through the most general gradient

expansion in the (chosen) equilibrium hydrodynamic variables. Typically these are taken to be a four-velocity u^a , the chemical potential μ and the temperature *T*:

$$\varepsilon = \varepsilon_{\rm eq} + \varepsilon_1 \dot{T} / T + \varepsilon_2 \partial_a u^a + \varepsilon_3 u^a \partial_a (\mu / T) , \qquad (2.77a)$$

$$\chi = \pi_1 \dot{T} / T + \pi_2 \partial_a u^a + \pi_3 u^a \partial_a (\mu / T) , \qquad (2.77b)$$

$$q^{a} = \theta_{1}\dot{u}^{a} + \frac{\theta_{2}}{T} \perp^{ab} \partial_{b}T + \theta_{3} \perp^{ab} \partial_{b}(\mu/T) , \qquad (2.77c)$$

$$\chi^{ab} = \eta \sigma^{ab} , \qquad (2.77d)$$

$$n = n_{\rm eq} + \nu_1 \dot{T} / T + \nu_2 \partial_a u^a + \nu_3 u^a \partial_a (\mu / T , \qquad (2.77e)$$

$$j^{a} = \gamma_{1}\dot{u}^{a} + \frac{\gamma_{2}}{T} \perp^{ab} \partial_{b}T + \gamma_{3} \perp^{ab} \partial_{b}(\mu/T) , \qquad (2.77f)$$

where the "dots" represent a derivative along u^a , that is $\dot{T} = u^a \partial_a T$. The causality and stability properties of these theories have been studied in a number of recent papers, such as, for example [40, 133, 117]. The results depend on the equation of state, but it has been demonstrated—at least for some simple cases, like conformal fluids—that one can derive a set of constraints on the expansion parameters, and guarantee stability and causality. Even more recently, an extension of this approach to General Relativity was explored in [42].

These results may come a bit of a surprise after the earlier discussion of first order theories, so let us briefly comment on why it is in fact possible to satisfy the stability and causality constraints at first order. A first and key ingredient for this is the larger number of free coefficients in the gradient expansion. This is ultimately motivated by the fact that quantities like the temperature or chemical potential are not uniquely defined out of equilibrium-different definitions are possible as long as they agree in equilibrium. This is essentially the reason why there are many more coefficients in eq. (2.77) when compared to the Landau or Eckart models. Moreover, we recall that, according to the effective field theory picture, hydrodynamics is a coarse grained theory whose validity is restricted to long wavelengths and low frequencies—and the gradient expansion makes sense as long as the gradients are in fact small. Given this, one is urged to respect the constraints coming from the second law of thermodynamics only in the regime of validity of the theory, namely "on-shell". The trick consists of stabilising the unstable modes that appear in first-order theories by allowing for violation of the second law of thermodynamics (out of the "hydrodynamic regime"). For more details on how this additional freedom can be used to ensure covariant stability, we point to section 4.8.1 where a similar strategy is used in the context of turbulence modelling.

Even though the discussion in this section is at a broad-brush level, it is clear that these theories represent a radical change with respect to the MIS paradigm. This is mainly with regards to the way they solve the instability by "killing" the unstable modes, and

also because of the change in perspective with respect to EIT. As such, they offer interesting prospects and we point to [88] for a pedagogical discussion of the differences with the EIT paradigm.

Chapter 3

Linearizing an action-based formalism for dissipative (multi-)fluids

The main goal of this chapter is to compare the action-based formulation of Andersson and Comer [20] with previous approaches—such as MIS (see section 2.2.2). The key point is that the Andersson and Comer [20] action principle does not reference any sort of chemical, dynamical, or thermal equilibrium, other than to start with the assumption that the physics can be modelled as fluid phenomena. Conversely, traditional strategies for dissipative fluids—and recent works based on an effective field-theory perspective (see section 2.4)—use an expansion to create an approximate set of field equations to describe dissipative phenomena. Since the action-based model already provides a set of equations (at least in principle) valid in every regime, we can make the comparison using standard perturbation techniques. The dissipation terms are assumed to generate first-order deviations away from equilibria obtained using the non-dissipative limit of the field equations. Working this way we hope to also understand better the role of length- and time-scales of fluid elements on the large scale behavior of the system; in particular, how to link the micro-scale dynamics of the many particles in a fluid element with the macro-scale dynamics between the fluid elements themselves, and the role of the Equivalence Principle in setting these scales. The results presented in this chapter have been published in Celora et al. [57].

3.1 Main assumptions of the model

Let us get started with an introductory section in which we expand on the central assumptions behind the fluid-modelling scheme and the nature of the non-closed three forms n_{ABC}^{x} introduced in section 2.3.3—central to the dissipative model of Andersson and Comer [20].

3.1.1 Flux definition

The crux of the fluid modelling scheme is to assume that knowledge of the total massenergy and momentum flux obtained by tracking the worldlines of individual particles can be replaced with tracking the worldlines of fluid elements. These are defined in the following way: Take a multi-particle system at some initial time having, say, total spatial size *V*, total number of particles *N*, total mass-energy *E*, and total entropy *S*. At the same time, fill-up side-to-side, top-to-bottom, and front-to-back the entire system with I = 1...M local conceptual boxes—the fluid elements. Each element has its own volume δV_I , number of particles δN_I , mass-energy δE_I , and entropy δS_I . Roughly speaking, if there are characteristic values $\delta V_I \sim \delta V$, $\delta N_I \sim \delta N$, etcetera, representative of the fluid elements, then $V \sim M \delta V$, $N \sim M \delta N$, $E \sim M \delta E$, and $S \sim M \delta S$. Clearly, as the number *M* is increased the ratios $\delta V/V$, $\delta N/N$, etcetera decrease, and the elements become "ultra-local", implying that the change in the spacetime metric across them is small.

Now consider the I^{th} -fluid element. It moves through spacetime and, if the element is small enough, the trajectory can be accurately represented by a single unit four-velocity u_I^a . When taken together, and in the limit $M \to \infty$, all the u_I^a form a vector field on spacetime and this field plays a role in the fluid system's degrees of freedom. If a local typical scattering length λ_I between the particles exists, and the size of fluid elements is commensurate with that length ($\delta V_I \sim \lambda_I^3$), then the average four-velocity of the δN_I particles will be u_I^a . In principle, we now have everything we need to define the actual fluid degrees of freedom, which are the particle fluxes $n_I^a = (\delta N_I / \delta V_I) u_I^a$.

Even though this may be familiar, we went through the details of the formal process to define fluid elements to point out what are the fundamental assumptions behind a fluid description. We have introduced typical scattering lengths and average velocities as part of our fluid element definition, and therefore we must assume that fluid elements contain enough particles to warrant a statistical/thermodynamical treatment. In the formal procedure there is no requirement of being close to thermodynamic equilibrium. It can be shown (see, for instance [181]) that fluid dynamics can be obtained as a limit of kinetic theory (via a Chapman-Enskog type expansion), but the realm of hydrodynamics is potentially vaster.

3.1.2 Matter space volume forms

All dissipative terms that enter the action-based equations are obtained by assuming that the fundamental current three-forms n_{abc}^{x} depend on an additional set of quantities which breaks their closure ($\nabla_{[a} n_{bcd]}^{x} \neq 0$). We now want to expand on how this can happen, but begin by introducing a bit of notation.

We need to distinguish between the Levi-Civita symbol η_{ABC} and a volume measure form ε_{ABC}^{x} on the matter space. The Levi-Civita symbol is defined as $\eta_{ABC} = [A B C]$ for every chosen set of coordinates (and thus is not a tensor but a tensor density) while the volume measure form ε_{ABC}^{x} can be defined¹ by means of the push-forward of the metric:

$$g_{x} = \frac{1}{3!} \eta_{ABC} \eta_{DEF} g_{x}^{AD} g_{x}^{BE} g_{x}^{CF} = \det(g_{x}^{AB}) , \qquad (3.1a)$$

$$\varepsilon_{ABC}^{\mathbf{x}} = \sqrt{g^{\mathbf{x}}} \eta_{ABC} = \sqrt{g^{\mathbf{x}}} [A B C] , \qquad (3.1b)$$

where $g^x = (g_x)^{-1}$ is the determinant of the inverse matrix g^x_{AB} ; i.e. $g^x_{AC} g^{CB}_x = \delta^B_A$. As a result, ε^x_{ABC} is a three form and transforms as a tensor under coordinates transformations on the matter space.

This volume measure form provides a way to measure the volume of "matter elements", infinitesimal volumes in the matter space manifold. We can relate these quantities to the current and momentum three-forms:

$$n_{ABC}^{\mathbf{x}} = \mathcal{N}_{\mathbf{x}} \varepsilon_{ABC}^{\mathbf{x}} = \bar{\mathcal{N}}_{\mathbf{x}} \eta_{ABC}^{\mathbf{x}} , \qquad (3.2a)$$

$$\mu_{\rm x}^{ABC} = \mathcal{M}_{\rm x} \, \varepsilon_{\rm x}^{ABC} = \bar{\mathcal{M}}_{\rm x} \, \eta_{\rm x}^{ABC} \,. \tag{3.2b}$$

The point we want to make here is that the barred quantities look more like scalar densities on the x-matter space, while the non-barred ones look more like scalars. The relation between the two normalizations is simply

$$\mathcal{N}_{\rm x} = \sqrt{g_{\rm x}}\,\bar{\mathcal{N}}_{\rm x}\,,\tag{3.3a}$$

$$\mathcal{M}_{\mathrm{x}} = \sqrt{g^{\mathrm{x}}}\,\bar{\mathcal{M}}_{\mathrm{x}}\,.\tag{3.3b}$$

We can use this to expedite our use of the variational principle by focusing the additional functional dependence of n_{ABC}^{x} into

$$\mathcal{N}_{x} = \mathcal{N}_{x}(X_{x}^{A}, X_{y}^{A}, g_{x}^{AB}, g_{y}^{AB}, g_{xy}^{AB}) .$$
(3.4)

¹This is tricky for a couple of reasons: It is well known from work on general relativistic elastic bodies [122] that this is not the only possible choice. Also, the projected metric g_x^{AB} is not "fixed" in the sense that the spacetime metric g_{ab} changes, in a general curved spacetime, as a fluid element moves from point-to-point along its worldline.

To make contact with proper quantities measured in spacetime—that is, with the rest frame density and rest frame momentum for each fluid component-it is useful to introduce an appropriate tetrad $e_a^{\hat{a}}$ for each species; an orthonormal basis whose timelike unit vector $\mathbf{e}_{\hat{0}} = \mathbf{u}_{x}$, so that $u_{x}^{\hat{a}} = (\mathbf{e}_{\hat{0}})^{\hat{a}} = \delta_{\hat{0}}^{\hat{a}} = (1,0,0,0)^{\top}$. The components of the spacetime measure form in this tetrad basis are²

$$\varepsilon^{\hat{a}\hat{b}\hat{c}\hat{d}} = \varepsilon^{abcd} e^{\hat{a}}_{a} e^{\hat{b}}_{b} e^{\hat{c}}_{c} e^{\hat{d}}_{d} = \eta^{\hat{a}\hat{b}\hat{c}\hat{d}}$$
(3.5)

, where $\eta^{\hat{a}\hat{b}\hat{c}\hat{d}} = -[\hat{a}\hat{b}\hat{c}\hat{d}]$ and we have omitted the chemical index. Now, since pushforward (and pull-back) is a linear map between vector spaces (the tangent space), it transforms as a linear map under coordinate changes, and we can write

$$A^{A} = \frac{\partial X^{A}_{\mathbf{x}}}{\partial x^{a}} A^{a} = \Psi^{A}_{\mathbf{x}\hat{a}} A^{\hat{a}} , \qquad (3.6)$$

where we have introduced the short-hand notation³

$$\Psi^{A}_{\mathbf{x}\hat{a}} \doteq \Psi^{A}_{\mathbf{x}a} e^{a}_{\hat{a}} = \frac{\partial X^{A}_{\mathbf{x}}}{\partial x^{a}} e^{a}_{\hat{a}} \,. \tag{3.7}$$

Making use of the fact that $0 = u_x^{\hat{a}} \Psi_{x\hat{a}}^A = \Psi_{x\hat{0}}^A$ we then get⁴

$$g_x^{AB} = \Psi_{x\hat{a}}^A \Psi_{x\hat{b}}^B \eta^{\hat{a}\hat{b}} \Longrightarrow g_x = \det(\Psi_{x\hat{i}}^A)^2 , \qquad (3.8)$$

which leads to⁵

$$\mathcal{M}_{x} = \frac{1}{3!} \mu_{x}^{ABC} \varepsilon_{ABC}^{x} =$$

= $\frac{1}{3!} \sqrt{g^{x}} \eta_{ABC} \Psi_{x\hat{a}}^{A} \Psi_{x\hat{b}}^{B} \Psi_{x\hat{c}}^{C} \varepsilon^{\hat{0}\hat{a}\hat{b}\hat{c}} \mu_{\hat{0}}^{x} = \mu_{x} , \qquad (3.9)$

where we have used $\mu_x = -\mu_a^x u_x^a = -\mu_x^{\hat{0}}$. This fact is important because it makes clear that only the (rest-frame) energy content of the four-momentum co-vector μ_a^{x} is stored in the normalization of the matter space momentum three-form μ_x^{ABC} . Similarly, one can show that $\mathcal{N}_{x} = n_{x}$

$$n_{\mathbf{x}} = -\frac{1}{3!} u_{a}^{\mathbf{x}} \varepsilon^{bcda} n_{bcd}^{\mathbf{x}} = -\frac{1}{3!} u_{\hat{a}}^{\mathbf{x}} \varepsilon^{\hat{b}\hat{c}\hat{d}\hat{a}} \Psi_{\mathbf{x}}^{B} \Psi_{\mathbf{x}\hat{c}}^{C} \Psi_{\mathbf{x}\hat{d}}^{D} \varepsilon_{BCD}^{\mathbf{x}} \mathcal{N}_{\mathbf{x}}$$
$$= -\frac{1}{3!} u_{\hat{0}}^{\mathbf{x}} \varepsilon^{\hat{b}\hat{c}\hat{d}\hat{0}} \varepsilon_{\hat{b}\hat{c}\hat{d}} \mathcal{N}_{\mathbf{x}} = -u_{\hat{0}}^{\mathbf{x}} \mathcal{N}_{\mathbf{x}} = \mathcal{N}_{\mathbf{x}} .$$
(3.10)

These relations are not surprising. It is, in fact, quite intuitive that the non-barred quantities are related to spacetime (rest-frame) densities given that the three-forms ε_{ABC}^{x} measure the volume of the matter space elements.

²Recall that, since $g_{ab} = e_a^{\hat{a}} e_b^{\hat{b}} \eta_{\hat{a}\hat{b}}$, the determinant of the tetrad $e = \sqrt{|g|}$. ³Following [50] we denote the inverse matrix of the tetrad as $e_{\hat{a}}^{\hat{a}}$. ⁴The index \hat{i} runs over the 1, 2, 3 components of the tetrad basis, and $\hat{a} = \hat{0}, \hat{i}$.

⁵Note that, because of the standard convention we use $\eta^{\hat{b}\hat{c}\hat{d}} = -\varepsilon^{\hat{b}\hat{c}\hat{d}}$ with $\hat{b}, \hat{c}, \hat{d} = 1, 2, 3$.

We can also use the tetrad formalism to prove another result that will be needed later on; the intimate connection between a non-zero particle creation rate and an extended functional dependence of the current three-form. In fact, we have (see eq. (2.45))

$$\Gamma_{\rm x} = \nabla_a n_{\rm x}^a = \frac{1}{3!} \varepsilon^{bcda} \Psi^B_{{\rm x}[b} \Psi^C_{{\rm x}c} \Psi^D_{{\rm x}d} \nabla_{a]} n^{\rm x}_{BCD} , \qquad (3.11)$$

where we used $\nabla_{[a} \Psi^{B}_{xb} \Psi^{C}_{xc} \Psi^{D}_{xd]} = 0$. Introducing (again) a tetrad comoving with the x-species, and multiplying by μ_x we have

$$\mu_{\rm x}\Gamma_{\rm x} = \frac{1}{3!}\mu_{\rm x}^{ABC} u_{\rm x}^a \nabla_a n_{ABC}^{\rm x} \equiv \frac{1}{3!}\mu_{\rm x}^{ABC} \frac{dn_{ABC}^{\rm x}}{d\tau_{\rm x}} \,. \tag{3.12}$$

As explained earlier, the right-hand-side of this equation vanishes identically if $n_{ABC}^{x} = n_{ABC}^{x}(X_{x}^{A})$, while it is in general non-zero if we assume the extended functional dependence given in eq. (3.4).

We can now use the introduced normalizations to slim the notation (with respect to that used in [20]) for the various pieces of R_a^x and D_{ab}^x . For instance, the "purely reactive" term from [20] becomes

$$\mathbf{R}_{a}^{\mathrm{xy}} = \frac{1}{3!} \mu_{\mathrm{x}}^{ABC} \frac{\partial n_{ABC}^{\mathrm{x}}}{\partial X_{\mathrm{y}}^{D}} \, \Psi_{\mathrm{y}a}^{D} = \mathcal{M}_{\mathrm{x}} \frac{\partial \mathcal{N}_{\mathrm{x}}}{\partial X_{\mathrm{y}}^{D}} \, \Psi_{\mathrm{y}a}^{D} \equiv \mathbf{R}_{D}^{\mathrm{xy}} \, \Psi_{\mathrm{y}a}^{D} \,. \tag{3.13}$$

Similarly we can write

$$s_{ab}^{xy} = \frac{1}{3} \mu_{x}^{ABC} \frac{\partial n_{ABC}^{x}}{\partial g_{y}^{DE}} \Psi_{ya}^{D} \Psi_{yb}^{E} = 2\mathcal{M}_{x} \frac{\partial \mathcal{N}_{x}}{\partial g_{y}^{DE}} \Psi_{ya}^{D} \Psi_{yb}^{E}$$

$$\equiv s_{DE}^{xy} \Psi_{ya}^{D} \Psi_{yb}^{E} , \qquad (3.14a)$$

$$\mathcal{S}_{ab}^{xy} = \frac{1}{3} \mu_{x}^{ABC} \frac{\partial n_{ABC}^{x}}{\partial g_{xy}^{DE}} \Psi_{xa}^{D} \Psi_{yb}^{E} = 2\mathcal{M}_{x} \frac{\partial \mathcal{N}_{x}}{\partial g_{xy}^{DE}} \Psi_{xa}^{D} \Psi_{yb}^{E}$$

$$\equiv \mathcal{S}_{DE}^{xy} \Psi_{xa}^{D} \Psi_{yb}^{E} , \qquad (3.14b)$$

where we have used the fact that the partial derivatives are performed, say, with respect to the metric g_y^{AB} keeping fixed g_x^{AB} and g_{xy}^{AB} . We will consider the validity of this assumption later. The remaining viscous stress tensor, S_{ab}^x , leads to a slightly more involved expression, because of the presence of g^x in eq. (3.3). We have

$$S_{ab}^{x} = \frac{1}{3} \mu_{x}^{ABC} \frac{\partial n_{ABC}^{x}}{\partial g_{x}^{DE}} \Psi_{xa}^{D} \Psi_{xb}^{E} = 2 \left(\frac{\mathcal{M}_{x}}{\sqrt{g^{x}}} \frac{\partial (\mathcal{N}_{x} \sqrt{g^{x}})}{\partial g_{x}^{DE}} \right) \Psi_{xa}^{D} \Psi_{xb}^{E} =$$
$$= 2 \left(\mathcal{M}_{x} \frac{\partial \mathcal{N}_{x}}{\partial g_{x}^{DE}} - \frac{1}{2} \mathcal{N}_{x} \mathcal{M}_{x} g_{DE}^{x} \right) \Psi_{xa}^{D} \Psi_{xb}^{E} =$$
$$\equiv S_{DE}^{x} \Psi_{xa}^{D} \Psi_{xb}^{E} . \tag{3.15}$$

It is also obvious, by looking at the respective definitions, that the reactive terms that stem from the fact that N_x can depend also on g_y^{AB} and g_{xy}^{AB} can now be written

$$r_{a}^{xy} = \frac{1}{2} s_{DE}^{xy} \nabla_{a} \left(g^{bc} \Psi_{yb}^{D} \Psi_{yc}^{E} \right) , \qquad (3.16a)$$

$$\mathcal{R}_a^{\mathrm{xy}} = \frac{1}{2} \mathcal{S}_{DE}^{\mathrm{xy}} g^{bc} \Psi_{\mathrm{x}b}^D \nabla_a (\Psi_{\mathrm{y}c}^E) . \qquad (3.16b)$$

Let us conclude this introductory section with a caveat on the slimmed notation introduced so far. In order to consider all the projected metrics to be independent, we need to make sure we are not actually adding degrees of freedom to the problem. Since the various projected metrics are ultimately combinations of Ψ_{xb}^A we need to make sure that the degrees of freedom associated with the metrics is less or equal to 12 l—where lis the number of constituents. Because the number of mixed projected metrics is easily found to be l(l - 1)/2 we have

$$6\frac{l(l+1)}{2} \le 12l \Longrightarrow l \le 3.$$
(3.17)

Therefore, the slimmed notation introduced in this section applies to cases with less than (or equal to) three species moving independently. In what follows, the machinery is developed in a general setting, but we will focus to cases with less than three species for specific applications.

3.2 The non-dissipative limit

We now begin to develop the process for comparing standard relativistic models for dissipative fluids with that provided by the action principle. Standard approaches [160, 118, 206, 119, 120] assume a reference equilibrium state and then build in dissipation via deviations away from this state. The action principle formally does not require any sort of equilibrium, but provides a fully non-linear set of field equations. Obviously, our first task must be to extract from the non-linear equations a notion of equilibrium. This is not straightforward for various reasons, a key one being that an arbitrary spacetime in General Relativity does not have global temporal, spatial, and rotational invariance. As a first step, we will recall features of the typical laboratory set-ups within which the laws of chemistry, dynamics, and thermodynamics were first established.

3.2.1 Laboratory vs. general relativistic set-up

A typical laboratory set-up is essentially local in the spacetime sense, implying there is—to a great deal of precision—temporal, spatial, and rotational invariance. Noether

symmetries exist, which lead to energy, momentum, and angular momentum conservation. A clean separation between internal and external influences can be made, and these influences themselves can be manipulated. The effect of long-range, nonscreenable forces on the system—for example, gravity—can be ignored. Well-defined (theoretical and experimental/observational) notions of total energy and entropy can be realized. Equilibrium can be defined in the broadest sense by saying the system evolves to a state where its total energy is minimized, or, equally, its total entropy is maximized.

Internal interactions are due to, say, chemical reactions, whereas external interactions are those which distort the system's volume or allow particles and heat to enter or leave through the volume's surface. If a system is in chemical equilibrium internally, we can say that the reactions inside it are running forwards and backwards at such a rate that constituent particle number ratios remain fixed in time. If the given system is in chemical equilibrium with another system, then the chemical potentials of the two will be equal. A system in dynamical equilibrium just sits there, with no temporal evolution. Any pressure acting on the system's surface will be balanced by an internal pressure of the same value. Finally, we can say that two systems are in thermal equilibrium when there is no heat flow between them, the end result being equality of their respective temperatures. Now, let us return to the problem at hand—equilibrium when General Relativity cannot be neglected.

A general relativistic set-up is problematic from the get-go, because one is hard-pressed to find properties of equilibrium like those just discussed which are workable at all time- and length-scales. Broadly speaking, there seems to be no general relativistic rules on how the local thermodynamics of local (intensive) parameters—chemical potential μ , pressure p, and temperature T—connect with some notion of global thermodynamics for global (extensive) parameters—such as the total energy E. An unambiguous extrapolation of the standard definitions of chemical, dynamical, and thermal equilibrium given above to General Relativity is not possible, for reasons to be explained below. There is also the well-known difficulty of identifying the total energy of a region in an arbitrary spacetime, since the Equivalence Principle precludes an ultra-local definition of gravitational energy density.⁶

The reason that the laboratory rules for chemical and thermal equilibrium are not viable in General Relativity was established long ago by Tolman and Ehrenfest [214, 215]: In General Relativity, all forms of energy react to gravity. Temperature and chemical potentials represent forms of energy and can undergo red-shift or blue-shift. There is no single temperature for an isolated system, and so saying "system A is in thermal equilibrium with system B if their temperatures are the same" becomes ambiguous;

⁶Of course, for asymptotically flat spacetimes, one can define quantities like the Schwarzschild mass. Gravitational-wave energy can be defined but only after averaging over wavelengths, see [145, 207].

similarly for chemical equilibrium. As for dynamical equilibrium, a standard undergraduate physics calculation shows that pressure increases with depth in water which nevertheless remains at rest.⁷

Even the use of the word "equilibrium" is tricky because it tends to imply that a system in thermal and chemical equilibrium is independent of time, because the total entropy and total particle number do not evolve. In General Relativity, a system which is independent of time occurs only for special spacetimes which have a global timelike Killing vector field. Strictly speaking, this immediately puts the non-dissipative fluid models of Cosmology—the Friedman-Lemaitre-Robertson-Walker solutions—out of the discussion, as the universe is expanding, making it time-dependent.

This points to another problem of the notion of total energy in General Relativity and arguments based on the standard understanding of energy conservation: In Special Relativity, the curvature is zero and there is a timelike Killing vector field leading directly to a Noether symmetry for the system and total energy conservation. (There are also Killing vector fields representing rotational and spatial invariance, which lead to Noether symmetries resulting in total angular and linear momentum conservation.) In an expanding universe this line of reasoning for energy conservation obviously breaks down.

The main message is this: Important issues remain unsettled even after a century's worth of debate. We will not resolve these issues here; instead, what we will do is take the action-based formalism and see how its internal machinery can be manipulated to produce a self-consistent notion of the non-dissipative limit, without trying to resolve the deeper issues about the nature of equilibrium⁸. Our way forward is to take advantage of the fact that the action-based field equations are fully non-linear and complete.

3.2.2 Multiple equilibrium states

The main mechanism for manipulating the machinery of the action-based field equations is to apply perturbation techniques similar to those used to determine, say, quasinormal modes of neutron stars. The general idea for neutron stars is to analyze linear perturbations of configurations having particular symmetries generated by Killing vectors. Among the most studied neutron star "ground-states" are those having Killing vectors which generate staticity and spherical symmetry, and those with Killing vectors that generate axisymmetry and stationarity; basically, non-rotating and rotating backgrounds, respectively.

⁷In this context, we can think of it as resulting from the breaking via gravity of spacelike Killing vectors which lead to space-translation invariance.

⁸We will still use the word "equilibrium" interchangeably with the non-dissipative limit.

In an analogous way, we can expect different options for generating the non-dissipative limit of a multi-fluid system. For example, we can take the limit where the different dissipation coefficients (such as shear and bulk viscosities) are effectively zero. Another possibility is the limit where the dissipation coefficients are non-zero but the fluid motion itself is such that the dissipation mechanisms are not acting. The formalism developed by Onsager [166] (recall discussion in section 2.1.1) is worthy of mention here, because the system of field equations it creates are more explicit in how the two limits can be implemented (see, for example, [18]). It is interesting also to note that the philosophy of the Onsager approach is not so much about how to expand away from an equilibrium, but rather how a non-equilibrium system gets driven back to the equilibrium state. Here, because the field equations are fully non-linear, they can, in principle, describe systems which are being driven toward or away from equilibrium.

Next, we will explore some of the different options for equilibrium states. We will use a global analysis which assumes that the second law of thermodynamics applies and that a knowledge of the fluxes throughout a region of spacetime is enough to determine whether or not dissipation is acting. A local analysis of the formalism will also be pursued, involving the field equations themselves.

3.2.3 Global analysis of the non-dissipative limit

Recall that the fundamental dynamical variables are the particle fluxes n_x^a and the entropy $s^a = n_s^{a,9}$ The formalism's linchpin is the breaking of the closure of the particle-flux three-forms, n_{abc}^x and s_{abc} , which leads to non-zero creation rates Γ_x and Γ_s . In turn, these non-zero creation rates lead to the resistive contribution R_x^a and the dissipation tensor D_{ab}^x terms in the equations of motion. The nice thing about fluxes, which we will exploit here, is that they can be integrated.

When we use the Einstein equations and the field equations of a multi-fluid system, our goal is to get solutions for the metric and fluxes on a "chunk" of spacetime, for a given set of initial/boundary conditions. Suppose we pick an ad hoc region \mathcal{M} of spacetime, as illustrated in fig. 3.1. The fact that it is a region implies there is a "conceptual boundary", meaning the whole spacetime is being divided up into smaller domains. Let u_B^a (collectively) denote the unit normal to the total boundary of the region, defined so that it always points "out". The boundary itself consists of two spacelike hypersurfaces $\partial \mathcal{M}_{\pm}$ (with unit normals $u_{B_{\pm}}^a$, $u_{B_{\pm}}^a u_a^{B_{\pm}} = -1$), and a timelike hypersurface $\partial \mathcal{M}_L$ (with unit normal $u_{B_L}^a$, $u_a^a u_a^{B_{\pm}} = +1$); in essence, think of $\partial \mathcal{M}_-$ as a 3D region of characteristic volume ΔL^3 on an initial time-slice of \mathcal{M} and $\partial \mathcal{M}_+$ as the same volume on the final time-slice, and then $\partial \mathcal{M}_L$ will be similar to the union of the surface of the same

⁹Because we impose the second law of thermodynamics below, we are specifically separating out the entropy flux in this discussion.



FIGURE 3.1: A depiction of the spacetime region \mathcal{M} , with one spatial axis suppressed. It has a characteristic spatial size ΔL and temporal size ΔT . Inside \mathcal{M} is a smaller region $\delta \mathcal{M}$ of characteristic spatial and temporal size δl and δt , respectively. The boundary $\partial \mathcal{M}$ consists of the initial and final time-slices $\partial \mathcal{M}_{-}$, $\partial \mathcal{M}_{+}$ and the time-like hypersurface $\partial \mathcal{M}_{L}$.

volume on each leaf of some spacelike foliation of \mathcal{M} between $\partial \mathcal{M}_{-}$ and $\partial \mathcal{M}_{+}$. The induced metric on $\partial \mathcal{M}_{\pm}$ is $h^{ab}_{\pm} = g^{ab} + u^{a}_{B_{\pm}} u^{b}_{B_{\pm}}$ and for $\partial \mathcal{M}_{L}$ it is $h^{ab}_{L} = g^{ab} - u^{a}_{B_{L}} u^{b}_{B_{L}}$.

There are three contributions to the total particle number change ΔN^x and total entropy change ΔS : (i) The total particle number N_{-}^x and entropy S_{-}^x which exist in $\partial \mathcal{M}_{-}$; (ii) The total particle number N_{+}^x and entropy S_{+}^x which exist in $\partial \mathcal{M}_{+}$; and, (iii) The number of particles ΔN_L^x and amount of entropy ΔS_L which enter/leave $\partial \mathcal{M}_L$. Each contribution is obtainable from its associated flux: If n_{\pm}^x (s_{\pm}) are the particle number (entropy) densities as measured with respect to the volumes $\partial \mathcal{M}_{\pm}$, and n_L^x (s_L) is the number of particles (amount of entropy) per unit area per unit time entering/leaving $\partial \mathcal{M}_L$, then¹⁰

$$N_{+}^{x} = \int_{\partial \mathcal{M}_{+}} d^{3}x \sqrt{h_{+}} \ n_{+}^{x} = \int_{\partial \mathcal{M}_{+}} d^{3}x \sqrt{h_{+}} \ \left(-u_{a}^{B_{+}} n_{x}^{a}\right) \ , \tag{3.18a}$$

$$N_{-}^{x} = \int_{\partial \mathcal{M}_{-}} d^{3}x \sqrt{h_{-}} \ n_{-}^{x} = \int_{\partial \mathcal{M}_{-}} d^{3}x \sqrt{h_{-}} \ \left(u_{a}^{B_{-}} n_{x}^{a}\right) , \qquad (3.18b)$$

$$\Delta N_L^{\rm x} = \int_{\partial \mathcal{M}_L} \mathrm{d}^3 x \sqrt{-h_L} \; n_L^{\rm x} = \int_{\partial \mathcal{M}_L} \mathrm{d}^3 x \sqrt{-h_L} \; \left(u_a^{\rm B_L} n_{\rm x}^a \right) \;, \tag{3.18c}$$

¹⁰We are denoting with h_{\pm} the determinant of h_{ab}^{\pm} . We have also taken into account the fact that $u_{B_{-}}^{a}$ points to the past, and that the signature of the induced metric h_{L}^{ab} is (-++) as $\partial \mathcal{M}_{L}$ is timelike.

and similarly for ΔS . The changes in the total x-particles ΔN_x and entropy ΔS over the region \mathcal{M} are therefore

$$\Delta N^{\rm x} = N^{\rm x}_{+} - N^{\rm x}_{-} + \Delta N^{\rm x}_{L} , \qquad (3.19a)$$

$$\Delta S = S_+ - S_- + \Delta S_L \,. \tag{3.19b}$$

If the length- and time-scales of spacetime region \mathcal{M} are those typical of terrestrial labs (read: its curvature is zero throughout), then we have great confidence in asserting the second law of thermodynamics; namely, the net change of the total entropy must satisfy $\Delta S \ge 0$. We could even be confident that we could determine the total energy E and volume V of the system, and have a working first law of thermodynamics which connects ΔE , ΔN^{x} , ΔV , and ΔS :

$$\Delta E = T\Delta S - p\Delta V + \sum_{x} \mu_{x} \Delta N^{x} . \qquad (3.20)$$

The temperature *T*, pressure *p*, and chemical potentials μ_x would be well-defined and calculable. We could even use the standard notions of chemical, dynamical, and thermal equilibrium and say that system A of spacetime region \mathcal{M}_A is in chemical, dynamical, and thermal equilibrium with system B of spacetime region \mathcal{M}_B if, respectively, their chemical potentials are equal, their pressures are equal, and their temperatures are equal.

Now, let us suppose we have a region large enough that spacetime curvature can no longer be ignored. Probably, it would be a safe bet to say that the second law still applies; i.e., $\Delta S \ge 0$. But, we are hard-pressed to employ the laboratory definitions of chemical, dynamical, and thermal equilibrium. Consequently, it is difficult to imagine a global first law of thermodynamics for general relativistic multifluid systems similar to that in eq. (3.20); again, the reason being that intensive parameters are spacetime dependent, and an extensive parameter like total energy may not even be definable. Still, our task is to explore any possible link between parameters which require scales where spacetime curvature is necessary (ΔN^x and ΔS) to the local fluid variables (n_x^a and s^a) which enter the fluid field equations. Fortunately, the divergence theorem provides such a link.

Applying it to the divergence of both the particle and entropy fluxes gives

$$\Delta N^{\mathrm{x}} = \int_{\mathcal{M}} \mathrm{d}^{4} x \sqrt{-g} \, \nabla_{a} n_{\mathrm{x}}^{a} = \int_{\mathcal{M}} \mathrm{d}^{4} x \sqrt{-g} \, \Gamma_{\mathrm{x}} \,, \qquad (3.21a)$$

$$\Delta S = \int_{\mathcal{M}} \mathrm{d}^4 x \sqrt{-g} \, \nabla_a s^a = \int_{\mathcal{M}} \mathrm{d}^4 x \sqrt{-g} \, \Gamma_\mathrm{s} \,. \tag{3.21b}$$

These are not new results, but they serve the purpose here of establishing a direct link between global and local variables, which we will use to formulate some aspects of the non-dissipative limit of our formalism. Consider an idealized situation of a spacetime region \mathcal{M} sub-divided into a region \mathcal{M}_A for which $\Delta N_A^x < 0$ and $\Delta S_A < 0$, and another region \mathcal{M}_B for which $\Delta N_B^x > 0$ and $\Delta S_B > 0$. The trick is that they are such that the total changes on \mathcal{M} vanish:

$$\Delta N^{\mathbf{x}} = \Delta N_A^{\mathbf{x}} + \Delta N_B^{\mathbf{x}} = 0 , \ \Delta S = \Delta S_A + \Delta S_B = 0 .$$
(3.22)

The point is that, even though Γ_x and Γ_s are not zero, this is an example of a global, fully general relativistic, non-dissipative system since there is no net total particle number or total entropy change. But is this realistic? Is this the kind of definition of the non-dissipative limit we are looking for? Probably not. What is more likely is that the non-dissipative limit is better understood by breaking up \mathcal{M} into many small spacetime regions $\delta \mathcal{M}$, with characteristic temporal and volume scales δt and $(\delta l)^3$, respectively, as illustrated in fig. 3.1.

Once again, let us imagine that $\delta \mathcal{M}$ is subdivided into two regions $\delta \mathcal{M}_A$ and $\delta \mathcal{M}_B$. It is conceivable that on these scales statistical fluctuations could lead to positive creation rates in one region and negative in the other. If the regions are small enough, we can assume that Γ_x and Γ_s vary slowly across them so that we can approximate the integrals for $\delta N^x_{\delta \mathcal{M}}$ and $\delta S_{\delta \mathcal{M}}$ as¹¹

$$\delta N_{\delta \mathcal{M}}^{\mathsf{x}} \approx \Gamma_{\mathsf{x}} \delta t \left(\delta l\right)^{3} , \ \delta S_{\delta \mathcal{M}} \approx \Gamma_{\mathsf{s}} \delta t \left(\delta l\right)^{3} .$$
 (3.23)

However, the random nature of statistical fluctuations for a system purported to be in equilibrium implies that any non-zero creation rates inside δM_A and δM_B must balance on average so that

$$\delta N_{\delta \mathcal{M}}^{\mathsf{x}} = \delta N_{\delta \mathcal{M}_{A}}^{\mathsf{x}} + \delta N_{\delta \mathcal{M}_{B}}^{\mathsf{x}} \approx \left(\Gamma_{\mathsf{x}}^{A} + \Gamma_{\mathsf{x}}^{B}\right) \delta t \ (\delta l)^{3} = 0 \implies \Gamma_{\mathsf{x}} = \Gamma_{\mathsf{x}}^{A} + \Gamma_{\mathsf{x}}^{B} = 0 \ , \ (3.24a)$$
$$\delta S_{\delta \mathcal{M}} = \delta S_{\delta \mathcal{M}_{A}} + \delta S_{\delta \mathcal{M}_{B}} \approx \left(\Gamma_{\mathsf{s}}^{A} + \Gamma_{\mathsf{s}}^{B}\right) \delta t \ (\delta l)^{3} = 0 \implies \Gamma_{\mathsf{s}} = \Gamma_{\mathsf{s}}^{A} + \Gamma_{\mathsf{s}}^{B} = 0 \ . \ (3.24b)$$

One conclusion from this exercise is that the characteristic time and volume scales of $\delta \mathcal{M}$ must be large enough that statistical fluctuations will, on average, balance out for a system in equilibrium. The second conclusion is that having $\delta N_{\delta \mathcal{M}}^{x} = 0$ ($\delta S_{\delta \mathcal{M}} = 0$) on the one hand means $\Gamma_{x} = 0$ ($\Gamma_{s} = 0$) on the other, and vice versa. Putting both together we will assume that the equilibrium state for multi-fluid systems must be such that regions like $\delta \mathcal{M}$ set the scales for fluid elements and we have $\Gamma_{x} = 0$ and $\Gamma_{s} = 0$ everywhere in \mathcal{M} .

¹¹We are also assuming that the δM to be small enough that we can transform away gravity by means of Riemann normal coordinates [155].

3.2.4 Local analysis of the non-dissipative limit

This next step begins where the previous one left off; that is, a necessary condition for a multi-fluid system to be in equilibrium is that the flux creation rates Γ_x (now including the entropy) vanish everywhere. We will use the field equations themselves to investigate three different ways for the action-based system to have zero particle creation rates: 1) The limit where the dissipation terms R_a^x and D_{ab}^x are zero, 2) the limit where the dissipation terms are non-zero but the fluid motion is such that the dissipative channels are dynamically suppressed, and 3) a combination of dynamical suppression with constraints between the dissipation terms that lead to Killing vector fields.

But before moving on with the analysis, it is advantageous to consider the simplest non-dissipative fluid model which can be derived from the action above—the ordinary perfect fluid, where all particle species and entropy flow together and the total particle numbers and entropy are conserved individually. The calculation is straightforward [67]. All the fluxes have the same four-velocity, say, u^a , and so $n_x^a = n_x u^a$. If each particle number flux is conserved individually, then

$$\nabla_a n_{\mathbf{x}}^a = \nabla_a \left(n_{\mathbf{x}} u^a \right) = u^a \nabla_a n_{\mathbf{x}} + n_{\mathbf{x}} \nabla_a u^a = 0 \implies u^a \nabla_a \ln n_{\mathbf{x}} = -\nabla_a u^a \,. \tag{3.25}$$

Obviously, the total particle flux $n^a = \sum_x n_x^a$ is also conserved and hence

$$u^{a}\nabla_{a}\ln n = -\nabla_{a}u^{a}, \ n = \sum_{\mathbf{x}}n_{\mathbf{x}}.$$
(3.26)

Therefore, we have

$$u^{a} \nabla_{a} \ln n_{x} - u^{a} \nabla_{a} \ln n = 0 \implies u^{a} \nabla_{a} \left(\frac{n_{x}}{n}\right) = 0.$$
(3.27)

The upshot is that each species fraction n_x/n must also be conserved along the flow, and this includes the entropy as well. This implies that only one matter space is required. In the action principle, this means that for each x—including the entropy—we have $\xi_x^a = \xi^a$, and there is only one Euler equation of the form

$$\sum_{\mathbf{x}} f_a^{\mathbf{x}} = 0 , \qquad (3.28)$$

where the f_a^x are exactly as in eq. (2.44).

With this example in mind we can now proceed with the description of three different non-dissipative limits consistent with the action-based model. Note that we will also impose another condition which defines the equilibrium. We will assume that all distinct fluids are comoving, so that we are not considering systems with superfluid/superconducting phases, or a perfect heat-conducting limit [52]. This means that there is a common four-velocity for all species, $u_x^a = u^a$. However, it is important to point out a subtlety about this comoving limit: For a multi-fluid system each species has its own evolution equation. Even in the comoving limit there are still x fluid equations. Now consider the field equations for a multi-species, single fluid system—as we see from eq. (3.28), it has only one fluid evolution equation. Therefore, the comoving limit of the multi-fluid system (x equations) is not equal to the single-fluid system (one equation). This is not an error, rather, it is a consequence of the fact that the number of independent field equations of the system is fixed by the number of independent fluids chosen before the action principle is applied. Note also that we can use the common four-velocity u^a to introduce a spatial covariant derivative D_a —acting in directions perpendicular to u^a —and a time derivative "'" = $u^a \nabla_a$. For a scalar A we have

$$D_a A = \perp_a^b \nabla_b A = \left(\delta_a^b + u_a u^b\right) \nabla_b A = \nabla_a A + \dot{A} u_a , \qquad (3.29)$$

and for a vector

$$D_a A_b = \perp_a^c \perp_b^d \nabla_c A_d . \tag{3.30}$$

Dynamical suppression of dissipation

We start by considering the consequences of the non-dissipative limit if the fluid flow is such that the dissipation mechanisms are not triggered. If we look at each species creation rate we have

$$\mu_{\rm x}\Gamma_{\rm x} = -R_a^{\rm x} \, u^a - D_{ab}^{\rm x} \nabla^a u^b = 0 \,, \qquad (3.31)$$

so that, summing over all species

$$\sum_{x} \mu_{x} \Gamma_{x} = -\left(\sum_{x} R_{a}^{x}\right) u^{a} - D_{ab} \nabla^{a} u^{b} = -D_{ab} D^{(a} u^{b)} = 0 , \qquad (3.32)$$

where we have used the identities $u_x^b D_{ab}^x = 0$, $\sum_x R_a^x = 0$ and the fact that D^{ab} is symmetric. Using the standard decomposition of the four velocity gradients it is easy to see that eq. (3.32) implies

$$D_{(a}u_{b)} = \perp_{(a}^{c} \perp_{b)}^{d} \nabla_{c}u_{d} = \nabla_{(a}u_{b)} + u_{(a}\dot{u}_{b)} = \sigma_{ab} + \frac{1}{3}\theta \perp_{ab} = 0.$$
(3.33)

In particular, this tells us that the (dynamically-suppressed) non-dissipative flow has zero expansion $\theta = 0$, and zero shear $\sigma_{ab} = 0$. What is left of the motion is captured by

$$\nabla_a u_b = \omega_{ab} - \dot{u}_b u_a \quad , \tag{3.34}$$

which is consistent with rigid rotation.

From the definition of creation rates, we can now write

$$\Gamma_{\rm x} = \nabla_a n_{\rm x}^a = \dot{n}_{\rm x} + n_{\rm x} \theta = \dot{n}_{\rm x} = 0.$$

$$(3.35)$$

Assuming a thermodynamical relation in the standard way, namely that the energy functional of the system is¹² $\varepsilon = \varepsilon(n_x)$, we see that the chemical potential of each species is $\mu_x = \mu_x(n_x)$ and likewise for the pressure p. Therefore, we have $\dot{\mu}_x = 0$ and $\dot{p} = 0$, as well. The proposed scenario is consistent with the minimum requirements for the system being non-dissipative—as explained above. This is not, however, the situation we will use as basis for the expansion.

The Euler limit

Later (in section 3.4) we will use thermodynamics arguments to show that the dissipative terms all vanish at equilibrium: $D_{ab}^{x, e}$ and $R_{a}^{x, e} = 0$ —where we have introduced the superscript "e" to stress that the dissipative terms are evaluated at equilibrium, consistently with the notation used later on. We now consider the non-dissipative limit with these additional constraints and show its compatibility with the Euler equations. Since the fluids are comoving at equilibrium we have for the fluxes $n_x^a = n_x u_{e}^a$, and so the four-momenta become

$$\mu_a^{\mathrm{x}} = \left(\mathcal{B}_{\mathrm{x}} n_{\mathrm{x}} + \sum_{\mathrm{y} \neq \mathrm{x}} \mathcal{A}_{\mathrm{xy}} n_{\mathrm{y}}\right) u_a^{\mathrm{e}} = \mu_{\mathrm{x}} u_a^{\mathrm{e}} \,. \tag{3.36}$$

and the equation of motion for the x-species is

$$f_a^{\rm x} = 2n_{\rm x}^b \nabla_{[b} \mu_{a]}^{\rm x} = n_{\rm x} \mu^{\rm x} \dot{u}_a^{\rm e} + n_{\rm x} \left(u_{\rm e}^b u_a^{\rm e} + \delta_a^b \right) \nabla_b \mu^{\rm x} = n_{\rm x} \mu^{\rm x} \dot{u}_a^{\rm e} + n_{\rm x} D_a \mu^{\rm x} = 0 .$$
(3.37)

The first term in f_a^x then looks like the mass/energy per volume times the acceleration while we can show that the second is a "pressure-like" term in the sense of being the gradient of a thermodynamic scalar. In fact, we have

$$\frac{\partial \Lambda}{\partial n_{\rm x}} = -\left(\mathcal{B}_{\rm x}n_{\rm x} - \sum_{\rm y \neq \rm x}\mathcal{A}_{\rm xy}n_{\rm y}^{\rm a}u_{\rm a}^{\rm x}\right) = -\mu_{\rm x} , \qquad (3.38)$$

and the sum of these terms provides the derivative of the total pressure Ψ :

$$\sum_{\mathbf{x}} n_{\mathbf{x}} D_a \mu^{\mathbf{x}} = D_a \left(\sum_{\mathbf{x}} n_{\mathbf{x}} \mu^{\mathbf{x}} + \Lambda \right) = D_a \Psi .$$
(3.39)

¹²Note that we are here using a compact notation, so that the chemical index x in $\varepsilon(n_x)$ runs over all the species/constituents in the system under consideration.
It is important to note that each individual term cannot (in general) be considered as the derivative of the x-species contribution to the total pressure. Partial pressures exist only when the various species do not interact.

Even though the comoving limit of the multi-fluid system is not the same as the single fluid, multi-species system, there is some overlap: Taking the sum over the chemical species of eq. (3.37) we find¹³ the standard relativistic Euler equation we derived in section 1.3. One can also show that eq. (3.28) can be written in this form. This is an important self-consistency check, but because the multi-fluid comoving limit is not the same as the single fluid limit, we need to go back to the individual fluid equations of the multi-fluid system.

We can rewrite the individual equations of motion as

$$\dot{u}_{b}^{e} = -D_{b}(\log \mu_{x});$$
 (3.40)

thus, for each combination of $x \neq y$,

$$D_a(\log \mu_x) = D_a(\log \mu_y) \implies D_a\left(\log \frac{\mu_x}{\mu_y}\right) = 0.$$
 (3.41)

This self-consistency therefore requires the various chemical potentials μ_x and μ_y (as functions on spacetime) to be proportional to each other by some factor C_y^x , which is constant in the spatial directions; namely,

$$\mu_{\rm x} = C_{\rm y}^{\rm x} \mu_{\rm y} , \ D_a C_{\rm y}^{\rm x} = 0 . \tag{3.42}$$

This is to be contrasted with the single-fluid case, where there is no such restriction—in the sense of being forced by the evolution equations—between the chemical potentials. Usually, one must provide additional information. For example, for neutron stars one typically imposes that beta decay and inverse beta decay are in equilibrium. If we combine this with the "dynamical suppression of dissipation", the factor C_y^x is in fact constant in all the space-time directions.

Dynamical suppression and Killing vectors

In a local region of spacetime, freely falling frames exist and the Killing equation will be satisfied approximately. In these local regions having an equilibrium will be consistent with the existence of Killing fields. However, local regions which are far removed from each other will not be (on the relevant dynamical timescale) in equilibrium with each other. This kind of "quasi-local" regression towards equilibrium has been discussed in the work of Fukuma and Sakatani [85], explicitly introducing two different spacetime

¹³We have used the standard Euler relation $\sum_{x} n_{x} \mu^{x} = p + \varepsilon$, where *p*, ε are the equilibrium pressure and energy density, respectively.

scales to describe the evolution of general relativistic dissipative systems. The hypothesis of Local Thermodynamic Equilibrium applies on the smaller scale—which is of the size of a fluid element—while the regression (in the sense of Onsager [166]) towards equilibrium takes place on the larger one, which can still be smaller than the body size.

A relation between the perfect fluid four-velocity and Killing vectors, for stationary axially symmetric rotating stars,¹⁴ has been discussed by Gourgoulhon [99]. A similar discussion about thermodynamic equilibrium in General Relativity and the existence of Killing vectors was provided by Becattini [37]. Specifically, he showed that there must be global Killing vector fields if the total entropy of the system is to be independent of the spacelike hypersurface over which the integration is performed. As for the work presented here, we will now show under what conditions the combination $\xi_x^a = \mu_x^{-1} u_e^a$ can be turned into Killing vector fields.

From eq. (3.40) it can be seen that

$$\nabla_a \xi_b^{x} + \nabla_b \xi_a^{x} = \frac{1}{\mu_x} \left(u_{(a}^e u_{b)}^e log \mu_x + D_{(a} u_{b)}^e \right) , \qquad (3.43)$$

so that, if dynamical suppression has worked, the right hand side becomes zero and the ξ_x^a will be a timelike Killing vector field, along which the local thermodynamical parameters n_x , μ^x , ε , and p become constants of motion. Put in different words, if we want the system to be (at least quasi-locally) at equilibrium—i.e. stationary— we also need to require rigid body motion.

3.2.5 A final comment on equilibrium

To conclude we will come full circle and consider again the change in total entropy given by eq. (3.19). The result only references spacelike hypersurfaces as part of the (ad hoc) choice of the boundary of the spacetime region for which the entropy change is being determined. There are no restrictions placed on the spacetime geometry in this construct; in particular, no requirement of global Killing vectors. As a matter of practice, the change in entropy of a system is clearly dependent on the spatial size and the amount of time the system has had to evolve. Coupling this with the fact that a separation of space from time is always a choice—an arbitrary spacetime has no preferred directions, no natural "moments-of-time"—we see that the ad hoc nature of the boundary in eq. (3.19) is not a drawback. This is precisely the freedom needed in order to incorporate a system's spatial extent and evolution time, and the fact that a separation of space from time in spacetime is always a choice.

The main reason why this is intriguing is that the second law of thermodynamics only refers to the change in total entropy, not the value of entropy itself at specific moments

¹⁴Note that Gourgoulhon [99] works with the enthalpy per particle instead of chemical potentials. However, this makes no difference for barotropic perfect fluids.

of time (i.e. spacelike hypersurfaces). It may be that questions of equilibrium are not to be settled by the "moment-to-moment" behaviour of three-dimensional integrals, but rather by global statements of the type represented by eq. (3.19). This is something we are currently investigating and hope to be able to give more detail on in a future work.

3.3 Perturbations with respect to equilibrium

With the equations of motion obtained from an action principle, we can consider perturbations away from equilibrium configurations (of the kind described above) in a way that is closely related—at least from the formal perspective—to standard hydrodynamical perturbation theory. The general approach to Lagrangian perturbation theory is perhaps best described by Friedman and Schutz [82]. Roughly speaking, the evolution equations for the perturbed fields can be obtained by perturbing the equations that follow from the action. It is also clear—at least in principle—how to construct a Lagrangian whose variation gives the perturbed equations (see §2 of [82]). However, since we are not focussing on a stability analysis of fluid oscillations we will not consider this additional aspect here.

To set the stage for the perturbative expansion, we consider the family of worldlines (not necessarily geodesics) that each constituent of a multifluid system traces out in spacetime. Our definition of equilibrium includes the assumption that all species are comoving. Therefore, our fiducial set of worldlines representing equilibrium are those the system would have followed if it were comoving throughout its history. This then allows us to view each of the "final" worldlines $x_f^a(\bar{\tau})$ as a curve in spacetime which is close to the equilibrium one $x_e^a(\tau)$, with $\bar{\tau}$ and τ being the proper times of the respective curves. See fig. 3.2 for an illustration of the idea. The unit four-velocities associated with the two worldlines are

$$u_{\rm f}^a = \frac{dx_{\rm f}^a}{d\bar{\tau}} , \qquad u_{\rm e}^a = \frac{dx_{\rm e}^a}{d\tau} . \tag{3.44}$$

Obviously, u_e^a represents the comoving frame introduced earlier.

We assume another family of curves $x_{ef}^a(\lambda)$, where λ is an affine parameter (say, the proper length), that connects the equilibrium worldline to the actual one. This means that for any point $x_e^a(\tau_e)$ on the equilibrium worldline, there is a unique point $x_f^a(\bar{\tau}_f)$ on the perturbed worldline, and a unique curve $x_{ef}^a(\lambda)$ between them having two points $x_{ef}^a(\lambda_e)$ and $x_{ef}^a(\lambda_f)$ such that

$$x_{\rm f}^a\left(\bar{\tau}_{\rm f}\right) = x_{\rm ef}^a\left(\lambda_{\rm f}\right) , \qquad x_{\rm e}^a\left(\tau_{\rm e}\right) = x_{\rm ef}^a\left(\lambda_{\rm e}\right) . \tag{3.45}$$



FIGURE 3.2: An illustration of worldlines associated with the fluid elements (solid vertical red lines, parameterized by τ , $\bar{\tau}$) and "Lagrangian displacements" which connect fluid elements (dashed horizontal blue lines, parameterized by λ).

Taylor expanding the perturbed worldline about the equilibrium up to second order, we get

$$\begin{aligned} x_{\rm f}^{a}(\bar{\tau}_{\rm f}) &= x_{\rm e}^{a}(\tau_{\rm e}) + \frac{dx_{\rm ef}^{a}}{d\lambda}\Big|_{\lambda_{\rm e}}(\lambda_{\rm f} - \lambda_{\rm e}) + \frac{1}{2}\frac{d^{2}x_{\rm ef}^{a}}{d^{2}\lambda}\Big|_{\lambda_{\rm e}}(\lambda_{\rm f} - \lambda_{\rm e})^{2} \\ &= x_{\rm e}^{a}(\tau_{\rm e}) + \zeta^{a}\Delta\lambda + \frac{1}{2}\Big(\zeta^{b}\partial_{b}\zeta^{a}\Big)\Delta\lambda^{2} , \end{aligned}$$
(3.46)

where we introduced the tangent vector

$$\frac{d}{d\lambda} = \frac{dx_{\rm ef}^a}{d\lambda} \Big|_{\lambda_{\rm e}} \frac{\partial}{\partial x^a} = \zeta^a \partial_a .$$
(3.47)

The first objects we want to perturb are the fluid element "names". That is, we attach a label X^A , where the index A = 1, 2, 3, to each of the worldlines used to cover the region of spacetime occupied by the fluid. By definition of the Lagrangian variation [82, 83] we then have

$$\Delta X^{A} = \left(\phi_{*}X^{A}(x_{f})\right)(x_{e}) - \bar{X}^{A}(x_{e}) = X^{A}(x_{f}) - \bar{X}^{A}(x_{e}) = 0, \qquad (3.48)$$

where ϕ is the diffeomorphism that connects the perturbed and unperturbed worldlines, via the flow lines x_{ef}^a , and and ϕ_* denotes the pull-back from perturbed to equilibrium manifold. The last equality then follows from the fact that the fluid label does not change as we follow it. As a result we have, to first order

$$\delta X^A = -\mathcal{L}_{\xi_{\mathbf{x}}} X^A = -\xi^a_{\mathbf{x}} \Psi^A_{\mathbf{e}\,a} = -\xi^A_{\mathbf{x}} \,, \tag{3.49}$$

where we introduced the Lagrangian displacement vector $\xi_x^a = x_f^a - x_e^a$.

It is important to note that these displacement vectors are different from the ones introduced when obtaining the equations of motion from the action principle (see eq. (2.56)), even though the mathematics appears the same. In the present case the displacement vector connects two configurations that are "close" in the space of physical solutions in field-theory parlance they are both "on-shell". We also note that, to compute the second order variation we cannot rely on the simple relation that exists between Lagrangian and Eulerian variation (at first order). We need to perform such calculations explicitly.

At this point, it is worth pausing to consider what is behind the perturbation scheme we are building. Since we assume the existence of a well defined equilibrium timelike congruence x_e^a with four velocity u_e^a , we may imagine riding along with the equilibrium fluid element observing the evolution of the system (towards equilibrium) from this perspective. This means that the x-species four-velocity u_x^a can be decomposed (in the usual way) as

$$u_{\rm x}^{a} = \gamma_{\rm x} \left(u_{\rm e}^{a} + w_{\rm x}^{a} \right)$$
, where $w_{\rm x}^{a} u_{a}^{\rm e} = 0$, $\gamma_{\rm x} = \left(1 - w_{\rm x}^{a} w_{a}^{\rm x} \right)^{-1/2}$. (3.50)

Moreover, since we are working up to first order we have

$$\gamma_{\rm x} = 1 + \frac{1}{2}w_{\rm x}^2 \approx 1 + \mathcal{O}(w_{\rm x}^2) \quad \Longrightarrow \quad u_{\rm x}^a = u_{\rm e}^a + w_{\rm x}^a \,. \tag{3.51}$$

. ...

We note that this linear expansion in the relative velocities, although in a different spirit, has also been discussed in the context of extensions to magneto-hydrodynamics [16, 24, 25].

Also, it is interesting in itself (and necessary for perturbing the full set of fluid equations) to understand the relation between the spatial velocity w_x^a as measured by the equilibrium observer and the Lagrangian displacement ξ_x^a . We consider the displacement to live in the local present of the equilibrium observer, i.e., to be such that $\xi_x^a u_a^e = \zeta_x^a u_a^e = 0.^{15}$ This implies that the vectors ξ_x^a and ζ_x^a are spacelike non-null vector fields in spacetime. As a result, if we consider the proper time of the perturbed worldline, we have

$$-d\bar{\tau}^2 = g_{ab} dx_{\rm f}^a dx_{\rm f}^b = g_{ab} dx_{\rm e}^a dx_{\rm e}^b + g_{ab} \left(dx_{\rm e}^a \zeta^b \Delta \lambda + dx_{\rm e}^b \zeta^a \Delta \lambda \right) = -d\tau^2 , \qquad (3.52)$$

where we used the fact that

$$x_{\rm e}^a = x_{\rm e}^a(\tau) \Longrightarrow dx_{\rm e}^a = u_{\rm e}^a d\tau .$$
(3.53)

¹⁵This is essentially a gauge choice, see [21] for discussion.

As a consequence, the proper time of the perturbed and equilibrium worldline is the same, so we have

$$u_{\rm x}^a = \frac{dx_{\rm f}^a}{d\bar{\tau}} \approx \frac{dx_{\rm e}^a}{d\tau} + \frac{d}{d\tau} \xi_{\rm x}^a = u_{\rm e}^a + \dot{\xi}_{\rm x}^a , \qquad (3.54)$$

where (again) the dot represents the covariant directional derivative in the direction of the equilibrium four-velocity.¹⁶ We observe that from the construction we have $w_x^a = \dot{\xi}_x^a$ and it is clear that when pushing the expansion to second order their relation will become more involved—both because the difference between the proper times ($\bar{\tau}$ versus τ) appears at second order and because the Taylor expansion gets more complicated.

We now aim to understand how to construct the expansion directly in matter space. We start by noting that, since we are considering each displacement ξ_x^a to be orthogonal to u_e^a there is no loss of information in projecting the Lagrangian displacements onto the equilibrium matter space and dealing with ξ_x^A . The general picture is thus as follows: in the general non-linear theory each matter space can be considered as an independent but interacting manifold, but this changes when we consider a perturbative expansion. In fact, the fundamental assumption of perturbation theory is that the two configurations (perturbed and unperturbed) are related by some diffeomorphism. This implies that the perturbed and unperturbed matter spaces¹⁷ are diffeomorphic, that is they are *the same* abstract manifold. Therefore we can use the same chart on the two manifolds X^A (label the worldlines in the same way) and the difference will be only in that $X_x^A(x^a) \neq X_e^A(x^a)$. The difference between the two will be exactly what we found above, namely $-\xi_x^A$. We also note that, by our definition of the unperturbed state, all the perturbed matter spaces are diffeomorphic to the same unperturbed one, and thus to each other.

Given this, we can work out how a general matter space tensor transforms under diffeomorphisms [50]. For instance, if we consider the projected metric g_x^{AB} we have¹⁸

$$\delta g_{\mathbf{x}}^{AB} = -\mathcal{L}_{-\xi_{\mathbf{x}}} g_{\mathbf{x}}^{AB} = \mathcal{L}_{\xi_{\mathbf{x}}} g_{\mathbf{x}}^{AB} = \xi_{\mathbf{x}}^{C} \partial_{C} g_{\mathbf{e}}^{AB} - g_{\mathbf{e}}^{CB} \partial_{C} \xi_{\mathbf{x}}^{A} - g_{\mathbf{e}}^{AC} \partial_{C} \xi_{\mathbf{x}}^{B} , \qquad (3.55)$$

where the partial derivatives are taken with respect to the equilibrium matter space coordinates. We now observe that, considering ξ_x^A as a scalar field in spacetime we can write

$$-g_{e}^{CB}\partial_{C}\xi^{A} = -g^{ab}\Psi_{ea}^{C}\Psi_{eb}^{B}\partial_{C}\xi_{x}^{A} = -\Psi_{eb}^{B}\nabla^{b}\xi_{x}^{A}.$$
(3.56)

¹⁶To be more precise, one should distinguish between $\frac{d}{d\tau} = u_e^b \partial_b$ and $\frac{D}{D\tau} = u_e^b \nabla_b$. Since we are introducing a decomposition of a vector as a sum of two, $\dot{\xi}_x^a$ must be a vector as well so that the dot represents a covariant directional derivative.

¹⁷Recall that the matter space is obtained by taking the quotient of the spacetime over the corresponding worldline, i.e. identifying the worldline as a single point.

¹⁸For the Lie derivative we use the formula with partial derivatives in order to avoid the possible confusion arising from the choice of the connection used on the matter space.

We also note that, since¹⁹ $\partial_C \Psi^A_{ea} = \partial_a \delta^A_C = 0$, we have

$$\partial_C g_e^{AB} = 2 g^{ab} \left(\frac{\partial}{\partial X_e^C} \Psi_{ea}^A \right) \Psi_{eb}^B = 0 .$$
(3.57)

As a result, the projected metrics transform as

$$\delta g_{\mathbf{x}}^{AB} = -\Psi^B_{\mathbf{e}a} \nabla^a \xi^A_{\mathbf{x}} - \Psi^A_{\mathbf{e}a} \nabla^a \xi^B_{\mathbf{x}} .$$
(3.58)

This also tells us that building the variation of the metric tensor in this way, we are only comparing the difference in the position of the particles, keeping fixed the spacetime metric.

We can now use the definition in eq. (3.29) to decompose the displacement gradients as

$$\nabla_a \xi^A_{\mathbf{x}} = -w^A_{\mathbf{x}} u^{\mathbf{e}}_a + D_a \xi^A_{\mathbf{x}} , \qquad (3.59)$$

and rewrite

$$\delta g_{x}^{AB} = \Psi_{ea}^{B}(w_{x}^{A}u_{e}^{a} - D^{a}\xi_{x}^{A}) + \Psi_{ea}^{A}(w_{x}^{B}u_{e}^{a} - D^{a}\xi_{x}^{B})$$

= $-D^{B}\xi_{x}^{A} - D^{A}\xi_{x}^{B}$, (3.60)

where we introduced the short-hand notation $D^A = \Psi^A_{a,b} g^{ab} D_a$. It is worth noting that eq. (3.60) is not a strain-rate tensor of the type usually introduced in fluid dynamics, because it involves gradients in the displacements instead of velocities. The usual strain rate tensor is in fact²⁰

$$\dot{g}_{x}^{AB} = -2 \Psi_{x(a}^{A} \Psi_{xb)}^{B} \left[-u_{x}^{b} \dot{u}_{x}^{a} + \omega_{x}^{ab} + \sigma_{x}^{ab} + \frac{1}{3} \theta_{x} \perp_{x}^{ab} \right] = \\ = -2 \Psi_{e(a}^{A} \Psi_{eb)}^{B} \left(\sigma_{x}^{ab} + \frac{1}{3} \theta_{x} \perp_{e}^{ab} \right) + \mathcal{O}(2) = -2 \left(\sigma_{x}^{AB} + \frac{1}{3} \theta_{x} g_{e}^{AB} \right) , \qquad (3.61)$$

We will comment on the implications of this difference later.

Even if it is not entirely obvious what kind of object the mixed projected metric g_{xy}^{AB} is in the general non-linear case, in the context of a perturbative expansion there is no real difference between the various matter spaces (they are all diffeomorphic to the equilibrium one). This means that we can use the same fundamental formula also for g_{xy} to get

$$\delta g_{xy}^{AB} = g_{xy}^{AB} - g_{e}^{AB} = g^{ab} \left(\delta \Psi_{xa}^{A} \Psi_{eb}^{B} + \delta \Psi_{yb}^{B} \Psi_{ea}^{A} \right) =$$
$$= -\Psi_{ea}^{B} \nabla^{a} \xi_{x}^{A} - \Psi_{ea}^{A} \nabla^{a} \xi_{y}^{B} .$$
(3.62)

¹⁹If this is not immediately convincing one can prove it by taking the explicit definition of a derivative on the coordinate functions $X^A(\bar{X}) = \delta^A_C \bar{X}^C = \bar{X}^A$ and using the linearity of the derivative. ²⁰To see this one has to use $\mathcal{L}_{u_x} \Psi^A_{\mathbf{x}a} = 0$.

It is interesting to note that since $\delta g^{ab} = 0$ we have

$$[\delta, \nabla_a] = [\delta, \partial_a] = 0.$$
(3.63)

That is, the variation commutes with both partial and covariant derivatives. This will become relevant when we need to work out the variation of the resistive terms that stem from a dependence of the N_x on g_{xy}^{AB} and g_y^{AB} .

As there has been a number of recent efforts aimed at building first-order dissipative hydrodynamic models starting from a field-theory perspective (cf. section 2.4), it makes sense to point out the differences between the present expansion and the field-theory-based ones. In that context, the models are said to be of first order if the constitutive equations involve all permissible terms with just one derivative—as when the system is close to equilibrium one can expect the gradients in temperature, chemical potential etc... to be small, so that terms with two or more derivatives are dominated by first-order ones. In contrast, we here assume the variables that define the physical state of the system to take values close to the equilibrium ones, and by "first order" we mean the deviations are expanded up to $\mathcal{O}(\xi_x)$. It is therefore clear that the present approach differs from the field-theory-based (gradient) expansions. The ultimate reason is that the action-based model provides the exact equations, which we then approximate, while in the field-theory approach one is trying to build the full equations as successive expansions.

3.4 Energy density is stationary at equilibrium

As discussed in section 2.1.2, in order to describe out-of-equilibrium systems with the Extended Irreversible Thermodynamic (EIT) paradigm, one postulates the existence of a generalized entropy—a function of a larger set of Degrees of Freedom than the corresponding equilibrium ones—which is maximized at equilibrium. The starting point for the formalism used here is a generalized energy where the only degrees of freedom are the fluxes. The action-based model provides the total stress-energy-momentum tensor T_{ab} of the system, so that we can easily extract the total energy density ε for some observer having four-velocity u^a via the projection $\varepsilon = u^a u^b T_{ab}$. We will now show that requiring the local energy density to be at a minimum in equilibrium means the viscous stress tensors have to be zero.

When specific modeling is carried out, such as a numerical evolution, we would need to provide an equation of state and specify values for the microphysical input parameters. From the phenomenological point of view, this corresponds to assuming the existence of a function—in our case, energy density—defined on some "thermodynamical manifold" whose coordinates are the relevant degrees of freedom. Practically speaking, the formalism developed here suggests we may identify the thermodynamical manifold

with the matter space used in the variational model. As the general discussion gets quite complex, we focus on the specific example of a two-component system, with the components representing matter and entropy (see [142, 22]).

Let us first consider the non-dissipative limit. Thermodynamics of a single fluid is described by some equilibrium energy $\varepsilon_e(n, s)$ such that

$$d\varepsilon_{\rm e} = Tds + \mu dn = \sum_{\rm x=n,s} \mu^{\rm x} dn_{\rm x} .$$
(3.64)

On the other hand, the conservative variational model is built using a master function $\Lambda(n_n^2, n_s^2, n_{ns}^2)$. Because of our assumption that all species are comoving in equilibrium there is no heat flux relative to the matter and therefore $n_{ns}^2 = -g_{ab}n_n^a n_s^b = +n_n n_s$, and the master function only depends on two variables, $\Lambda_e = \Lambda_e(n_n, n_s)$. It is indeed easy to see that the equilibrium energy density, as measured by the equilibrium observer, is

$$\varepsilon_{\rm e} = T^{\rm e}_{ab} \, u^a_{\rm e} u^b_{\rm e} = \left[\Psi_{\rm e} g_{ab} + (\Psi_{\rm e} - \Lambda_{\rm e}) u^a_a u^e_b \right] u^a_{\rm e} u^b_{\rm e} = -\Lambda_{\rm e} \,. \tag{3.65}$$

Since we have already identified the matter space normalizations of the three-forms with the rest frame densities $N_x = n_x$, we can think of the thermodynamic energy as a function defined on the matter space, and write

$$\varepsilon_{\rm e} = \varepsilon_{\rm e}(\mathcal{N}_{\rm n}, \mathcal{N}_{\rm s}) = -\Lambda_{\rm e}(\mathcal{N}_{\rm n}, \mathcal{N}_{\rm s}) . \tag{3.66}$$

The equilibrium case suggests that we could try to extend this identification to the nonequilibrium setting, and "build" the thermodynamics on the matter space. This raises the (difficult) question of what the global matter space is in the full non-linear case. We will not address that issue here. Instead, we focus on the near-equilibrium case, where we only have to deal with the equilibrium matter space.

Because of the way we have built the expansion, it is natural to project tensor quantities —fluxes, stress-energy-momentum tensor, etcetera—into the frame of the equilibrium observer, as defined by the equilibrium worldlines congruence u_e^a . Quantities measured in this frame will be indicated by a "hat" in the following. Objects without a hat are measured in fluid rest frames, which are defined by the u_x^a . The equilibrium value of a quantity in the equilibrium frame will be indicated with a "bar". For instance, the particle density measured in the equilibrium frame is $\hat{n}_x = -u_a^e n_x^a$; in the x-fluid rest frame it is $n_x = -u_a^x n_x^a$; and the equilibrium value in the equilibrium frame is $\bar{n}_x = \hat{n}_x|_e$.

The "out-of-equilibrium" energy density $\hat{\epsilon}_{o.e.}$ of the system as determined in the equilibrium rest frame is given by

$$\hat{\varepsilon}_{\text{o.e.}} = \left(T_{\text{n.d.}}^{ab} + \sum_{x} D_{x}^{ab}\right) u_{a}^{e} u_{b}^{e} = \hat{\varepsilon}_{\text{o.e.}}^{n.d.} + D^{ab} u_{a}^{e} u_{b}^{e} , \qquad (3.67)$$

where we have separated the contribution from the viscous stress tensor D_{ab} from those having the "non-dissipative" form

$$T_{n.d.}^{ab} = \left(\Lambda - \sum_{x} n_{x}^{c} \mu_{c}^{x}\right) g^{ab} + \sum_{x} n_{x}^{a} \mu_{x}^{b} = \Psi g^{ab} + \sum_{x} n_{x}^{a} \mu_{x}^{b} .$$
(3.68)

The expression for $\hat{\varepsilon}_{o.e.}$ can be made more explicit by means of eq. (3.38), which leads to $\Psi = \Lambda + \sum_{x} n_{x} \mu_{x}$ and

$$\hat{\varepsilon}_{\text{o.e.}}^{\text{n.d.}} = u_a^{\text{e}} u_b^{\text{e}} T_{\text{n.d.}}^{ab} = -\Lambda - \sum_{\mathbf{x}=n,s} \left(n_{\mathbf{x}} \mu_{\mathbf{x}} - \hat{n}_{\mathbf{x}} \hat{\mu}_{\mathbf{x}} \right) \,. \tag{3.69}$$

Because the flux is a vector, the two densities \hat{n}_x and n_x are easily shown to be related by

$$\hat{n}_{x} = -n_{x}^{a}u_{a} = -n_{x}u_{x}^{a}u_{a} = (1 - w_{x}^{a}w_{a}^{x})^{-1/2}n_{x} = \left(1 + \frac{1}{2}w_{x}^{2}\right)n_{x} + \mathcal{O}(w_{x}^{3}).$$
(3.70)

Meanwhile, the corresponding momentum relation is a bit more involved because of entrainment:

$$\mu_{x} = -\mu_{b}^{x}u_{x}^{b} = -\gamma_{x}(u^{b} + w_{x}^{b})\left(\mathcal{B}_{x}n_{x}u_{b}^{x} + \sum_{y \neq x}\mathcal{A}_{xy}n_{y}u_{y}^{b}\right)$$
$$= \gamma_{x}\left(\hat{\mu}_{x} - \mathcal{B}_{x}n_{x}\gamma_{x}w_{x}^{2} - \sum_{y \neq x}\mathcal{A}_{xy}n_{y}\gamma_{y}w_{x}^{a}w_{a}^{y}\right).$$
(3.71)

We can rearrange this as

$$\hat{\mu}_{x} = \mu_{x} - \frac{1}{2}\bar{\mu}_{x}w_{x}^{2} + \bar{\mathcal{B}}_{x}\bar{n}_{x}w_{x}^{2} + \sum_{y \neq x}\bar{\mathcal{A}}_{xy}\bar{n}_{y}w_{x}^{a}w_{a}^{y} , \qquad (3.72)$$

and, wrapping up, we get

$$\hat{\varepsilon}_{\text{o.e.}}^{n.d.} = -\Lambda + \bar{\mathcal{B}}_{n}\bar{n}_{n}^{2}w_{n}^{2} + \bar{\mathcal{B}}_{s}\bar{n}_{s}^{2}w_{s}^{2} + 2\bar{\mathcal{A}}_{ns}\bar{n}_{s}\bar{n}_{n}w_{n}^{a}w_{a}^{s}$$

$$= -\Lambda + \bar{\mu}_{n}\bar{n}_{n}w_{n}^{2} + \bar{\mu}_{s}\bar{n}_{s}w_{s}^{2} - \mathcal{A}_{ns}\bar{n}_{n}\bar{n}_{s}w_{ns}^{2} , \qquad (3.73)$$

where

$$w_{\rm xy}^2 = g_{ab} \left(w_{\rm x}^a - w_{\rm y}^a \right) \left(w_{\rm x}^b - w_{\rm y}^b \right) \,. \tag{3.74}$$

It is now clear that, in order to proceed, we need an expansion for the master function, Λ .

Note that the dissipative action model assumes Λ depends on $(X_n^A, X_s^A, g_n^{AB}, g_s^{AB}, g_{ns}^{AB})$ through the scalar product of the fluxes n_n^2, n_s^2, n_{ns}^2 . Therefore, we can expand Λ up to

second order in the standard way (see [21]). We thus have

$$\Lambda = \Lambda_{\rm e} - \frac{1}{2} \sum_{{\rm x}=n,s} \mathcal{B}_{\rm x} \delta n_{\rm x}^2 - \mathcal{A}_{\rm ns} \delta n_{\rm ns}^2 - \frac{1}{4} \sum_{{\rm x}=n,s} \frac{\partial \mathcal{B}_{\rm x}}{\partial n_{\rm x}^2} (\delta n_{\rm x}^2)^2 - \frac{1}{2} \frac{\partial \mathcal{A}_{\rm ns}}{\partial n_{\rm ns}^2} (\delta n_{\rm ns}^2)^2 - \frac{1}{2} \frac{\partial \mathcal{B}_{\rm n}}{\partial n_{\rm s}^2} (\delta n_{\rm n}^2) (\delta n_{\rm s}^2) - \frac{\partial \mathcal{A}_{\rm ns}}{\partial n_{\rm n}^2} (\delta n_{\rm ns}^2) (\delta n_{\rm ns}^2) - \frac{\partial \mathcal{A}_{\rm ns}}{\partial n_{\rm s}^2} (\delta n_{\rm s}^2) (\delta n_{\rm ns})^2 .$$
(3.75)

To make contact with the previous expansion on the matter space we need explicit expressions for δn_x^2 and all other similar terms that appear in this expression.

For the four-current we have

$$\delta n_{\rm x}^{a} = n_{\rm x}^{a} - \bar{n}_{\rm x}^{a} = (\bar{n}_{\rm x} + \delta n_{\rm x}) \left[\left(1 + \frac{1}{2} w_{\rm x}^{2} \right) u^{a} + w_{\rm x}^{a} \right] - \bar{n}_{\rm x} u^{a} = \frac{1}{2} \bar{n}_{\rm x} w_{\rm x}^{2} u^{a} + \bar{n}_{\rm x} w_{\rm x}^{a} + \delta n_{\rm x} u^{a} + \delta n_{\rm x} w_{\rm x}^{a} , \qquad (3.76)$$

and we see that it—quite intuitively—changes both as the density and the four-velocity change. By means of eq. (3.76) we get

$$\delta n_{\rm x}^2 = -\left(2\bar{n}_{\rm x}^a \delta n_a^{\rm x} + \delta n_{\rm x}^a \delta n_a^{\rm x}\right) = 2\bar{n}_{\rm x} \delta n_{\rm x} + (\delta n_{\rm x})^2 . \tag{3.77}$$

Similarly, we have

$$\delta n_{\rm xy}^2 = -\left(\bar{n}_{\rm x}^a \delta n_a^{\rm y} + \bar{n}_{\rm y}^a \delta n_a^{\rm x} + \delta n_{\rm x}^a \delta n_a^{\rm y}\right)$$
$$= \bar{n}_{\rm x} \delta n_{\rm y} + \bar{n}_{\rm y} \delta n_{\rm x} + \delta n_{\rm x} \delta n_{\rm y} + \frac{1}{2} \bar{n}_{\rm x} \bar{n}_{\rm y} w_{\rm xy}^2 \,. \tag{3.78}$$

In order to complete the second order expansion of Λ we also need the products (for every possible combination) of eq. (3.77) and eq. (3.78). These are found to be

$$(\delta n_{\rm x}^2)^2 = 4\bar{n}_{\rm x}^2 (\delta n_{\rm x})^2$$
, (3.79a)

$$(\delta n_{xy}^2)^2 = \bar{n}_x^2 (\delta n_y)^2 + \bar{n}_y^2 (\delta n_x)^2 + 2\bar{n}_x \bar{n}_y \delta n_x \delta n_y , \qquad (3.79b)$$

$$(\delta n_x^2)(\delta n_y^2) = 4\bar{n}_x \bar{n}_y \delta n_x \delta n_y , \qquad (3.79c)$$

$$(\delta n_{\rm xy}^2)(\delta n_{\rm x}^2) = 2\bar{n}_{\rm x}(\delta n_{\rm x}) \left(\bar{n}_{\rm y} \delta n_{\rm x} + \bar{n}_{\rm x} \delta n_{\rm y} \right) \,. \tag{3.79d}$$

Plugging these expressions into eq. (3.75) we find (up to second order)

$$\hat{\varepsilon}_{\text{o.e.}}^{n.d.} = \varepsilon_{e}(\bar{n}_{n}, \bar{n}_{s}) + \bar{\mu}_{n}\delta n_{n} + \bar{\mu}_{s}\delta n_{s} + \frac{1}{2}(\bar{\mathcal{B}}_{n}\bar{c}_{n}^{2} - \bar{\mathcal{A}}_{uu}^{nn})(\delta n_{n})^{2} \\
+ \frac{1}{2}(\bar{\mathcal{B}}_{s}\bar{c}_{s}^{2} - \bar{\mathcal{A}}_{uu}^{ss})(\delta n_{s})^{2} - (\bar{\chi}_{uu}^{sn} + \bar{\mathcal{A}}_{uu}^{ns})(\delta n_{n})(\delta n_{s}) + \bar{\mu}_{n}\bar{n}_{n}w_{n}^{2} \\
+ \bar{\mu}_{s}\bar{n}_{s}w_{s}^{2} - \frac{1}{2}\bar{\mathcal{A}}_{ns}\bar{n}_{n}\bar{n}_{s}w_{ns}^{2},$$
(3.80)

where we have made use of eq. (3.73) and defined (see Andersson and Comer [21])

$$\bar{c}_{\rm x}^2 = 1 + 2\frac{\bar{n}_{\rm x}^2}{\bar{\mathcal{B}}_{\rm x}}\frac{\partial\bar{\mathcal{B}}_{\rm x}}{\partial n_{\rm x}^2}\,,\tag{3.81a}$$

$$\bar{\mathcal{A}}_{ab}^{xx} = -\left(\bar{n}_{y}^{2} \frac{\partial \bar{\mathcal{A}}_{xy}}{\partial n_{xy}^{2}} + 4\bar{n}_{x}\bar{n}_{y} \frac{\partial \bar{\mathcal{A}}_{xy}}{\partial n_{x}^{2}}\right) u_{a}^{e} u_{b}^{e} \doteq \bar{\mathcal{A}}_{uu}^{xx} u_{a}^{e} u_{b}^{e} , \qquad (3.81b)$$

$$\bar{\mathcal{A}}_{ab}^{\mathrm{ns}} = \bar{\mathcal{A}}^{ns} \perp_{ab} - \left(\bar{\mathcal{A}}_{\mathrm{ns}} + 2\bar{n}_{\mathrm{n}}^{2} \frac{\partial \bar{\mathcal{A}}^{\mathrm{ns}}}{\partial n_{\mathrm{n}}^{2}} + 2\bar{n}_{\mathrm{s}}^{2} \frac{\partial \bar{\mathcal{A}}^{\mathrm{ns}}}{\partial n_{\mathrm{s}}^{2}} + \bar{n}_{\mathrm{n}} \bar{n}_{\mathrm{s}} \frac{\partial \bar{\mathcal{A}}^{\mathrm{ns}}}{\partial n_{\mathrm{ns}}^{2}}\right) u_{a}^{e} u_{b}^{e}$$
$$\stackrel{:}{=} \bar{\mathcal{A}}^{\mathrm{ns}} + u_{a}^{e} u_{b}^{e} u_{b}^$$

$$= \mathcal{A}^{-} \perp_{ab} + \mathcal{A}_{uu} u_{a} u_{b}^{-} , \qquad (3.81c)$$

$$\bar{\chi}_{uu}^{\rm ns} = -2\bar{n}_{\rm n}\bar{n}_{\rm s}\frac{\partial \mathcal{B}^{\rm n}}{\partial n_{\rm s}^2} = -2\bar{n}_{\rm n}\bar{n}_{\rm s}\frac{\partial \mathcal{B}^{\rm s}}{\partial n_{\rm n}^2}\,.$$
(3.81d)

Noting that the quantity δn_x is the variation of the rest frame density, we can relate it to a variation of \mathcal{N}_x and "close the loop". Since the \mathcal{N}_x are functions on matter space of the variables (X_n , X_s , g_n^{AB} , g_s^{AB} , g_{ns}^{AB}) the expression for the energy is actually a second order expansion in terms of those variables. We note also that, because of the "two-layer structure", the δn_x above contain second-order terms.

A priori, the expression in eq. (3.80) does not provide the total out-of-equilibrium energy because we also need to account for the dissipative terms. However, we will now show that these actually do not contribute. To do this, we assume an expansion for all the viscous stress tensors of form

$$S_{AB} = S_{AB}^{e} + S_{AB}^{1} + S_{AB}^{2} + \mathcal{O}(\xi^{3}) , \qquad (3.82)$$

without providing (for now) the explicit expressions. Recalling $\Psi_{ea}^{A}u_{e}^{a} = 0$, we can write

$$S_{ab}u_{e}^{a}u_{e}^{b} = S_{AB}(X_{e}^{A} + \delta X^{A})_{,a}(X_{e}^{B} + \delta X^{B})_{,b}u_{e}^{a}u_{e}^{b} = S_{AB}^{e}\,\delta X_{,a}^{A}\,\delta X_{,b}^{B}u_{e}^{a}u_{e}^{b}\,, \qquad (3.83)$$

where the expansion is up to second order. It is clear that this argument is valid for each viscous stress tensor, and for $D_{ab}u_e^a u_e^b$ as well, so that the dissipative contributions to the off-equilibrium energy are, at least, of second order. Assuming that the energy is stationary, that is

$$\hat{\varepsilon}_{\text{o.e.}}^{n.d.} - \varepsilon_{\text{e}}(\bar{n}_{\text{n}}, \bar{n}_{\text{s}}) = \mathcal{O}(\xi^2) , \qquad (3.84)$$

we then have

$$\bar{\mu}_{\rm n}\delta n_{\rm n} + \bar{\mu}_{\rm s}\delta n_{\rm s} = \mathcal{O}(\xi^2) , \qquad (3.85)$$

which has a clear thermodynamical interpretation and is consistent with the EIT picture, since, up to first order, the generalized energy is a function of the n_x only.

We want to translate the above result into conditions for the matter space functions N_x . We start by observing that in the conservative case, the three-form n_{ABC}^x is a function of the X_x^A coordinates only. Therefore, \bar{N}_x is just a function of X_x^A , while, because $N_x =$ $\bar{\mathcal{N}}_x \sqrt{g^x}$, the latter depends also on the projected metric

$$\frac{\partial \mathcal{N}_{x}}{\partial g_{x}^{AB}} = \frac{1}{2} \sqrt{g^{x}} \bar{\mathcal{N}}_{x} g_{AB}^{x} = \frac{1}{2} \mathcal{N}_{x} g_{AB}^{x} .$$
(3.86)

When considering the expansion of n_x (and hence \mathcal{N}_x) we assume that we can write

$$\mathcal{N}_{\mathrm{x}} = \mathcal{N}_{\mathrm{x}}^{\mathrm{e}} + \mathcal{N}_{\mathrm{x}}^{\mathrm{d}} , \qquad (3.87)$$

where \mathcal{N}_x^e is the same as in the non-dissipative limit while the dissipative contribution \mathcal{N}_x^d is a function also of the additional variables that encode the dissipation. Given the separation of \mathcal{N}_x into two pieces it is natural to assume that \mathcal{N}_x^d , but not its derivatives, vanishes at equilibrium.

Since the equilibrium evolves in a conservative fashion, we can write

$$\delta n_{x} \equiv \mathcal{N}_{x} - \mathcal{N}_{x}^{e} = \mathcal{N}_{x}^{d} = \mathcal{N}_{x}^{d} - \mathcal{N}_{x}^{d}\Big|_{e}$$
$$= \frac{\partial \mathcal{N}_{x}^{d}}{\partial X_{x}^{A}} \delta X_{x}^{A} + \frac{\partial \mathcal{N}_{x}^{d}}{\partial X_{y}^{A}} \delta X_{y}^{A} + \frac{\partial \mathcal{N}_{x}^{d}}{\partial g_{x}^{AB}} \delta g_{x}^{AB} + \frac{\partial \mathcal{N}_{x}^{d}}{\partial g_{y}^{AB}} \delta g_{y}^{AB} + \frac{\partial \mathcal{N}_{x}^{d}}{\partial g_{xy}^{AB}} \delta g_{xy}^{AB} + \mathcal{O}(2) , \quad (3.88)$$

where here, and in similar expansions below, each quantity is to be evaluated at equilibrium. With this assumption it is easy to read off from eq. (3.85) the first order relation

$$\mathcal{M}_{n}\delta\mathcal{N}_{n}^{d} + \mathcal{M}_{s}\delta\mathcal{N}_{s}^{d} = 0.$$
(3.89)

This leads to

$$\mathcal{M}_{n}\frac{\partial \mathcal{N}_{n}^{d}}{\partial X_{n}^{A}} + \mathcal{M}_{s}\frac{\partial \mathcal{N}_{s}^{d}}{\partial X_{n}^{A}} = 0 , \qquad (3.90)$$

and analogous results for variations with respect to X_s^A , g_n^{AB} , g_s^{AB} and g_{ns}^{AB} follow immediately. In particular, this shows that the total viscous stress tensor, acting on each component $D_{ab'}^x$, vanishes when the energy is stationary.

To see this explicitly we note that (see eqs. (3.14) and (3.15))

$$S_{AB}^{\rm xy,e} \equiv 2\mathcal{M}_{\rm x} \frac{\partial \mathcal{N}_{\rm x}}{\partial g_{\rm xy}^{AB}} = -2\mathcal{M}_{\rm y} \frac{\partial \mathcal{N}_{\rm y}}{\partial g_{\rm xy}^{AB}} = -\mathcal{S}_{BA}^{\rm yx,e} , \qquad (3.91)$$

where we made use of the symmetry property of the mixed metric, namely $g_{xy}^{AB} = g_{yx}^{BA}$. Similarly,

$$S_{AB}^{x,e} \equiv 2\mathcal{M}_{x} \left(\frac{\partial \mathcal{N}_{x}}{\partial g_{x}^{AB}} - \frac{1}{2} \mathcal{N}_{x} g_{AB}^{x} \right)$$
$$= 2\mathcal{M}_{x} \left[\frac{\partial \mathcal{N}_{x}^{d}}{\partial g_{x}^{AB}} - \frac{1}{2} (\mathcal{N}_{x} - \mathcal{N}_{x}^{e}) g_{AB}^{x} \right]$$
$$= 2\mathcal{M}_{x} \frac{\partial \mathcal{N}_{x}^{d}}{\partial g_{x}^{AB}} = -2\mathcal{M}_{y} \frac{\partial \mathcal{N}_{y}^{d}}{\partial g_{x}^{AB}} = -s_{AB}^{yx,e} .$$
(3.92)

It is now clear that, by means of eq. (3.91) and eq. (3.92), the x-species viscous stress tensor vanishes:

$$D_{AB}^{x,e} = S_{AB}^{x,e} + s_{AB}^{yx,e} + \frac{1}{2}(S_{AB}^{xy,e} + S_{BA}^{yx,e}) = 0.$$
(3.93)

We have considered the fully general case with all the additional dependences in \mathcal{N}_x and all the viscous tensors S_{ab}^x , \mathcal{S}_{ab}^{xy} and s_{ab}^{xy} . The same result—that is, each $D_{ab}^{x,e}$ vanishes —holds even in a less rich situation when the model is built from fewer viscous tensors. In that case we have to go back to eq. (3.89) and modify the argument accordingly. It is important to stress that we have shown that the full stress-energy-momentum tensor at equilibrium is made out of just the non-dissipative part, and that the dissipative parts of the total stress-energy-momentum tensor do not contribute to the total energy density at second order.

However, we note that the energy minimum conditions in eq. (3.89) do not set the purely resistive terms to zero (eq. (3.13)). In fact, it only leads to

$$\mathcal{M}_{n}\frac{\partial \mathcal{N}_{n}^{d}}{\partial X_{n}^{A}} = -\mathbf{R}_{A}^{\mathrm{sn}, e} , \qquad (3.94a)$$

$$\mathcal{M}_{\rm s} \frac{\partial \mathcal{N}_{\rm s}^{\rm d}}{\partial X_{\rm s}^{\rm A}} = -\mathbf{R}_{A}^{\rm ns, e} \,. \tag{3.94b}$$

The reason for this is pretty clear as these terms do not enter the expression for the energy density. We nonetheless might want to consider the case where the equilibrium equations are exactly as the conservative ones. The motivation for this can be found in the derivation of the purely resistive terms. If the different species are comoving at the action level, there is no distinction between the different X_x^A and no resistive term of this form would appear. We can enforce consistency with this observation in two ways: either we assume that we use the complete dependence on X_x^A in the conservative part, in which case

$$\frac{\partial \mathcal{N}_{x}^{d}}{\partial X_{x}^{A}}\Big|_{e} = 0 \Longrightarrow R_{A}^{xy,e} = 0 , \qquad (3.95)$$

or, we just set the terms R_a^x to zero, so that

$$\mathcal{M}_{n} \frac{\partial \mathcal{N}_{n}^{d}}{\partial X_{n}^{A}}\Big|_{e} = \mathcal{M}_{s} \frac{\partial \mathcal{N}_{s}^{d}}{\partial X_{s}^{A}}\Big|_{e} .$$
(3.96)

The latter, less restrictive assumption reminds us of the dynamical nature of chemical equilibrium in nature. Reactions happen also at equilibrium, although they do so in such a way that there is no net particle production.

Finally, it is quite easy to see that if we choose a different observer, such as the ones associated with the Eckart or Landau frame, the differences in the energy density will be of second order. Crucially, the equilibrium conditions in eq. (3.89) do not depend on the choice of frame.

3.5 The last piece of the puzzle

In order to work out the perturbative expressions we need to expand the various dissipative terms. It should now be clear that for the viscous stress tensors we can write²¹

$$\delta s_{AB}^{xy} = 2 \frac{\partial \mathcal{N}_x^d}{\partial g_y^{AB}} \delta \mathcal{M}_x + 2 \mathcal{M}_x \delta \left(\frac{\partial \mathcal{N}_x^d}{\partial g_y^{AB}} \right) , \qquad (3.97a)$$

$$\delta S_{AB}^{xy} = 2 \frac{\partial \mathcal{N}_{x}^{d}}{\partial g_{xy}^{AB}} \delta \mathcal{M}_{x} + 2 \mathcal{M}_{x} \delta \left(\frac{\partial \mathcal{N}_{x}^{d}}{\partial g_{xy}^{AB}} \right) , \qquad (3.97b)$$

$$\delta S_{AB}^{x} = 2 \frac{\partial \mathcal{N}_{x}^{d}}{\partial g_{x}^{AB}} \delta \mathcal{M}_{x} + 2 \mathcal{M}_{x} \delta \left(\frac{\partial \mathcal{N}_{x}^{d}}{\partial g_{x}^{AB}} \right) - \mathcal{M}_{x} (\delta \mathcal{N}_{x}^{d}) g_{AB}^{e} , \qquad (3.97c)$$

where we recall that

$$s_{AB}^{\rm xy} = 2\mathcal{M}_{\rm x} \frac{\partial \mathcal{N}_{\rm x}^{\rm d}}{\partial g_{\rm y}^{\rm AB}} ,$$
 (3.98a)

$$S_{AB}^{\rm xy} = 2\mathcal{M}_{\rm x} \frac{\partial \mathcal{N}_{\rm x}^{\rm d}}{\partial g_{\rm xy}^{AB}} ,$$
 (3.98b)

$$S_{AB}^{\mathsf{x}} = 2\left(\mathcal{M}_{\mathsf{x}}\frac{\partial\mathcal{N}_{\mathsf{x}}^{\mathsf{d}}}{\partial g_{\mathsf{x}}^{AB}} - \frac{1}{2}\mathcal{M}^{\mathsf{x}}\mathcal{N}_{\mathsf{x}}^{\mathsf{d}} g_{AB}^{\mathsf{x}}\right).$$
(3.98c)

Similarly, for the "purely resistive" terms we have

$$\delta \mathbf{R}_{A}^{\mathrm{xy}} = \frac{\partial \mathcal{N}_{x}^{\mathrm{d}}}{\partial X_{y}^{A}} \delta \mathcal{M}_{\mathrm{x}} + \mathcal{M}_{\mathrm{x}} \delta \left(\frac{\partial \mathcal{N}_{x}^{\mathrm{d}}}{\partial X_{y}^{A}} \right) \,. \tag{3.99}$$

Since \mathcal{N}_x^d is a function of $(X_x, X_y, g_x^{AB}, g_y^{AB}, g_{xy}^{AB})$, its derivatives are as well, so that we have

$$\delta\left(\frac{\partial\mathcal{N}_{x}^{d}}{\partial X_{y}^{A}}\right) = \frac{\partial^{2}\mathcal{N}_{x}^{d}}{\partial X_{x}^{B}\partial X_{y}^{A}}\delta X_{x}^{B} + \frac{\partial^{2}\mathcal{N}_{x}^{d}}{\partial X_{y}^{B}\partial X_{y}^{A}}\delta X_{y}^{B} + \frac{\partial^{2}\mathcal{N}_{x}^{d}}{\partial g_{x}^{BC}\partial X_{y}^{A}}\delta g_{x}^{BC} + \frac{\partial^{2}\mathcal{N}_{x}^{d}}{\partial g_{y}^{BC}\partial X_{y}^{A}}\delta g_{y}^{BC} + \frac{\partial^{2}\mathcal{N}_{x}^{d}}{\partial g_{xy}^{BC}\partial X_{y}^{A}}\delta g_{xy}^{BC} .$$
(3.100)

Similar results hold for the other variations that were not explicitly written in eq. (3.97) and eq. (3.99).

Concerning the purely resistive term we note that $u_x^a R_a^{yx} = 0$ by construction. Because we are doing an expansion with undetermined coefficients, we need to impose this by

²¹All the derivatives are intended to be evaluated at equilibrium.

hand at every order; specifically, at the linear level. This then leads to

$$\delta\left(u_{\mathbf{x}}^{a}\mathbf{R}_{a}^{\mathbf{y}\mathbf{x}}\right) = \mathbf{R}_{A}^{\mathbf{y}\mathbf{x},\mathbf{e}}\left(w_{\mathbf{x}}^{A} - \dot{\boldsymbol{\xi}}_{\mathbf{x}}^{A}\right) = 0 , \qquad (3.101)$$

so that not only do we have $w_x^a = \dot{\xi}_x^a$ but also $w_x^A = \dot{\xi}_x^A$. This then means that we must have²² $u_e^a \xi_x^b X_{e;ba}^D = 0$, which in turn implies that the orthogonality conditions for the viscous stress tensors

$$S_{ab}^{x}u_{x}^{a} = S_{ab}^{xy}u_{x}^{a} = s_{ab}^{xy}u_{y}^{a} = 0 , \qquad (3.102)$$

are automatically satisfied at linear order.

From eq. (3.97) we can find the expansion for the spacetime viscous tensors through

$$\delta S_{ab}^{x} = \delta S_{DE}^{x} \Psi_{ea}^{D} \Psi_{eb}^{E} - S_{DE}^{x} \left(\xi_{x,a}^{D} \Psi_{eb}^{E} + \Psi_{ea}^{D} \xi_{x,b}^{E} \right) , \qquad (3.103a)$$

$$\delta s_{ab}^{xy} = \delta s_{DE}^{xy} \Psi_{ea}^{D} \Psi_{eb}^{E} - s_{DE}^{xy} \left(\xi_{y,a}^{D} \Psi_{eb}^{E} + \Psi_{ea}^{D} \xi_{y,b}^{E} \right) , \qquad (3.103b)$$

$$\delta \mathcal{S}_{ab}^{\mathrm{xy}} = \delta \mathcal{S}_{DE}^{\mathrm{xy}} \Psi_{ea}^{D} \Psi_{eb}^{E} - \mathcal{S}_{DE}^{\mathrm{xy}} \left(\xi_{\mathrm{x},a}^{D} \Psi_{eb}^{E} + \Psi_{ea}^{D} \xi_{\mathrm{y},b}^{E} \right) , \qquad (3.103c)$$

while for the resistive terms associated with s_{ab}^{xy} and \mathcal{S}_{ab}^{xy} we have

$$\delta r_{a}^{\rm xy} = \frac{1}{2} \delta s_{DE}^{\rm xy} \nabla_{a} g_{\rm e}^{DE} - \frac{1}{2} s_{DE}^{\rm xy} \partial_{a} \left[g^{bc} (\xi_{\rm y,b}^{D} \Psi_{\rm e\,c}^{E} + \Psi_{\rm e\,b}^{D} \xi_{\rm y,c}^{E}) \right] , \qquad (3.104a)$$

$$\delta \mathcal{R}_{a}^{\mathrm{xy}} = \frac{1}{4} \delta \mathcal{S}_{DE}^{\mathrm{xy}} \nabla_{a} g_{\mathrm{e}}^{DE} - \frac{1}{2} \mathcal{S}_{DE}^{\mathrm{xy}} g^{bc} \left(\xi_{\mathrm{x},b}^{D} \nabla_{a} \Psi_{\mathrm{e}\,c}^{E} + \Psi_{\mathrm{e}\,b}^{D} \nabla_{a} \xi_{\mathrm{y},c}^{E} \right) , \qquad (3.104b)$$

where we made use of the fact that $[\delta, \nabla_a] = 0$ because of $\delta g^{ab} = 0$ (see the discussion at the end of section 3.3).

Having "understood" how we may perturb the terms R_a^x and D_{ab}^x , let us focus on the remaining pieces of the equation of motion. A quick look back at eq. (2.70) reveals that the only terms we still have to discuss are $\delta\Gamma_x$ and $\delta\mu_a^x$. For the particle creation rate we have (see eq. (3.76))

$$\delta\Gamma_{\rm x} = \nabla_a \delta n_{\rm x}^a = \dot{\delta n}_{\rm x} + \nabla_a (\bar{n}_{\rm x} w_{\rm x}^a) , \qquad (3.105)$$

while for the x-species momentum, we get

$$\delta\mu_a^{\mathbf{x}} = \delta(\mathcal{B}_{\mathbf{x}}n_{\mathbf{x}})u_b^{\mathbf{e}} + \bar{\mathcal{B}}_{\mathbf{x}}\bar{n}_{\mathbf{x}}w_b^{\mathbf{x}} + \sum_{\mathbf{y}\neq\mathbf{x}}\delta(\mathcal{A}_{\mathbf{xy}}n_{\mathbf{y}})u_b^{\mathbf{e}} + \bar{\mathcal{A}}_{\mathbf{xy}}\bar{n}_{\mathbf{y}}w_b^{\mathbf{y}} \,. \tag{3.106}$$

Using the fact that we identified M_x with μ_x we have

$$\delta \mathcal{M}_{\mathbf{x}} = \delta \left(-\mu_{\mathbf{x}}^{a} u_{a}^{\mathbf{x}} \right) = -\left(\bar{\mu}_{a}^{\mathbf{x}} w_{\mathbf{x}}^{a} + \delta \mu_{\mathbf{x}}^{a} u_{a}^{\mathbf{e}} \right)$$
$$= \delta \left(\mathcal{B}_{\mathbf{x}} n_{\mathbf{x}} + \sum_{\mathbf{y} \neq \mathbf{x}} \mathcal{A}_{\mathbf{xy}} n_{\mathbf{y}} \right), \qquad (3.107)$$

²²Here the semicolon is, as usual, a short-hand notation for covariant derivative $A_{;a} = \nabla_a A$.

and since \mathcal{B}_x and \mathcal{A}_{xy} are ultimately functions of n_x^2 and n_{xy}^2 , we may use

$$\delta \mathcal{B}_{x} = \left(2n_{x}\frac{\partial \mathcal{B}_{x}}{\partial n_{x}^{2}} + n_{y}\frac{\partial \mathcal{B}_{x}}{\partial n_{xy}^{2}}\right)\delta n_{x} + \left(2n_{y}\frac{\partial \mathcal{B}_{x}}{\partial n_{y}^{2}} + n_{x}\frac{\partial \mathcal{B}_{x}}{\partial n_{xy}^{2}}\right)\delta n_{y}, \qquad (3.108a)$$

$$\delta \mathcal{A}_{xy} = \left(2n_x \frac{\partial \mathcal{A}_{xy}}{\partial n_x^2} + n_y \frac{\partial \mathcal{A}_{xy}}{\partial n_{xy}^2}\right) \delta n_x + \left(2n_y \frac{\partial \mathcal{A}_{xy}}{\partial n_y^2} + n_x \frac{\partial \mathcal{A}_{xy}}{\partial n_{xy}^2}\right) \delta n_y , \qquad (3.108b)$$

in eq. (3.107). This way, making use of definitions in eq. (3.81), we arrive at

$$\delta \mathcal{M}_{\mathrm{x}} = \left(\bar{\mathcal{B}}_{\mathrm{x}} c_{\mathrm{x}}^{2} - \bar{\mathcal{A}}_{uu}^{\mathrm{xx}}\right) \delta n_{\mathrm{x}} - \left(\bar{\mathcal{A}}_{uu}^{\mathrm{ns}} + \bar{\chi}_{uu}^{\mathrm{ns}}\right) \delta n_{\mathrm{y}} , \qquad (3.109)$$

and we see that the parameters that enter the dissipative fluid equations are the entrainment coefficients (and first derivatives; that is, second order derivatives of $\Lambda(n_x^2, n_{xy}^2)$) and the (up to second order) derivatives of the function $\mathcal{N}_x(X_x, X_y, g_x^{AB}, g_y^{AB}, g_{xy}^{AB})$.

Having outlined the perturbative framework, it is natural to ask how many dissipative channels the (general) model contains. Or, to be more specific, how many "dissipation coefficients" would have to be determined from microphysics? According to the expansion scheme we have developed so far, the perturbative expressions for the dissipative terms will ultimately involve all second and first order derivatives of the \mathcal{N}_x^d when considered as functions of X_x , X_y , g_x^{AB} , g_y^{AB} , g_{xy}^{AB} . Also, to make use of the model we need to specify the entrainment coefficients and their derivatives in the combinations from eq. (3.81). This means that the most general model one can think of contains a large number of coefficients. However, they should be, in general, known once a specific model is chosen; that is, once the explicit functional forms of Λ and the \mathcal{N}_x^d have been provided. For example, if nuclear physics calculations are used to determine these explicit forms, they must be done in such a way that the constraints which arise from requiring a meaningful equilibrium configuration are taken into account, and they must ensure that the second law of thermodynamics is obeyed. If Onsager-type reasoning is invoked to ensure Γ_s is positive (up to second order), then explicit use of

$$T\Gamma_{\rm s} = -D^{\rm s}_{ba} \nabla^b u^a_{\rm s} - u^a_{\rm s} R^{\rm s}_a , \qquad (3.110)$$

where $T = -u_s^a \mu_a^s$ is the temperature, would have to be made.

3.6 Model comparison

As an intuitive application of the formalism we have developed, it is useful to make contact with existing models for general relativistic dissipative fluids, in particular, the classic work of Landau-Lifschitz and Eckart and the second-order Müller-Israel-Stewart model. Specifically, we want to understand how standard quantities (like shear and bulk viscosity) enter the present formalism. Therefore, we need to see if the dissipative terms of the existing models can be matched with terms in the action-based description. This procedure is fairly straightforward.

The action-based model provides the total fluid stress-energy-momentum tensor, so we only have to decompose it in the usual way (cf. eq. (2.15)):

$$T^{ab} = (\bar{p} + \chi) \perp^{ab} + \varepsilon u^a u^b + 2q^{(a} u^{b)} + \chi^{ab} , \qquad (3.111)$$

In this expression, the fluxes are defined with respect to some observer with fourvelocity u^a . In order to be consistent with the perturbative expansion outlined above, we take this observer to be associated with the thermodynamical equilibrium, i.e. $u^a = u_e^a$. Finally, we have split the isotropic pressure into an equilibrium contribution (denoted as \bar{p} as discussed above) and a non-equilibrium one.

3.6.1 Equating the flux currents

Let us first consider the heat. We can read off the heat flux from the total stress-energymomentum tensor as

$$q^a = -\varepsilon u^a_{\mathbf{e}} - T^{ab} u^{\mathbf{e}}_b = -\perp^a_b T^{bc} u^{\mathbf{e}}_c .$$

$$(3.112)$$

First, let us note that there is no contribution at linear order coming from the dissipative part of the stress-energy-momentum tensor D^{ab} . In fact, making use of eqs. (3.91), (3.92) and (3.103), it is easy to show that $D^{ab}u_a^e = (\delta D^{ab})u_a^e = O(2)$. Let us therefore consider the non-dissipative part of T^{ab} . For the generalized pressure we have to first order

$$\Psi = \Lambda + \sum_{x} n_{x} \mu_{x} = -\bar{\varepsilon}_{e} + \bar{\mu}\bar{n} + \bar{T}\bar{s} + \sum_{x=n,s} \bar{n}_{x}\delta\mu_{x} = \bar{p} + \sum_{x=n,s} \bar{n}_{x}\delta\mu_{x} , \qquad (3.113)$$

where we have used the minimum energy condition (eq. (3.85)) and the equilibrium Euler relation. Using

$$\sum_{x} n_{x}^{a} \hat{\mu}_{x} = \sum_{x} n_{x}^{a} \mu_{x} + \mathcal{O}(2) \approx \sum_{x} \left[\bar{n}_{x} \bar{\mu}_{x} u_{e}^{a} + \bar{n}_{x} \delta \mu_{x} u_{e}^{a} + \bar{\mu}_{x} \left(\delta n_{x} u_{e}^{a} + \bar{n}_{x} w_{x}^{a} \right) \right], \quad (3.114)$$

we then identify the heat flux as

$$q^{a} = \sum_{x} \bar{\mu}_{x} \delta n_{x}^{a} = \bar{\mu} \bar{n} \, w_{n}^{a} + \bar{T} \bar{s} \, w_{s}^{a} \,. \tag{3.115}$$

Here, we have repeatedly used the Euler relation and the minimum energy condition eq. (3.85). We note that this quantity is consistent with the definition used in the classic models, see [21].

Let us now move on to the other fluxes and, as before, first focus on the non-dissipative contribution. It is easy to check that

$$T_{n.d.}^{ab} = \left(\bar{p} + \sum_{x} \bar{n}_{x} \delta \mu_{x}\right) g^{ab} + (\bar{p} + \bar{\epsilon}_{e}) u_{e}^{a} u_{e}^{b} + \sum_{x} \left[\bar{\mu}_{x} \bar{n}_{x} u_{e}^{b} w_{x}^{a} + \bar{n}_{x} u_{e}^{a} \left(\delta \mu_{x} u_{e}^{b} + \mathcal{B}_{x} \bar{n}_{x} w_{x}^{b} + \sum_{y \neq x} \mathcal{A}_{xy} \bar{n}_{y} w_{y}^{b}\right)\right], \qquad (3.116)$$

so that, using the standard decomposition above one arrives at

$$(\bar{p} + \chi) \perp^{ab} + \chi^{ab} = \perp^{a}_{c} \perp^{b}_{d} T^{cd} = T^{ab} + T^{ad} u^{e}_{d} u^{b}_{e} + T^{cb} u^{e}_{c} u^{a}_{e} + \varepsilon u^{a}_{e} u^{b}_{e} .$$
(3.117)

If we now use the non-dissipative contribution $T_{n.d.}^{ab}$ in this equation, we get

$$(\bar{p} + \chi) \perp^{ab} + \chi^{ab} = (\bar{p} + \sum_{x} \bar{n}_{x} \,\delta\mu_{x}) \perp^{ab} = (\bar{p} + \delta\Psi) \perp^{ab}$$
 (3.118)

That is, there may be a first-order correction in the pressure coming from $T_{n.d.}^{ab}$. Next, let us consider the contribution due to the non-dissipative part. From eq. (3.103) we see that

$$\perp^a_c \perp^b_d D^{cd} = D^{ab} = \delta D^{ab} . \tag{3.119}$$

Putting everything together, we have identified

$$\hat{\chi} = \delta \Psi + \frac{1}{3} g_{ab} \delta D^{ab} , \qquad (3.120a)$$

$$\hat{\chi}^{ab} = \delta D^{\langle ab \rangle}$$
 , (3.120b)

$$\hat{q}^a = \bar{\mu}\bar{n}w_n^a + \bar{T}\bar{s}w_s^a , \qquad (3.120c)$$

where we reintroduced the "hat" to stress that these fluxes are measured by the equilibrium observer while the angle brackets mean that we are taking the trace-free symmetric part of the tensor.

3.6.2 Example: A viscous single fluid

We now consider the specific example of a two-component, single viscous fluid. The two species are matter, with non-equilibrium flux $n^a = nu_f^a$, and entropy, with non-equilibrium flux $s^a = su_f^a$. In this simple case, we assume that the non-equilibrium

fluxes remain parallel²³, meaning $w_n^a = w_s^a = w^a$ and therefore

$$n_{\rm n}^a = n u_{\rm f}^a = n (u_{\rm e}^a + w^a)$$
, (3.121a)

$$n_{\rm s}^a = s u_{\rm f}^a = s (u_{\rm e}^a + w^a)$$
, (3.121b)

where again u_e^a is the equilibrium flow. In this case we do not have resistive terms because the two fluids are locked together from the beginning. Dissipation enters by assuming both currents depend on the (single) projected metric

$$\mathcal{N}_{n} = \mathcal{N}_{n}(X^{A}, g^{AB}) , \qquad (3.122a)$$

$$\mathcal{N}_{\rm s} = \mathcal{N}_{\rm s}(X^A, g^{AB}) \,. \tag{3.122b}$$

In practice, this means that we will have additional terms due to S_{ab}^{s} and S_{ab}^{n} in the equations of motion.

Also, the creation rate Γ_n has to vanish²⁴; this implies

$$\Gamma_{\rm n} = -\frac{1}{\mu_{\rm n}} S^{\rm n}_{ab} \nabla^a u^b_{\rm f} = 0 \Longrightarrow S^{\rm n}_{ab} = 0 , \qquad (3.123)$$

as, by construction, the viscous stress tensor S_{ab}^n must be orthogonal to u_f^a (see eq. (3.15) and [20] for further details). As a result, the final form of the non-linear equation of motion is

$$2n_{n}^{a}\nabla_{[a}\mu_{b]}^{n} + 2n_{s}^{a}\nabla_{[a}\mu_{b]}^{s} + \Gamma_{s}\mu_{b}^{s} = -\nabla^{a}S_{ab}^{s}.$$
(3.124)

Note that, when we linearize, the term involving Γ_s will not appear in the equations, because Γ_s has no linear contributions—entropy is expanded around a maximum, leaving only second-order terms.

Our next step is to use the expansion formalism developed in the previous sections to determine the explicit form of the viscous stress tensor S_{ab}^{s} . Let us start by considering the equilibrium (minimum energy) conditions. Clearly, we should have

$$S_{AB}^{s,e} = 2\mathcal{M}_s \frac{\partial \mathcal{N}_s^d}{\partial g^{AB}} = 0 \Longrightarrow \frac{\partial \mathcal{N}_s^d}{\partial g^{AB}} = 0$$
 (3.125)

²³We note that a "real" two-fluid model would involve two independent fluid degrees of freedom n_n^a , n_s^a . By forcing them to move together we are imposing quite strong constraints on the model. Basically, we are assuming that the timescale over which the entropy current relaxes to the particle flow is short enough that it may be neglected.

²⁴The matter particle flux n_n^a is conserved as it is identified with the baryon current.

It also makes sense to assume $\partial N_s^d / \partial X^A = 0$. To see why, let us forget for the moment that the two species are locked together and consider:

$$R_{A}^{s} = \mathcal{M}_{n} \frac{\partial \mathcal{N}_{n}^{d}}{\partial X_{s}^{A}} - \mathcal{M}_{s} \frac{\partial \mathcal{N}_{s}^{d}}{\partial X_{n}^{A}} = -\mathcal{M}_{s} \frac{\partial \mathcal{N}_{s}^{d}}{\partial X_{s}^{A}} - \mathcal{M}_{s} \frac{\partial \mathcal{N}_{s}^{d}}{\partial X_{n}^{A}}$$
$$= -2\mathcal{M}_{s} \frac{\partial \mathcal{N}_{s}^{d}}{\partial X_{s}^{A}} = -2\mathcal{M}_{s} \frac{\partial \mathcal{N}_{s}^{d}}{\partial X^{A}} = 0 , \qquad (3.126)$$

where we initially distinguished between the two constituents' matter-space coordinates, and used the equilibrium condition. The condition $\partial N_x^d / \partial X^A = 0$ is thus motivated by the fact that the resistive term vanishes (because the two currents are effectively locked).

As a result of these constraints we have $\delta N_x^d = O(2)$, $\delta \Psi = O(2)$ and the viscous stress tensor becomes (see eqs. (3.97) and (3.103))

$$\delta S_{ab}^{s} = \left[2\mathcal{M}_{s}\delta\left(\frac{\partial\mathcal{N}_{s}^{d}}{\partial g^{AB}}\right) \right] \Psi_{ea}^{A} \Psi_{eb}^{B}$$
$$= 2\bar{T} \left[\frac{\partial\mathcal{N}_{s}^{d}}{\partial X^{C} \partial g^{AB}} \delta X^{C} + 2 \frac{\partial\mathcal{N}_{s}^{d}}{\partial g^{DE} \partial g^{AB}} \delta g^{DE} \right] \Psi_{ea}^{A} \Psi_{eb}^{B} , \qquad (3.127)$$

Let us now rewrite the terms within the square brackets as

$$A_{CAB} = 2 \frac{\partial \mathcal{N}_{\rm s}^{\rm d}}{\partial X^C \partial g^{AB}} , \qquad (3.128a)$$

$$\Sigma_{DEAB} = 4 \frac{\partial \mathcal{N}_{s}^{d}}{\partial g^{DE} \partial g^{AB}} . \qquad (3.128b)$$

We further assume that A_{CAB} is zero as this involves degrees of freedom we do not need to recover Navier-Stokes equations.

Before moving on with the model specification, let us introduce a slight generalization to the original model from [20]. The idea is to take the normalizations \mathcal{N}_x^d as functionals—instead of functions—of the additional variables. This does not constitute a major difference as the equations of motion, and particle production rate formulae, remain unchanged. Still, the step can be taken subject to the following caveat: The (functional) integration should extend (at most) to the spacetime region that is causally connected with each point. Here, we will assume that the analysis is done locally in space but not necessarily in time, i.e. on the world-tube formed by the spatial part of the region $\delta \mathcal{M}$ in fig. 3.1. In the present example this would mean

$$\mathcal{N}_{x}^{d}[g^{AB}] = \mathcal{N}_{x}^{d}[g_{e}^{AB}] + \int \frac{\delta \mathcal{N}_{x}^{d}}{\delta g^{AB}(x)} \delta g^{AB}(x) d^{4}x + \frac{1}{2} \int \frac{\delta^{2} \mathcal{N}_{x}^{d}}{\delta g^{AB}(x) \delta g^{CD}(y)} \delta g^{AB}(x) \delta g^{CD}(y) d^{4}x d^{4}y$$
(3.129)

where the first two terms vanish because (i) \mathcal{N}_x^d vanishes at equilibrium, and (ii) the minimum energy condition. The key step then is to replace the ordinary partial derivatives with functional derivatives in the various expressions we have discussed, so that the viscous stress tensor will be

$$S_{AB}^{s}(x) = 2\bar{T}\delta\left(\frac{\delta\mathcal{N}_{s}^{d}}{\delta g^{AB}}\right)(x) = 2\bar{T}\int\frac{\delta^{2}\mathcal{N}_{s}^{d}}{\delta g^{AB}(x)\delta g^{CD}(y)}\delta g^{CD}(y)d^{4}y.$$
(3.130)

We can now formally introduce a set of spatial coordinates \bar{x} comoving with the equilibrium observer and attached to the world-tube, and take the time coordinate to be the equilibrium worldline's proper time τ . Also, to enforce locality in space we let

$$\frac{\delta^2 \mathcal{N}_{\rm s}^{\rm d}}{\delta g^{AB}(x) \delta g^{CD}(y)} = \frac{1}{4} \Sigma_{ABCD}(\bar{x}, \tau_x - \tau_y) \,\delta^3(\bar{x} - \bar{y}) \,. \tag{3.131}$$

where the causality condition $\tau_x - \tau_y \ge 0$ is assumed to be encoded within Σ_{ABCD} .

Let us first of all, as a consistency check, show that the formula for the particle production rate remains unaltered by these modifications. We have (see eq. (3.12))

$$\mu_{\rm x}\Gamma_{\rm x} = \frac{1}{3!}\mu_{\rm x}^{ABC}\frac{dn_{ABC}^{\rm x}}{d\tau_{\rm x}} = \bar{\mathcal{M}}_{\rm x}\frac{d}{d\tau_{\rm x}}\left(\bar{\mathcal{N}}_{\rm x}^{\rm e} + \bar{\mathcal{N}}_{\rm x}^{\rm d}\right)$$
$$= \mathcal{M}_{\rm x}\left(\frac{d\mathcal{N}_{\rm x}^{\rm d}}{d\tau_{\rm x}} + \frac{1}{2}\mathcal{N}_{\rm x}^{\rm d}g_{AB}^{\rm x}\frac{dg_{\rm x}^{AB}}{d\tau_{\rm x}}\right).$$
(3.132)

where, for a single viscous fluid the result simplifies to

$$\Gamma_{\rm x} = \frac{d\mathcal{N}_{\rm x}^{\rm d}}{d\tau} + \mathcal{O}(3) . \tag{3.133}$$

If in particular we consider the entropy production rate Γ_s we have

$$\mathcal{N}_{\rm s}^{\rm d} = \frac{1}{8} \int \Sigma_{ABCD}(\bar{x}, \tau - \tau') \delta g^{AB}(\bar{x}, \tau) \delta g^{CD}(\bar{x}, \tau') \mathrm{d}^3 \bar{x} \, \mathrm{d}\tau \, \mathrm{d}\tau' \,, \tag{3.134}$$

To compute the entropy creation rate we have to use the chain rule (generalized to functionals) on $\mathcal{N}_{s}^{d}[g^{AB}(x)]$

$$\frac{d\mathcal{N}_{\rm s}^{\rm d}}{d\tau} = \int \frac{\delta N_{\rm s}^{\rm d}}{\delta g^{AB}(y)} \frac{\delta g^{AB}(y)}{\delta \tau(x)} {\rm d}^4 y \,. \tag{3.135}$$

But, because g^{AB} is a "normal" function of the spacetime coordinates

$$\frac{\delta g^{AB}(y)}{\delta \tau(x)} = -2\delta^4(x-y)D^{(A}w^{B)}(x) , \qquad (3.136)$$

so that we are left with

$$\Gamma_{\rm s} = -\frac{1}{\bar{T}} S_{AB}(\bar{x},\tau) D^{(A} w^{B)}(\bar{x},\tau) . \qquad (3.137)$$

We can now make use of this "functional generalization" to recover the Navier-Stokes model for a bulk- and shear viscous fluid. To focus on the key point, let us consider first the purely bulk-viscous case

$$S(\tau) = \int K(\tau - \tau')A(\tau')d\tau', \qquad (3.138)$$

where *S* represents the trace of the viscous-stress tensor while A stands for the trace tr δg^{AB} . Because of the difference between δg^{AB} and \dot{g}^{AB} —the former involves gradients in the displacement while the latter depends on the velocity—in order to recover a Navier-Stokes model we need to take

$$K(\tau - \tau') = -T\zeta \partial_{\tau'} \delta(\tau - \tau') , \qquad (3.139)$$

since this would give

$$S(\tau) = T\zeta \int [-\partial_{\tau'}\delta(\tau - \tau')]A(\tau')d\tau' = T\zeta \int \delta(\tau - \tau')\partial_{\tau'}A(\tau')d\tau' = T\zeta \frac{dA}{d\tau}, \quad (3.140)$$

as desired. We now implement this for the bulk- and shear model. We can do this using the standard decomposition of the bulk and shear response as

$$\Sigma_{ABCD} = \Sigma^{\rm b}_{ABCD} + \Sigma^{\rm s}_{ABCD} , \qquad (3.141)$$

with

$$\Sigma^{b}_{ABCD} = \frac{\zeta(x)}{\bar{T}} g^{e}_{AB} g^{e}_{CD} \delta^{3}(\bar{x} - \bar{y}) q_{b}(\tau_{x} - \tau_{y}) , \qquad (3.142a)$$

$$\Sigma^{\rm s}_{ABCD} = 2\frac{\eta(x)}{\bar{T}} \left(g^{\rm e}_{A(C} g^{\rm e}_{B)D} - \frac{2}{3} g^{\rm e}_{AB} g^{\rm e}_{CD} \right) \delta^3(\bar{x} - \bar{y}) \, q_{\rm s}(\tau_x - \tau_y) \,, \tag{3.142b}$$

where the two kernels would be²⁵ the same $q_b = q_s = -\partial_{\tau_y}\delta(\tau_x - \tau_y)$. It then follows that the only viscosity tensor of the model is

$$S_{ab}^{s} = \chi_{ab} + \chi \perp_{ab} = \frac{1}{3} \zeta \theta \perp_{ab} + \eta \sigma_{ab} . \qquad (3.143)$$

²⁵Note that we have chosen to separate the bulk- and shear channels as usual, even though the present construction allows for anisotropic response in the velocity gradients to viscosity relation. We have also introduced two independent kernels to allow for different response to bulk and shear strain rates.

It is also easy to check that we can enforce compatibility with the second law by fixing the sign of the bulk-and shear viscosity coefficients. In fact,

$$\Gamma_s = \frac{\zeta}{\bar{T}} \,\theta^2 + \frac{\eta}{\bar{T}} \,\sigma^{ab} \sigma_{ab} \ge 0 \,. \tag{3.144}$$

where $\zeta, \eta \ge 0$. With these relations we have recovered the usual relativistic Navier-Stokes equations (the Landau-Lifschitz-Eckart model for a viscous fluid).

Let us conclude by pointing out that, to write down the full set of equations one should also expand the "Euler part" of the equation of motion, i.e. the left-hand-side of eq. (3.124). We have provided all the ingredients necessary for the explicit calculation, but leave it out here—and point to [21] for further details—as the result is not new and not key to the present discussion.

3.7 Cattaneo-type equations

As a practical example of the first-order expansion we outlined the model for a single (bulk and shear) viscous fluid, and showed how this leads to the expected form of the relativistic Navier-Stokes equations. The derivation shows that the action-based formalism encodes the previous models. It is also clear that the formalism allows us to consider much more complicated settings, should we need to do so. However, the discussion of the first-order results is clearly not complete, because the final set of equations is widely known to suffer from causality/stability issues (see discussion in section 2.2) and it is then natural to wonder how we may fix this. When it comes to heat-conducting systems, the way forward has been discussed in [142, 22], where it is demonstrated that one can resolve the stability/causality issues at first order by properly accounting for the entrainment between matter and entropy currents—retaining the compatibility with the second law. However, for the single viscous fluid under consideration, the problem must be addressed in a different way, as the key ingredient used to solve the heat-flux case accounting for the entropy inertia, will not work for the present case as our model setting does not involve relative flows. We now show how we can make progress in the single viscous fluid case by using a functional form that is different from eq. (3.142). Notably, the argument stresses the importance of the "principle of memory or heredity" (see [121]).

Let us first focus on the bulk viscosity case, and then extend the results to the bulk- and shear viscous model. Recalling eq. (3.138), the first step would again be to assume

$$K(\tau - \tau') = -\partial_{\tau'}g(\tau - \tau') , \qquad (3.145)$$

so that

$$S(\tau) = \int K(\tau - \tau')A(\tau')d\tau' = \int g(\tau - \tau')\frac{dA(\tau')}{d\tau'}d\tau' .$$
(3.146)

We can then look for the convolution kernel *g* such that the bulk-viscous scalar *S* satisfies an equation of the Cattaneo type

$$t_{\rm b}\dot{S} = -S - \zeta \frac{dA}{d\tau} , \qquad (3.147)$$

where t_b is the relaxation time-scale of the bulk-viscosity response. In terms of the convolution kernel *g* this would mean

$$g(\tau - \tau') = -\frac{\zeta}{t_{\rm b}} e^{-(\tau - \tau')/t_{\rm b}} \theta(\tau - \tau') , \qquad (3.148)$$

and one can check by direct computation this leads to eq. (3.147) as

$$t_{\rm b}\partial_{\tau}g(\tau-\tau')+g(\tau-\tau')=-\zeta\delta(\tau-\tau'). \qquad (3.149)$$

By inspecting the last expression, we also see that in the "fast relaxation limit" $t_b \rightarrow 0$ we recover a Navier-Stokes-type response as we would intuitively expect. We also point to chapter 6 where this fast relaxation limit is discussed in light of the (perhaps inevitable) resolution limitations faced in numerical implementations.

We are now ready to go back to the full bulk- and shear viscous model. We will retain the structure and symmetries from before (see eqs. (3.141) and (3.142)), but introduce two different convolutions q_b and q_s to account for retarded response to bulk and shear strain rates. In essence, we have shown how we can implement a retarded response of the Cattaneo type in the action-based model by assuming that S_{AB} (and therefore D_{ab} as well) is an integral function of g^{AB} . The question then is, does this mean that the final fluid equations are integro-differential equations? Fortunately the answer is no. In fact, we have shown that, by a suitable choice of the response function $q(\tau - \tau')$, the fluxes satisfy an equation of the Cattaneo-type. Therefore, instead of solving an integro-differential equation, one should treat $S_{ab}^s = \chi \perp_{ab} + \chi_{ab}$ as an unknown in eq. (3.124), and add two equations to the system

$$\chi + t_{\rm b}\,\dot{\chi} = -\zeta\,\theta\,\,,\tag{3.150a}$$

$$\chi_{ab} + t_{\rm s} \, \dot{\chi}_{ab} = -\eta \, \sigma_{ab} \, . \tag{3.150b}$$

This means that, at the end of the day, to actually solve a set of differential dissipative equations at first order, we have to treat the fluxes as additional unknowns, for which one has to provide equations that are not given by the stress-energy-momentum conservation law $\nabla_a T^{ab} = 0$. This is reminiscent of the EIT paradigm, where one postulates from the beginning an entropy function that depends on an additional set of quantities—the thermodynamical fluxes. The difference is that the microphysical origin of the equation for the fluxes is now more clear. It is also worth noting that equations of the Cattaneo-type for the fluxes cannot be obtained in the field-theory-based models, as the constitutive equations are given in terms of the usual equilibrium variables (like μ , *T*) and their derivatives—so that terms with derivatives of the fluxes (like $\dot{\chi}$) do not appear.

Equations 3.150 are (formally) the same as in the linearized version of Müeller-Israel-Stewart model, which has been shown to be stable and causal. In theory, nothing prevents us from choosing a different form for the retarded response q which could lead to acausal/unstable behaviour. However, the form of q suggested above has a clear physical interpretation and is microphysically motivated. If one wants to come up with an alternative, this would need to be motivated by microphysical arguments, as well.

Let us now consider the implications of the Cattaneo laws for the entropy production rate. As in the Navier-Stokes model sketched above we have

$$\Gamma_{\rm s} = \int \Sigma_{ABCD}(x,\tau-\tau') D^{(C} w^{D)}(\bar{x},\tau') D^{(A} w^{B)}(\bar{x},\tau) d\tau' = = \frac{1}{\bar{T}} S_{AB}(\bar{x},\tau) D^{(A} w^{B)}(\bar{x},\tau) .$$
(3.151)

Again, let us use the bulk-viscous case to highlight the relevant features. We then have

$$\Gamma_{\rm s} = \frac{\zeta}{t_{\rm b}} \int_{-\infty}^{\tau} e^{-(\tau - \tau')/t_{\rm b}} \theta(\tau) \theta(\tau') \, \mathrm{d}\tau' \,, \qquad (3.152)$$

where θ is the expansion rate. It is clear that, because the expansion rate is evaluated at different times, we cannot guarantee the positivity of the entropy production rate just by fixing the sign of the bulk-viscosity coefficient as in the previous Navier-Stokes case. However, we will argue this result is not as dramatic as it may appear at first sight—at least not in the regimes relevant for physical predictions. Because of the exponential in the integral, we can assume that the values for the expansion at times τ' can be neglected for τ' that is a few t_b away from τ . As a result we can expand $\theta(\tau')$ as

$$\theta(\tau') = \theta(\tau) + \frac{d\theta(\tau)}{d\tau}(\tau' - \tau) = \theta(\tau) + \dot{\theta}(\tau)(\tau' - \tau) , \qquad (3.153)$$

so that we obtain

$$\Gamma_{\rm s} = \zeta \left[\theta^2(\tau) - t_{\rm b} \theta(\tau) \dot{\theta}(\tau) \right] \,. \tag{3.154}$$

Again, because of the product of the expansion rate with its time derivative, the entropy production rate cannot be made generally positive simply by fixing the sign of the bulk-viscosity coefficient ζ . However, as discussed in [90, 140], physical fluid states relax—on a timescale characteristic of the microscopic particle interactions—to ones that are essentially indistinguishable from the simple relativistic Navier-Stokes description. Translated to the present context this would mean

$$t_{\rm b}\dot{\theta}(\tau) \approx t_{\rm b} \frac{\Delta\theta}{\tau_{\rm hydro}} \to 0$$
. (3.155)

because the ratio between the relaxation timescale t_b and the hydrodynamical timescale τ_{hydro} is effectively negligible. As a result, for actual physical applications we can neglect the second term in the entropy production rate and compatibility with thermodynamic second law is restored. In a way, this would mean that for bulk- and shear viscous fluids, we can introduce Cattaneo laws for the fluxes—to fix the causality and stability issues of the Navier-Stokes model—while the physical content will be precisely that of Navier-Stokes.

3.8 Summary and outlook

We have considered the close-to-equilibrium regime of the action-based model of Andersson and Comer [20] for dissipative multi-fluid systems. In particular, we have shown that, starting from a set of fully non-linear dynamical equations with only the fluxes as the degrees of freedom, an expansion with respect to (a self-consistently defined) equilibrium can be introduced in a clear fashion, with the line of reasoning being similar to that of usual hydrodynamical perturbation theory.

After discussing the aspects of equilibrium which can be inferred from the action-based model itself, we established how to construct an expansion in deviations away from equilibrium in a general setting, so that the framework is of wider relevance. In the process we demonstrated the importance of the frame-of-reference of the equilibrium observer. We also noted that the construction promotes the role of the matter space: Instead of it being a mathematical "trick" to facilitate a constrained variation, it might well be the arena where the microphysical details are encoded. This is a novel perspective that needs further discussion and consideration.

We then focused on a particular first-order viscous fluid model, with shear- and bulkviscosity, paying particular attention to the key causality issues. We showed that causal behaviour can be linked to a retarded response function that keeps track of a system's history. The specific form of the response function can be modelled in a phenomenological way—as we did—but should ideally be provided by specific microphysical calculations, for instance by means of the fluctuation-dissipation theorem (see [181] for a general discussion and [121] for comments on its role from the EIT perspective).²⁶ In a sense, the action-based model provides the "context", determining the geometric structure and form of the equations of motion, while the detailed microphysics is encoded in the specific response function.

Building the first-order expansion we made this connection clear, and showed how and where the microphysics enters the discussion. An interesting outcome of this analysis

²⁶We also note that there have been recent efforts to make explicit use of the fluctuation-dissipation theorem to compute response coefficients through Green-Kubo-like formulae, see [156, 157]—although in a special relativistic setting. Nevertheless, there are similarities with the way we deal with causality issues here.

is that we showed that there is no need to go to second order in deviations from equilibrium to implement a causal response in the model. This has already been demonstrated for the heat-flux problem (see [22, 142]), where the Cattaneo-type equation for the heat flux is ultimately related to the multi-fluid nature of the problem. The entrainment effect (through which the entropy current gains an effective mass [19]) results in an inertial heat response. The case of a single viscous fluid is different since its retarded response cannot be associated with the multifluid nature of the problem.

As the variational model is designed for dealing with multi-fluids, the route to further extensions is—at least at the formal level—quite clear. A natural next step would be the modelling of a viscous fluid allowing for the heat to flow differently from the matter. This application should be fairly straightforward since the two main issues of the problem have now been studied separately. A more challenging step will be the inclusion of superfluidity. The presence of currents that persist for very long times changes drastically the non-dissipative limit. The model would require the use of more than one equilibrium worldline congruence [20, 87], one for each "superfluid condensate" and one for all the remaining constituents.

Part II

A covariant approach to large-eddy filtering

Chapter 4

Filtering relativistic hydrodynamics

In the first part of this thesis we focused on modelling dissipative fluids in relativity, presenting the different ideas/schemes currently on the market, and then focusing on the only scheme that naturally lends itself to multi-fluid extensions. However, any discussion of dissipative relativistic fluids, would not be complete unless it touches upon (at the very least) the issues that arise when modelling turbulent flows. Turbulent flows are not only ubiquitous in the real-world—as turbulence is a manifestation of highly non-linear behaviour intrinsic to hydrodynamic equations—but also known to transport quantities like energy and momentum at a much faster rate than we would expect from microphysical transport mechanisms.

We start with a brief introduction to hydrodynamic turbulence in section 4.1, whilst the rest of this chapter is devoted to address some of the issues that arise when modelling turbulent flows in relativity. We focus on the formal aspects associated with averaging/filtering the fluid dynamics which enter most of the recent developments/discussions. To make progress, we develop a new covariant framework for filtering/averaging based on the fibration of spacetime associated with fluid elements and the use of Fermi coordinates to facilitate a meaningful local analysis. We demonstrate how "effective" dissipative terms arise because of the coarse-graining, paying particular attention to the thermodynamical interpretation of the resolved quantities. In particular, as the smoothing of the fluid dynamics inevitably leads to a closure problem, we discuss a new closure scheme inspired by the recent progress in modelling dissipative relativistic fluids that we briefly touched upon in section 2.4. The results presented in this chapter have been published in Celora et al. [58].

We continue in chapter 5 by discussing the first steps towards extending the framework to charged multi-fluids mixtures. In particular, we argue it is somewhat natural to begin with a discussion of magneto-hydrodynamics (MHD). We will do so after having introduced and discussed the main differences between hydrodynamics and MHD turbulence. There we will also derive the relativistic MHD equations, given that electromagnetic aspects have not been discussed in the earlier parts of this thesis.

4.1 A brief introduction to hydrodynamic turbulence

As a warm up, we begin with a very brief introduction to hydrodynamic turbulence, while referring to monographs such as [138] for an exhaustive discussion. Everyday life gives an intuitive understanding of hydrodynamic turbulence: the flow of a river downstream of an obstacle or atmospheric/oceanic currents are just two examples. Turbulence typically involves an abrupt change in, say, the velocity field (or better, changes over very small scales) and is often driven by the development of fluid instabilities. A turbulent flow is also often associated with chaotic behaviour and a lack of predictability. While "chaos" can be precisely defined for simple mechanical systems—and turbulent flows are not necessarily chaotic in this sense—it is true nonetheless that uncertainties in the observation of the initial state forbid a precise prediction at later times. This is, in fact the reason why turbulent flows appear chaotic. On the other hand, there is ample evidence that Navier-Stokes equations describe well turbulent flows. An important quantity in the characterization of turbulence is the Reynolds number

$$\operatorname{Re} = \frac{\rho L^2 / \eta}{L/V} = \frac{L V \rho}{\eta} , \qquad (4.1)$$

where *L* and *V* are the characteristic length and velocity of the flow, while ρ , η are the mass density and (dynamic) shear-viscosity coefficient—which has units (in cgs) of g cm⁻¹s⁻¹. The Reynolds number is formed as the ratio of the timescales over which flow properties are transferred by molecular diffusion as compared to macroscopic convection. It then quantifies the importance of inertial over viscous effects¹. As viscosity tends to make neighbouring fluid elements move together, it is intuitively clear that for high Reynolds numbers we expect the fluid flow to change over very small scales.

From these considerations, we can conclude that fluid turbulence is (or can be described as) a deterministic phenomenon, although the evolution in time is very much complicated by the non-linearities in the fluid equations. Having said that, while a precise definition of turbulence does not exist, we will try to define it anyway pointing out some of the common properties of turbulent flows. First, a turbulent flow is disordered in (space and time) and often presents well-organized structures such as vortices. Second, turbulent flows are able to mix transported quantities—like energy and momentum—much faster than if only molecular diffusion processes were involved. Thirdly, it involves a wide range of spatial wavelengths. The latter are typically divided into three broad ranges (see fig. 4.1) :

¹Similarly one can define the Peclet number $Pe = LV/\kappa$ for turbulent heat diffusion, where κ is the heat conductivity.



FIGURE 4.1: Cartoon of the scales of the turbulent energy spectrum. Figure adapted from McDonough [151].

- the *large scale* is defined by the problem domain geometry.
- the *integral scale* is a fraction of the large scale, often associated with a single wavelength *k*_{*I*}. It is defined as the maximum of the energy spectrum, and often associated with the scale at which energy is put into the system to sustain turbulence.
- the *inertial range*, which covers a wider range of length scales, is characterized by the fact that viscous effects are negligible.
- the *dissipation range*, typically associated with the Kolmogorov scale *k*_K, where viscosity effects are dominant over inertial ones.

Because of the complexity of turbulent flows, a deterministic analytical description is practically excluded. One option, not further developed in the present work, is to resort to statistical analysis. A very important result in this direction was obtained by Kolmogorov in 1941 (see [128, 129, 130] and the more recent discussion by Frisch [84]). Assuming statistical isotropy and homogeneity, Kolmogorov derived a simple set of scaling laws that are in very good agreement with observations—the most famous being the "5/3 law". Using simple dimensional arguments he showed that

$$E(k) = C_K \varepsilon^{2/3} k^{-5/3} , (4.2)$$

where E(k) is the energy spectrum, ε is the rate with which energy is pumped into the system at large scales, and C_K is a constant—the Kolmogorov constant. For stationary,

homogeneous and isotropic turbulence ε is also the constant energy flux from scale to scale, and hence the rate at which energy is dissipated once we hit the Kolmogorov scale.

An alternative option, which is often preferred, especially with the increase of computing power, is to rely on numerical simulations. For relatively simple flows one can try to evolve directly the Navier-Stokes equations. However, it has been shown that the number of grid points needed to fully resolve the flow scale as² $\text{Re}^{9/4}$. This means that most of the interesting turbulent flows, characterized by very high Reynolds numbers of the order of 10^4 or bigger, are in fact out of reach for direct numerical simulations now and in the not so near future. Then, what do we do? The strategy, which motivates the analysis in the rest of this chapter, is to use mean flow models or large-eddysimulations. This boils down to averaging/smoothing the fluid dynamics, and resolving scales only up to some value in the inertial sub-range, while taking into account the smaller scales through sub-grid models.

4.2 Averaging turbulent flows

Fluid models inevitably involve aspects of averaging—we have to average over a large number of particles in order to describe a fluid system in terms of a small number of macroscopic quantities (in the thermodynamic sense). However, in the simplest settings we do not have to worry (too much) about the actual process of averaging. For example, the transition from particle kinetic theory to a fluid model follows intuitively when the momentum distribution develops a well-defined peak. Similarly, the notion of a fluid element enters naturally on scales much larger than the individual particle mean-free paths (cf. discussion in section 3.1.1). However, the story changes when we turn to dynamical simulations and problems involving, for example, turbulence. When we consider the problem from a simulation point of view, we have to consider the scale associated with the numerical resolution. This numerical scale tends to be vast compared to (say) the size of a fluid element. For example, in the high-density core of a neutron star we would typically deal with mean-free paths of a fraction of a millimeter, while the best current large-scale numerical simulations of neutron-star mergers involve a resolution of order 10 meters (see [126]). This scale discrepancy has "uncomfortable" implications. In a highly dynamical situation we may not be able to resolve the full range of scales involved. Quite a lot of action can be hidden inside each computational cell. As we discussed in section 4.1, this is a well-known fact that motivates the (considerable) effort going into developing "large-eddy" simulations schemes in computational fluid dynamics (the subject of numerous textbooks, see for example [136, 151, 138]).

²This is for the three dimensional case, the two dimensional scaling is Re² [136].

The astrophysical significance of the problem is obvious given that many relevant situations involve/require the modelling of hydro(-magnetic) turbulence. Again, the dynamics of neutron stars comes to mind. A topical example is the turbulent flow caused by the Kelvin-Helmholtz instability, which in turn drives the amplification of the magnetic field (through the dynamo effect) in binary neutron star mergers ([173, 125, 15, 229]). Traditionally—in the context of Newtonian theory—turbulent flows have been studied in terms of Reynolds averaged equations, which, roughly speaking, are obtained via time-averaging the fluid dynamics. This smoothing requires the modelling of features that are not captured by the resolved flow, bringing in the need to introduce suitable "closure conditions" (necessary as the averaged scheme introduces more degrees of freedom than there are equations of motion). Most recent work replaces the averaging with spatial filtering, leading to what is known as Large Eddy Simulations (LES). This strategy is often preferred because it involves less modelling [151], although Reynolds-type averaging is still widely used in the context of magnetic dynamos, see the reviews by Brandenburg and Subramanian [47] and Rincon [187].

Because of the relevance for, in particular, binary neutron-star merger simulations, there have been several recent efforts to extend the "familiar logic" from the Newtonian setting to relativity. These range from the more formal discussion in [77] to the actual simulations in [176, 177]. Also worth noting is the recent work in [72] in which the results from [176] are contrasted with those obtained modelling the turbulent flow as effectively viscous on a larger scale (see e.g. [202]). The general relativistic magnetohydrodynamics (MHD) merger simulations of [94] are another relevant example. Most of these results are based on spatial averaging, with subgrid models tailored to account for small-scale dynamo action. Recently, a more refined gradient subgrid-scale model for general relativistic simulations was developed in [219, 49, 220] and applied to binary neutron-star mergers with impressive results [4, 169, 5].

In short, while there has been notable effort to carry out large-eddy simulations in relativity, the formal underpinnings for this effort are not as firmly established as one might like—the exception being the discussion in [77]. This is the gap we are trying to bridge here. Starting from the beginning, we bring to the fore the fundamental issues associated with any effort to "average" or "filter" in a curved spacetime.

We want to consider the problem from a covariant spacetime point of view, a key point being that the underlying principles—for both time-averaging and space filtering should be the same (or at least "similar"). Both strategies combine "smoothing" with suitable closure relations to determine contributions that may not be "directly" calculable (i.e. represented on the resolved scale). The issues we are interested in can be approached at the level of an "effective theory" based on fairly simple rules, avoiding a detailed discussion of the underlying averaging/filtering process. This is a useful strategy as it leads to a relatively straightforward derivation of the dynamical equations. At the same time, one has to pay attention to the details as a number of issues come into play when we consider the problem from the covariant perspective of General Relativity. In essence, we want to establish what a consistent spacetime scheme for averaging or filtering should look like, and highlight issues relating to the formulation of such a scheme. As the final aim is to develop a consistent set-up for simulations, there are important numerical issues to be discussed; e.g. the implicit filtering associated with numerical discretisation. In order not to confuse these with the foundational issues, we leave them out of the present discussion.

As we need to keep track of the relevant scales and quantities associated with different "observers", the notation easily gets somewhat messy. This may be inevitable, but let us try to ease the pain by explaining the notation used in this chapter from the outset. First of all, we need to distinguish between fine-scale and coarse-scale quantities. To do so, we use bars and angle brackets—i.e. \overline{A} and $\langle A \rangle$ —for averaged and filtered quantities (respectively), obtained from the fine-scale one, A. However, as we will see, these quantities are not necessarily the most natural to evolve. Thus, we use tildes, e.g. \tilde{A} , to identify the evolved/resolved quantities. Finally, while discussing the linear stability of the proposed closure scheme (section 4.8.1), we drop the tilde notation—as all the quantities are then assumed to be evolved and there is no need to make the distinction—and use instead sub/superscripts to represent quantities evaluated on the background, like A_0 . This subscript should not be confused with the spacetime indices, which are represented by latin letters $a, b, c \ldots = 0, 1, 2, 3$ throughout.

4.3 Averaging vs filtering

In order to provide the appropriate context and establish the general strategy, it is useful to briefly summarise the standard approach for (typically incompressible) fluid dynamics in Newtonian gravity. Traditionally, small scale fluctuations are considered in terms of averaging, following the pioneering work of Reynolds and others (see [136]). In effect, this means that we have $A = \overline{A} + \delta A$, with the fluctuations represented by δA at each spacetime point. Introducing this formal split has the advantage of providing a straightforward derivation of the dynamical equations and a relatively clear interpretation of the involved quantities. One may also resort to an expansion for "small" δA (see [72] for a relevant example of this). Typically, progress is made by assuming that the average of the linear fluctuations vanishes, which may not be a faithful representation of the physics the model aims to describe (see [151] for a more extensive discussion). Noting this, the typical strategy for spatial filtering-forming the basis for modern large-eddy simulations—is different. In particular, the filtered fluctuations are not taken to vanish, nor does the argument involve expanding in the fluctuations. Instead, one typically proceeds by introducing a new set of variables to represent the filtered dynamics. From the conceptual point of view, each of the two strategies has

attractive features and—as we are interested in the formal aspects of the problem—we will consider both of them in the following.

Let us first consider the standard averaging problem. The standard strategy is to derive the averaged equations by applying a simple set of rules. In essence, one would *assume* that (using a bar over quantities to represent averaging)

$$\overline{c} = c$$
,for constants ,(4.3a) $\overline{A + B} = \overline{A} + \overline{B}$,linearity of the procedure ,(4.3b) $\overline{\partial_a A} = \partial_a \overline{A}$,averaging commutes with derivatives .(4.3c)

It is immediately clear that, while the last of these relations is intuitive for a time average in Newtonian physics, we need to tread carefully when we turn to the relativistic setting. First of all, we need to face the fact that we do not have an observer-independent space-time split. Secondly, the derivatives we need to consider will be covariant, and hence we must consider the spacetime curvature. Simply noting these reservations for the moment (they will be discussed in section 4.4), the stated rules imply that

$$\overline{cA} = c \,\overline{A} \,. \tag{4.4}$$

Moreover—and this is where the main distinction from large-eddy models comes in—it is common to further assume that the average of the fluctuations vanishes so we have

$$\overline{\delta A} = 0. \tag{4.5}$$

It then follows that

$$\overline{A} = \overline{A}$$
, (4.6)

which means that the field \overline{A} remains unchanged after the averaging. In effect, this additional rule leads to

$$\overline{A}B = \overline{A}\,\overline{B} \,. \tag{4.7}$$

This simplifies the discussion considerably as we can ignore all linear fluctuation terms in the averaged equations.

Time averaging is the (conceptually) simplest approach to the problem, but (strictly speaking) it removes dynamical features associated with the fluctuations³, which is unlikely to be realistic. A faithful representation of the physics may require a different prescription. One option would be to not introduce the assumption from (4.5). The typical description then involves (spatial) filtering, using some specified kernel to define the separation of scales (see [136]). This (effectively) leads to the same kind of rules as before—with the exception of eqs. (4.5) and (4.6)—although we now have (indicating

³The averaged quantity is strictly speaking not time dependent, even though the time derivative term is usually retained in the evolution of Reynolds-Averaged Navier-Stokes equations (see [151]).
filtering by angle brackets)

$$\langle \langle A \rangle B \rangle \neq \langle A \rangle \langle B \rangle , \tag{4.8}$$

which means that filtered fluctuations do not have to vanish. That is, in general we have $\langle \delta A \rangle \neq 0$.

4.4 The spacetime view: Fermi Coordinates

As a first—and essential—step towards a relativistic model for averaging/filtering, we have to consider the spacetime aspects of the problem. In particular, we need to introduce an unambiguous space-time decomposition—otherwise we cannot meaningfully consider "time" averages or "space" filtering. This is more than semantics [77]. An interesting discussion of the problem (mainly from the cosmology perspective) has been provided by Ellis [75], and it is evident that the issue is conceptually problematic since the notions of time and space are observer dependent. The problem is particularly vexing for a foliation based approach to spacetime (as assumed in numerical relativity [72], where the spacetime foliation is manifestly gauge dependent). However, for a fluid there does exist a natural fibration of spacetime [21]. If we take the associated fluid frame as our starting point, we can introduce a meaningful "local analysis" which allows us to make progress. Moreover, it is natural to use the fluid frame to make the all-important connection with the microphysics and the equation of state [21]. The strategy also allows us to consider thermodynamical aspects of the averaging/filtering scheme.

Let us explore the steps involved in an averaging/filtering procedure based on a spacetime fibration. In particular, we want to establish under which conditions we may assume that the covariant derivative commutes with the averaging/filtering procedure. Intuitively, we need to assume from the outset that there is a separation of scales between the metric fluctuations and the fluid fluctuations. The natural approach to the problem then involves Fermi-type coordinates (cf. appendix A). In order to establish the logic, consider the following situation: The fluid four-velocity (and other physical properties) varies over a resolved spacetime region (this can be thought of as a numerical cell, even though such numerical cells would typically be defined in terms of a foliation). However, we assume that it is still possible to identify a family of observers associated with a four-velocity vector field U^a which can be taken to be constant over the resolved region and which is "close enough" to the actual fluid four-velocity. (The latter assumption is not strictly required for the definition of an averaging procedure, nor for the spacetime decomposition, but it helps develop the logic). Then we can use the worldlines with tangent U^a to construct Fermi-type coordinates and explore the details of a given averaging/filtering procedure.

Fermi coordinates were first introduced by Fermi in 1922 [78, 79] and then developed by, in particular, Manasse and Misner [148] (see also, for example, [208, 179]). We will not dwell on the construction itself here as this is not a new result (we point to appendix A for more details). Instead, we focus on the properties and region of validity of the associated coordinate system. The set of coordinates is essentially built from a spacetime tetrad transported along a central worldline (naturally taken to be timelike in our case). This is convenient because the metric and the Christoffel symbols take a very simple form along the central curve.

Let us introduce Fermi coordinates $x^{\hat{a}} = \{x^{\hat{0}}, x^{\hat{1}}, x^{\hat{2}}, x^{\hat{3}}\}$ (distinguished by hats on the indices) such that, on the central worldline *G*, the metric reduces to the Minkowski form $g_{\hat{a}\hat{b}} = \eta_{\hat{a}\hat{b}}$ while its first derivatives can be obtained from the Christoffel symbols (see [155])

$$g_{\hat{a}\hat{b},\hat{0}} = 0$$
 , (4.9a)

$$g_{\hat{0}\hat{0},\hat{j}} = -2a_{\hat{j}},$$
 (4.9b)

$$g_{\hat{0}\hat{l}\hat{k}} = 0$$
, (4.9c)

$$g_{j\hat{k},\hat{m}} = 0 , \qquad (4.9d)$$

where the commas represent partial derivatives. We have introduced the non-vanishing piece of the four acceleration, a_{j} , of the worldline⁴ and chosen to construct the tetrad in such a way that the associated observer is non-rotating (which seems natural). With this construction we can formulate an expansion of the metric in the neighbourhood of the worldline. This leads to

$$g_{\hat{0}\hat{0}} = g_{\hat{0}\hat{0}}|_{G} + g_{\hat{0}\hat{0},\hat{a}}x^{\hat{a}} = -(1 + 2a_{\hat{j}}x^{\hat{j}}) + \mathcal{O}(x^{\hat{j}})^{2} , \qquad (4.10a)$$

$$g_{\hat{0}\hat{j}} = g_{\hat{0}\hat{j}}|_{G} + g_{\hat{0}\hat{j},\hat{a}}x^{\hat{a}} = \mathcal{O}(x^{\hat{j}})^{2} , \qquad (4.10b)$$

$$g_{\hat{i}\hat{j}} = g_{\hat{i}\hat{j}}\big|_{G} + g_{\hat{i}\hat{j},\hat{a}}x^{\hat{a}} = \eta_{\hat{i}\hat{j}} + \mathcal{O}(x^{\hat{j}})^{2} , \qquad (4.10c)$$

where $|_G$ indicates that the quantity is evaluated on the worldline. This is just a Taylor expansion for the metric where the "small parameter", s (say), is taken to be the proper distance from the central curve. That is, we have $s^2 = (x^1)^2 + (x^2)^2 + (x^3)^2$. We see that, if the worldline is a geodesic then $a^{\hat{j}} = 0$ and there are no corrections up to second order in the metric. However, there will always be corrections at second order due to the spacetime curvature. These corrections can be expressed in terms of the Riemann tensor (again evaluated on the worldline *G*), but we will not need the explicit results here. Because we are assuming that the metric fluctuations happen on a larger scale (with respect to the fluid variations), we can make use of these expansions in the following.

Next, we can use the coordinates we have introduced to define a formal averaging or filtering procedure. Focusing on time-averaging first, we may use the spacetime split

⁴That is, the four acceleration is $a_b = U^a \nabla_a U_b$ here.

associated with the coordinates and define the procedure as

$$\overline{A}(\hat{x}) = \lim_{T \to \infty} \frac{1}{T} \int_0^T d\hat{\tau} A(\hat{x}, \hat{\tau}) .$$
(4.11)

That is, given a point on the fluid element trajectory we average in the proper time associated with the worldline. In terms of the Fermi coordinates, the time coordinate is exactly the proper time of the central curve. Note that there is no problem in taking the limit $T \to \infty$ since the Fermi coordinates are formally defined over the entire worldline. The region of validity is only limited in the spatial directions orthogonal to the central curve. From the definition, it is clear that time-averaged quantities must be time-independent, and it immediately follows that $\overline{A} = \overline{A}$ (making contact with (4.6)). We stress that this property follows if and only if we take the limit $T \to \infty$ in the definition, exactly as in the Newtonian case. As such, this implies that we should (strictly) neglect time derivatives in the averaged equations. The upshot is that the time-averaging strategy is (formally) valid for stationary flows only. The same is true in the Newtonian context—even though time derivatives are typically retained in the equations⁵. In fact, this is one of the main motivations in favour of spatial filtering and large-eddy models. In the following, we follow the "tradition" and retain terms involving time derivatives, as the main point of our discussion of the time-averaging case is pedagogical.

As an application, let us consider the averaged metric. From the metric expansion above, we immediately see that the "time-time" component gains a correction due to the acceleration, which does depend on the proper time. However, we only need to integrate over points on the central worldline, where we have the Minkowski metric (in Cartesian coordinates) by construction. The situation is similar for all the remaining components.

As a result, each component of the averaged metric takes the non-averaged value from the central worldline. That is, the metric is constant (in the sense described in section 4.3), and we have

$$\overline{g}_{\hat{a}\hat{b}} = g_{\hat{a}\hat{b}} . \tag{4.12}$$

Similarly, U^a is constant under averaging. To see this it is sufficient to note that (in terms of the Fermi coordinates) we have $U^{\hat{a}} = (1, 0, 0, 0)^{\top}$ so that $\overline{U}^a = U^a$.

Analogously, we can use the spacetime split to define a space-filtering. We first have to assume that the width *L* of the region over which we are filtering—the "resolved box"—is small enough (in terms of the distance λ) that the Fermi coordinates are well defined on it. For instance, one such condition is L < 1/a where *a* is the magnitude

⁵See [151] for a more extensive discussion on this.

of $a^{\hat{j}}$ (see [163] for a detailed discussion). In this case, the filtering procedure may be defined through

$$\langle A \rangle(\hat{x}, \hat{t}) = \int dV A(\hat{x} + \hat{y}, \hat{t}) f(\hat{y}) , \qquad (4.13)$$

where we introduced the filter f (normalised over the resolved box). The expression simplifies further if we note that, by construction, the spacetime point (\hat{x}, \hat{t}) lies on the central worldline $(x^{\hat{0}} = \hat{t} = \tau, x^{\hat{i}} = 0)$ where τ is the relevant proper time. Also, in terms of the Fermi coordinates, the volume element is

$$dV = U^{\hat{0}} \sqrt{-g} dy^{\hat{1}} dy^{\hat{2}} dy^{\hat{3}} = (1 + 2a_{\hat{j}}y^{\hat{j}})^{1/2} dy^{\hat{1}} dy^{\hat{2}} dy^{\hat{3}} \approx (1 + a_{\hat{j}}y^{\hat{j}}) d^{\hat{3}} \hat{y} .$$
(4.14)

Note that we have not specified the exact filter to be used. This is not required at this stage, but we will assume the filter to be an even function (noting that this is the case for the three most common filters used in large-eddy models, see [136] for instance) and normalized over the resolved box such that we have⁶

$$\int d^3\hat{y} f(\hat{y}) = 1 \quad , \quad \int d^3\hat{y} \, y^{\hat{j}} f(\hat{y}) = 0 \,. \tag{4.15}$$

Again, let us first apply this to the filtered metric. Since there are no first-order corrections in the expansion in eq. (4.10), $g_{\hat{l}\hat{i}}$ and $g_{\hat{i}\hat{i}}$ are constant over the box, and we have

$$\langle g_{\hat{0}\hat{i}} \rangle = 0 = g_{\hat{0}\hat{i}} |_{G}$$
, (4.16a)

$$\langle g_{\hat{i}\hat{j}} \rangle = \int d^3 \hat{y} (1 + a_{\hat{j}} y^{\hat{j}}) f(\hat{y}) \eta_{\hat{i}\hat{j}} = \eta_{\hat{i}\hat{j}} = g_{\hat{i}\hat{j}} \big|_G .$$
 (4.16b)

We also have

$$\langle g_{\hat{0}\hat{0}} \rangle = -\int d^3 \hat{y} (1 + 2a_{\hat{i}} y^{\hat{i}})^{3/2} f(\hat{y})$$

= $-1 - 3a_{\hat{i}} \int d^3 \hat{y} \, y^{\hat{i}} f(\hat{y}) = -1 = g_{\hat{0}\hat{0}} \big|_G ,$ (4.17)

where the last integral vanishes because of the assumed symmetry of the kernel. Once again, each component of the filtered metric takes the non-averaged value from the central worldline throughout the region under consideration. That is, the metric is constant

$$\langle g_{\hat{a}\hat{b}} \rangle = g_{\hat{a}\hat{b}}.\tag{4.18}$$

We also note that, by construction U^a is constant over the box so we have $\langle U^a \rangle = U^a$.

Finally, since we have shown that the metric can (effectively) be taken to be constant under both averaging and filtering, it is easy to show that partial derivatives commute

⁶If the filter has a sharp boundary—i.e. vanishes at the boundary of the resolved box—the argument does not involve extending the spatial integral to infinity. However, one can think of filters with no sharp boundary, like a Gaussian filter (see [151]), which may give rise to formal issues. In practice though, the exponential tail of the Gaussian should suppress anything beyond the Fermi coordinate boundary.

with each procedure. That is (obviously connecting with (4.3))

$$\partial_a \langle A \rangle = \langle \partial_a A \rangle$$
 and $\partial_a \overline{A} = \overline{\partial_a A}$. (4.19)

We now show that these relations hold given the definitions of the average/filter above. Let us start with the time-averaging case. When we consider the partial derivative with respect to the spatial coordinates the argument is straightforward, as the partial derivative (in the spatial direction) can be brought inside the integral. For the time derivative, we have

$$\overline{\partial_{\hat{t}}A}(\hat{x},\hat{t}) = \lim_{T \to \infty} \int_0^T d\hat{\tau} \partial_{\hat{\tau}}A(\hat{x},\hat{\tau}) = \lim_{T \to \infty} \frac{A(T) - A(0)}{T} = 0.$$
(4.20)

On the other hand, if we take the time derivative of the averaged quantity, this trivially vanishes as it is time-independent.

The argument for the time derivative in the filtering case is similarly straightforward. For spatial derivatives, we have on the one hand

$$\langle \partial_{\hat{i}} A \rangle(\hat{x}, \hat{t}) = \int dV \left(\partial_{\hat{i}} A \right) (\hat{x} + \hat{y}, \hat{t}) f(\hat{y}) , \qquad (4.21)$$

while, on the other hand,

$$\frac{\partial}{\partial x^{\hat{i}}} \langle A \rangle(\hat{x}, \hat{t}) \big|_{\hat{x}_0} = \int dV \frac{\partial}{\partial x^{\hat{i}}} A(\hat{x} + \hat{y}, \hat{t}) \big|_{\hat{x}_0} f(\hat{y}) .$$
(4.22)

Using the chain-rule in the last equation, we see that

$$\frac{\partial}{\partial x^{\hat{i}}} A(\hat{x} + \hat{y}) \big|_{\hat{x}_0} = \frac{\partial A(\hat{z})}{\partial z^{\hat{i}}} \big|_{\hat{z} = \hat{x}_0 + \hat{y}}, \qquad (4.23)$$

and it is clear that the two relations lead to the same result.

We now have all the ingredients we need to prove that covariant derivatives commute with the averaging/filtering procedure. In fact, given that the metric is constant (in the sense of eq. (4.3)), we have

$$\langle \partial_c g_{ab} \rangle = \partial_c \langle g_{ab} \rangle = \partial_c g_{ab}$$
, (4.24a)

$$\partial_c g_{ab} = \partial_c \overline{g_{ab}} = \partial_c g_{ab} .$$
 (4.24b)

As a result, the Christoffel symbols—which are obtained from combinations of first derivatives of the metric—are (locally) constant under the procedure, as well. Therefore, we have

$$\langle \nabla_a A^b \rangle = \partial_a \langle A^b \rangle + \Gamma^b_{ac} \langle A^c \rangle = \nabla_a \langle A^b \rangle , \qquad (4.25)$$

with an analogous result for the time-averaging case. At the end of the day, the argument is quite intuitive.

4.4.1 On covariance and the Einstein equations

It makes sense now, before we move on, to spell out the covariance of the proposed averaging/filtering procedure, and comment on its compatibility with the field equations of General Relativity. It is, in fact, clear that the integrals we used to define the averaging/filtering procedure are not of the usual type. We have to define each procedure in such a way that the integration preserves the tensorial nature of the input⁷. Given this, we define the procedure for scalar quantities and then apply it to each component⁸ of, say, the metric tensor. We then require the averaged/filtered quantity to transform as a tensor on the resolved scale. The proposed definition reduces to the one of [77] in the special relativity context and it also leads back to the Newtonian ones [136, 151]. The main difference is that, in special relativity such integrals are chart independent—as long as the integration is performed on each component using a fixed basis (see [101])—while this is not the case in General Relativity. However, even though the special relativistic integrals are chart independent, the results are not, because the notions of length- and time scales are observer dependent. We also note that, as shown in [77], the observer-dependence cannot be resolved by the introduction of some kind of "spacetime" filter. However, we are setting up the averaging/filtering using Fermi coordinates defined from the fibration. As this is naturally associated with the fluid motion, the "gauge" dependence of the procedure is more physical. We execute the smoothing in "some" local frame U^a which we can choose to "associate" to the "microscale" fluid motion.

Let us now discuss the averaging/filtering of the Einstein equations, focusing on the geometry. First of all, consider the Einstein tensor G^{ab} . From eqs. (4.19) and (4.24) it follows that

$$\langle g_{ab,cd} \rangle = g_{ab,cd} , \qquad (4.26a)$$

$$\overline{g_{ab,cd}} = g_{ab,cd} . \tag{4.26b}$$

Since the Einstein tensor is ultimately a combination of the metric and its (up-to-second order) derivatives $G = G(g, \partial g, \partial^2 g)$, this implies that we must have

$$\overline{G(g, \partial g, \partial^2 g)} = G(\overline{g}, \partial \overline{g}, \partial^2 \overline{g}) = G(g, \partial g, \partial^2 g) , \qquad (4.27)$$

and analogously for the filtering case. The net result is that the coarse-grained theory remains consistent with General Relativity. In particular, the Einstein equations become

⁷Recall that the usual integral on a (sub-)manifold of dimension p takes a p-form as input and outputs a scalar [101].

⁸Intuitively, because the procedures are based on the Fermi-coordinates construction in terms of a non-rotating tetrad with respect to U^a , we can effectively think of the components as scalars.

$$G^{ab} = 8\pi \overline{T^{ab}} , \qquad (4.28a)$$

$$G^{ab} = 8\pi \langle T^{ab} \rangle . \tag{4.28b}$$

These results follow from the Fermi-coordinate construction, and the assumed separation of scales in the metric fluctuations with respect to the fluid variables. This should be a safe assumption for binary neutron star merger applications, but not necessarily for problems relating to the very early universe (where quantum fluctuations in the gravitational field may play an important role). In this sense the Fermi-coordinate construction should be regarded as a pragmatic argument rather than a mathematical proof. Having said that, we can now focus the discussion on the matter side, i.e. the stress-energy-momentum tensor T^{ab} .

4.5 Averaging in the fluid frame

Backed up by the Fermi-coordinate argument, let us first explore the problem of spacetime averaging. We start by introducing a fine-grained congruence of wordlines with tangent vector field u^a , e.g. associated with individual fluid elements. Then, working on a slightly larger (coarse grained) scale, we introduce another vector field U^a —the one used to define Fermi coordinates—such that small scale features are smoothed. In effect, we can then use the decomposition

$$u^a = \gamma (U^a + \delta v^a) , \qquad (4.29)$$

with

$$U_a \delta v^a = 0 , \qquad (4.30)$$

and

$$\gamma = \left(1 - \delta v^2\right)^{-1/2} \approx 1 + \frac{1}{2} g_{ab} \delta v^a \delta v^b .$$
(4.31)

At this point we take the view that it is natural to assume $\delta v \ll 1$ (the speed of the fluctuations is well below that of light) as this should be a safe assumption for the problems we are interested in [72]. This allows us to develop the logic more explicitly, even though we will drop this assumption later. One may view the linear assumption as an additional constraint (alongside the assumptions of the Fermi frame) on the size of the region we average over, although we will not try to make this statement precise. Finally, let us assume that it makes sense to work with an ordered expansion in the fluctuations. Working to second order—throughout the discussion of averaging but not in the filtering case that follows, where the expressions are not expanded in this sense—we then have

$$u^{a} \approx \left(1 + \frac{1}{2}g_{bc}\delta v^{b}\delta v^{c}\right)U^{a} + \delta v^{a}.$$
(4.32)

Let us now consider the average of this four velocity. Given the set-up we have $\overline{U}^a = U^a$ and one might expect to have $\overline{u}^a = U^a$, as well. However, the problem turns out to be a little bit more intricate than that. First of all, from the discussion of time-averaging in section 4.3 we assume

$$\overline{\delta v^a} = 0. \tag{4.33}$$

It is also worth noting that the averaging procedure preserves directionality; that is

$$U_a \delta v^a = 0 \implies U_a \overline{\delta v^a} = 0 , \qquad (4.34)$$

This holds as long as we satisfy the conditions laid out in section 4.4—not because of eq. (4.33). We also get (to second order)

$$\overline{\gamma} = 1 + \frac{1}{2} g_{ab} \overline{\delta v^a \delta v^b} , \qquad (4.35)$$

and it follows that

$$\overline{u}^{a} = \left(1 + \frac{1}{2}g_{bc}\overline{\delta v^{b}\delta v^{c}}\right)U^{a} = \overline{\gamma}U^{a}.$$
(4.36)

At this point we reach an impasse. It is clear that \overline{u}^a is not (automatically) normalised and therefore can not serve as a four velocity. We would have to re-calibrate co-moving clocks to depend on the averaged fluctuations. This is problematic as we need a projection to effect the space-time split and one would expect this to involve the fluid four velocity. There seems to be two ways to proceed. First, we could (perhaps pragmatically) opt to work with U^a as the variable representing the flow. Alternatively, we may constrain the fluctuations to ensure that the averaging procedure returns $\overline{u}^a = U^a$. This would follow if we were to assume the fluctuations to be such that

$$g_{ab}\delta v^a \delta v^b = 0 \implies \overline{\gamma} = 1.$$
(4.37)

This allows us to move on, working with \overline{u}^a to represent the flow, which might seem the natural generalisation of the Newtonian logic. However, considering the expected nature of small-scale turbulence, condition (4.37) seems too restrictive. By assuming that the variance of the velocity fluctuations vanishes, we effectively remove the small scale kinetic energy that links to large-scale features in standard eddy-based models for turbulence. The kinetic energy (per particle) of the fluctuations is defined as

$$k = \frac{1}{2} g_{ab} \overline{\delta v^a \delta v^b} , \qquad (4.38)$$

which clearly vanishes if we impose (4.37). The second of the suggested approaches thus seems unattractive and we will not pursue it further. A third—indeed, likely preferred—possibility will become apparent when we consider the equation for the conserved matter flux.

4.5.1 Baryon number conservation

Having discussed the issues associated with averaging the four-velocity, the natural next step is to consider baryon number conservation. Letting $n = \overline{n} + \delta n$ represent the baryon number, the matter flux takes the form

$$n^{a} = nu^{a} \approx \left(\overline{n} + \delta n\right) \left(\gamma U^{a} + \delta v^{a}\right), \qquad (4.39)$$

such that (since $\overline{\delta n} = 0$ from the averaging)

$$\overline{n}^a = \overline{n}\,\overline{\gamma}\,U^a + \overline{\delta n \delta v^a} \,. \tag{4.40}$$

The number density measured by an observer moving along with U^a is then given by

$$n_0 = -U_a \overline{n}^a = \overline{n} \,\overline{\gamma} \,, \tag{4.41}$$

and we can write the averaged flux as

$$\overline{n}^a = n_0 U^a + \overline{\delta n \delta v^a} . \tag{4.42}$$

The continuity equation then becomes

$$\overline{\nabla_a n^a} = \nabla_a \overline{n}^a = \nabla_a (n_0 U^a) + \nabla_a \left(\overline{\delta n \delta v^a}\right) = 0 , \qquad (4.43)$$

or

$$U^{a}\nabla_{a}n_{0} + n_{0}\nabla_{a}U^{a} = -\nabla_{a}\overline{\delta n\delta v^{a}} .$$

$$(4.44)$$

This last equation shows that there is particle diffusion at second order (relative to U^a). Fluctuations lead to drift from large scale elements to their neighbours.

While nothing prevents us from taking this as given and moving on to the stressenergy-momentum tensor, the other equations of motion and the equation of state, it is useful to consider a density-weighted velocity, now starting from the flux n^a . The advantage of this is that we can arrange things in such a way that the new four velocity is normalised, while the non-linear fluctuations are hidden in its definition. In essence, we use the weighting with the number density to adjust the co-moving clocks in the desired way. Suppose we define

$$\overline{n}^a = \tilde{n}\tilde{u}^a , \qquad (4.45)$$

while insisting that $\tilde{u}_a \tilde{u}^a = -1$. This immediately leads to⁹

$$\tilde{n} = \overline{n}\,\overline{\gamma}\,,\tag{4.46}$$

⁹From eqs. (4.45) and (4.46) we see that this density-weighted average corresponds to the Favre-type averaging often used in the Newtonian context (see, for instance, [138, 193, 70]).

and

$$\tilde{u}^a = U^a + \frac{1}{\tilde{n}} \overline{\delta n \delta v^a} \,. \tag{4.47}$$

It is easy to see that this retains the required normalisation as long as we ignore terms beyond second order. Crucially, this would remain true (by construction) if we did not expand in small fluctuations. We can now meaningfully introduce the projection with respect to the "Favre"-filtered observer \tilde{u}^a , namely

$$\tilde{\perp}^a_b = \delta^a_b + \tilde{u}^a \tilde{u}_b \,. \tag{4.48}$$

As anticipated, it also follows that

$$\tilde{n} = -\tilde{u}_a \overline{n}^a , \qquad (4.49)$$

and the continuity equation takes the form

$$\nabla_a(\tilde{n}\tilde{u}^a) = \dot{\tilde{n}} + \tilde{n}\nabla_a\tilde{u}^a = 0, \qquad (4.50)$$

where the "dot" represent the covariant derivative with respect to \tilde{u}^a , i.e. $\tilde{n} = \tilde{u}^a \nabla_a \tilde{n}$. This is attractive because—as the fluctuations are "hidden"—we are left with a conservation law of the pre-averaged form. Note that we can remove the need of a closure in this equation, but this forces us to use the \tilde{u}^a observer. One can imagine making a different choice, which would lead to drift terms entering the continuity equation, as in (4.44).

4.5.2 Averaged matter dynamics

Next, consider the perfect fluid stress-energy-momentum tensor (derived in section 1.3). Starting from

$$T^{ab} = (p+\varepsilon)u^a u^b + pg^{ab} , \qquad (4.51)$$

we make use of (4.32) (noting that, to second order in the fluctuations, we have $\overline{\gamma^2} = \overline{\gamma}^2$), then after averaging we introduce \tilde{u}^a according to (4.47). This leads to

$$\overline{T}^{ab} = (\overline{p} + \overline{\varepsilon})\overline{\gamma}^2 \tilde{u}^a \tilde{u}^b + \overline{p}g^{ab} + 2\tilde{u}^{(a}q^{b)} + s^{ab} , \qquad (4.52)$$

with

$$q^{a} = -\frac{\overline{p} + \overline{\varepsilon}}{\overline{n}} \overline{\delta n \delta v^{a}} + \overline{(\delta p + \delta \varepsilon) \delta v^{a}} , \qquad (4.53)$$

and

$$s^{ab} = (\overline{p} + \overline{\varepsilon})\overline{\delta v^a \delta v^b} . \tag{4.54}$$

It is convenient to work with the energy density measured by an observer moving along with \tilde{u}^a . This follows from

$$\tilde{\varepsilon} = \tilde{u}_a \tilde{u}_b \overline{T}^{ab} = \overline{\gamma}^2 \overline{\varepsilon} + (\overline{\gamma}^2 - 1) \overline{p} .$$
(4.55)

As a result, we have

$$\overline{p} + \tilde{\varepsilon} = (\overline{p} + \overline{\varepsilon})\overline{\gamma}^2$$
, (4.56)

which means that we can rewrite the stress-energy-momentum tensor as

$$\overline{T}^{ab} = (\overline{p} + \tilde{\varepsilon})\tilde{u}^a\tilde{u}^b + \overline{p}g^{ab} + 2\tilde{u}^{(a}q^{b)} + s^{ab}.$$
(4.57)

The equations of motion

$$\nabla_a \overline{T}^{ab} = 0 , \qquad (4.58)$$

then lead to the energy equation;

$$\dot{\tilde{\varepsilon}} + (\overline{p} + \tilde{\varepsilon}) \nabla_a \tilde{u}^a = \tilde{u}_b \tilde{u}^a \nabla_a q^b - \nabla_a q^a + \tilde{u}_b \nabla_a s^{ab} , \qquad (4.59)$$

and the momentum equation;

$$(\overline{p}+\tilde{\varepsilon})\,\tilde{a}^b+\tilde{\bot}^{ab}\nabla_a\overline{p}=-\tilde{\bot}^b_c\tilde{u}^a\nabla_aq^c-q^b\nabla_a\tilde{u}^a-q^a\nabla_a\tilde{u}^b-\tilde{\bot}^b_c\nabla_as^{ac}\,,\qquad(4.60)$$

where $\tilde{a}^b = \tilde{u}^a \nabla_a \tilde{u}^b$. It is worth remarking that with this choice of resolved variables, the final equations of motion resemble those of a general dissipative fluid with viscosity and heat-flux (cf. discussion in chapter 2).

4.5.3 The equation of state

A key step of any fluid model involves the connection to the microphysics as represented by the equation of state. As a first stab at this, let us outline the logic for the simple barotropic case (and then return to the issue in section 4.7, with a more realistic model in mind). In a barotropic model the starting point is a one-parameter energy density $\varepsilon = \varepsilon(n)$, which leads to the thermodynamical (Gibbs) relation

$$p + \varepsilon = n\mu$$
, (4.61)

where the chemical potential is defined as

$$\mu = \frac{d\varepsilon}{dn} . \tag{4.62}$$

Introducing fluctuations in all the scalars, as before (eg. $p = \overline{p} + \delta p$), we have

$$\overline{p} + \delta p + \overline{\varepsilon} + \delta \varepsilon = (\overline{n} + \delta n)(\overline{\mu} + \delta \mu) = \overline{n} \,\overline{\mu} + \overline{n}\delta\mu + \overline{\mu}\delta n + \delta n\delta\mu \,. \tag{4.63}$$

Since the average of linear fluctuations are all taken to vanish in the present case, this leads to

$$\overline{p} + \overline{\varepsilon} = \overline{n} \,\overline{\mu} + \overline{\delta n \delta \mu} , \qquad (4.64)$$

which shows that the number density fluctuations impact on the equation of state inversion required to evolve the system. In order to proceed, we need to provide a closure relation for $\overline{\delta n \delta \mu}$.

Noting that the energy was assumed to depend only on the number density, the fluctuations in (4.64) should not be independent. In order to take a closer look at this, we may Taylor expand for small fluctuations δn (recalling that we have already considered expanding the Lorentz factor in this way). This then leads to

$$\mu \approx \mu(\overline{n}) + \mu'(\overline{n})\delta n + \frac{1}{2}\mu''(\overline{n})\delta n^2 .$$
(4.65)

That is, we identify

$$\overline{\mu} = \mu(\overline{n}) + \frac{1}{2}\mu''(\overline{n})\overline{\delta n^2} , \qquad (4.66)$$

and

$$\delta\mu = \mu'(\overline{n})\delta n . \tag{4.67}$$

The averaged Gibbs relation then becomes

$$\overline{p} + \overline{\varepsilon} = \overline{n}\mu(\overline{n}) + \left[\frac{\overline{n}}{2}\mu''(\overline{n}) + \mu'(\overline{n})\right]\overline{\delta n^2} , \qquad (4.68)$$

showing that—in addition to the derivatives of the chemical potential— we need to provide a (closure) relation for $\overline{\delta n^2}$. The other fluctuations are similarly slaved to δn . We get

$$\varepsilon(n) \approx \varepsilon(\overline{n}) + \varepsilon'(\overline{n})\delta n + \frac{1}{2}\varepsilon''(\overline{n})\delta n^2$$
, (4.69)

which leads to

$$\bar{\varepsilon} = \varepsilon(\bar{n}) + \frac{1}{2}\varepsilon''(\bar{n})\overline{\delta n^2} , \qquad (4.70)$$

and

$$\delta \varepsilon = \varepsilon'(\overline{n}) \delta n = \mu(\overline{n}) \delta n .$$
(4.71)

In summary, we have equations describing the (proper time) evolution of \tilde{n} , $\tilde{\varepsilon}$ and \tilde{u}^a , represented by equations (4.50), (4.59) and (4.60). In order to be able to solve these equations, we need to provide closure relations for the fluctuations involved in $\overline{\gamma}$, q^a and s^{ab} , that is $\overline{\delta n \delta v^a}$ and $\overline{\delta v^a \delta v^b}$. Once these, and $\overline{\delta n^2}$, are provided, we can work out \overline{n} from \tilde{n} and—assuming that we have access to the derivatives of the chemical potential—then we have $\overline{\varepsilon}$, as well. Using either (4.68) or (4.56) we can then rewrite \overline{p} in terms of resolved variables and the closure terms, which completes the set of quantities

we need to close the system and carry on the evolution. In principle, the fibration based averaging model is complete.

4.6 Fluid element filtering

Having established a workable procedure for averaging in the spacetime setting, let us turn to the issue of filtering. This is important because most actual numerical simulations assume filtering rather than averaging. Formally, we expect the two problems to be similar, but we know from section 4.3 that the filtering problem involves a slightly different logic. In particular, it is less natural—and also not desirable since the filtered fluctuations are unlikely to vanish—to consider an expansion in terms of the fluctuations. This is an important difference, so we need to carefully consider the steps in the analysis.

The natural way to proceed is to use a weighted average, in the spirit of (4.47). We would then start from¹⁰

$$\langle n^a \rangle \equiv \tilde{n} \tilde{u}^a$$
, (4.72)

with (by construction) $\tilde{u}_a \tilde{u}^a = -1$, leading to

$$\tilde{n} = -\tilde{u}_a \langle n^a \rangle \,. \tag{4.73}$$

As in the averaging case, it is easy to see that the continuity equation then becomes

$$\langle \nabla_a n^a \rangle = \nabla_a \langle n^a \rangle = 0 \implies \dot{\tilde{n}} + \tilde{n} \nabla_a \tilde{u}^a = 0.$$
 (4.74)

The main lesson is that, formally, the equation we arrive at takes the same form as in the averaging case; equation (4.50). Still, there are differences relating to i) the nonlinear quantities that have to be provided by a closure model and ii) the interpretation of the resolved/evolved variables.

Deferring the discussion of more general aspects of the equation of state for a moment (these will be discussed in section 4.7), let us move on to write down the equations of motion consistent with (4.74). The filtered version of the perfect fluid stress-energy-momentum tensor can be written as

$$\langle T^{ab} \rangle = (\langle p \rangle + \langle \varepsilon \rangle) \tilde{u}^a \tilde{u}^b + \langle p \rangle g^{ab} + \tau^{ab} , \qquad (4.75)$$

where

$$\tau^{ab} = \langle (p+\varepsilon)u^a u^b \rangle - (\langle p \rangle + \langle \varepsilon \rangle) \tilde{u}^a \tilde{u}^b , \qquad (4.76)$$

¹⁰It is important to note that quantities like \tilde{n} are not the same in the averaging and filtering cases. Still, we are using the same notation because they play the same role in each evolution scheme. Note also that our definition does not mean that $\langle \tilde{u}^a \rangle = \tilde{u}^a$.

requires a closure relation.

Introducing, as before, the energy density measured by \tilde{u}^a :

$$\tilde{\varepsilon} = \tilde{u}_a \tilde{u}_b \langle T^{ab} \rangle = \langle \varepsilon \rangle + \tilde{u}_a \tilde{u}_b \tau^{ab} , \qquad (4.77)$$

we can rewrite the filtered stress-energy-momentum tensor as

$$\langle T^{ab} \rangle = (\langle p \rangle + \tilde{\varepsilon}) \tilde{u}^a \tilde{u}^b + \langle p \rangle g^{ab} + 2 \tilde{u}^{(a} q^{b)} + s^{ab} , \qquad (4.78)$$

with

$$q^{a} = -\tilde{\perp}^{a}_{b}\tilde{u}_{c}\tau^{cb} = -\tilde{\perp}^{a}_{b}\tilde{u}_{c}\left\langle (p+\varepsilon)u^{a}u^{b}\right\rangle, \qquad (4.79)$$

and

$$s^{ab} = \tilde{\perp}^a_c \tilde{\perp}^b_d \tau^{cd} = \tilde{\perp}^a_c \tilde{\perp}^b_d \langle (p+\varepsilon)u^a u^b \rangle .$$
(4.80)

It is easy to see that the energy and momentum equations take exactly the same form (once we change $\langle p \rangle \rightarrow \overline{p}$) as in the averaged case (cf. equations (4.59) and (4.60)). As in that case, the equations of motion provide all the information we need to solve the system once the relevant closure relations are provided. The key to a workable model is to provide appropriate closure relations, so it is important to understand what this involves.

4.7 Filtered Thermodynamics

So far we have only outlined the argument for the simple case of a barotropic fluid. As the equation of state is a central issue for any realistic model—and, obviously, any numerical simulation—let us rethink this. A quick look back at eq. (4.64) shows that, even if we start from a barotropic Gibbs relation at the fine scale, the averaged/filtered result is effectively "non-barotropic". In fact, the $\delta n \delta \mu$ closure term could be interpreted as an "entropy-like" contribution associated with the fluctuations. This suggests that it makes sense to start straight away from a non-barotropic model, i.e. with a Gibbs relation of the form

$$p + \varepsilon = \mu n + Ts$$
, (4.81)

where *s* is the entropy density and *T* is the associated temperature.

The barotropic example also shows that the "effective" term that stems from the averaging or filtering procedure does not relate to the actual entropy, in the sense that it is not associated with some dissipative process and/or entropy production rate. This is evident from the fact that we have freedom in the choice of the averaging/filtering observer, and therefore in the variables to be evolved, and one might choose to frame the model in such a way that the fluctuation terms are reabsorbed in the definition of the variables themselves, just like we did for the weighted four-velocity \tilde{u}^a in eq. (4.47). Let us try to make these points more concrete, by considering a model that is nonbarotropic from the get-go. Because there is no formal difference in the resulting equations, we will do this without distinguishing between the averaging and filtering cases (although, as we are now familiar with the logic, using the slightly more abstract notation from the latter).

We start by discussing some subtleties of the barotropic example that we previously left aside. The energy density of a barotropic fluid is a function of the matter density only, which means there is no need to evolve both quantities separately. On the fine-scale, the energy equation contains the same information as the continuity equation:

$$\frac{d\varepsilon}{d\tau} + (p+\varepsilon)\nabla_a u^a = \mu \frac{dn}{d\tau} + \mu n \nabla_a u^a = \mu(\nabla_a n^a) = 0 , \qquad (4.82)$$

where τ is the proper time associated with the "actual" fluid worldlines with tangent u^a . Note that the Gibbs relation eq. (4.61) is crucial for this argument. However, the situation changes on the coarse-grained scale. The link between the micro-scale equation of state and the resolved energy is not trivial—it may have to be established by a set of high resolution simulations, even though this may not be practical/feasible. In effect, the equation of state we are working with here is not the one you get from nuclear physics. This is, in fact, true also in the simpler case where we set to zero the contribution coming from the τ^{ab} residual as the evolved density is not obtained by averaging the fine-scale one $\tilde{n} \neq \langle n \rangle$. The net result is that we have to treat the resolved energy $\tilde{\varepsilon}$ and the resolved density \tilde{n} as independent variables, and evolve both of them.

4.7.1 The effective entropy

This subtle difference in the counting of independent variables between the fine- and coarse scale models obviously no longer exists for a non-barotropic fluid. For a two-parameter equation of state, the energy and particle density can be taken as independent variables already at the fine scale. This is, indeed, standard practice in numerical relativity simulations. Let us also recall that, if the fluid is ideal there is no additional information gained from evolving the entropy current. In fact, we have seen in section 1.3 that the entropy current is automatically advected as a consequence of the perfect fluid equations, provided this is a function of the energy and particle number densities, $s = s(n, \varepsilon)$.

To complete the model set-up, however, we still have to clarify how the filtered pressure relates to the evolved variables. The barotropic model has been discussed in section 4.5.3 for the averaging case, and the filtering case would work analogously. We now look at the non-barotropic fluid case. We can work with a resolved entropy defined as the usual thermodynamic potential $\tilde{s} \doteq s(\tilde{\varepsilon}, \tilde{n})$. The resolved temperature and chemical potential then follow from the standard definitions

$$\frac{1}{\tilde{T}} \doteq \left(\frac{\partial \tilde{s}}{\partial \tilde{\varepsilon}}\right)_{\tilde{n}} (\tilde{\varepsilon}, \, \tilde{n}) , \qquad (4.83a)$$

$$-\frac{\tilde{\mu}}{\tilde{T}} \doteq \left(\frac{\partial \tilde{s}}{\partial \tilde{n}}\right)_{\tilde{\varepsilon}}(\tilde{\varepsilon},\,\tilde{n}) \,. \tag{4.83b}$$

We stress that we chose to use as thermodynamic potential the entropy, as it is a function of the chosen independent variables in the equations of motion, \tilde{n} and $\tilde{\epsilon}$. Note also that, \tilde{s} does not represent the true entropy and \tilde{T} is not the actual temperature, either. We are simply assuming that the usual thermodynamical definitions "make sense" at the filtering scale. Then, following the same logic we used for the barotropic case (see section 4.5.3) we filter eq. (4.81) and rewrite it as

$$\langle p \rangle = -\tilde{\varepsilon} + \tilde{\mu}\tilde{n} + \tilde{T}\tilde{s} + M$$
, (4.84)

with

$$M = \left(\langle Ts \rangle - \tilde{T}\tilde{s} \right) + \left(\langle \mu n \rangle - \tilde{\mu}\tilde{n} \right) - \left(\langle \varepsilon \rangle - \tilde{\varepsilon} \right).$$
(4.85a)

The argument is now complete. We have explained how to express the averaged/filtered pressure that enters the equations of motion in terms of the resolved variables $\tilde{\epsilon}$, \tilde{n} and the (new) residual *M*.

Having considered the resolved thermodynamics, we can turn to the "entropy production" associated with the averaging/filtering procedure. The final equations (i.e. eqs. (4.59) and (4.60)) clearly remind us of the result for a dissipative fluid (see [21]) so we are motivated to consider possible constraints stemming from the second law of thermodynamics. To do this we can work through steps analogous to eq. (1.12) to establish the impact of the averaging/filtering procedure. Because the resolved entropy \tilde{s} is taken to be a function of the resolved energy $\tilde{\varepsilon}$ and the number density \tilde{n} we have

$$\tilde{T}\nabla_a(\tilde{s}\tilde{u}^a) = \tilde{T}\tilde{s}\nabla_a\tilde{u}^a + \tilde{T}\dot{\tilde{s}} = \tilde{T}\tilde{s}\nabla_a\tilde{u}^a + \dot{\tilde{\varepsilon}} - \tilde{\mu}\dot{\tilde{n}} .$$
(4.86)

Now, by means of eqs. (4.50) and (4.59) we obtain

$$\tilde{T}\nabla_{a}(\tilde{s}\tilde{u}^{a}) = (\tilde{T}\tilde{s} + \tilde{\mu}\tilde{n} - \langle p \rangle - \tilde{\varepsilon})\nabla_{a}\tilde{u}^{a} - q^{a}\tilde{a}_{a} - \nabla_{a}q^{a} - s^{ab}\nabla_{a}\tilde{u}_{b}$$
$$= -M\nabla_{a}\tilde{u}^{a} - q^{a}\tilde{a}_{a} - \nabla_{a}q^{a} - s^{ab}\nabla_{a}\tilde{u}_{b} .$$
(4.87)

This shows that the entropy is no longer advected at the coarse scale, as a result of the averaging/filtering procedure. However, the fine scale ("exact") theory is ideal, so the actual entropy is advected. For this reason, the model is not constrained by the second law at the coarse-grained scale. This is a very important point as it impacts on the closure relations (see below), which (evidently) can be discussed without considering the thermodynamical restrictions for "real" dissipative fluids. Effectively, the heat-flux

term in eq. (4.53) or eq. (4.79) is associated with energy transfer from large eddies to small ones (or vice versa) rather than being a faithful heat transfer. We also note that this would not change even if we started from a fluid that is dissipative already at the fine scale. Restrictions stemming from the second law of thermodynamics apply only at the fine-scale level, not at the coarse one.

4.7.2 Energy cascade argument

Having discussed the thermodynamical interpretation of the quantities that enter the equations of motion, it makes sense to consider the involved energy cascade. This is relevant because an analogous argument is used in standard work on turbulence (see, for instance, [139, 137]) to motivate the closure of the fluid equations. It is useful to spell out the relativistic analogue of the classical argument.

The starting point is the energy equation (see eq. (4.59)) rewritten as

$$\underbrace{\check{\tilde{\varepsilon}} + (\langle p \rangle + \tilde{\varepsilon}) \nabla_a \tilde{u}^a}_{\text{macro}} = \overbrace{-q^b \tilde{a}_b + \nabla_a q^a + \tilde{u}_b \nabla_a s^{ab}}^{\text{mixed}} . \tag{4.88}$$

Here, we have highlighted that the terms on the left-hand side can be considered as macroscopic, in the sense that they involve only resolved quantities and describe an ideal evolution—intended to correctly capture the large-scale dynamics. In contrast, the terms on the right-hand side are "mixed" as they involve unresolved quantities—the residuals—and couple macro- and micro-scale terms. In effect, they can be thought of as transferring energy from one scale to another.

To see this, we may, for a moment, assume a steady state evolution. As a consequence of the matter continuity equation, we then have $\nabla_a \tilde{u}^a = 0$ and therefore rewrite the energy equation as (setting to zero terms involving time derivatives with respect to \tilde{u}^a)

$$\nabla_a q^a = s^{ab} \nabla_a \tilde{u}_b . \tag{4.89}$$

In this relation, the term on the left-hand side should represent the energy sink (source) due to the (inverse) energy cascade—subtracting energy from the macro-scale into the micro one (or vice versa). In analogy with Newton's law of viscosity, Boussinesq suggested that one should relate the turbulent stress to the mean shear flow (see, e.g., [151]). In our case, this leads to

$$s^{ab} \propto \tilde{\sigma}^{ab}$$
, (4.90)

where the shear rate $\tilde{\sigma}^{ab}$ is defined in the usual way but in terms of the filtered four velocity, namely

$$\tilde{\sigma}_{ab} = \left[\tilde{\bot}^{c}_{(a}\tilde{\bot}^{d}_{b)} - \frac{1}{3}\tilde{\bot}^{cd}\tilde{\bot}_{ab}\right]\nabla_{c}\tilde{u}_{d}.$$
(4.91)

Motivated by this argument we move on to develop a closure scheme to complete the "fibration framework" we are proposing.

4.8 An explicit closure model

Our ultimate aim is to develop a consistent scheme for large-eddy simulations in relativity. Even though this involves numerical aspects which we will not touch upon here, we need to provide a strategy for closing the system of equations already at the fibration level. This is the problem we focus on now.

As we have seen, in order to carry out an evolution we need to provide some prescription for the residual terms, that is τ^{ab} , M in the filtering case or, equivalently, $\overline{\delta n \delta n}$, $\overline{\delta n \delta v^a}$, $\overline{\delta v^a \delta v^b}$ in the averaging one. In classical computational fluid dynamics, one of the earliest closures proposed—still widely used—is due to Smagorinsky [205]. This model effectively boils down to retaining only the s^{ab} term and modelling it as a traceless tensor proportional to the (resolved) shear-flow. Such a closure is motivated by arguments of the kind we provided in the previous section. However, this model may be too simplistic to capture all relevant features of a turbulent flow¹¹. In view of this, we aim to set up a scheme that can be used to describe turbulent flows for which the Smagorinsky model gives unsatisfactory results. Finally, it is important to note that the Smagorinsky model is typically implemented—both in recent relativistic numerical work as well as in the Newtonian context—in the Eulerian frame associated with a foliation. The simple fact that the translation between fibration and foliation leads to a "mixing" of the different terms in the stress-energy-momentum tensor, suggests that we need to consider a more general closure model.

In effect, we propose to model the residuals in terms of a general expansion in derivatives of the resolved variables, \tilde{n} , \tilde{u}^a and $\tilde{\epsilon}$. For practical reasons we halt the derivative expansion at first order, and decompose the gradients of the resolved quantities as

$$\nabla_a \tilde{n} = \tilde{\perp}_a^b \nabla_b \tilde{n} - \tilde{u}_a \dot{\tilde{n}} , \qquad (4.92a)$$

$$\nabla_a \tilde{\varepsilon} = \tilde{\perp}_a^b \nabla_b \tilde{\varepsilon} - \tilde{u}_a \dot{\tilde{\varepsilon}} , \qquad (4.92b)$$

$$\nabla_a \tilde{u}_b = -\tilde{a}_b \tilde{u}_a + \tilde{\omega}_{ab} + \tilde{\sigma}_{ab} + \frac{1}{3} \tilde{\theta} \tilde{\perp}_{ab} , \qquad (4.92c)$$

where, as before, $\dot{n} = \tilde{u}^a \nabla_a \tilde{n}$ (similarly for $\dot{\tilde{\epsilon}}$) and the filtered four velocity gradients $\nabla_a \tilde{u}_b$ are decomposed as usual. This closure scheme is analogous, although in a different spirit, to the most general constitutive relations discussed for dissipative hydrodynamics (at the linear level), see section 2.4.

¹¹We do not want to comment on the validity of the Boussinesq hypothesis here, so simply refer to [151] where it is discussed, albeit in a non-relativistic setting.

Because there is no formal difference in the modelling of the sub-filter scale terms between the averaging and filtering cases, let us set up the closure scheme for the filtering case. We also immediately consider the case of a two-parameter equation of state, as the barotropic limit can be easily recovered from the more general results. We then have to model the residuals q^a and s^{ab} . Recalling the definitions eqs. (4.79) and (4.80) we express these as

$$s^{ab} = -\eta \tilde{\sigma}^{ab} + (\pi_1 \tilde{\theta} + \pi_2 \dot{\tilde{n}} + \pi_3 \dot{\tilde{\varepsilon}}) \tilde{\bot}^{ab} , \qquad (4.93a)$$

$$q^{a} = \theta_{1}\tilde{a}^{a} + \theta_{2}\tilde{\perp}^{a}_{b}\nabla^{b}\tilde{n} + \theta_{3}\tilde{\perp}^{a}_{b}\nabla^{b}\tilde{\varepsilon}.$$
(4.93b)

In order to evolve the system, we also need to express $\langle p \rangle$ in terms of the resolved variables. To do so, we have to provide *M*. As this is a scalar, we model it as (see eq. (4.85))

$$M = \chi_1 \tilde{\theta} + \chi_2 \dot{\tilde{n}} + \chi_3 \dot{\tilde{\varepsilon}} , \qquad (4.94a)$$

and the filtered pressure then takes the form

$$\langle p \rangle = -\tilde{\varepsilon} + \tilde{T}\tilde{s} + \tilde{n}\mu + M.$$
 (4.95)

We have now introduced a total of 10 parameters to be used in the actual large-eddy model. These parameters—potentially validated/calibrated through high-resolution simulations—can be considered as functions of the resolved energy, density etcetera. Therefore, when we focus on a small region of the fluid they can be treated as simple constants.

4.8.1 Stability Analysis

Let us turn to the issue of linear stability, as this is a necessary condition for the system of equations to be (numerically) solved. Moreover, the fact that the "effective" theory we arrive at the resolved scale resembles that of a dissipative fluid further motivates this analysis. After all, it is well known that the standard/textbook relativistic viscous hydrodynamics equations are unstable (cf. chapter 2).

The linear stability of the effective theory obviously depends on the closure used, so let us focus on the specific relations proposed above. However, the aim is not to discuss the stability of the closure model in full generality, only to provide a "proof of principle" argument.

As the averaging/filtering residuals have been expressed in terms of gradients of the evolved variables and we are considering a local region, it makes sense to assume that the background configuration—the stability of which we want to study—is that of a

homogeneous fluid at rest. We will also consider, as usual, a flat background spacetime and ignore metric perturbations. This is justified—even in the general relativistic context—since the stability analysis is intended to be local, so that the effects of gravity can be transformed away (using a local inertial frame argument, in the spirit of the Fermi frame logic). Finally, we simplify the notation in order not to clutter up the equations. We drop the "tildes" used to identify the resolved variables, as we no longer need to make the distinction. Instead, we identify background quantities with a subscript (0). For instance, we write the background four-velocity as u_0^a and the chemical potential in the background configuration as μ_0 .

Let us start by expanding the perturbed fields (indicated by a δ) in Fourier modes

$$\delta u^a = \mathcal{B}^a e^{ik_d x^d} , \qquad (4.96a)$$

$$\delta n = \mathcal{A}e^{ik_d x^d} , \qquad (4.96b)$$

$$\delta \varepsilon = \mathcal{E} e^{ik_d x^a} . \tag{4.96c}$$

where \mathcal{B}^a is orthogonal to u_0^a because $u_0^a \delta u_a = 0$ as a result of the four-velocity normalization. We also decompose the wave-vector k^a as

$$k^a = \omega u_0^a + k\hat{k}^a , \qquad (4.97)$$

where ω is the frequency, k is the wavelength and \hat{k}^a is a unit four-vector orthogonal to u_0^a which describes each mode's direction. Because of the metric signature convention (+2), the system will be linearly stable (in time) if all solutions to the dispersion relation—written as $\omega = \omega(k)$ —have a negative (or vanishing) imaginary part. We will also use $\hat{k}^a \hat{k}^b$ and $^{12} \delta^{ab} - \hat{k}^a \hat{k}^b$ to decompose the momentum equation as well as $\mathcal{B}^a = (0, B_L, \mathcal{B}_{T1}, \mathcal{B}_{T2})^{\top}$ into its longitudinal and transverse part (with respect to wave direction).

In order to write the linearized equations in terms of the perturbed fields, we have to clarify how to perturb the pressure. As can be seen from eq. (4.95), its explicit expression depends on *M*. Let us first focus on the non-residual contribution and come back to *M* later:

$$\delta p = (\mathcal{CA} + \mathcal{DE})e^{ik_d x^d} + M , \qquad (4.98)$$

where we have defined

$$C = \left(\frac{\partial p}{\partial n}\right)_{\varepsilon}(n_0, \varepsilon_0) , \qquad (4.99a)$$

$$\mathcal{D} = \left(\frac{\partial p}{\partial \varepsilon}\right)_n (n_0, \varepsilon_0) , \qquad (4.99b)$$

to simplify the expressions that follow.

¹²Recall the definition of orthogonal projection with respect to a time-like vector. The sign difference stems from \hat{k}^a being a unit space-like vector.

Let us start by linearizing first the non-residual part of the equations of motion. The result is

$$-i\omega \mathcal{A} + ikn_0 \mathcal{B}_L = 0$$
, (4.100a)

$$-i\omega\mathcal{E} + ih_0k\mathcal{B}_L = 0 , \qquad (4.100b)$$

$$-ih_0\omega\mathcal{B}_L + ik(\mathcal{C}\mathcal{A} + \mathcal{D}\mathcal{E}) = 0, \qquad (4.100c)$$

$$-ih_0\omega\mathcal{B}_{T1}=0, \qquad (4.100d)$$

$$-ih_0\omega\mathcal{B}_{T2}=0, \qquad (4.100e)$$

where we have introduced the usual enthalpy density, $h_0 = p_0 + \varepsilon_0$. To work out the full linearized system of equations, let us consider each residual at a time. We start by looking at the trace-free part of s^{ab} , as this would correspond to the (fibration version of the) model proposed by Smagorinsky in the Newtonian context. A straightforward calculation leads to

$$\begin{split} \delta\sigma^{ab} &= \delta \left(\perp_0^{ac} \perp_0^{bd} \partial^{(c} u^{d)} - \frac{1}{3} (\partial_c u^c) \perp_0^{ab} \right) \\ &= ik \left(\hat{k}^{(a} \mathcal{B}^{b)} - \frac{1}{3} \mathcal{B}_L \perp_0^{ab} \right) e^{ik_d x^d} , \end{split}$$
(4.101)

where $\perp_0^{ab} = \eta^{ab} + u_0^a u_0^b$ is the projection orthogonal to the background velocity and η^{ab} is the Minkowski metric. As for the trace part of s^{ab} we have

$$\delta \left[\left(\pi_1 \partial_c u^c + \pi_2 u^c \partial_c n + \pi_3 u^c \partial_c \varepsilon \right) \perp_0^{ab} \right] =$$

= $\left(i \pi_1 k \mathcal{B}_L - i \pi_2 \omega \mathcal{A} - i \pi_3 \omega \mathcal{E} \right) \perp_0^{ab}$, (4.102)

and it is easy to see that these additional terms only affect the longitudinal projection of the momentum equation. Next we have the heat-flux q^a . It is fairly easy to see that only two (out of five) terms will contribute to the linearized equations. These terms lead to

$$\delta(\partial_a q^a) = \theta_1 \omega k \mathcal{B}_L - \theta_2 k^2 \mathcal{A} - \theta_3 k^2 \mathcal{E} , \qquad (4.103)$$

which enters the energy equation, while

$$\delta(\perp_c^b u^a \partial_a q^c) = -\theta_1 \omega^2 B^b + \theta_2 \omega k \hat{k}^b \mathcal{A} + \theta_3 \omega k \hat{k}^b \mathcal{E} , \qquad (4.104)$$

affects the momentum equation. We note that the last two terms in the expression above affect only the longitudinal projection, while the first term modifies both the longitudinal and transverse components. Last but not least, we consider the residuals that arise from the Gibbs relation:

$$M = \chi_1 \theta + \chi_2 \dot{n} + \chi_3 \dot{\varepsilon} . \tag{4.105}$$

It is easy to see that this residual will not affect the (linearized) energy equation, while it contributes to the longitudinal momentum equation as

$$\delta(\perp^{ab} \partial_a M) = \left[-\chi_1 k^2 \mathcal{B}_L + \chi_2 k \omega \mathcal{A} + \chi_3 k \omega \mathcal{E}\right] \hat{k}^b .$$
(4.106)

Collecting everything together, we can write the linearized equations¹³ as

$$\begin{pmatrix} \boldsymbol{L} & \boldsymbol{0} \\ \boldsymbol{0} & \boldsymbol{T} \end{pmatrix} \cdot \begin{pmatrix} \mathcal{A} & \mathcal{E} & \mathcal{B}_{L} & \mathcal{B}_{T1} & \mathcal{B}_{T2} \end{pmatrix}^{\top} = 0, \qquad (4.107)$$

where

$$\boldsymbol{L} = \begin{pmatrix} -i\omega & 0 & in_0k \\ -\theta_2k^2 & -i\omega - \theta_3k^2 & ih_0k + \theta_1k\omega \\ ik\mathcal{C} + (\zeta_2 + \theta_2)k\omega & i\mathcal{D}k + (\zeta_3 + \theta_3)k\omega & -(ih_0\omega - \frac{2}{3}\eta k^2 + \zeta_1k^2 + \theta_1\omega^2) \end{pmatrix},$$
(4.108)

and

$$\mathbf{T} = \begin{pmatrix} -(ih_0\omega - \frac{\eta}{2}k^2 + \theta_1\omega^2) & 0\\ 0 & -(ih_0\omega - \frac{\eta}{2}k^2 + \theta_1\omega^2) \end{pmatrix},$$
(4.109)

and we have introduced $\zeta_i = \chi_i + \pi_i$ with i = 1, 2, 3. A similar analysis is carried out, for instance, in [133]. However, the different gradient expansions (cf. eq. (2.77) and eqs. (4.93) and (4.94)) lead to slightly different results.

The stability analysis for the general case is perhaps best considered numerically. We also stress (again) that, because the entropy \tilde{s} does not represent the true one, we are allowed to violate the second law of thermodynamics. This will give us more freedom (with respect to faithful dissipative fluids) to control the stability of the closure.

4.8.2 Smagorinsky model

As a first step, let us consider the simple case where the only non-vanishing parameter in eqs. (4.93) and (4.94) is η . This would correspond to the (fibration version of the) model proposed by Smagorinsky in the Newtonian context. Starting from eq. (4.107) we easily obtain the linearized equations for this case

$$\begin{pmatrix} -i\omega & 0 & in_0k & 0 & 0 \\ 0 & -i\omega & ih_0k & 0 & 0 \\ ik\mathcal{C} & i\mathcal{D}k & -ih_0\omega + \frac{2}{3}\eta k^2 & 0 & 0 \\ 0 & 0 & 0 & -ih_0\omega + \frac{\eta}{2}k^2 & 0 \\ 0 & 0 & 0 & 0 & -ih_0\omega + \frac{\eta}{2}k^2 \end{pmatrix} \cdot \begin{pmatrix} \mathcal{A} \\ \mathcal{E} \\ \mathcal{B}_L \\ \mathcal{B}_{T1} \\ \mathcal{B}_{T2} \end{pmatrix} = 0. \quad (4.110)$$

¹³Note that the coefficient matrix depends only on $\omega = -u^a k_a$ and $k^2 = k^a k_a + \omega^2$, in accordance with Lorentz invariance.

The required dispersion relations are obtained by setting to zero the determinant of the coefficient matrix. Working this out, we find that the transverse modes decouple, and the corresponding dispersion relation is

$$\omega = -i\frac{\eta}{2h_0}k^2 \,. \tag{4.111}$$

The other non-trivial modes are longitudinal, with dispersion relation

$$h_0\omega^2 + i\frac{2}{3}\eta k^2\omega - (h_0\mathcal{D} + \mathcal{C}n_0)k^2 = 0.$$
(4.112)

Solving this equation we obtain

$$\omega = \pm c_s k - i \frac{\eta}{3h_0} k^2 + \mathcal{O}(k^3) , \qquad (4.113)$$

where

$$c_s^2 = (\mathcal{D} + Cn_0/h_0)$$
, (4.114)

is the usual sound speed. The longitudinal modes represent sound waves, while the transverse modes are not propagating. Both sets of modes are stable for $\eta > 0$. As a simple consistency check, it is easy to see that in the ideal limit, when $\eta = 0$, one obtains a single non-trivial solution representing an undamped sound wave.

The result demonstrates that the simple Smagorinsky model is stable according to the fibration observer. This is a key conclusion for cosmological applications, as these tend to involve a cosmological time associated with a co-moving observer; in essence, a fibration. The situation is different for numerical relativity simulations, which tend to be based on a spacetime foliation. The matter description (formally) involves a fibration associated with fluid element worldlines, but the evolution is carried out in a different frame. Our stability demonstration does not (yet) cover this case. In order to complete the argument, we have to consider the stability issue in a different frame. We therefore introduce (as usual) the Eulerian observer N^a as

$$u^{a} = W(N^{a} + v^{a})$$
, with $W = (1 - v^{2})^{-1/2}$, (4.115)

and note that the two frames are related by a Lorentz boost. This turns out to cause trouble. The simple Smagorinsky model, while stable in the fibration frame, becomes unstable in the boosted frame.

In order to demonstrate this result, we start by noting that (using primes to indicate boosted quantities)

$$\partial_a^\prime \varepsilon^\prime = \Lambda_a^b \partial_b \varepsilon = 0$$
, (4.116a)

$$\partial_a' n' = \Lambda_a^b \partial_b n = 0 , \qquad (4.116b)$$

$$\partial'_a u'_b = \Lambda^c_a \Lambda^d_b \partial_c u_d = 0 , \qquad (4.116c)$$

where Λ is the Lorentz boost matrix. As we are linearizing with respect to a homogeneous (in spacetime) background this confirms that the gradient-based closure scheme we are proposing still makes sense. We have to work out the dispersion relations in a non-comoving frame, but because these are expressed in terms of $\omega = -k^a u_a$ and $k^2 = k^a k_a + \omega^2$, we just have to boost these quantities. We can then take (without loss of generality) v^a to be in the *x*-direction, while \hat{k} lies in the x - y plane. Then we introduce the angle ϕ between the wave-vector and v^a as $\hat{k}^a v_a = v \cos \phi$ and write the Lorentz boost as:

$$\omega = W(\omega' - vk'\cos\phi), \qquad (4.117a)$$

$$k_x = W(k'\cos\phi - v\omega'), \qquad (4.117b)$$

$$k_y = k' \sin \phi = k'_y , \qquad (4.117c)$$

$$k_z = 0 = k'_z \,. \tag{4.117d}$$

Applying this to the transverse dispersion relation in eq. (4.111) we obtain (dropping the primes for clarity)

$$(\eta W^2 v^2) \omega^2 - 2 (ih_0 W - \eta W^2 v k \cos \phi) \omega - - (\eta W^2 k^2 \cos^2 \phi + \eta k^2 \sin^2 \phi - 2ih_0 \cos \phi W v k) = 0.$$
(4.118)

We note that eq. (4.111) was a first order polynomial, while the boost made it second order, thus generating an additional solution. For long wavelengths, the two solutions are

$$\omega = vk\cos\phi - i\frac{\eta}{2h_0W^3} \left(\cos^2\phi + W^{-2}\sin^2\phi\right)k^2 + \mathcal{O}(k^3)$$
(4.119a)

$$\omega = i \frac{2h_0}{\eta W v^2} + \mathcal{O}(k) . \tag{4.119b}$$

The first solution is the boosted version of the mode we obtained in the fibration. It is stable for $\eta > 0$ (as in the comoving frame), propagating with phase velocity $v \cos \phi$ and the decay rate reduces to the original value as $v \to 0$, $W \to 1$. There is, however, an additional solution which is non-vanishing for k = 0 (in [133, 117] these are referred to as "gapped" modes). This second mode is evidently unstable for $\eta > 0$.

This result demonstrates that the simple Smagorinsky model is unstable when "observed" from a non-comoving frame. This is a well-known problem of Eckart-Landau models for dissipative fluids (cf. chapter 2). As we are not dealing with a dissipative model that describes linear deviation from a thermodynamical equilibrium state, this is not intrinsically problematic. For real dissipative systems, stability of equilibrium is not only required for numerical implementations, but also guided by intrinsic consistency. A system slightly out of equilibrium must evolve, "by definition," toward thermodynamical equilibrium, no matter if the fluid in equilibrium is at rest or not. Our case is different. However, because we are setting up the filtering scheme in the fibration—in order to retain consistency with the covariance of General Relativity, as discussed in section 4.4.1—while the simulations will be carried out in the foliation, we have to ensure that the model is "covariantly" stable. The LES model of [176] gets away with a simple Smagorinsky closure because it is directly implemented in the foliation frame, where the simulation is then performed.

4.8.3 Fixing the Smagorinsky instability

The aim now is to show how we can fix the instability problem (in the boosted frame) by introducing more parameters in the closure. Focusing first on the transverse modes, we see from eq. (4.109) that the only way to fix the problem is by considering a non-zero θ_1 . The co-moving transverse dispersion relation then becomes

$$2ih_0\omega - \eta k^2 + \theta_1 \omega^2 = 0 , \qquad (4.120)$$

with solutions for long wavelengths (i.e. small *k*):

$$\omega_{+} = -i\frac{\eta}{2h_{0}}k^{2} + \mathcal{O}(k^{4}) , \qquad (4.121a)$$

$$\omega_{-} = -2i\frac{h_0}{\theta_1} + \mathcal{O}(k^2)$$
 (4.121b)

Because the dispersion relation is now quadratic we obtain two solutions: the "ungapped" mode from the Smagorinsky model, and an additional gapped mode that appears already in the co-moving frame. The (long wavelength) stability in the unboosted frame is guaranteed by taking $\eta > 0$ (as before) alongside $\theta_1 > 0$. We can further check the stability in the co-moving frame at all wavelengths by means of the Routh-Hurwitz criterion (cf. appendix E and [131]). In order to do so we introduce $\Delta = -i\omega$ (to deal with real algebraic equations) and rewrite the dispersion relation as

$$\theta_1 \Delta^2 + 2h_0 \Delta + \eta k^2 = 0 . (4.122)$$

Stability requires the solutions to have negative real part Re $\Delta < 0$. The Routh-Hurwitz criterion then guarantees the stability (at all wavelengths) as long as $\theta_1 > 0$ and $\eta > 0$. These conditions are identical to the ones obtained at long wavelengths.

In order to check the stability in the boosted frame, we boost the transverse modes dispersion relation (as before) to get

$$(\theta_1 - \eta v^2) W^2 \omega^2 + 2 (ih_0 W + (\eta + \theta_1) W^2 v k \cos \phi) \omega - - (\eta W^2 k^2 \cos^2 \phi + \eta k^2 \sin^2 \phi + 2ih_0 W v k - \theta_1 W^2 v^2 k^2 \cos^2 \phi) = 0.$$
 (4.123)

To work out the long wavelength stability conditions, we may solve this equation perturbatively. This means that we introduce

$$\omega = \omega_0 + \omega_1 k + \omega_2 k^2 + \omega_3 k^3 , \qquad (4.124a)$$

$$\omega^{2} = \omega_{0}^{2} + 2\omega_{0}\omega_{1}k + (\omega_{1}^{2} + 2\omega_{0}\omega_{2})k^{2} + (2\omega_{1}\omega_{2} + 2\omega_{0}\omega_{3})k^{3}, \qquad (4.124b)$$

$$\omega^{3} = \omega_{0}^{3} + 3\omega_{0}^{2}\omega_{1}k + (3\omega_{0}\omega_{1}^{2} + 3\omega_{0}^{2}\omega_{2})k^{2} + (\omega_{1}^{3} + 6\omega_{0}\omega_{1}\omega_{2} + 3\omega_{0}^{2}\omega_{3})k^{3}, \quad (4.124c)$$

and solve order by order. Solving eq. (4.123) to lowest order we find two solutions: the first is the un-gapped mode ($\omega_0 = 0$) while the second is given by

$$\omega_0 = -i\frac{2h_0W^{-1}}{\theta_1 - \eta v^2}.$$
(4.125)

We may focus on the un-gapped mode as we already have the imaginary part (to lowest order) for the gapped mode. Working to first order we obtain the phase velocity, while at second order we get the damping rate. Collecting the results, the small *k* solutions to the boosted dispersion relation can be written

$$\omega = -i\frac{2h_0W^{-1}}{\theta_1 - \eta v^2} + \mathcal{O}(k) , \qquad (4.126a)$$

$$\omega = vk\cos\phi + i\frac{\eta}{2h_0W^3}\left(\cos^2\phi + W^{-2}\sin^2\phi\right) + \mathcal{O}(k^3).$$
 (4.126b)

We see that stability in the boosted frame requires $\eta > 0$ and $\theta_1 > \eta v^2$. To the best of our knowledge, stability at all wavelengths cannot be studied analytically in the boosted case. This is because the Routh-Hurwitz criterion applies to real polynomials only. The stability is then perhaps best studied numerically, once a specific equation of state model has been chosen. However, our demonstration shows that the LES model passes the key stability tests.

Next, we turn our attention to the longitudinal modes, assuming again that the only non-vanishing parameters in eq. (4.108) are η and θ_1 . The (comoving) longitudinal modes dispersion relation is then

$$\omega \left[\theta_1 \omega^3 + ih_0 \omega^2 - \left(\frac{2\eta}{3} + \theta_1 \mathcal{D} \right) k^2 \omega - ih_0 c_s^2 k^2 \right] = 0.$$
(4.127)

As before, we can work out the non-trivial longitudinal modes using eq. (4.124). Again, working to lowest order we find that one mode is "gapped" and two are not. These long wavelength modes are

$$\omega = -i\frac{h_0}{\theta_1} + \mathcal{O}(k^2) , \qquad (4.128a)$$

$$\omega = \pm c_s k - \frac{i}{3h_0} \left(\eta - \frac{3}{2} (c_s^2 - D) \theta_1 \right) k^2 + \mathcal{O}(k^3) .$$
 (4.128b)

In order to make sure the longitudinal modes are also stable we have to take $\theta_1 > 0$, $\eta > 0$ and

$$\frac{3}{2}(c_s^2 - \mathcal{D}) < \frac{\eta}{\theta_1} < \frac{1}{v^2} .$$
(4.129)

We can then study the stability condition at all wavelengths using the Routh-Hurwitz criterion. To do so, we again introduce $\Delta = -i\omega$ in eq. (4.127) to make it a real algebraic equation:

$$\theta_1 \Delta^3 + h_0 \Delta^2 + A \Delta k^2 + h_0 c_s^2 k^2 = 0 , \qquad (4.130)$$

where $A = 2/3\eta + \theta_1 D$. From this it is easy to see that the Routh-Hurwitz criterion guarantees stability for eq. (4.129). Again, the general *k* case does not change the stability requirements.

As in the case of transverse modes, the story does not end here. We still have to establish the stability in the boosted frame. To do so, we "boost" eq. (4.127) using eq. (4.117) to obtain

$$a\,\omega^3 + b\,\omega^2 + c\,\omega + d = 0$$
, (4.131)

where

$$a = W^3 \left(\theta_1 - A v^2 \right) ,$$
 (4.132a)

$$b = -W^{2} \left[(3\theta_{1} - 2A)Wvk\cos\phi - ih_{0} - WAv^{3}k\cos\phi + ih_{0}c_{s}^{2}v^{2} \right] , \qquad (4.132b)$$

$$c = W \left[(3\theta_1 - 2A) W^2 v^2 k^2 \cos^2 \phi - Ak^2 \left(1 + W^2 v^2 \cos^2 \phi \right)$$
(4.132c)

2*ik* Werk(1 - z^2) cos z

$$-2ih_0 Wvk(1-c_s^2)\cos\phi],$$

$$d = -\left[\theta_1 v^3 W^3 \cos^3\phi - WAvk^3 \cos^3\phi - W^3 v^3 Ak^3 \cos^3\phi - ih_0 k^2 \left(W^2(v^2 - c_s^2)\cos^2\phi - c_s^2(1 - \cos^2\phi)\right)\right].$$
(4.132d)

Again we solve the problem using the expansion in eq. (4.124). At lowest order we find two "un-gapped" modes and one "gapped" solution. The latter is given by

$$\omega = -i\frac{h_0(1 - c_s^2 v^2)}{(\theta_1 - Av^2)W} + \mathcal{O}(k) , \qquad (4.133)$$

which is stable for

$$\eta < \frac{3}{2}\theta_1 \frac{1 - \mathcal{D}v^2}{v^2} \,. \tag{4.134}$$

As in the case of the "un-gapped" modes, working to $O(k^2)$ (as the O(k) problem is trivial) we obtain the boosted sound speed $\omega_1 = C_s$. This is found by solving

$$W^{2}(1-v^{2}c_{s}^{2})\mathcal{C}_{s}^{2}-2W^{2}v\cos\phi(1-c_{s}^{2})\mathcal{C}_{s}+W^{2}(v^{2}-c_{s}^{2})\cos^{2}\phi-c_{s}^{2}\sin^{2}\phi=0.$$
 (4.135)



FIGURE 4.2: Left: shaded region resulting from combining the stability constraints obtained for the i) transverse modes (gapped, un-gapped, boosted, un-boosted), ii) longitudinal un-boosted modes (gapped and un-gapped), and iii) longitudinal boosted gapped modes. Right: sound speeds (solid) and damping rates (dashed) for the ungapped longitudinal modes in the boosted frame. Colours match sound speeds with the corresponding damping rates. For illustrative purposes we used: $c_s = 0.16$, v =0.4, $\eta = 1.2$, $\theta_1 = 0.5$ and a barotropic equation of state (both figures).

To see that this result actually makes sense, we provide the solution for the two cases where v^a is parallel/orthogonal to \hat{k}^a (respectively)

$$\mathcal{C}_s = \frac{v \pm c_s}{1 \pm v \, c_s} \,, \tag{4.136a}$$

$$C_s = \pm \frac{\sqrt{1 - v^2}}{\sqrt{1 - v^2 c_s^2}} c_s .$$
 (4.136b)

The solution for different values of ϕ is best understood by considering specific examples, see fig. 4.2, noting that it only depends on the thermodynamic speed of sound c_s and the relative velocity v.

In order to work out the longitudinal mode damping we work at $O(k^3)$. This leads to a purely imaginary $\omega_2 = \Gamma(\phi)$. The solution involves the boosted sound speed C_s , and it is not particularly illuminating, so it is also best understood by specific examples, see fig. 4.2.

As we see from the illustrations in fig. 4.2 and fig. 4.2 there are regions of the (η, θ) parameter space where all modes are stable. As we are only aiming at a proof of principle (not a comprehensive stability analysis) this concludes the argument. The stability conditions depend on the equation of state (which enters through c_s , \mathcal{D} , \mathcal{C}) and the relative velocity v. Therefore, an exhaustive study of the stability in the general case is best done once a specific equation of state has been chosen, following the logic outlined here.

4.9 Summary

Averaging and filtering are the standard strategies for dealing with the problem of simulating (computationally demanding) turbulent flows. Both approaches are complicated by the covariance of General Relativity, where the split between space and time is an observer-dependent notion. As the problem is beginning to be considered from the point of view of numerical relativity (relevant examples are [94, 176, 177, 4, 220, 72]), it is important to understand the underpinning theory. Hence, we decided to start from the beginning and considered how the different strategies should be implemented in the curved spacetime setting relevant for, say, binary neutron-star mergers.

After clarifying the sense in which consistency with the principles of General Relativity poses interesting foundational questions, we argued that it is natural to set up the analysis in the "fibration" associated with individual fluid elements. This then allowed us to introduce a meaningful local analysis via the use of Fermi coordinates ([78, 79]), which defines the covariant averaging/filtering procedure. Building on this, we worked out the coarse-grained fluid dynamics, and considered the impact averaging/filtering has on the (thermodynamical) interpretation of the resolved variables. Finally, because smoothing the fluid dynamics inevitably introduces a closure issue, we proposed a closure scheme and discussed its linear stability. This completed the formal development of the fibration-based model. In order for this work to have practical relevance, however, we need to make contact with actual simulations. Whilst this goes beyond the scope of this thesis, we will go back to (briefly) comment on this at the end of chapter 5.

Chapter 5

Filtering relativistic magneto-hydrodynamics

In this chapter we take the first steps towards extending the covariant filtering scheme discussed in chapter 4 to charged multi-fluids. The relevance of this effort is clear in the context of neutron star astrophysics, as the discussion in the previous chapter left out aspects that are important for (if not crucial to) neutron stars modelling—from electromagnetism to superfluidity/superconductivity. Keeping in mind the bigger picture and final aim, it is natural to start by focusing on magneto-hydrodynamics (MHD), which can be derived as the single-fluid limit of an underlying two-fluid plasma model.

We will start in section 5.1 with a brief introduction to magneto-hydrodynamics turbulence, mainly focusing on the differences with hydrodynamics. The discussion is based on the recent reviews by Schekochihin [192] and Beresnyak [44], which we refer to for more details (and references to the relevant original papers). We then introduce the relativistic magneto-hydrodynamics equations in section 5.2. As anticipated, these can be derived starting from a two-fluid plasma model, but we here take a shortcut, namely we start from the Euler-Maxwell system instead. We conclude this chapter (and this part) by discussing in section 5.3 the first steps towards a covariant filtering scheme for magneto-hydrodynamics based, as the approach discussed in chapter 4, on the fibration associated with individual fluid elements.

5.1 A brief introduction to magneto-hydrodynamic turbulence

In a similar fashion as for hydrodynamic turbulence—discussed in section 4.1—useful quantities in the characterization of MHD turbulence are the magnetic Reynolds number Re_m and the Lundquist number *S*. These are defined as

$$\operatorname{Re}_{m} = \frac{\rho V L}{\beta}, \quad S = \frac{LB}{\sqrt{\mu_{0}\rho}\beta} = \frac{Lv_{A}}{\beta},$$
(5.1)

where β is the magnetic diffusivity (or resistivity, proportional to the inverse of the conductivity) and v_A is the Alfvén velocity, whilst ρ , V, L are characteristic values for the density, velocity and lengthscale of the flow (as in section 4.1). The magnetic Reynolds number measures the importance of fluid convection over resistive diffusion, while the Lundquist number has a similar interpretation, but compares the Alfvén wave crossing time to the timescale of resistive diffusion. When these numbers¹ (together with the Reynolds number defined in eq. (4.1)) are very large, we expect the MHD flow to be turbulent.

MHD turbulence, however, is expected to be substantially different from hydrodynamics. This is intuitive, as we can eliminate a mean flow by a proper choice of frame, while a mean magnetic field cannot be removed by such a transformation. In essence, we may say that the magnetic field is the only large-scale feature that does not "go away" at small scales. This simple fact is what makes MHD turbulence *a priori* different from hydrodynamics: the large-scale mean magnetic field makes the system anisotropic. This simple fact means we should expect MHD turbulence to be "more complicated", as it would not be possible to, for example, justify scaling laws of the form in eq. (4.2) which assume (statistical) homogeneity and isotropy.

Dynamically, we expect transport along the magnetic field lines (on a scale $l_{//}$) to be associated with Alfvén wave, while transport across the field lines (on a scale λ) to be associated with non-linear interactions. This leads us to a key feature of MHD turbulence, the so-called *critical balance*. Originally conjectured by Goldreich and Sridhar [96, 97], critical balance boils down to postulating a balance between parallel and perpendicular transport in the inertial range. To see where this brings us, we introduce the Alfvén time and the non-linear time

$$\tau_A = \frac{l_{\prime\prime}}{v_A}, \qquad \tau_{nl} = \frac{\lambda}{\delta u_\lambda},$$
(5.2)

where δu_{λ} is the typical velocity increment between points separated by λ , and assume the two timescales are of the same order $\tau_A \sim \tau_{nl} \sim \tau_c$ —where τ_c is the cascade time, that is the typical time it takes to transfer energy from one (perpendicular) scale to

¹One can also define the magnetic Prandtl number $Pm = Re_m/Re = \eta/\beta$, quantifying the importance of fluid viscosity over magnetic diffusivity.

the next. Considering perpendicular transport first, and noting that on dimensional grounds we expect the energy spectrum $E \sim \delta u_{\lambda}^2 \lambda$, we have on the one hand,

$$\varepsilon \sim \frac{\delta u_{\lambda}^2}{\tau_c} \sim \frac{\delta u_{\lambda}^2}{\tau_{nl}} \Longrightarrow \delta u_{\lambda} \sim (\varepsilon \lambda)^{1/3} \Longrightarrow E(k_{\perp}) \sim \varepsilon^{2/3} k_{\perp}^{-5/3} .$$
 (5.3)

that is, the Kolmogorov spectrum. If we focus instead on transport along the magnetic field, a similar logic leads to

$$\varepsilon \sim \frac{\delta u_{l_{//}}^2}{\tau_c} \sim \frac{\delta u_{l_{//}}^2}{\tau_A} \Longrightarrow \delta u_{l_{//}} \sim (\varepsilon \tau_A)^{1/2} \Longrightarrow E(k_{//}) \sim \frac{\varepsilon}{v_A} k_{//}^{-2} .$$
(5.4)

In essence, we find that critical balance leads to an anisotropic spectrum, with a Kolmogorov type spectrum in the perpendicular directions.

Critical balance can be justified as follows. On the one hand we have the observation (based on theoretical arguments and numerical evidence) that weak MHD turbulence—where perturbation amplitudes are small enough that non-linear interactions are negligible ($\tau_{nl} \gg \tau_A$) and the perpendicular spectrum presents the same scaling as the parallel one—leads naturally to the strong turbulence regime where $\tau_{nl} \sim \tau_A$. On the other hand, the opposite regime with $\tau_A \gg \tau_{nl}$ is unsustainable as information in MHD propagates predominantly along the magnetic field lines with velocity v_A (the Alfvén waves) and hence no structure with $l_{l/l}$ larger than $v_A \tau_{nl}$ can be kept coherent and will break up. In fact, critical balance appears to be a robust property of MHD turbulence.

Whilst the parallel spectrum, with scaling k_{ll}^{-2} , appears to be robust, there is currently no definitive consensus in the community with regards to the perpendicular spectrum. Initially, in fact, solar wind observations favoured the -5/3 Kolmogorov scaling, but then sets of (direct) numerical simulations started to suggest a different scaling $E(k_{\perp}) \sim k_{\perp}^{-3/2}$ —although issues with such results have been raised and some authors claim to see a better convergence with the Kolmogorov scaling (e.g. Beresnyak [43]). This leads us to another important feature of MHD turbulence, dynamic alignment, according to which the velocity and magnetic field tend to shear each other into alignment (in the plane perpendicular to the mangetic field) and hence form sheet-like structures as we approach smaller scales. Phenomenological models based on dynamic alignment (and critical balance) can, in fact, reproduce both the parallel spectrum we found above and the -3/2 scaling in the perpendicular spectrum. The key point, however, is that such sheet-like configurations are not sustainable asymptotically at ever smaller scales due to different known instabilities-including, for example, a magnetized version of the Kelvin-Helmoltz instability (cf. chapter 7)-and other non-ideal processes like magnetic reconnection, thus eventually leading to the break up of these sheets into islands. When enough of these islands form the flow becomes isotropic, and the cascade starts up again in Kolmogorov form. This isotropic-to-sheet-to-islands transition



FIGURE 5.1: Cartoon of the perpendicular MHD turbulence spectrum. At scales larger than critical balance (i.e. $k < k_{CB}$) is shown the same scaling as in weak turbulence (i.e. $\propto k_{\perp}^{-2}$). At scales smaller than critical balance is shown the aligned cascade (i.e. $\propto k_{\perp}^{-3/2}$) periodically interrupted at k_1, k_2 and so on, while k_D represents the scale at which dissipative/resistive effects begin to prevail over inertial ones. Figure adapted from Schekochihin [192].

is expected to repeat, giving a periodically interrupted, or intermittent, turbulent cascade (cf. fig. 5.1).

In essence, it appears that a complete theory of MHD turbulence should contain (to some degree) a theory of reconnection, thus making the story even more complicated². As for what matters for the present discussion, it is fair to say that sub-grid models of MHD turbulence constitute a very challenging problem because of the local anisotropy and complicated dissipative processes like reconnection [193].

²Not to mention the fact that we have here discussed the so-called balanced MHD turbulence regime where the averaged/total cross helicity $\mathbf{v} \cdot \mathbf{B}$, where \mathbf{v} is the velocity and \mathbf{B} is the magnetic field, vanishes and did not touch upon the link between turbulence and dynamo processes leading to the amplification of the magnetic field [47, 187].

5.2 Magneto-hydrodynamics in the fibration: a shortcut

We argued that, from a theory perspective, it would be natural to start the discussion at the level of a charged multi-fluid model, and derive the magneto-hydrodynamics equations as the single-fluid limit of a two-fluid plasma model. The underlying multi-fluid nature is, in fact, also important for the development of consistent models beyond ideal. This motivates several discussions in the literature, both connecting to the variational (dissipative) multi-fluid framework discussed in section 2.3 (see [16, 23, 24]) as well as linking to "3+1 formulations" geared towards numerical implementations (see [25, 26]). Our aim here is, however, less ambitious. We aim to provide a streamlined derivation of the magneto-hydrodynamics equations to set the stage for discussing extensions of the covariant filtering scheme developed in chapter 4. With this aim in mind, we will start from the Euler-Maxwell system as a proxy for magneto-hydrodynamics. We will mainly focus on the electromagnetic degrees of freedom, while the fluid equations are obtained from the stress-energy-momentum conservation law (as before).

The total stress-energy-momentum tensor is written as the sum of a perfect fluid part

$$T_{p.f.}^{ab} = \varepsilon u^{a} u^{b} + p(g^{ab} + u^{a} u^{b}) , \qquad (5.5)$$

where u^a is the fluid four-velocity, p is the pressure and ε is the energy density (not to be confused with the energy flux from scale to scale in the previous section), augmented with the Maxwell stress-energy-momentum tensor

$$T_{EM}^{ab} = \frac{1}{\mu_0} \left[F^{ad} F^b_{\ d} - \frac{1}{4} g^{ab} F^{cd} F_{cd} \right] , \qquad (5.6)$$

where F^{ab} is the Faraday tensor. The fluid equations are then obtained from³

$$\nabla_a T^{ab}_{\text{p.f.}} = -\nabla_a T^{ab}_{EM} = -j_a F^{ab} , \qquad (5.7)$$

where the term on the right-hand side is the Lorentz four-force and j^a is the charge fourcurrent. We consider the pressure to be a function of two thermodynamic variables recall the discussion in section 4.7 to see why barotropic models are of lesser interest for the present discussion—which we conveniently take as the energy ε and the baryon density *n*. As such, we also need an equation for *n*, given by the continuity equation

$$u^a \nabla_a n + n \nabla_a u^a = 0. \tag{5.8}$$

Before we move on to focus on the electromagnetic degrees of freedom, let us stress that there is a formal inconsistency in the fluid equations we wrote down. Deriving the equations of magneto-hydrodynamic as the single fluid limit of a two-fluid plasma

³The last step here involves using the Maxwell equations, eq. (5.9).

model, one can show that a perfect fluid form for the matter stress-energy-momentum tensor holds in the "centre of momentum" frame—the analogue of Landau frame in this context—but this would lead to a continuity equation for the baryon current with drift terms⁴. As discussed in detail in [26], working with eqs. (5.7) and (5.8) involves additional steps such as assuming that the electron mass is much smaller than the baryon mass. This assumption may or may not apply depending on the physical system under consideration. It is certainly well motivated for non-relativistic systems, but less obvious for a neutron star core model where the electron effective mass may be up to ten per cent of baryon rest mass [66]. Let us further note that an evolution equation (or a constraint) for the charge four-current is missing from the fluid equations obtained—although the charge current is intuitively associated with a drift velocity between the two species constituting the plasma. Simply noting these reservations here, we will work with eqs. (5.7) and (5.8) as the fluid equations, and move on to focus on the electromagnetic degrees of freedom.

Assuming we want to work with the electric and magnetic fields, and because the filtering scheme of chapter 4 is tied to the fluid, it is natural to write down the Maxwell equations in the "fluid frame" [21]—leading to a (fibration) formulation commonly used in cosmology [35]. We start from Maxwell equations in covariant form⁵

$$\nabla_a F^{ba} = \mu_0 j^b , \qquad \nabla_{[a} F_{bc]} = 0 , \qquad (5.9)$$

and introduce a four-velocity U^a associated with a generic observer. We decompose the Faraday tensor and the charge current as

$$F_{ab} = 2U_{[a}e_{b]} + \varepsilon_{abc}b^c , \qquad j^a = \sigma U^a + J^a .$$
(5.10)

where the electric and magnetic field (as measured by U^a) are defined as

$$e_a = F_{ab}U^b$$
, $b_a = \frac{1}{2}\varepsilon_{abc}F^{bc}$, and $\varepsilon_{abc} = \varepsilon_{dabc}U^d$. (5.11)

⁴To get rid of drift terms we would have to work in the (analogue of) Eckart frame [16, 24], but this in turn would lead to momentum flux terms in the fluid stress-energy-momentum tensor.

⁵Here we are working using a "mixture" of SI units and geometric units. In the SI system, the electric constant (or vacuum permittivity) is given by $\varepsilon_0 = 1/\mu_0 c^2$ —where *c* is the speed of light and μ_0 is the magnetic constant (or vacuum permeability)—and the electric and magnetic field have different units. This would appear to be in contrast with eq. (5.10) as e^a and b^a seem to have the same dimension. However, this is due to the fact that we are also using (at the same time) geometric units, where the speed of light c = 1 and suppressed, so that $\varepsilon_0 = 1/\mu_0$. This detail is however irrelevant for the present discussion.

With these definitions we can work out the parallel and orthogonal projections—with respect to U^a —of eq. (5.9) and rewrite the Maxwell equations as [74, 16, 21]

$$\perp^a_b \nabla_a e^b - \mu_0 \sigma = 2W^a b_a , \qquad (5.12a)$$

$$\perp^a_b \nabla_a b^b = -2W^a e_a , \qquad (5.12b)$$

$$\perp_{ab} \dot{e}^b - \varepsilon_{abc} \nabla^b b^c + \mu_0 J_a = e^b \left(\sigma_{ba} + \omega_{ba} - \frac{2}{3} \theta \perp_{ba} \right) + \varepsilon_{abc} a^b b^c , \qquad (5.12c)$$

$$\perp_{ab} \dot{b}^b + \varepsilon_{abc} \nabla^b e^c = b^b \left(\sigma_{ba} + \omega_{ba} - \frac{2}{3} \theta \perp_{ba} \right) - \varepsilon_{abc} a^b e^c , \qquad (5.12d)$$

where dots-which stand for co-moving time derivatives-and the projection operator refer to the observer⁶ U^a . In eq. (5.12), the terms on the left-hand side should be familiar, while those on the right-hand side are associated with gradients of the observer four-velocity. As such, they vanish identically for an inertial observer, and hence do not appear in most textbook discussions. We also note that the system of eqs. (5.7), (5.8) and (5.12) is not closed. We need an additional equation linking the charge four-current to the other quantities. This can be derived starting from a two-fluid model [16, 24, 26]. For ideal models, however, this additional equation is often given by a phenomenological Ohm's law based on the standard argument that in a perfect conductor—where charges easily flow—one would expect the electric field to "short out" as the matter becomes locally charge neutral [38]. Although this is sufficient for getting to a workable (i.e. closed) set of equations [93, 65], we here want to tread a bit more carefully. We do so for two reasons: First, our discussion will allow us to better appreciate the assumptions involved as it connects directly to the Newtonian argument for dropping the displacement current. Second, we will derive the induction equation according to a general (i.e. non inertial) observer, which will be used later in chapter 7.

While both the electric and magnetic fields are dynamical degrees of freedom at the level of Maxwell equations, in textbook magneto-hydrodynamics the electric field is demoted to a lesser role and the magnetic field evolution is described by the induction equation. The Newtonian argument (see, for example, [39]) for deriving the induction equation from Maxwell involves assuming that the dynamics is associated with characteristic length- and timescales, *L* and *T*, leading to an associated velocity $V \sim L/T$ which is much smaller than the speed of light. With this in mind, we can "massage" the Faraday equation in the form

$$\underbrace{\perp_{ab} \dot{b}^{b} - b^{b} \left(\sigma_{ba} + \omega_{ba} - \frac{2}{3}\theta \perp_{ba}\right)}_{\sim b/T} + \underbrace{\varepsilon_{abc} \left(\nabla^{b}e^{c} + a^{b}e^{c}\right)}_{\sim e/L} = 0 , \qquad (5.13)$$

to see that $e \sim Lb/T \sim Vb$. As long as the electric and magnetic fields are slowly evolving, a similar dimensional analysis then leads us to neglecting terms involving

⁶The same is true for the observer four velocity gradients, and we also note for clarity that W^a is the vorticity vector, defined as $W^a = \frac{1}{2} \varepsilon^{abc} \omega_{bc}$.
the electric field (i.e. the displacement current) in the Ampère law, so that

$$J^{a} = \frac{1}{\mu_{0}} \left(\varepsilon_{abc} \nabla^{b} b^{c} + \varepsilon_{abc} a^{b} b^{c} \right) .$$
(5.14)

We then immediately see where this is going to take us. By effectively working with the pre-Maxwell form of Ampère law (leaving out the displacement current) the charge current is slaved to the magnetic field. At the same time, we no longer have an evolution equation for the electric field e^a , so we need an additional relation between the electric and magnetic fields. This is where the issue of (infinite) conductivity enters the discussion. Connecting the local fluid four-velocity u^a to the observer U^a as⁷

$$u^{a} = \gamma (U^{a} + v^{a})$$
, $U^{a} v_{a} = 0$, $\gamma = (1 - v^{a} v_{a})^{-1/2}$, (5.15)

where v^a is the spatial fluid velocity as measured by the observer U^a , we see that the electric field as measured by the fluid is linked to e^a , b^a (measured by U^a) as

$$F_{ab}u^{b} = \gamma \left[e_{a} + \varepsilon_{abc}v^{b}b^{c} + U_{a}(v^{b}e_{b}) \right] .$$
(5.16)

Assuming this vanishes yields

$$e_a + \varepsilon_{abc} v^b b^c = 0. ag{5.17}$$

With this constraint, we can derive the induction equation from Faraday's law. To do so, note that

$$\varepsilon_{abc}\varepsilon^{cde} = U^{f}U_{g}\varepsilon_{fabc}\varepsilon^{gdec} = -3!U^{f}U_{g}\delta_{f}^{[g}\delta_{a}^{d}\delta_{b}^{e]}$$
$$= \left(\delta_{a}^{d}\delta_{b}^{e} - \delta_{a}^{e}\delta_{d}^{b}\right) - \left(//_{a}^{d}\delta_{b}^{e} - //_{a}^{e}\delta_{b}^{d}\right) - \left(\delta_{a}^{d}//_{b}^{e} - \delta_{a}^{e}//_{d}^{b}\right) , \quad (5.18)$$

where we introduced the parallel projection $//{a}_{b} = -U^{a}U_{b}$. When this is contracted with a spatial tensor (with respect to U^{a}) the last two terms in eq. (5.18) can be dropped. It follows that

$$\varepsilon_{abc}\varepsilon^{cde}\left(a^{b}v_{d}b_{e}\right) = \left(a^{b}b_{b}\right)v_{a} - \left(a^{b}v_{b}\right)b_{a}.$$
(5.19)

We also need to take care of the curl of e^a term in the Faraday equation. This can be written

$$-\varepsilon_{abc}\nabla^{b}\left(\varepsilon^{cde}v_{d}b_{e}\right) = \left[-\varepsilon_{abc}\left(\nabla^{b}\varepsilon^{cde}\right)v_{d}b_{e}\right] - \left[\varepsilon_{abc}\varepsilon^{cde}\nabla^{b}\left(v_{d}b_{e}\right)\right], \qquad (5.20)$$

where it is convenient to consider the two terms separately. We start from the second term and, even if U^a is not necessarily surface forming (i.e. has non-vanishing vorticity), we introduce a "spatial" covariant derivative D in the usual way (projecting each

⁷It would seem natural, in writing down the Maxwell equations in the fibration formulation, to identify the fibration observer as the local fluid four-velocity [35, 21]. However, as we want here to make contact with the discussion in chapter 4, it is natural to distinguish the two (cf. discussion in section 4.5).

index in the sub-space orthogonal to U^a). Then, it is easy to see that

$$\varepsilon_{abc}\varepsilon^{cde}D^b\left(v_db_e\right) \doteq \varepsilon_{abc}\varepsilon^{cde}\left(\bot_f^b \bot_d^g \bot_e^h\right)\nabla^f\left(v_gb_h\right) = \varepsilon_{abc}\varepsilon^{cde}\nabla^b\left(v_db_e\right) , \qquad (5.21)$$

and hence

$$-\varepsilon_{abc}\varepsilon^{cde}\nabla^{b}\left(v_{d}b_{e}\right) = D^{b}\left(v_{b}b_{a}\right) - D^{b}\left(v_{a}b_{b}\right) .$$
(5.22)

As for the other term, writing it it as

$$-\varepsilon_{abc} \left(\nabla^{b} \varepsilon^{cde}\right) v_{d} b_{e} = -U^{g} \varepsilon_{gabc} \varepsilon^{fcde} \left(\nabla^{b} U_{f}\right) v_{d} b_{e}$$
$$= -U^{g} \delta_{g}^{[f} \delta_{a}^{d} \delta_{b}^{e]} g_{fh} \left(-U^{b} a^{h} + \omega^{bh} + \sigma^{bh} + \frac{1}{3} \theta \perp^{bh}\right) v_{d} b_{e} , \quad (5.23)$$

we see that—given the anti-symmetrization—it vanishes identically. In summary, the induction equation according to a generic observer⁸ can be written as

$$\perp^{ab} \dot{b}_b + D_b(v^b b^a) - D_b(v^a b^b) = \left(\sigma^{ab} - \omega^{ab} - \frac{2}{3}\theta \perp^{ab}\right)b_b + v^a(a_b b^b) - b^a(a_b v^b) ,$$
(5.24)

where the terms on the left should be familiar, while those on the right vanish for an inertial observer.

Let us now pause for a second, and ponder the implications of the argument we put forward—which is the intuitive extension to the curved spacetime setting of the Newtonian one. Clearly, the argument is non-controversial at the Newtonian/non-relativistic level. At the relativistic level, however, dropping the displacement current might not be fully justified, at least not in general. Whilst this may be seen as a flaw in the logic, it is actually the reason why we derived the induction equation this way. We could have, in fact, argued for a relation of the form in eq. (5.17) from the beginning, thus arriving at the same induction equation but "sweeping under the rug" the controversial issue of dropping the displacement current. The derivation provided suggests that, in a sense, magneto-hydrodynamics is intrinsically-with a slight abuse of nomenclaturea "post Newtonian" theory as it necessarily involves a low-frequency/low-velocity approximation (the timescale over which the local electric field shorts out is tiny but not zero). In essence, we have to apply the magneto-hydrodynamics approximation with some level of caution. We also point to the discussion in [26] where the same argument is detailed in the so-called "3+1 formulation"—thus showing in which sense the approximation may still be used (on a case by case basis) and how gauge issues (in the choice of lapse and shift) play a key role for the validity of the approximation.

We conclude this section by showing how the derived equation further simplifies in the Newtonian limit. On dimensional grounds, we observe that the last two terms on

⁸The worldlines of the generic observer U^a constitute a fibration of the spacetime, hence we may call this the ideal induction equation in the fibration framework, as opposed to the corresponding "3+1 form" derived by, for example, Andersson et al. [26].

the right hand side of eq. (5.24) contain an extra factor of $1/c^2$ (with respect to the rest, where *c* is the speed of light), and will as a result be negligible in the non-relativistic limit ($c^2 \rightarrow \infty$). Similarly, let us consider the absence of monopoles constraint. From eq. (5.12) and eq. (5.17) we immediately obtain

$$\perp^a_b \nabla_a b^b = 2W^a \varepsilon_{abc} v^b b^c , \qquad (5.25)$$

and we observe that the term on the left hand side is $\sim b/L$ while that on the right is $\sim bL/T^2$. Dimensional consistency implies the term on the right-hand side contains an extra factor of $1/c^2$ and should be neglected in the Newtonian limit. In essence, non-inertial effects do not affect the absence of monopoles constraint at the Newtonian level. When it comes to the Lorentz force, we expect it not to change at the Newtonian level, but let us nonetheless check this for consistency. The Lorentz four-force can be written

$$-j_b F^{ba} = -\left(\sigma U_b + J_b\right) \left(U^b e^a + U^a e^b + \varepsilon^{bacd} U_c b_d \right)$$
$$= -U^a \left(J_b \varepsilon^{bcd} v_c b_d \right) + \varepsilon^{abc} \left(J_b - \sigma v_b \right) b_c , \quad (5.26)$$

where we used the ideal magneto-hydrodynamics relation (5.17) in the last step. The Lorentz three force corresponds to the second term, where the charge density is measured by the observer, hence does not (in general) vanish. However, if we insist on the local charge density to be zero (consistently with (5.17)), then we have

$$-u^{a}j_{a} = W(\sigma - v_{a}J^{a}) = 0 \Longrightarrow J_{b} - \sigma v_{b} = (g^{a}_{b} - v^{a}v_{b})J_{a}.$$

$$(5.27)$$

Re-inserting the factor of $1/c^2$ we see that the second term is negligible with respect to the first. As also the second term in eq. (5.14) is negligible in the Newtonian limit, we see that the Lorentz force in the Euler equation is unchanged (as expected).

5.3 MHD covariant filtering: first steps

We now want to make contact with the filtering strategy developed in chapter 4. Whilst this is very much unfinished business, we discuss some of the issues that arise along the way, and what seems to be the right strategy to make progress.

First of all, we note that it does not seem to make much sense to take the MHDapproximation—that is, neglecting the displacement current—first and then applying the filtering procedure. The main reason for this has to do with the choice of filtering observer—recall the discussion of the normalization issue in section 4.5—and the simple fact that electric and magnetic fields are observer dependent quantities. That such a strategy would be problematic can also be appreciated by considering the filtered charge current, noting that the issue of local charge neutrality and vanishing electric field are intrinsically linked [26]. Using the filtering notation introduced in chapter 4, we can consider the filtered charge four-current and decompose it as

$$\langle j^a \rangle = \tilde{\sigma} \tilde{u}^a + \tilde{J}^a$$
, $\tilde{\sigma} = \tilde{u}_a \langle j^a \rangle$ and $\tilde{J}^a = \tilde{\perp}^a_b \langle j^b \rangle$, (5.28)

where \tilde{u}^a is the filtered four-velocity (say, the Favre-filtered velocity). Using an analogous decomposition of the fine-scale charge-current with respect to the fine-scale fluid velocity we get

$$\tilde{\sigma} = \tilde{u}_a \left(\left\langle \sigma u^a \right\rangle + \left\langle J^a \right\rangle \right) \ . \tag{5.29}$$

The upshot is that even if we assume the system to be charge neutral at the fine-scale (i.e. $\sigma = 0$), we are left with

$$\tilde{\sigma} = \tilde{u}_a \langle J^a \rangle , \qquad (5.30)$$

which clearly does not have to vanish. Conversely, if we impose charge neutrality on the filtered flow then this condition may not hold on the small scale (which might be an issue for equation of state inversion in numerical relativity simulations).

The upshot is that the correct strategy seems to involve a filtering at the level of the Maxwell equations. We can then use the fact that the filtering procedure defined in chapter 4 commutes with partial derivatives to arrive at

$$\nabla_a \langle F^{ba} \rangle = \mu_0 \langle j^b \rangle . \tag{5.31}$$

These equations can then be written in terms of the coarse-grained electromagnetic fields—defined decomposing $\langle F^{ab} \rangle$ as in eq. (5.10), now in terms of the filtered velocity \tilde{u}^a . The net result is that the filtered Maxwell equations retain the usual form (as the Maxwell system is linear). In particular, the filtered electric and magnetic field are given in terms of the fine scale ones as

$$\tilde{E}^{a} = \tilde{u}^{b} \langle u_{a} e_{b} \rangle - \tilde{u}^{b} \langle u_{b} e_{a} \rangle - \tilde{\varepsilon}_{abc} \langle u^{b} b^{c} \rangle , \qquad (5.32)$$

and

$$\tilde{B}^{a} = \tilde{u}^{b} \langle u_{a} b_{b} \rangle - \tilde{u}^{b} \langle u_{b} b_{a} \rangle + \tilde{\varepsilon}_{abc} \langle u^{b} e^{c} \rangle .$$
(5.33)

From this, we immediately see that even if we take the fine-scale electric field to vanish (due to the assumption on local charge neutrality) we would still have a non-vanishing coarse grained electric field:

$$\tilde{E}^{a} = -\tilde{\varepsilon}_{abc} \langle u^{b} b^{c} \rangle$$
, $\tilde{B}^{a} = \tilde{u}^{b} \left(\langle u_{a} b_{b} \rangle - \langle u_{b} b_{a} \rangle \right)$. (5.34)

Nonetheless, we may, pragmatically, move on and take the MHD-approximation at the filtered level (i.e. neglecting the displacement current in the filtered equations) and provide a closure relation for the filtered electric field. For example, we could model the filtered electric field using an algebraic decomposition in terms of the filtered variables as

$$\tilde{E}^a = \gamma_1 \tilde{B}^a + \gamma_2 \tilde{J}^a + \gamma_3 \tilde{\varepsilon}^{abc} \tilde{B}_b \tilde{J}_c .$$
(5.35)

Whilst this is a very simple closure scheme, it is attractive as it allows for an immediate physical interpretation of the expansion coefficients. It is, in fact natural to interpret the second term as an effective resistivity and the third one as an effective Hall term. The first term instead reminds us of the "classic" alpha-dynamo term, which enters many mean-field-dynamo models [48, 47, 187].

We conclude this section by commenting on the filtering of the fluid equations. When it comes to the baryon continuity equation, it is natural to work with the Favre-filtered four-velocity as in section 4.6 so that it retains the pre-filter form. As for the (fluid part of the) stress-energy-momentum tensor, we know already from the hydrodynamic analysis in chapter 4 that filtering will introduce additional terms in the equations that can be interpreted as effective dissipative terms. In addition to these, we also have terms coming from filtering the Lorentz four-force

$$\mathcal{F}^{a} = \langle j_{b} F^{ba} \rangle - \langle j_{b} \rangle \langle F^{ba} \rangle .$$
(5.36)

This contribution also requires modelling.

5.4 Outlook

As anticipated at the end of the previous chapter, an important point missing from the current discussion is the link with actual numerical relativity simulations. These tend to involve a foliation of spacetime and the associated "3+1" spacetime split [100, 184], thus adding a new observer (the "Eulerian" observer) to the game. In principle, one might make progress simply by "translating" our results to the foliation picture, but it is by no means clear that this is the most sensible way to proceed. Moreover, any discussion of numerical simulations should consider a number of additional issues, such as the role of numerical discretization errors. Models developed directly in the foliation—such as the gradient sub-grid model developed in [219, 49]—appear to be more suited for dealing with issues like the lack of numerical resolution. At the same time, the fibration scheme we developed provides a clear link with the underlying thermodynamics and the equation of state information we would like to extract from numerical simulations. Moreover, the covariance of our framework, as well as the fact that the filtering observer acquires a clear physical meaning, allow us to "lift" the results of box simulations to any space-time (at least in principle).

In essence, the scheme we have discussed suggests that the large-eddy strategy can be enhanced to a tool for linking models valid at "mesoscopic" and "macroscopic" scales. This is relevant as it appears that large-eddy simulations based on gradient subgrid models need to minimally resolve the relevant physical effects in order to describe turbulence at even smaller scales—as demonstrated by Miravet-Tenés et al. [154] for the specific case of the magneto-rotational instability (cf. chapter 7). The (although minimal and incomplete) discussion provided here is a first step in this direction.

Part III

Binary neutron-star merger applications

Chapter 6

Formulating bulk-viscosity for neutron star simulations

In the previous parts of this thesis, we focused on modelling dissipation and turbulence in general relativistic fluids. This part is instead dedicated to neutron star merger applications. In particular, we will consider in this chapter the role of bulk viscosity associated with nuclear reactions—which may, or may not, leave an observable imprint on (say) the gravitational-wave signal [158, 105, 159, 178, 227]—and ask how this mechanism can be implemented in nonlinear simulations. We highlight the formal aspects of the problem and establish how the inevitable "limitations" of a numerical simulation (in terms of resolution) enter the discussion. The aim is to establish to what extent simulations based on an effective bulk viscosity are viable and (perhaps more importantly) when they are not. This understanding will be crucial for future numerical implementations. The results discussed in this chapter have been published in [59].

In the following chapter we discuss the magneto-rotational instability (MRI) using a (Newtonian) local analysis. We do so as the MRI is thought to play a key role in the development of (MHD) turbulence in the outer envelope of a merger remnant. In particular, the discussion we provide is suited for highly-dynamical systems such as mergers, highlighting the importance of global properties for the standard results (and criteria) to be valid.

6.1 Simplifications must be made

The underlying physical model of a neutron star merger is expected to be a system of multiple interacting "fluids" of different charged particle species, coupled to an electromagnetic field and radiation through, for example, neutrinos, all evolving on a dynamical relativistic spacetime [21]. The complexity of this model makes it impractical

for use in either theoretical calculations or numerical simulations. Instead, simplifications must be made, with heuristic arguments needed to justify each assumption that is introduced.

To illustrate the key argument with a simple toy model, consider the problem of heat propagation. When the underlying model is required to be *causal*, the starting point is often taken to be the Cattaneo equation (see section 2.1.2)

$$\tau \frac{\partial q}{\partial t} + q = -\kappa \frac{\partial T}{\partial x} . \tag{6.1}$$

In the "fast relaxation limit", when the relaxation time $\tau \to 0$, we recover Fourier's law, relating the heat flux to temperature gradients, and leading to the familiar heat equation. While the underlying model is hyperbolic, the fast relaxation limit is parabolic and hence not causal. Specifically, working out the characteristic velocities in the problem¹ one finds that the Cattaneo equation (6.1) is causal with finite propagation speeds bounded by $\pm (\kappa/\tau)^{1/2}$. At the same time, there is a critical wavenumber $\gtrsim (\kappa\tau)^{-1/2}$ below which the behaviour is purely parabolic and the solution is diffused away. An illustration of the transition from second sound to diffusion can be found in Figure 16 of [21].

From a theory point of view it would be natural to argue that we should base our models on the Cattaneo formulation, but from a numerical perspective this may be problematic. We would need to resolve the (presumably fast) relaxation towards equilibrium and this may not be possible/practical. In this sense, the parabolic prescription may be preferable.

A heuristic argument for using the parabolic heat equation within a relativistic model for which causality would be a prerequisite—would be as follows. We assume that on the length scales *L* relevant for our model we have $\tau \sim L/c$, where *c* is the speed of light. By causality there can be no (propagating!) scales of physical relevance faster than *c*, hence with timescales smaller than τ or (equivalently) frequency scales larger than τ^{-1} . Therefore the only relevant behaviour for heat propagation is the purely parabolic case where heat fluctuations are rapidly damped, which is well modelled by the standard heat equation. It is possible to use the internal consistency of the underlying model to check when this heuristic argument is valid. For example, to be consistent with causality the dispersion relation at low frequencies requires that $\kappa \leq \tau c^2 \sim cL$.

The key point here is the existence of a single scale (length, time, or frequency) at which some physical effect acts or changes type. This issue is often considered for turbulence (for example, are the length scales probed sufficient to trigger the magneto-rotational instability [71, 204, 124, 102]), for reactions (are physical regions of parameter space probed so that out-of-equilibrium physics, such as a bulk viscosity has to be accounted

¹Computed, for example, taking the large wavenumber limit of the (real part of the) phase speed.

for [13]), and for radiation (are neutrinos propagating or trapped, see e.g. [104]). When this key scale is outside that which can be included in the model then the physical effect is either ignored (by using a purely ideal model, or by assuming instantaneous relaxation to an equilibrium) or modelled (by approximating the additional physics through a closure term, such as an effective equation of state, or a large-eddy approach [104, 94, 176, 49]).

The fundamental issue regards how the physics, and hence the model, should behave when key scales cross or overlap at different parts of the required parameter space. This is particularly relevant for nonlinear numerical simulations, where the discretization length introduces a scale (or multiple scales with uneven grids or mesh refinement), and nonlinearity can lead to the relevant physical scales varying over many orders of magnitude. The interaction between the different scales makes heuristic simplifications dubious.

With these issues in mind we will study, analytically and numerically, issues relating to bulk viscosity in reactive fluids. Here the underlying model involves nuclear reactions, specifically the direct and modified Urca reactions (although the analytical calculations presented apply more generally). These microphysical reactions can, in some regimes, be approximated as a (resonant) bulk viscosity [194]. However, the timescales on which the reactions take place strongly depend on (for example) the temperature and may as a result be close to those that can be captured in numerical calculations. The timescales may also interact with, for example, large eddy closure terms required for turbulent regions (cf. section 6.4.3).

6.2 The reactive system

We want to consider the hydrodynamics of an isotropic reactive system consisting of comoving neutrons, electrons and protons, the simplest meaningful matter composition for a neutron star core.² Further, because we assume charge neutrality, there are only two independent number densities (or, equivalently, one density and one species fraction). These are conveniently taken as n, representing the baryon number density, and $Y_e = n_e/n$ the electron (lepton) fraction. The proton fraction then follows as $Y_p = n_p/n = Y_e$ while the neutron fraction is given by $Y_n = n_n/n = 1 - Y_e$. Because we assume isotropy, the stress-energy-momentum tensor takes the perfect fluid form (cf. section 1.3)

$$T^{ab} = \varepsilon u^a u^b + p(g^{ab} + u^a u^b) , \qquad (6.2)$$

²This may seem somewhat reductionist, given that the high density region is likely to bring other matter constituents into play, but our main interest is to establish the principles involved. If richer matter models are required then the extension of our discussion will be conceptually straightforward.

where the four-velocity u^a is uniquely defined as the one associated with the flow of all particle species. The pressure and energy are identified with the corresponding thermodynamical quantities. As we will not impose that the system is in chemical equilibrium (we allow reactions) we assume the equation of state to involve three parameters. That is, the pressure follows from $p = p(n, \varepsilon, Y_e)$. This relation is assumed to be provided in tabulated form (suitable for a numerical simulation, see appendix C and [210]). The energy-momentum conservation laws and baryon continuity are the same as in section 1.3. Nonetheless, we repeat them here for convenience:

$$u^b \nabla_b \varepsilon = -(p+\varepsilon)\theta , \qquad (6.3a)$$

$$(p+\varepsilon)a_b = -\perp_b^c \nabla_c p , \qquad (6.3b)$$

and

$$u^a \nabla_a n + n\theta = 0 . ag{6.4}$$

As we are presently working with a three parameter equation of state, we need an evolution equation for the electron fraction. We write this as

$$u^a \nabla_a Y_{\rm e} = \frac{\Gamma_{\rm e}}{n} , \qquad (6.5)$$

where the rate Γ_e is generally non-vanishing as we are considering a reactive system. Once the reaction rate is provided by the microphysics (and tabulated as a function of the other variables), these equations constitute a closed system.

From the perspective of thermodynamics, it is natural to introduce the affinity

$$\beta = \mu_{\rm n} - \mu_{\rm p} - \mu_{\rm e} , \qquad (6.6)$$

as it quantifies how far the system is out of cold beta equilibrium. This has the advantage that we can work within the so-called Fermi surface approximation [9] and express relevant quantities, like the reaction rates, as expansions for small values of β (with the coefficients in the expansion evaluated at equilibrium, $\beta = 0$). In the following, and notably in the illustrations we provide, we assume that this strategy is appropriate. Pragmatically, this makes sense as we are only aiming to establish a proof of principle and these assumptions allow us to work out all required parameters for the model from a standard tabulated equation of state. However, it is important to keep in mind that the assumptions will not be appropriate for much of the parameter space (temperature and density) explored in the binary neutron star merger/post-merger phase, and they completely exclude any neutrino effects. At finite temperatures, the true notion of beta equilibrium is more complex, and may require the addition of an isospin chemical potential in the definition of β (see [9, 10, 14, 6, 8, 104]). A complete model should account for the correct notion of equilibrium, but this will require the equation of state table to be extended to include all necessary information. As such data is not yet available for simulations, we are (pragmatically) doing the best we can given the information at hand.

As we will see in section 6.2.1, the affinity is (thermodynamically) conjugate to Y_e , meaning that either of the two variables can be used "equivalently" in the discussion. This is important because state-of-the-art simulations tend to involve Y_e while the theory is somewhat more transparent when expressed in terms of β . In the following we will develop the model both in terms of β and Y_e , showing the expected consistency, and emphasising the different perspectives the two complementary approaches bring on the problem. The evolution equation for β is easily obtained by considering it as a function of (ε, n, Y_e) —which follows from β and Y_e being thermodynamically conjugated. We arrive at

$$u^{a}\nabla_{a}\beta = \left(\frac{\partial\beta}{\partial Y_{e}}\right)_{n,\varepsilon}\frac{\Gamma_{e}}{n} - n\mathcal{B}\theta , \qquad (6.7)$$

with

$$\mathcal{B} = \left(\frac{\partial\beta}{\partial n}\right)_{\varepsilon, Y_{e}} + \frac{p+\varepsilon}{n} \left(\frac{\partial\beta}{\partial\varepsilon}\right)_{n, Y_{e}}.$$
(6.8)

We again see that the system of equations is closed once the reaction rate (as well as the relevant thermodynamical coefficients) is provided.

A useful simplification occurs when the system is sub-thermal, when³ $\beta/T \ll 1$. Then we can expand the rate with respect to chemical equilibrium $\beta = 0$ to write it as⁴ $\Gamma_{\rm e} = -\gamma\beta$. The evolution equation for the affinity β then simplifies to

$$u^a \nabla_a \beta = -\mathcal{A}\beta - n\mathcal{B}\theta , \qquad (6.9)$$

where we introduced the new coefficient (with units of inverse time)

$$\mathcal{A} = \frac{\gamma}{n} \left(\frac{\partial \beta}{\partial Y_{\rm e}}\right)_{n,\varepsilon} \,. \tag{6.10}$$

The information encoded in the reaction rate Γ_e is now "stored" in γ . We can then make progress and compute γ from the equation of state tables provided in the compOSE database [210] *as long as* we ignore finite temperature effects [9, 10, 14, 6, 8] (see appendix C and [106] for more details). While the coefficient \mathcal{B} can be introduced without reference to an expansion around equilibrium, this is not the case for \mathcal{A} . In the sub-thermal limit we retain only terms linear in β , so that \mathcal{A} must be evaluated at $\beta = 0$.

For completeness, let us also comment on the entropy density, viewed as a function of (ε, n, Y_e) . Using the equations of motion for these quantities (as well as the generalized

³Let us note for clarity that we use units where the Boltzmann constant $k_B = 1$.

⁴There is a sign convention here, and we are following [11]. The logic is, if $\beta > 0$ then $\mu_n > \mu_e + \mu_p$ and neutron decay is favoured (over electron-capture) as this will release energy. Therefore, we want the electron rate to be positive when β is positive and vice versa. The sign of γ should then be negative.

Gibbs relation provided in section 6.2 below) we arrive at

$$T\nabla_a \left(s u^a \right) = \beta \Gamma_{\rm e} \,. \tag{6.11}$$

As long as β has the same sign as Γ_e the entropy increases. This is guaranteed to be the case in the sub-thermal limit if $\gamma < 0$. Note that this assumes that a negligible amount of energy is deposited in neutrinos by the reactions, which will be a poor approximation at high temperatures. This important caveat will quantitatively affect our results without changing the qualitative conclusions.

6.2.1 Thermodynamics of a reactive system

Having discussed the hydrodynamics, let us turn to the associated thermodynamics. Because the underlying model is required to be causal, and hence based on Cattaneotype laws, it is natural to set the discussion within the Extended Irreversible Thermodynamics (EIT) framework (cf. section 2.1.2). We here provide a streamlined discussion, and refer to the monograph [121] and references therein (see also [89] for a recent analysis focused specifically on bulk viscosity). The first step is to <u>assume</u> that the Gibbs relation takes the usual form—noting that the various quantities may not be in thermodynamical equilibrium

$$p + \varepsilon = \sum_{\mathbf{x}=\mathbf{n}, \mathbf{p}, \mathbf{s}, \mathbf{e}} n_{\mathbf{x}} \mu_{\mathbf{x}} = n \mu_{\mathbf{n}} - n_{\mathbf{e}} \beta + Ts , \qquad (6.12)$$

and

$$dp = \sum_{\mathbf{x}=\mathbf{n},\mathbf{p},\mathbf{s},\mathbf{e}} n_{\mathbf{x}} d\mu_{\mathbf{x}} = n d\mu_{\mathbf{n}} - n_{\mathbf{e}} d\beta + s dT.$$
(6.13)

As we will be working at the fluid level with either (n, ε, Y_e) or (n, ε, β) , we can use the entropy as a thermodynamical potential. This is convenient because if we also assume that the system is close to thermodynamical equilibrium, we can expand the entropy as

$$s = s^{\text{eq}}(n,\varepsilon) + \frac{1}{2}s_2(n,\varepsilon)\beta^2$$
, where $s_2 = \left(\frac{\partial^2 s}{\partial\beta^2}\Big|_{\beta=0}\right)_{n,\varepsilon}$. (6.14)

From this we can compute the out-of-equilibrium expansion of the thermodynamical variables. Linearizing in deviations from equilibrium, and assuming that the equation

of state is expressed in terms of Y_e rather than β , we obtain

$$T = T^{\text{eq}} \left[1 - n \left(\frac{\partial \beta}{\partial \varepsilon} \right)_{n, Y_{\text{e}}} \left(\frac{\partial \beta}{\partial Y_{\text{e}}} \right)_{n, \varepsilon}^{-1} \beta \right] + \mathcal{O}(\beta^2) , \qquad (6.15a)$$

$$\mu_{\rm n} = \mu^{\rm eq} - \left[n\mu^{\rm eq} \left(\frac{\partial \beta}{\partial \varepsilon} \right)_{n, Y_{\rm e}} + n \left(\frac{\partial \beta}{\partial n} \right)_{\varepsilon, Y_{\rm e}} - Y_{\rm e} \right] \left(\frac{\partial \beta}{\partial Y_{\rm e}} \right)_{n, \varepsilon}^{-1} \beta + \mathcal{O}(\beta^2) , \qquad (6.15b)$$

$$s = s^{\text{eq}} + \frac{1}{2} \frac{n}{T^{\text{eq}}} \left(\frac{\partial \beta}{\partial Y_{\text{e}}}\right)_{n,\varepsilon}^{-1} \beta^2 .$$
(6.15c)

Note that the thermodynamical requirement on the entropy reaching a maximum at equilibrium, namely $s_2 < 0$, implies that $\partial \beta / \partial Y_e$ must be negative. Recalling eqs. (6.9) and (6.10), and the fact that the restoring term $\gamma < 0$, we see that $\mathcal{A} > 0$ and therefore plays the role of an (inverse) relaxation rate.

Now that we have expansions (in β) of the thermodynamical variables, we can use the Gibbs relation to work out the pressure. To linear order in the deviation from equilibrium we then have

$$p = p(n,\varepsilon,\beta) = p^{eq} + p_1\beta$$
, where $p^{eq} = p(n,\varepsilon,\beta=0)$, and $p_1 = \left(\frac{\partial p}{\partial \beta}\Big|_{\beta=0}\right)_{n,\varepsilon}$,
(6.16)

or, explicitly,

$$p^{\rm eq} = -\varepsilon + n\mu^{\rm eq} + T^{\rm eq}s^{\rm eq} , \qquad (6.17)$$

and

$$p_{1} = -n^{2} \left(\frac{\partial \beta}{\partial Y_{e}}\right)_{n,\varepsilon}^{-1} \left[\frac{p^{eq} + \varepsilon}{n} \left(\frac{\partial \beta}{\partial \varepsilon}\right)_{n,Y_{e}} + \left(\frac{\partial \beta}{\partial n}\right)_{\varepsilon,Y_{e}}\right] = \frac{\gamma}{n} \frac{\mathcal{B}}{\mathcal{A}}.$$
 (6.18)

In essence, the thermodynamical expansion provides us with an expression for the outof-equilibrium contribution to the pressure, which would naturally be interpreted as a bulk viscosity. We identify

$$\chi_t = \left(\frac{\partial p}{\partial \beta}\Big|_{\beta=0}\right)_{n,\varepsilon} \beta = p_1 \beta .$$
(6.19)

Note that, even though we have outlined the derivation in the simplest case (where the system is subthermal and close to equilibrium), the argument applies more generally. A broader discussion would rely on a detailed description of the out-of-equilibrium physics, which is not included in equation of state tables currently used for numerical simulations. Specifically, as described in appendix C, starting from a three-parameter equation of state from the compOSE database, the derivatives we need can be worked out before carrying out a simulation. We are relying on this in the specific example discussed later.

6.2.2 Thermodynamics working with the equilibrium electron fraction

As mentioned earlier, we may equivalently work with the electron fraction Y_{e} . Given this, we want to revisit the path that we just followed, now working with the electron fraction. As we will see, this also results in a correction term to the pressure, which will be naturally expressed in terms of derivatives involving a notion of equilibrium electron fraction.

In order to define the equilibrium electron fraction we consider the following thermodynamical potential

$$g = s - n_{\rm e} \frac{\beta}{T} , \qquad (6.20)$$

such that

$$dg = \frac{1}{T}d\varepsilon - \frac{\mu_{\rm n}}{T}dn - n_{\rm e}d\left(\frac{\beta}{T}\right) \ . \tag{6.21}$$

Exactly as we did for the entropy, we can then expand *g* around equilibrium. There will now be a first order term in β , which provides the "formal definition" of the equilibrium electron number density, $n_{\rm e}^{\rm eq}$. We get

$$g(n,\varepsilon,\beta/T) = s^{\rm eq}(n,\varepsilon) - n_{\rm e}^{\rm eq}\frac{\beta}{T} + \frac{1}{2}g_2\left(\frac{\beta}{T}\right)^2 , \qquad (6.22)$$

where

$$n_{\rm e}^{\rm eq} = \left(\frac{\partial g}{\partial(\beta/T)}\Big|_{\beta=0}\right)_{n,\varepsilon} , \quad \text{and} \quad g_2 = \left(\frac{\partial^2 g}{\partial(\beta/T)^2}\Big|_{\beta=0}\right)_{n,\varepsilon} . \tag{6.23}$$

We can then work out the expansion (to first order in β) for n_e as

$$n_{\rm e}(n,\varepsilon,\beta) = n_{\rm e}^{\rm eq}(n,\varepsilon) - g_2 \frac{\beta}{T} , \qquad (6.24)$$

and use it in the definition of *g* to arrive at

$$g = s^{eq} + \frac{1}{2}s_2\beta^2 - n_e^{eq}\frac{\beta}{T} + g_2\left(\frac{\beta}{T}\right)^2 .$$
 (6.25)

Comparing this with the expansion above we identify

$$g_2(n,\varepsilon) = -T_{\rm eq}^2 s_2 = -T_{\rm eq} \left(\frac{\partial\beta}{\partial n_{\rm e}}\right)_{n,\varepsilon}^{-1} \,. \tag{6.26}$$

Using this result and working out the expansion for the thermodynamical variables from *g*, we obtain (after linearizing in β and introducing $Y_e^{eq} = n_e^{eq}/n$)

$$T = T^{\rm eq} + n \left(\frac{\partial Y_{\rm e}^{\rm eq}}{\partial \varepsilon}\right)_n T^{\rm eq} \beta , \qquad (6.27a)$$

$$\mu_{n} = \mu^{eq} + \left[\mu^{eq}n\left(\frac{\partial Y_{e}^{eq}}{\partial \varepsilon}\right)_{n} + n\left(\frac{\partial Y_{e}^{eq}}{\partial n}\right)_{\varepsilon} - Y_{e}^{eq}\right]\beta, \qquad (6.27b)$$

$$Y_{\rm e} = Y_{\rm e}^{\rm eq} - \left(\frac{\partial\beta}{\partial Y_{\rm e}}\right)_{n,\varepsilon}^{-1} \beta .$$
(6.27c)

Combining the result with the Gibbs relation we can work out the pressure (and the "thermodynamical" bulk viscosity)

$$p(n,\varepsilon,\beta) = p^{\rm eq}(n,\varepsilon) + p_1\beta = p^{\rm eq} + \chi_t , \qquad (6.28)$$

with

$$p_1 = n \left[\left(p^{\text{eq}} + \varepsilon \right) \left(\frac{\partial Y_{\text{e}}^{\text{eq}}}{\partial \varepsilon} \right)_n + n \left(\frac{\partial Y_{\text{e}}^{\text{eq}}}{\partial n} \right)_{\varepsilon} \right] \,. \tag{6.29}$$

The take home message is that we have two equivalent expressions for the bulk-viscosity purely from thermodynamical arguments: this is always written as $\chi_t = p_1\beta$, where p_1 can be either written as in eq. (6.15) or (6.29). These results are thermodynamically correct as long as the linearization in β is valid (i.e., the system is close to chemical equilibrium), independently of the modelling of the relaxation towards equilibrium.

6.3 Approximating the reactive system

At this point we have formulated the relaxation problem for the reactive system. The equations we have written down are, in principle, all we need to evolve the system. However, it may not be numerically practical to solve the full nonlinear system, for example when the physical reactions are fast compared to numerically resolvable timescales. Given this, it is natural to consider approximations.

To set up the discussion let us introduce the (proper) time derivative $d/dt = u^a \nabla_a$. We can then write the hydrodynamical equations in non-dimensional form (assuming for a moment that we work with the lepton fraction instead of β , an assumption that makes

no practical difference here)

$$\frac{d\varepsilon}{dt} = -\frac{1}{\epsilon_{St}} (\varepsilon + c_r^2 p) \theta , \qquad (6.30a)$$

$$a_b = -\frac{1}{\epsilon_{St}} \frac{1}{\epsilon_{Ma}^2} \frac{1}{\epsilon + c_r^2 p} \perp_b^c \nabla_c p , \qquad (6.30b)$$

$$\frac{dn}{dt} = -\frac{1}{\epsilon_{St}} n\theta , \qquad (6.30c)$$

$$\frac{dY_{\rm e}}{dt} = -\frac{1}{\epsilon_A} (Y_{\rm e} - Y_{\rm e}^{\rm eq}) , \qquad (6.30d)$$

where we have defined the dimensionless parameters

$$\epsilon_{St} = \frac{l_r}{u_r t_r}, \quad \epsilon_{Ma} = \frac{u_r}{c_r}, \quad \epsilon_A = \frac{1}{\mathcal{A}t_r}, \quad (6.31)$$

and introduced a reference sound speed c_r as well as l_r, t_r, u_r as reference lengthscale, timescale and fluid velocity—so that a_b has dimensions u_r/t_r . To write the equations in non-dimensional form as above, we have first taken the sub-thermal limit $\Gamma_{\rm e} = -\gamma\beta$ and then expanded around equilibrium⁵. From this we see that n, ε, u^a evolve on similar timescales (in terms of the proper time associated with u^a), while the electron fraction evolution timescale is given by ϵ_A . Assuming—as expected for large regions in neutron star merger simulations—that reactions occur on a fast timescale (so that $\epsilon_A \ll 1$), we may consider three different regimes: i) The expansion rate⁶ θ varies only (or primarily) on slow timescales-which makes sense for numerical simulations where the spatial dynamics are resolvable but the reaction timescale is not; ii) The expansion θ varies on the fast timescale and hence this must be resolved in a simulation; iii) The flow is turbulent and therefore all scales are coupled. In the last two cases, we cannot analytically simplify the problem much—expensive direct numerical simulations are required, although the large-eddy strategy [151, 136, 49, 58] (see also section 6.4.3) may provide a useful alternative. As our main interest here is to consider the regime where progress can be made through approximations, we focus on the first case, where we can use standard multi-scale methods (see, e.g. [223]) to "integrate out" the fast behaviour.

6.3.1 Multi-scale arguments and the reactive system

Bringing the multiscale argument from [170] to bear on the problem (see appendix B for more details on the results used throughout the rest of this chapter) and assuming that we continue to work with β , we have to compute the late-time behaviour for the affinity by integrating the β equation considering the other variables as parameters, and then taking the limit $t \rightarrow \infty$. The approximated equations for the remaining variables

⁵Let us note that, as is clear from the last of eq. (6.27), $Y_e = Y_e^{eq}$ if and only if $\beta = 0$, and that in the sub-thermal limit this means $\Gamma_e(Y_e = Y_e^{eq}) = 0$.

⁶As can be seen from e.g. eq. (6.7), the expansion rate is a "source-term" in the affinity equation.

are then unchanged to lowest order, but we have to evaluate every function of β using the late-time result. This is intuitively motivated by the underlying assumption that the affinity evolves on a faster timescale, so that the remaining degrees of freedom are approximately frozen on short timescales. The inclusion of the first order corrections then guarantees the approximated equations to be correct to $\mathcal{O}(\epsilon_A^2)$ up to times $\mathcal{O}(1)$.

The equations of motion using the affinity can be written

$$\frac{d}{dt} \begin{pmatrix} n \\ \varepsilon \\ \beta \end{pmatrix} = \begin{pmatrix} -n\theta \\ -(p+\varepsilon)\theta \\ -\mathcal{A}\beta - n\mathcal{B}\theta \end{pmatrix}.$$
(6.32)

The non-dimensional form in eq. (6.30) indicates that the reaction timescale ϵ_A enters only through A—and it can be easily seen that B relates to how quickly the equilibrium electron fraction adjusts to a change in number and energy density. Assuming only Ato be fast, and using the results from appendix B.4 we split the affinity into fast and slow pieces, writing

$$\frac{d}{dt} \begin{pmatrix} n \\ \varepsilon \\ \beta^{\text{slow}} \end{pmatrix} = \begin{pmatrix} -n\theta \\ -(p+\varepsilon)\theta \\ -n\mathcal{B}\theta \end{pmatrix}, \qquad (6.33a)$$

$$\frac{d}{dt}\left(\beta^{\text{fast}}\right) = \left(-\epsilon_A^{-1}\mathcal{A}\beta\right),\tag{6.33b}$$

where $\beta^{\text{fast}} + \beta^{\text{slow}} = \beta$. By construction, \mathcal{A} is independent of β —as it is defined at $\beta = 0$ —so the fast dynamics are now linear and the results of appendix B.3 apply. We see that the invariant manifold—describing the "late-time behaviour"—is given by $\beta^{\text{fast}} = -\beta^{\text{slow}}$, which is equivalent to saying $\beta = 0$ on the invariant manifold. That is, the invariant manifold corresponds to beta equilibrium, as expected.

The reduced system is

$$\frac{d}{dt} \begin{pmatrix} n\\ \varepsilon\\ \beta^{\text{slow}} \end{pmatrix} = \begin{pmatrix} -n\theta\\ -(p+\varepsilon)\theta\\ -n\mathcal{B}\theta \end{pmatrix} - \epsilon_A \frac{n\mathcal{B}\theta}{\mathcal{A}} \begin{pmatrix} 0\\ \frac{\partial p}{\partial \beta}\theta\\ 0 \end{pmatrix} , \qquad (6.34)$$

where all terms (pressure and its derivatives, and \mathcal{B}) have to be evaluated at beta equilibrium. The final equation decouples as all quantities in the first two equations depend on the total β evaluated at equilibrium, which is $\beta = 0$. We see from the energy equation that the pressure correction appears as

$$\chi_d = -\epsilon_A \left(\frac{\partial p}{\partial \beta}\right)_{n,\epsilon} \frac{n\mathcal{B}\theta}{\mathcal{A}} = -\left(\frac{\partial p}{\partial \beta}\right)_{n,\epsilon} \frac{n\mathcal{B}\theta}{\mathcal{A}},\tag{6.35}$$

where in the second equality we have re-absorbed the scaling parameter into the reaction rate $\epsilon_A^{-1} \mathcal{A} \to \mathcal{A}$.

If we instead make the assumption that \mathcal{B} is also a fast parameter, then it is easy to see that the affinity would be an entirely fast variable, that is $\beta^{\text{slow}} = 0$ automatically. Repeating the analysis, the invariant manifold would now be $\varphi(X) = -n\mathcal{B}\theta/\mathcal{A}$, and as a consequence the pressure in the energy (as well as Euler) equation

$$p = p\left(n, \varepsilon, -\frac{n\mathcal{B}\theta}{\mathcal{A}}\right) = p^{\text{eq}} + \chi_d .$$
(6.36)

This shows that, in the case when \mathcal{B} is fast the bulk-viscous correction enters already at lowest order. The corrections to this are second order in β , beyond the regime of validity of the theory (linear in β), and hence cannot be trusted. In essence, whatever assumption we make for \mathcal{B} (being fast or slow), the result is the same.

The "interpretation" of eq. (6.35) as the Navier-Stokes bulk-viscosity is supported by the analysis in section 6.2 (cf. eq. (6.19))—recall also the argument around eq. (6.1), now specified to eq. (6.9). This result, while in line with "expectations", is non-trivial. In fact, common arguments in favour of a representation of the net effect of under-resolved reactions via a bulk-viscous pressure are typically perturbative in nature [11, 7].

6.3.2 Invariant manifold method with the electron fraction

Let us now run through the invariant manifold argument working with the electron fraction. Consistently with the dimensional analysis of eq. (6.30), we assume that the electron fraction evolves on faster timescales than the other variables. This means we are effectively considering the system of equations

$$\frac{d}{dt} \binom{n}{\varepsilon} = -\binom{n\theta}{(p+\varepsilon)\theta} , \qquad (6.37a)$$

$$\frac{d}{dt}\left(Y_{\rm e}\right) = -\left(\epsilon^{-1}\mathcal{A}(Y_{\rm e} - Y_{\rm e}^{\rm eq})\right) \ . \tag{6.37b}$$

where we have made explicit that the fast behaviour arise from \mathcal{A} as explained in appendix B.4. Again, the linear fast case of appendix B.3 is relevant, giving that the invariant manifold is the equilibrium surface. This immediately tells us that, to lowest order, the approximated equations describe a reactive fluid for which chemical equilibrium is restored immediately on the dynamical timescale. Including the first order corrections to the approximated equations—simply drawing on the results from [170] and appendix B.3—we obtain

$$\frac{d}{dt} \binom{n}{\varepsilon} = -\binom{n\theta}{(p^{\text{eq}} + \varepsilon + \chi_d)\theta} , \qquad (6.38a)$$

where

$$\chi_d = \frac{n}{\gamma} \left(\frac{\partial \beta}{\partial Y_e}\right)_{n,\varepsilon}^{-1} \left(\frac{\partial p}{\partial Y_e}\right)_{n,\varepsilon} \left[\left(p^{eq} + \varepsilon\right) \left(\frac{\partial Y_e^{eq}}{\partial \varepsilon}\right)_n + n \left(\frac{\partial Y_e^{eq}}{\partial n}\right)_{\varepsilon} \right] \theta .$$
(6.39)

Now, using the fact that

$$\left(\frac{\partial p}{\partial \beta}\right)_{n,\varepsilon} = \left(\frac{\partial p}{\partial Y_{\rm e}}\right)_{n,\varepsilon} \left(\frac{\partial \beta}{\partial Y_{\rm e}}\right)_{n,\varepsilon}^{-1} , \qquad (6.40)$$

along with the two alternative formulae for p_1 eqs. (6.18) and (6.29) we observe the (pleasing) consistency with the result in eq. (6.35). Exactly as when working with the affinity, by integrating out the fast variable we pick up an additional contribution to the pressure that corresponds to a bulk-viscous response.

6.3.3 Partially resolved reactions and double counting

Let us now consider the situation where some part of the reactions are slow enough that we can capture them, and the rest is not. We consider the case where we can split the reactions in two families, fast ones (representative of, say, direct Urca processes) that cannot be captured by the numerics, and slow ones (representative of, say, modified Urca processes) that can be resolved. Even though this might be somewhat artificial, the key point is that we assume a clear scale-separation between two types of reactions. We do so as we are only interested in a proof of principle argument here. We want to show the problems in constructing a consistent bulk viscous approximation in this case.

It would seem reasonable, after the discussion we just had, to model the impact of unresolved reactions as a bulk-viscosity. Whilst this may be a valid strategy, one has to be careful because the multi-scale methods suggest that the resolvable reaction rates should pick up a correction term as well. This also provides a proof of principle argument for the "double counting" issue raised by Hammond et al. [104]. The discussion in [104] points out that the effects of under-resolved reactions are already accounted for in schemes aimed at modelling neutrinos, and hence adding a bulk-viscous pressure on top of that may lead to a double-counting.

Let us start from the equation for the electron fraction, written in terms of the reaction rate. We define the fast/slow part of the total creation rate as discussed in appendix B.4

$$\Gamma_f = \lim_{\epsilon \to 0} (\epsilon \Gamma_e) , \qquad \Gamma_s = \Gamma_e - \frac{1}{\epsilon} \Gamma_f , \qquad (6.41)$$

and hence split the electron fraction into its fast and slow contributions $Y_e = Y_s + Y_f$ according to

$$\frac{d}{dt}Y_s = \frac{\Gamma_s}{n}$$
, and $\frac{d}{dt}Y_f = \frac{1}{\epsilon}\frac{\Gamma_f}{n}$. (6.42)

We can think of Γ_f as a function of $(n, \varepsilon, Y_s + Y_f)$ and define a fast equilibrium fraction Y_f^{eq} such that $\Gamma_f(n, \varepsilon, Y_s + Y_f^{\text{eq}}) = 0$. It is then possible to expand the equation for the fast variable as

$$\frac{d}{dt}Y_f = \frac{1}{\epsilon} \frac{\partial \Gamma_f}{\partial Y_e} \Big|_{Y_e = Y_s + Y_f^{eq}} (Y_f - Y_f^{eq}) .$$
(6.43)

Note that, because the two fractions must add up to the total electron fraction, we have $Y_f^{eq} = Y_e - Y_s$, namely the equilibrium fast fraction $Y_f^{eq} = Y_f^{eq}(n, \varepsilon, Y_s)$. As for the slow reaction rate, we do not need to expand it around equilibrium because this is assumed to be resolved in the simulation. Applying the results of appendix B (including the first order corrections) we obtain

$$\frac{d}{dt} \begin{pmatrix} n \\ \varepsilon \\ Y_s \end{pmatrix} = \begin{pmatrix} -n\theta \\ -(p+\varepsilon+\chi_d)\theta \\ \Gamma_s - \frac{1}{n} \left(\frac{\partial\Gamma_s}{\partial Y_e}\right)_{n,\varepsilon} \left(\frac{\partial p}{\partial Y_e}\right)_{n,\varepsilon}^{-1} \chi_d \end{pmatrix},$$
(6.44)

with

$$\chi_{d} = \left\{ \left[\left(p + \varepsilon \right) \left(\frac{\partial Y_{f}^{\text{eq}}}{\partial \varepsilon} \right)_{n, Y_{s}} + n \left(\frac{\partial Y_{f}^{\text{eq}}}{\partial n} \right)_{\varepsilon, Y_{s}} \right] \theta - \left(\frac{\partial Y_{f}^{\text{eq}}}{\partial Y_{s}} \right)_{n, \varepsilon} \frac{\Gamma_{s}}{n} \right\} \left(\frac{\partial \Gamma_{f}}{\partial Y_{e}} \right)_{n, \varepsilon}^{-1} \left(\frac{\partial p}{\partial Y_{e}} \right)_{n, \varepsilon}, \tag{6.45}$$

and everything evaluated at $Y_f = Y_f^{eq}$. This argument then shows that, if there are both fast and slow reactions in the system, and we are trying to capture the effect of the fast/unresolved ones via a bulk-viscosity like contribution, we need to tread carefully, as the introduction of the bulk viscosity also impacts on the resolved reaction rates, and the rates impact on the definition of equilibrium.

6.4 Making contact with simulations

Having discussed the approximate equations we obtain from the multi-scales approach, it makes sense to "step back" and ask to what extent we expect this approximation to make sense for numerical simulations. To set the stage for the discussion, let us rewrite eq. (6.9) as

$$\frac{d\beta}{dt} = -\mathcal{A}\beta + \mathcal{B}\frac{dn}{dt} \,. \tag{6.46}$$

The parabolic limit—which corresponds to neglecting the time derivative of the affinity $d\beta/dt$ —in the Fourier domain is

$$\beta_{NS}(\omega) = \frac{\mathcal{B}}{\mathcal{A}} n\omega , \qquad (6.47)$$



FIGURE 6.1: Illustrating the behaviour of $\beta(\omega)$ in two cases: the solution to the Cattaneo equation (i.e. the full extended irreversible thermodynamics (EIT) relaxation towards the Navier-Stokes limit, blue solid curve) and the parabolic case (the Navier-Stokes limit, orange dashed line). The shaded region indicates frequencies we assume we may not have "access" to numerically (this region moves towards higher frequencies when the numerical resolution is increased).

while the extended irreversible thermodynamics (EIT) result takes the form

$$\beta_{EIT}(\omega) = n\mathcal{B}\mathcal{A}\frac{\omega}{\omega^2 + \mathcal{A}^2} = \beta_{NS}(\omega)\frac{\mathcal{A}^2}{\mathcal{A}^2 + \omega^2}.$$
(6.48)

The two results are compared in fig. 6.1, from which we see that the EIT behaviour shows the "expected" resonance feature [191, 11, 10, 7, 6, 8]. We note, however, that this is not the usual figure (see, for example, figure 2 in [10]), as we are plotting the affinity β and not the bulk-viscosity coefficient. To link the results, we need to recall that $\chi = p_1\beta$ and then define ζ via $\chi = -\zeta\theta$. Then we can use eq. (6.4) to write $\theta = -\dot{n}/n$ to obtain

$$\zeta_{NS} = p_1 \frac{\beta}{\mathcal{A}} n$$
, and $\zeta_{EIT} = \zeta_{NS} \frac{\mathcal{A}^2}{\omega^2 + \mathcal{A}^2}$. (6.49)

The difference is subtle as both β_{EIT} and ζ_{EIT} present resonant features. However, ζ_{EIT} does so when the frequency ω is kept fixed and A is varied, while β_{EIT} exhibits the resonance even if we fix A and vary the frequency.

The difference may not seem particularly relevant, but the illustration in fig. 6.1 allows us to draw useful conclusions. The figure shows that the parabolic limit (i.e. the limit we represent with the multi-scale argument) is a "good" approximation at low frequencies (we provide a quantitative argument in the next section 6.4.1), but becomes less accurate at higher frequencies⁷. Keeping in mind that we consider the problem in the context of numerical simulations, we may also note that there will inevitably be frequencies that we do not have access to. This high-frequency cut-off is (schematically) represented by the shaded region in fig. 6.1. Specifically, the resolution is limited by $\omega \sim \Delta t^{-1}$ where Δt is the numerical timestep. Now, because the resonance frequency is given by $\omega = A$, we are left with two options: either the numerical timestep is large enough ($\Delta t^{-1} < A$) that the peak is not resolved, or the simulation is precise enough that the region to the right of the resonant peak is (at least partially) resolved. In the first case, our simulation cannot resolve the (fast) relaxation towards the Navier-Stokes limit (and it is also likely that the expected instability [115] associated with the Navier-Stokes behaviour will be suppressed). In the second case, the relaxation towards Navier-Stokes can be resolved by the numerics, so working with the Navier-Stokes approximation is just wrong.

6.4.1 What bulk viscous pressure approximation is suitable?

In the regime where the reactions need to be approximated as a bulk viscous pressure there are many possible ways in which such an approximation can be constructed. The standard first order form could be used, where the bulk pressure depends on a (data dependent) coefficient multiplied by the expansion associated with the fluid motion [158]. Alternatively, in frequency space the bulk pressure can be written as a term depending on the thermodynamics multiplied by a function of frequency. In this case the "true" (Cattaneo) result differs from the first order ("Navier Stokes") result only in the form of the frequency term.

Intuitively we have argued that these two different bulk viscous approximations should be close to each other, as long as we are considering frequencies below the resonant peak. Here we make that argument more quantitative.

We are interested in the low frequency part of the pressure that can be captured in a numerical simulation. We will therefore assume that we are only interested in frequencies $\hat{\omega} = \omega/\mathcal{A} < 1$ ("to the left of the peak"), and that there is a hard cut-off at $\hat{\omega}_{\Delta} \sim 2\pi/(\mathcal{A}\Delta t)$ (the numerical scheme is spectral like and captures all frequencies available on the grid). Therefore

$$\chi = \int_{-\infty}^{\infty} \mathrm{d}\hat{\omega} \, H\left(1 - \left|\frac{\hat{\omega}}{\hat{\omega}_{\Delta}}\right|\right) p_1 \beta = 2 \int_0^{\hat{\omega}_{\Delta}} \mathrm{d}\hat{\omega} \, p_1 \beta \,. \tag{6.50}$$

⁷The illustration also provides an intuitive demonstration of why the high-frequency behaviour is noncausal and associated with a linear instability [115].

We want the relative difference between the two bulk viscous pressure approximations, which is

$$\mathcal{E} = \frac{|\chi_{\text{Cattaneo}} - \chi_{\text{NS}}|}{|\chi_{\text{NS}}|} .$$
(6.51)

To compute this we *assume* that we can bound the correction terms as powers of the frequency, as

$$C_{-}\hat{\omega}^{a} < n\mathcal{B}p_{1} < C_{+}\hat{\omega}^{b} , \qquad (6.52)$$

where 0 > a > b > -2 is needed for the results to converge. This seems reasonable for Kolmogorov turbulence, but the range of these coefficients (C_{\pm} , a, b) has an impact on the result.

From this assumption we have

$$2\int_{0}^{\hat{\omega}_{\Delta}} \mathrm{d}\hat{\omega} \, C_{-}\hat{\omega}^{1+a} < |\chi_{\mathrm{NS}}| < 2\int_{0}^{\hat{\omega}_{\Delta}} \mathrm{d}\hat{\omega} \, C_{+}\hat{\omega}^{1+b} \,, \tag{6.53}$$

giving

$$\frac{2C_{-}}{2+a}\hat{\omega}_{\Delta}^{2+a} < |\chi_{\rm NS}| < \frac{2C_{+}}{2+b}\hat{\omega}_{\Delta}^{2+b} .$$
(6.54)

To bound the pressure difference, first write

$$\begin{aligned} |\chi_{\text{Cattaneo}} - \chi_{\text{NS}}| &= \left| n \mathcal{B} p_1 \hat{\omega} \left(\frac{\hat{\omega}^2}{1 + \hat{\omega}^2} \right) \right| \\ &= \left| n \mathcal{B} p_1 \hat{\omega}^3 \right| + \mathcal{O}(\hat{\omega}^5) . \end{aligned}$$
(6.55)

From this we get

$$\frac{2C_{-}}{4+a}\hat{\omega}_{\Delta}^{4+a} < |\chi_{\text{Cattaneo}} - \chi_{\text{NS}}| < \frac{2C_{+}}{4+b}\hat{\omega}_{\Delta}^{4+b} .$$
(6.56)

Using the appropriate bound for numerator and denominator in \mathcal{E} , it follows that

$$\mathcal{E} < \frac{C_+}{C_-} \frac{2+a}{4+b} \hat{\omega}_{\Delta}^{2+b-a} \,. \tag{6.57}$$

As a > b and a - b < 2 we see the difference between the two approximations is, in the frequency range ($\hat{\omega}_{\Delta} < 1$) of interest, of order $\mathcal{E} < \mathcal{O}(\hat{\omega}_{\Delta}^c)$ where $c \in (0, 2)$. If we expect $a \simeq b$ then $\mathcal{E} < \mathcal{O}(\hat{\omega}_{\Delta}^2)$. Therefore the difference between the two approximations will be small until we are close to $\hat{\omega}_{\Delta} = 1$, which is the resonant frequency peak. This supports the arguments made in connection with the results in fig. 6.1.

6.4.2 How relevant is bulk viscosity in mergers?

In the previous sections we have described how a reactive system near equilibrium can be approximated as a single fluid model with a bulk viscous pressure. One question to tackle is whether this approximation is of any relevance for, for instance, a neutron star merger simulation. In order for the argument to be of interest we need multiple criteria to be met. First, the timescales that we can resolve in the simulation must not include the key timescales for the reactions. If we can resolve the reactions then we should, as the bulk viscosity approximation (as well as the Cattaneo-type law eq. (6.9)) only includes the linear effects (in deviations from thermodynamic equilibrium), whilst the reactions account for the full nonlinear behaviour. Besides, the illustration in fig. 6.1 shows that the approximation would simply break down in a resolved simulation. Second, we would require that the correction to the total pressure from the bulk viscous approximation should not be negligible. If the bulk viscous term due to the reactions is tiny in comparison to the standard fluid pressure then it is pointless to include the reactions in the numerical simulation. Instead we should impose chemical equilibrium directly, and solve without reactions or a bulk viscosity approximation.

Figure 6.2 combines both these criteria in a single plot (based on data for the APR equation of state [195, 210] used in the simulations discussed in [104]). In order for the bulk viscous approximation to be useful the timescale from the simulation must be slower, or at a lower angular frequency, than that given by the peak frequency of the resonance. This peak is defined by A. Therefore every point in the (T, n) plane below a contour of fixed A has reactions acting on time or frequency scales that can (and hence should) be resolved by the numerical simulation. The contours given show that as the numerical grid resolution is improved, more of the (T, n) plane should be modelled by solving directly for the reactions. The bottom two contours in the figure bracket frequencies relevant for bulk gravitational wave generation (1 - 10 kHz), and so must be captured by any numerical simulation. The top contour lines are indicative of the grid frequencies achievable by current simulations (an angular frequency of 10^7s^{-1} roughly corresponds to a grid spacing of 200m) and the state of the art in maybe ten years (an angular frequency of 10^9s^{-1} roughly corresponds to a grid spacing of 2m). However, there remain points at densities above $10^{-2}n_{sat}$ and temperatures above 10MeV where reactions can only be modelled using the bulk viscous approximation, even with the best resolution available in the near future. Current simulations such as [104] do see substantial amounts of matter in the post-merger remnant within this part of phase space, meaning that the bulk viscous approximation will remain necessary for the foreseeable future.

In addition, fig. 6.2 shows the maximum relative magnitude of the bulk viscosity pressure, again in the (T, n) plane, with contours of fixed A overlaid. In this calculation the bulk viscous approximation requires a range of additional approximations, including assuming the Fermi surface approximation is valid in order to compute the rates. These approximations are laid out in detail in appendix C. However, for the purposes of our argument, the key is to note that we are interested in regions of phase space where the bulk viscosity approximation may be valid (above the contours), and where the bulk viscosity makes a significant contribution to the pressure. We see that this holds again in the region with densities above $10^{-2}n_{sat}$ and temperatures above 10MeV.

Figure 6.2 also shows a number of additional features. First, on the right hand side we see a vertical line. As this corresponds to a density just above saturation, it is natural to associate it with the APR phase transition. To the left of this there are two features that start vertically at similar densities. The first one goes up until about 10MeV before turning left and going down to just below 1MeV. This is expected to be an artefact due to the way the equation of state is constructed since different techniques are used in different parts of the parameter space and then cobbled together. The second feature also turns left, but then goes down to about 2MeV before turning right again and going to the top right of the figure. Even though we have not explored this feature in detail, we believe this has to do with points in parameter space where $\partial \beta / \partial Y_e$ goes to zero. We also stress that we see similar features when analogous figures are produced using different (but similarly constructed) equations of state.

As a final important point we note that there is no sharp transition between regions where the effects of the bulk viscous approximation are sizeable and where they are not, when considering contours of fixed \mathcal{A} (which can be linked to the numerical resolution). Unless the numerical resolution can be increased by many orders of magnitude beyond the current state of the art, there will always be regions in spacetime where the bulk viscosity is significant but the approximation itself is debatable. Therefore we have to consider a model that is able to model reactions by directly evolving, for example, the species fraction in some parts of spacetime, makes the bulk viscous pressure approximation in other parts of the spacetime, and transitions between the two appropriately. This is difficult to do correctly.

6.4.3 The impact of large-eddy filtering

We have demonstrated how the Navier-Stokes limit for bulk viscosity due to reactions can be obtained through a multi-scale argument, effectively integrating out the fast timescales of the problem. The argument is quite intuitive. However, the discussion is not yet complete. In the context of numerical simulations we also have to consider other "filtering" aspects. In particular, we need to explore the link to (or conflict with) the large-eddy strategy. This problem is not straightforward. In a sense, the large-eddy approach is "complementary" to the approximate scheme as it aims to represent the regimes where multi-scale arguments do not apply—namely turbulent flows and when all dynamics take place on fast timescales. One may be tempted to view the multi-scale



FIGURE 6.2: The maximum (for each point we assume that $\omega = A$) potential relative contribution of the bulk viscous approximation $\chi^{\text{max}}/p^{\text{eq}}$ at each point in phase space using the APR [210, 195] equation of state. We see that the bulk viscous pressure contribution can be large for most temperatures when $n \gtrsim 10^{-3}n_{\text{sat}}$. However, the bulk viscous approximation should only be used where the reaction rate cannot be resolved by the numerical simulation, which is where the grid frequency is greater than A. Also shown are contours at $A = \{10^3, 10^4, 10^7, 10^9\} \text{ s}^{-1}$ (solid, dashed, dotdash, dot). For current simulations, frequencies of $\sim 10^6 \text{ s}^{-1}$ are resolvable. This shows that the bulk viscous approximation should only be used for $T \gtrsim 10$ MeV, and as the grid resolution improves becomes less necessary.

method and large-eddy space-filtering as "orthogonal". A space filter cuts off short length-scales, while the invariant manifold method integrates out the fast dynamics (removing short timescales). However, the two issues are linked.

An actual numerical implementation introduces an implicit filtering associated with the discretized numerical grid. On a grid with fixed spacing Δx , the implicit filtering requires that any spatial feature on shorter lengthscales cannot be captured and must be modelled by some closure relation, as discussed in [176, 177, 49, 220, 72]. Equally, the CFL bound (linked to causality) imposes that the timestep $\Delta t \propto \Delta x$, and so any physical feature happening on shorter timescales cannot be captured. Thus, increasing the accuracy by modifying the grid spacing automatically means the Δt in fig. 6.1 decreases and therefore the amount of the fast reactions that cannot be captured, shown by the grey area, decreases. There is a direct link between the amount of physics that must be modelled via a spatial filtering and via a time filtering or multi-scale argument.

In order to examine how the filtering associated with the large-eddy strategy impacts on the discussion of bulk viscosity, let us frame the argument using the "fibration" framework developed in chapter 4. Let us first consider what happens when filtering is applied to the continuity equations for baryons and electrons, eqs. (6.4) and (6.5). As discussed in section 4.5.1, the equation for the baryons remains unchanged if (and only if!) we choose to work with the density-weighted four-velocity:

$$\nabla_a n^a = 0 \Longrightarrow \tilde{u}^a \nabla_a \tilde{n} + \tilde{n} \nabla_a \tilde{u}^a = 0 , \qquad (6.58)$$

where $\langle n^a \rangle = \tilde{n}\tilde{u}^a$ defines the filtered baryon number density current (indicated with a tilde). In the case of the electrons, we need to introduce the "electron fraction residual"

$$\tau_{Y_{\rm e}}^a = \langle Y_{\rm e} n^a \rangle - \langle Y_{\rm e} \rangle \langle n^a \rangle , \qquad (6.59)$$

and work with a coarse-grained electron fraction, defined by $\tilde{n}_e = -\tilde{u}_a \langle n_e u^a \rangle$ and leading to

$$\tilde{Y}_{\rm e} = \frac{\tilde{n}_{\rm e}}{\tilde{n}} = -\frac{\tilde{u}_a}{\tilde{n}} \langle n_{\rm e} u^a \rangle .$$
(6.60)

The filtered equation for the electron fraction then becomes⁸

$$\tilde{n}\tilde{u}^a \nabla_a \tilde{Y}_{\rm e} = \langle \Gamma_{\rm e} \rangle - \nabla_a \tilde{v}_{\rm e}^a , \qquad (6.61)$$

where

$$\tilde{Y}_{\rm e} = \langle Y_{\rm e} \rangle - \frac{1}{\tilde{n}} \tilde{u}_b \tau_{Y_{\rm e}}^b, \quad \text{and } \tilde{v}_{\rm e}^a = \tilde{\perp}_b^a \langle n_{\rm e} u^b \rangle = \tilde{\perp}_b^a \tau_{Y_{\rm e}}^b.$$
(6.62)

To avoid confusion, let us explicitly state that the orthogonal projection is here defined with respect to the filtered velocity \tilde{u}^a . The take home message is simple. The large-eddy filtering introduces an effective creation rate in addition to the faithful microphysical one. As the effect of the reactions can be modelled as a bulk-viscosity this may clearly have an impact on the analysis (cf. eq. (6.7)).

Now turn to the remaining equations of motion, e.g. eq. (6.3), which follow from the conservation of the stress-energy-momentum tensor. As this retains exactly the same form as in the perfect fluid case, we can simply draw on the results from sections 4.5.2 and 4.6. The only difference is in the pressure and the Gibbs relation, as we are now

⁸Note that we could choose a different coarse-grained observer in such a way that the effective creation rate is re-absorbed in the macroscopic fluid four-velocity. However, this would come at a cost since the equation for the filtered baryon current would then have an additional diffusion term on the right-hand-side.

considering a reactive system:

$$\langle p \rangle = \langle -\varepsilon + Ts + \mu_{\rm n} n - n_{\rm e} \beta \rangle .$$
 (6.63)

The $n_e\beta$ term was not considered in section 4.7 as the fine-scale model considered there did not account for reactions. Nonetheless, we may simply adapt the same strategy: introduce an effective three-parameter equation of state at the resolved scale, and use it to define the macroscopic thermodynamic variables

$$\frac{1}{\tilde{T}} \doteq \left(\frac{\partial \tilde{s}}{\partial \tilde{\epsilon}}\right)_{\tilde{n}, \tilde{Y}_{e}} , \qquad (6.64a)$$

$$-\frac{\tilde{\mu}_{n}}{\tilde{T}} \doteq \left(\frac{\partial \tilde{s}}{\partial \tilde{n}}\right)_{\tilde{\epsilon}, \tilde{Y}_{e}} - \frac{\tilde{Y}_{e}}{\tilde{n}} \left(\frac{\partial \tilde{s}}{\partial \tilde{Y}_{e}}\right)_{\tilde{n}, \tilde{\epsilon}}, \qquad (6.64b)$$

$$\frac{\tilde{\beta}}{\tilde{T}} \doteq \frac{1}{\tilde{n}} \left(\frac{\partial \tilde{s}}{\partial \tilde{Y}_{e}}\right)_{\tilde{n},\tilde{\varepsilon}} .$$
(6.64c)

With these definitions in hand, we can rewrite the filtered Gibbs relation as

$$\langle p \rangle = -\tilde{\varepsilon} + \tilde{n}\tilde{\mu}_{n} + \tilde{T}\tilde{s} - \tilde{n}\tilde{Y}_{e}\tilde{\beta} + M$$
, (6.65)

with the enhanced closure term

$$M = \left[\left(\langle n\mu_{n} \rangle - \tilde{n}\tilde{\mu}_{n} \right) + \left(\langle Ts \rangle - \tilde{T}\tilde{s} \right) - \left(\langle \varepsilon \rangle - \tilde{\varepsilon} \right) - \left(\langle n_{e}\beta \rangle - \tilde{n}\tilde{Y}_{e}\tilde{\beta} \right) \right] .$$
(6.66)

It makes sense to introduce the effective pressure $\tilde{p} = \langle p \rangle - M$ as this will satisfy a Gibbs relation of the pre-filtered form (but now in terms of the coarse-grained equation of state and the associated variables). Then, because the filtered energy and Euler equations will contain $\langle p \rangle$, the *M* term will enter the final equations as a correction to the pressure—effectively providing a bulk-viscous-like contribution. Let us stress that while we are only providing a minimal discussion of a large-eddy model, this is everything we need here. To make real progress we would need to introduce an explicit closure scheme and perform numerical experiments, both of which go beyond the scope of this thesis.

The essence of the argument is that the large-eddy filtering introduces an "effective bulk-viscosity" contribution to the coarse-grained equations. This happens in (at least) two ways: i) through the residual term M stemming from filtering the Gibbs relation ii) by adding an effective creation rate, and the two effects are not (necessarily) linked as they depend on the introduced closure relations. In particular, the effective creation rate (which follows from the four-divergence of v_e^a) affects the effective restoring term γ , and in turn \mathcal{A} —as is evident from inspection of eq. (6.10). As the resonance frequency in fig. 6.1 is given by $\omega = \mathcal{A}$, and this essentially dictates whether or not the Navier-Stokes approximation is applicable, it makes sense to consider applying the filtering and the multi-scale approach at the same time.

We can intuitively see (and check explicitly) that the coarse-grained equations will have a bulk-viscous contribution stemming from having integrated out the electron fraction degrees of freedom, and one from the filtering. The analytic expressions of these two terms depend on the order with which we take the steps: either we apply the multiscale methods first and then filter, or the other way around. The results are unlikely to be the same. To see this, simply note that the multi-scale/invariant manifold method essentially boils down to an approximation of the equations around the equilibrium surface $Y_e = Y_e^{eq}$. If we take the filtering step first, the notion of equilibrium changes both because the (potentially different) equation of state is evaluated in terms of the coarse grained variables, and because of the effective rate. This highlights the importance of including all the relevant physics when constructing the closure terms in a large eddy model. This is problematic as the best closure terms require direct fine scale numerical simulations, and the analysis in this paper shows these expensive simulations need repeating each time additional physics is added. It is bound to be an expensive endeavour.

6.5 Summary and Outlook

Binary neutron star mergers offer a unique opportunity to explore several extremes of physics, but we need to improve our numerical simulation technology if we want to realize the discovery potential of future gravitational-wave instruments (required to catch the high-frequency dynamics of the merger events). In particular, we need to make sure that efforts to infer the detailed physics are not stumped by systematic errors associated with the numerical implementation. An important step towards realism involves dealing with nuclear reactions. Motivated by this, we have considered the issue of reactions from the perspective of numerical simulations (and the associated limited resolution), aiming to provide "new insights on an old problem". Specifically, we have discussed to what extent it makes sense to capture the net effect of reactions via a bulk viscosity prescription, taking explicitly into account the issue of resolution limitations. In essence, we assessed the impact of the reaction timescales on the way we should frame the modelling, and represent the net effect of reactions.

Our key messages link the reaction timescales to the numerical (grid) timescales. When the reactions are slow the ground truth result is found by evolving the reactions directly. When the timescales are comparable, particularly when the physical timescales are (slightly) faster, this is not numerically practical. Instead the evolution system must be approximated. To leading order the reactions relax the system to equilibrium instantly. However, the error incurred is proportional to the ratio in the timescales. The first order (in the ratio of scales) approximation introduces correction terms that act as a bulk viscosity. This demonstrates how a multi-component reactive system can be approximated as a dissipative single fluid. This bulk viscous structure emerges regardless of whether we formulate the problem in terms of β (which would be natural from a thermodynamics perspective) or Y_e (to connect more directly with simulations).

In a neutron star merger simulation the ratio of scales covers all ranges, with the scales being comparable (and hence a bulk viscosity approximation necessary) particularly in the core shortly after merger (see [12, 10] and cf. fig. 6.2). We also showed that the prescription for the bulk viscosity—either algebraically in a "Navier-Stokes" like fashion, or by providing an equation of motion in a "Cattaneo" or "Israel-Stewart" like fashion—is irrelevant, as they agree (to the order of the approximation) in the regime where the approximation is useful (at low frequencies where the ratio of scales is comparable). Finally, we demonstrated explicitly how the equations of motion are modified when some, but not all, of the scales are comparable. This flags up how the introduction of a bulk viscosity (either directly through approximating reactions, or indirectly through using large-eddy simulation techniques) can conflict with the definition of equilibrium. Consistently accounting for these issues to avoid "double-counting" is possible, as outlined in section 6.3.3, but difficult.

Our main conclusions are intuitive but this is the first time that they have been spelled out in detail. Contrasting theoretical work against the simulations in [104], we find that different regions of the parameter space relevant for mergers would require different prescriptions. In particular, there are regions of the density-temperature phase space where reactions are slow enough that they can (and hence, should) be captured directly, and other regions where they are not—even with the best resolution available, now and in the foreseeable future—with no sharp boundaries between the two regimes. This indicates that, to properly account for reactions, we need to develop numerical codes capable of handling both regimes (reactions fast/slow compared to the resolved timescales), and the transition between them on the fly.

The issue of bulk viscosity is also closely linked to the role of large-eddy filtering which enters the discussion implicitly or explicitly. Noting this, we provided general filtering arguments that help set the stage for further work, and highlighted the coupling between bulk viscosity and large-eddy modelling. A "definitive" prescription for reactive systems will require explicit numerical experiments and the introduction of an appropriate closure model.

As a final comment, let us note that there have been recent papers supporting the idea that bulk viscosity effects can be important for gravitational waves [105, 159], and also papers suggesting it has no impact at all [178, 227]. In particular, different approaches are used to tackle the stiffness issue, where reactions happen faster than the numerical scheme can stably capture them. Even though we would have to know the fineprints of

the different numerical implementations to fully understand the origin of such discrepancies, we can add a final comment that may be part of the story. By checking numerically the formal accuracy of the (bulk viscous) approximation detailed above, we found that the expected accuracy results are recovered when the system is started close to the equilibrium manifold. However, for the accuracy result to hold, additional boundary layer effects need to be accounted for if the system is kicked far out of equilibrium—as suggested, for example, by the simulations of [104].

Chapter 7

Magneto-rotational instability in mergers: a local Newtonian analysis

In the previous chapter we focused on modelling bulk viscosity from the perspective of numerical simulations. We also touched upon the links to turbulence and filtered models. In this chapter, instead, we will focus on the magneto-rotational instability or, as will become clear as we proceed, on more general magneto-shear instabilities. We do so as we often see this kind of instabilities in action—the Kelvin-Helmholtz instability is, for example, responsible for the formation of the billow clouds—and because these provide one of the most important mechanisms for developing and sustaining turbulence. Binary neutron star mergers are no exception. Before we march on, let us point out that the analysis in this chapter is *Newtonian*, in marked contrast with the rest of this work. The reason, quite naturally, is that the nature of the magneto-rotational instability is typically discussed in a Newtonian setting, with almost no exception to the best of our knowledge.

The magneto-rotational instability was discovered by Balbus and Hawley in the early 1990s [31, 108, 33] (linking to earlier ideas from, for example, Chandrasekhar [63] and Velikhov [217]). Due to the fast instability growth rate, this mechanism is considered the most promising candidate for developing/sustaining magneto-hydrodynamic turbulence in accretion disks as well as explaining enhanced angular momentum transfer [33, 198]. The instability is due to an interplay between the magnetic field and a sheared background flow. With few exceptions (see for example [147] and [197]) and due to its "local" nature, the magneto-rotational instability is discussed in the so-called "shearing sheet approximation" [95, 113]. That is, the instability is established in a frame that corotates with a fiducial point in the mid-plane of the undisturbed disk (see also [98]). This is convenient both for analytical studies as well as numerical analysis since local simulations can reach much higher resolution than global ones (see [109, 228] and references therein).

Although originally discussed in the context of accretion disks, the magneto-rotational instability is expected to play a role also in neutron-star mergers [71, 204, 169, 150, 112, 127], especially for sustaining a magneto-turbulent state in the outer envelope of the remnant, where the Kelvin-Helmholtz instability is less significant or, indeed, not active [126]. To assess whether or not the magneto-rotational instability is active and resolved in merger simulations, criteria discussed/established in the context of accretion disks [110, 111, 200] are often used. However, because binary neutron star mergers are highly dynamical environments, framing a discussion of the magneto-rotational instability using criteria that exploits restrictive symmetry conditions might be misleading. Motivated by this, we aim to discuss the impact of relaxing common assumptions—well-motivated in the accretion disks scenario, like an axi-symmetric and circular background flow—on the magneto-shear instability. The results presented in this chapter will be published in [60].

7.1 Background gradients and plane-wave expansion

Let us begin by observing that the magneto-rotational instability is, in some sense, a "global instability analyzed with local tools". The local nature is evident since the instability is established by means of a dispersion relation (hence involves a plane-wave expansion and, by assumption, a short-wavelength approximation). At the same time, one may appreciate the "global nature" of the instability by recalling the key aspects of the instability: The addition of a weak magnetic field turns axisymmetric modes (which would otherwise be hydrodynamically stable) unstable. The global axisymmetry of the background, then, plays a crucial role as the relevant hydrodynamic stability criterion—the Rayleigh criterion [180]—applies to axisymmetric modes only. Although the standard derivation of the instability does not highlight this subtlety, this aspect becomes apparent if we formulate the problem using a co-rotating local frame (cf. the discussion in appendix D).

With these points in mind, let us spell out how we intend to discuss the magnetoshear instability without referring to a given axisymmetric and circular background. Consistently with the shearing box idea [95, 113], the strategy is to zoom in on a small region of fluid—small enough for the analysis to be local but large enough to allow for a meaningful hydrodynamic description. We then set up a local Cartesian frame comoving with the background flow—so that the background velocity vanishes at the origin of the local box. As this frame moves around with the flow—and hence cannot be expected to be inertial—we need to consider the (at this point Newtonian) ideal magneto-hydrodynamics equations in a non-inertial frame. The non-inertial equations will then be perturbed—retaining gradients in the background quantities as explained below—and a local WKB-type dispersion relation will be derived and studied. This way we can account for the effects of a sheared background and its interplay with the magnetic field in a general setting.

Strictly speaking, the plane-wave expansion only makes sense for a homogeneous background —that is, the plane-wave amplitude is assumed to vary over the same scales as the background. At the same time, we know that a sheared background is key to the magneto-rotational instability. Therefore, given any quantity/field a, we first write it as a sum of background plus perturbations

$$a = A + \delta A , \qquad (7.1)$$

and then introduce a WKB-type expansion of the form [213, 28]

$$\delta A = \bar{\delta} \left(\sum_{q=0} \epsilon^q \bar{A}_q \right) e^{i\theta/\epsilon} , \qquad (7.2)$$

with book-keeping parameters $\bar{\delta}$ and ϵ (see also [168]). The former ($\bar{\delta}$) is introduced to measure the relative magnitude of background vs. perturbations, while the latter (ϵ) is given by $\epsilon \approx \lambda/L$ where λ is the typical wavelength of the waves and L is the typical lengthscale over which the wave amplitude, polarization and wavelength vary. Having split the perturbations into amplitude and phase, we follow the standard convention [155] and stick all "post-geometric optics" corrections into the amplitude \bar{A}_q . With this Ansatz, the background equations are obtained by collecting all terms of order $\mathcal{O}(\bar{\delta}^0, \epsilon^0)$, while the perturbation equations are obtained collecting terms of order $\mathcal{O}(\bar{\delta}^1, \epsilon^0)$. Terms of higher order in ϵ correspond to post-geometric optics, while those of higher order in $\bar{\delta}$ represent non-linear perturbations.

Along with this WKB-type Ansatz, we need to introduce the concept of fast and slowly varying quantities. Given a specific choice of coordinates, a quantity is slow in the variable x if A = A(X) where $X = \epsilon x$ while it is fast if A = A(x). Deciding which quantities are fast or slow corresponds to specifying (in a qualitative manner) the background configuration. As an illustration, consider the simple toy problem

$$a(\partial_x b + \partial_x c) = 0 , \qquad (7.3)$$

together with the Ansatz from eq. (7.2). Let us first assume that both *B* and *C* are fast, so that $\partial_x B \approx \mathcal{O}(\bar{\delta}^0, \epsilon^0)$ and similarly for *C*. The background equation is then

$$A\left(\partial_x B + \partial_x C\right) = 0. \tag{7.4}$$

If we instead assume that, say, *B* is fast while *C* is slow, then $\partial_x B \approx \mathcal{O}(\bar{\delta}^0, \epsilon^0)$ while $\partial_x C \approx \mathcal{O}(\bar{\delta}^0, \epsilon)$ and the background equation becomes

$$A\partial_x B = 0. (7.5)$$

Clearly, the two problems are different already at the background level. Let us now turn to the linear perturbations. Because we have explicitly introduced the book-keeping parameter ϵ in eq. (7.2), we take all amplitude terms as well as the phase to be slowly varying. Then, to order $O(\bar{\delta}, \epsilon^0)$ we have

$$\left(A + \bar{\delta}\bar{A}_0 e^{i\theta/\epsilon}\right)\partial_x \left[B + \bar{\delta}\bar{B}_0 e^{i\theta/\epsilon} + C + \bar{\delta}\bar{C}_0 e^{i\theta/\epsilon}\right] = 0.$$
(7.6)

Assuming again that the background quantity *B* is fast, while *C* is slow, the perturbation equation becomes

$$\bar{A}_0(\partial_x B) e^{i\theta/\epsilon} + A\left(\bar{B}_0 \partial_x e^{i\theta/\epsilon} + \bar{C}_0 \partial_x e^{i\theta/\epsilon}\right) = 0.$$
(7.7)

Next, Taylor expanding the phase—which is slowly varying—we get

$$\frac{\theta(x)}{\epsilon} \approx \frac{\theta(0)}{\epsilon} + \frac{\partial\theta}{\partial X}\Big|_{X=0} x + \dots = \theta(0)/\epsilon + k_x x + \mathcal{O}(\epsilon) , \qquad (7.8)$$

where we define the wave-vector $k_x = \partial \theta / \partial X$ from the first order term in the expansion, while the overall constant can be neglected. Then, introducing an analogous expansion for the fast background gradients $\partial_x B(x) = \partial_x B(0) + \mathcal{O}(\epsilon)$ we end up with

$$\bar{A}_0(\partial_x B) + A \left(ik_x \bar{B}_0 + ik_x \bar{C}_0 \right) = 0 , \qquad (7.9)$$

where both $\partial_x B$ and A are evaluated at a point (conveniently chosen as the origin of the coordinate system). Therefore, if all background quantities are "slow", we get back the dispersion relation we would have obtained ignoring all background gradients. This is quite intuitive. However, the strategy also allows us to account for the impact that "fast" background gradients have on the dispersion relation. In short, as long as these terms are treated as constants, we may retain them and work out a dispersion relation in the usual way.

7.2 The slowly evolving background

The starting point for any hydrodynamic perturbation analysis is the choice/identification of a stationary background flow configuration, which is then perturbed in order to establish stability (or not). Here, we want to frame the analysis of the magneto-shear instability without considering a specific background configuration (with constraining symmetries) stated from the outset. Nonetheless, we need to clarify how we can refer to a suitable "background" in highly dynamical environments like binary neutron star mergers. Given real numerical simulation data, this discussion will inevitably involve some kind of filtering operation. Anticipating that this can be done in a meaningful way, we consider perturbations evolving rapidly with respect to the evolution timescale of an unspecified "background" flow.
To make this statement more precise, let us consider the inertial ideal MHD equations and introduce reference values for each quantity (indicated with an "r" subscript) such as $\rho = \rho_r \tilde{\rho}$. We introduce the (dimensionless) Strouhal, Mach, Froude and magnetic interaction numbers as

$$\varepsilon_{\rm St} = \frac{l_r}{t_r v_r}$$
, $\varepsilon_{\rm Ma} = \frac{v_r}{c_r}$, $\varepsilon_{\rm Fr} = \frac{v_r}{\sqrt{\Phi_r}}$, $\varepsilon_B = \frac{B_r^2}{\mu_0 \rho_r v_r^2}$, (7.10)

where l_r , t_r , v_r are characteristic lengthscale, timescale and velocity (respectively) while B_r , ϕ_r , ρ_r are reference values for the magnetic field, gravitational potential and density and c_r is the (adiabatic) speed of sound. This way, the non-dimensional inertial ideal MHD equations read (now dropping the "tilde"s for notational clarity)

$$\varepsilon_{\rm St} \,\partial_t \rho = -\rho \nabla_i v^i - v^i \nabla_i \rho \,\,, \tag{7.11a}$$

$$\varepsilon_{\rm St} \,\partial_t B^i = -v^j \nabla_j B^i + B^j \nabla_j v^i - B^i \nabla_j v^j \,, \tag{7.11b}$$

$$\varepsilon_{\rm St} \partial_t v^i = -v^j \nabla_j v^i - \frac{1}{\epsilon_{\rm Ma}^2} \frac{1}{\rho} \nabla^i \rho - \frac{1}{\epsilon_{\rm Fr}^2} \nabla^i \Phi - \epsilon_B \frac{1}{\rho} \left[B^j \nabla_j B^i - \nabla^i \left(\frac{B^2}{2} \right) \right] .$$
(7.11c)

From this we see that a generic flow configuration can be considered slowly evolving (in time) as long as the corresponding Strouhal number is small. In practice, given a characteristic lengthscale l_r and velocity v_r of a generic flow, we consider disturbances evolving on timescales t_r such that $\epsilon_{St} \ll 1$ —over which the background can be effectively taken as stationary. In turn, this determines the time-scales over which we expect the following results to be reliable.

7.2.1 Velocity gradient decomposition

In the following we will consider the impact that gradients in the background flow velocity have on the time evolution of perturbations. It is then convenient to introduce the standard decomposition of the velocity gradient into expansion, shear and vorticity. Even though this has been used many times before in this thesis, it makes sense to write it down explicitly here as we are working at a Newtonian setting. The three vector velocity gradient decomposition in the Newtonian case reads

$$\nabla_i v_j = \frac{1}{3} \theta g_{ij} + \sigma_{ij} + \omega_{ij} , \qquad (7.12)$$

where

$$\theta = \nabla_i v^i , \quad \sigma_{ij} = \nabla_{(i} v_{j)} - \frac{1}{3} \theta g_{ij} = \frac{1}{2} \left(\nabla_i v_j + \nabla_j v_i \right) - \frac{1}{3} \theta g_{ij} , \qquad (7.13a)$$

and

$$\omega_{ij} = \nabla_{[i} v_{j]} = \frac{1}{2} \left(\nabla_i v_j - \nabla_j v_i \right) . \tag{7.13b}$$

In order to bring out the magneto-shear nature of the instability, we will consider the impact of having a background with non-negligible shear and vorticity separately. We will, however, not consider the impact of a background expansion rate as exact non-linear results are sufficient to predict this. In fact, due to the Alfvén theorem, we know that the magnetic intensity must grow in a (ideal magneto-)fluid undergoing compression as the field lines are squeezed together. Similarly, the field will get weaker in an expanding fluid. In essence, we expect—and have verified explicitly—this non-linear prediction to emerge in the analysis as a generic "instability". The background magnetic field cannot grow in time as it is assumed to be slowly evolving by construction, so the required growth must be represented by perturbations.

In what follows, we will first analyse the problem analytically, and come back to discuss the link to/relevance for numerical simulations in the concluding remarks in section 7.6. Before we move on though, it is useful to take a brief detour and consider a realization of a flow with only non-negligible shear. Because we are interested in flows that are "slowly evolving" we can start by assuming that $v^i = v^i(\epsilon t, x, y, z)$ and suppress the time dependence in the following. We then take the velocity vector as mainly two-dimensional, specifically in the x - y plane of a set of local Cartesian coordinates

$$\mathbf{v} = v^{x}(x, y, z)\hat{x} + v^{y}(x, y, z)\hat{y} + \mathcal{O}(\epsilon) , \qquad (7.14)$$

where, in order to make sure the expansion is small, we take $\partial_x v^x = -\partial_y v^y + O(\epsilon)$. The shear matrix is then given by

$$\sigma = \begin{pmatrix} \partial_x v^x & \frac{1}{2} \left(\partial_x v^y + \partial_y v^x \right) & \frac{1}{2} \partial_z v^x \\ \frac{1}{2} \left(\partial_x v^y + \partial_y v^x \right) & -\partial_x v^x & \frac{1}{2} \partial_z v^y \\ \frac{1}{2} \partial_z v^x & \frac{1}{2} \partial_z v^y & 0 \end{pmatrix} + \mathcal{O}(\epsilon) , \qquad (7.15)$$

while the curl of v becomes

$$\nabla \times \mathbf{v} = -\left(\partial_z v^y\right) \hat{x} + \left(\partial_z v^x\right) \hat{y} + \left(\partial_x v^y - \partial_y v^x\right) \hat{z} .$$
(7.16)

This has to be of $\mathcal{O}(\epsilon)$ for background flows with only non-negligibile shear, in which case

$$\partial_z v^x = \partial_z v^y = \mathcal{O}(\epsilon) , \qquad \partial_y v^x = \partial_x v^y + \mathcal{O}(\epsilon),$$
(7.17)

and as result the determinant of the shear matrix vanishes (more precisely, is of order $\mathcal{O}(\epsilon)$). This is equivalent to saying that two eigenvalues of the shear matrix are opposite and the third is zero (to order $\mathcal{O}(\epsilon)$). In essence, a mainly two-dimensional flow with negligible expansion and vorticity is characterized by a shear matrix with one zero eigenvalue, and hence a vanishing determinant. This will turn out to be a useful observation later on. We also note that having negligible expansion (although relevant for the present analysis) is not strictly necessary to the argument.

7.3 Non-inertial equations and the local frame

The ideal magneto-hydrodynamic equations above hold in an inertial frame. As an observer locally comoving with the fluid cannot be expected to be inertial (in general) we need to consider the equations according to a non-inertial observer. The non-inertial MHD equations have been derived earlier, so that we will here build on the results of section 5.2. However, as we are interested in a local analysis, we now take a step further and make contact with the concept of local frame associated with an observer (see [101, 155]). In doing so we, for a moment, go back to use concepts and notations that are common in relativity. This is required as we need to make contact with the non-inertial induction equation derived in section 5.2 using a relativistic language. Given an observer worldline with tangent U^a , the local frame is constructed by considering three spatial unit vectors that complete U^a to an orthonormal basis on the tangent space at a point (see appendix A for more details). These three spatial vectors are then transported along the worldline as

$$\nabla_U e_{\hat{a}} = \Omega^{\hat{b}}_{\ \hat{a}} e_{\hat{b}} , \quad \Omega_{\hat{a}\hat{b}} = U_{\hat{a}} a_{\hat{b}} - a_{\hat{a}} U_{\hat{b}} - \epsilon_{\hat{a}\hat{b}\hat{c}\hat{d}} U^{\hat{c}} W^{\hat{d}} , \qquad (7.18)$$

where $a^{\hat{a}}$ is the four-acceleration of U^{a} (an intrinsic property of the worldline) and W^{d} is the *arbitrary* four-rotation of the local frame. Let us then look at the non-inertial induction equation, which we report here for convenience¹

$$\perp^{\hat{a}\hat{b}} \dot{b}_{\hat{b}} + D_{\hat{b}}(v^{\hat{b}}b^{\hat{a}}) - D_{\hat{b}}(v^{\hat{a}}b^{\hat{b}}) = \left(\sigma^{\hat{a}\hat{b}} - \omega^{\hat{a}\hat{b}} - \frac{2}{3}\theta \perp^{\hat{a}\hat{b}}\right) b_{\hat{b}} .$$
(7.19)

Focusing on the first term, and using eq. (7.18)

$$\dot{b}_{\hat{b}} = (\nabla_{U}b)_{\hat{b}} = u^{\hat{c}}\partial_{\hat{c}}b_{\hat{b}} + (a^{\hat{c}}b_{\hat{c}}) u_{\hat{b}} + \varepsilon^{u}_{\hat{b}\hat{c}\hat{c}}W^{\hat{c}}b^{\hat{c}} .$$
(7.20)

The second term vanishes due to the orthogonal projection, while²

$$\perp^{\hat{a}\hat{b}}\dot{b}_{\hat{b}} + \omega^{\hat{a}\hat{b}}b_{\hat{b}} = \perp^{\hat{a}\hat{b}}\left(u^{\hat{c}}\partial_{\hat{c}}b_{\hat{b}}\right) .$$
(7.21)

In practice, the term involving the four rotation of the frame drops out of the induction equation. We also note that, because we are now considering the non-inertial equations in the local frame of a single observer, there is no shear or expansion and the induction equation in the Newtonian limit simplifies to

$$\partial_t B^{\hat{i}} + \nabla_{\hat{j}} \left(v^{\hat{j}} B^{\hat{i}} - v^{\hat{i}} B^{\hat{j}} \right) = 0 .$$
 (7.22)

¹We drop the last two terms in equation eq. (5.24), cf. discussion at the end of section 5.2.

²Note that we are here identifying the vorticity of the fibration observer with the four-rotation of the local frame chosen.

At the Newtonian level then, the induction equation in the local frame of a generic observer retains the same form as for an inertial one. This is similar to the case of the Lorentz force (entering the Euler equation) and the Ampère law. Let us nonetheless stress that additional terms involving the four-acceleration of the observer worldline do appear at the special relativistic level, even though working with the ideal MHD induction equation may be somehow controversial in a fully-relativistic regime (cf. discussion in section 5.2).

When it comes to the non-inertial terms in the Euler equations, these are obviously well known: we have to account for fictitious accelerations. We refer to [101] for a rigorous derivation of the fictitious forces in special relativity, showing also how additional terms involving the observer four-acceleration enter the relativistic expressions. We also stress that working with a rotating or non-rotating local frame is entirely a matter of choice (see [155]). At the local Newtonian level then, we can always get rid of the non-inertial terms associated with the frame rotation. As the linear acceleration of the observer drops out of the perturbation equations, this means we can effectively work with the inertial equations.

We conclude this section by noting that, as previously anticipated, some kind of filtering operation is key to separate between background and fluctuations in a highly dynamical environment. Postponing a discussion of this to section 7.6, let us simply note at this point that the notion of local frame discussed here is clearly linked to the covariant filtering procedure discussed in chapter 4.

7.4 Going back to hydrodynamics

As briefly hinted at in section 7.1, the magneto-rotational instability relies on the *hy-drodynamic* stability of axisymmetric modes. The generic instability problem is more involved. If we relax the symmetry assumptions on the background, we need to consider the fact that hydrodynamic shear flows tend to be unstable. That is, we expect to find instabilities to appear already at the hydrodynamic level. Clearly, such instabilities would be affected by a magnetic field but not caused by it in the first place. This is an important distiction seeing as the magneto-rotational instability is *due to* the presence of the magnetic field.

With this observation in mind, let us first consider the fluid problem. This will be useful for two reasons: First, it will allow us to get a better grasp on the magnetic field impact on the instability. Second, it will allow us to make contact with the Rayleigh criterion (and ultimately the magneto-rotational instability). As the fluid problem is much simpler than the magneto-fluid one, we will study the case where both shear and vorticity gradients are retained, and also discuss the impact of shear viscosity—either of microphysical origin or due to filtering as in the Smagorinsky model (or, more in general, in the so-called eddy-viscosity type models, see section 4.7.2). Shear viscosity is introduced in the usual way (see, for example, Landau and Lifshitz [134]), and the shear viscosity coefficient η will be considered constant consistently within the local analysis.

Before we move on to discuss the perturbation equations and the resulting dispersion relation(s) though, it is worth stressing that, in many situations of interest the relevant dynamics is either sub- or supersonic. As such, for these problems it is worth considering models that filter out modes that are either faster or slower than the sound waves. This can be done starting from a fully compressible dispersion relation and taking either of two limits: either we assume the speed of sound to be very large, in which case the model becomes sound-proof (we point to [216] for more details), or very small. In the following, we typically work in the sound-proof limit, noting that the MRI is commonly discussed within the so-called Boussinesq approximation [34], thus removing fast magneto-sonic waves [31].

Starting from the continuity equation, perturbing it and introducing the plane-wave expansion we readily obtain

$$\partial_t \delta \rho + \delta \rho \nabla_i v^i + v^i \nabla_i \delta \rho + \rho \nabla_i \delta v^i = 0 \Longrightarrow -i\omega \delta \rho + i\rho k_i \delta v^i = 0 , \qquad (7.23)$$

where ω and k_i are defined as in section 7.1. Note that we set $v^i = 0$ as we evaluate the relation at the centre of the local box, and assume that the background expansion rate $\nabla_i v^i$ can be neglected. In a similar fashion, the perturbed Euler including a shearviscous term gives

$$\partial_t \delta v_i + \delta v^j \nabla_j v_i + \frac{1}{\rho} \nabla_i \delta P - \delta \left(\eta \nabla^j \tau_{ji} \right) = 0$$

$$\implies -i\omega \delta v_i + i \frac{c_s^2}{\rho} k_i \delta \rho + \sigma_{ij} \delta v^j + \epsilon_{ijk} W^j \delta v^k - \delta \left(\eta \nabla^j \tau_{ji} \right) = 0 , \quad (7.24)$$

where τ_{ji} is the rate-of-strain/shear tensor and $W^i = 1/2\epsilon^{ijk}\omega_{jk}$. Working this out we retained gradients in the background flow only, used the velocity gradient decomposition (section 7.2.1), introduced the adiabatic speed of sound $c_s^2 = \partial P/\partial \rho$, and considered the gravitational potential to be externally sourced (hence neglecting its perturbations).

In order to derive the dispersion relation and study the effects of a sheared background, it is convenient to choose a basis that is adapted to it. Because the shear is a trace-free symmetric matrix, we know there exists a basis (in the tangent space) whereby

$$\sigma^{ij} = \operatorname{diag}\left(\sigma_1, \sigma_2, -(\sigma_1 + \sigma_2)\right) . \tag{7.25}$$

We will make use of this basis to write down the coefficient matrix of the linearized system. Before doing so, however, it is reasonable to wonder whether this change of basis has any impact on the perturbation equations. We are, always free to choose a basis in the tangent space that is not associated with the coordinates chosen, but this (in general) introduces additional terms in the covariant derivative. Let us spell out why this is not the case here. Working with a non-coordinate basis, we need to account for spin-coefficients when a derivative acts on vectors and tensors. The spin coefficients are given by the sum of two terms (see the formula in the Notation chapter, or [50] for more details). The first involves the Christoffel symbols associated with the coordinates chosen, and thus vanish as we are working with a non-rotating Cartesian frame. The second term instead stems from the fact that the change of basis matrix (translating the coordinates base into the shear-adapted one) may be different from point to point. In the context of this analysis, however, we are looking at scales smaller than those over which background quantities vary. In essence, also this second term vanishes as the shear matrix is (by construction) constant over the local region of fluid we are zooming on.

Working in the shear adapted basis, we write the coefficients matrix of the linearized system as

$$\mathbf{M} = \begin{pmatrix} -\omega & \rho k_1 & \rho k_2 & \rho k_3 \\ \frac{c_s^2}{\rho} k_1 & -\omega - i\sigma_1 - i\eta L_1 & iW^3 - \frac{i}{6}\eta k_1 k_2 & -iW^2 - \frac{i}{6}\eta k_1 k_3 \\ \frac{c_s^2}{\rho} k_2 & -iW^3 - \frac{i}{6}\eta k_2 k_1 & -\omega - i\sigma_2 - i\eta L_2 & iW^1 - \frac{i}{6}\eta k_2 k_3 \\ \frac{c_s^2}{\rho} k_3 & iW^2 - \frac{i}{6}\eta k_3 k_1 & -iW^1 - \frac{i}{6}\eta k_3 k_2 & -\omega + i\sigma_2 + i\sigma_2 - i\eta L_3 \end{pmatrix} , \quad (7.26)$$

where $L_1 = \frac{2}{3}k_1^2 + \frac{1}{2}k_2^2 + \frac{1}{2}k_3^2$ and L_2 , L_3 are similarly defined. The dispersion relation is computed taking the determinant of this matrix and equating it to zero. In order to keep the discussion as general as possible (i.e. without having to refer to a specific background configuration) we will decompose the coefficients of the characteristic polynomial in terms of scalars built from background quantities. In the simplest cases this can be done "by eye", but the procedure can easily become quite messy. The logic is nonetheless simple: we group the different terms in each coefficients according to the power of the various background quantities, for example we group all the terms quadratic in the shear and wave-vector components. We then build all the possible scalars that are quadratic in the shear and wave-vector, and look for the correct linear combination of them. This logic can be easily implemented on a computer algebra program such as Mathematica³. We now discuss the dispersion relations obtained by retaining only shear terms, both shear and viscous terms, and lastly shear and vorticity terms. Before doing so, we observe that the coefficients will involve scalars constructed

³https://www.wolfram.com/mathematica/

from the shear matrix only. As any 3 \times 3 matrix, the shear matrix σ has three invariants

$$I_1 = \text{Tr}(\sigma) , \quad I_2 = \frac{1}{2} \left[\text{Tr}(\sigma^2) - (\text{Tr}(\sigma))^2 \right] , \quad I_3 = \det(\sigma) ,$$
 (7.27)

related via the Cayley-Hamilton theorem as

$$\sigma^{3} - I_{1}\sigma^{2} + I_{2}\sigma - I_{3}\mathbb{I} = 0 , \qquad (7.28)$$

where I is the 3 × 3 identity matrix. Because the shear matrix is trace-free, we will write the coefficients in terms of 1/2Tr(σ^2) and det(σ).

We begin our analysis by considering a background with negligible vorticity, and set the shear viscosity to zero. The resulting dispersion relation is

$$\omega^4 + a_2\omega^2 + a_1\omega + a_0 = 0 , \qquad (7.29)$$

with

$$a_2 = -c_s^2 k^2 + \frac{1}{2} \operatorname{Tr}(\sigma^2)$$
, (7.30a)

$$a_1 = i \left[c_s^2 \sigma_{ij} k_i k_j - \det(\sigma) \right] , \qquad (7.30b)$$

$$a_{0} = c_{s}^{2} \left[\sigma_{ij}^{2} k_{i} k_{j} - \frac{1}{2} \operatorname{Tr}(\sigma^{2}) k^{2} \right] .$$
 (7.30c)

We then take the sound-proof limit (i.e. we retain only terms proportional to the speed of sound) and consider the case det(σ) = 0, looking for modes such that $\sigma^{ij}k_j$ = 0. Recalling that, as discussed in section 7.2.1, a mainly two dimensional flow with negligible vorticity is characterized by having a shear matrix with vanishing determinant—that is det(σ) ~ $O(\epsilon)$ —and noting that we can choose the orientation of the local axes in such a way that the background flow is, say, along the \hat{x} , \hat{y} directions, we can always consider the determinant to be zero. This means that there always exists a wave-vector living in the eigen-space corresponding to the zero eigenvalue. Then we end up with⁴

$$\omega^{2} = -\frac{1}{2} \operatorname{Tr}(\sigma^{2}) \Longrightarrow \omega = \pm i \sqrt{\frac{1}{2} \operatorname{Tr}(\sigma^{2})} .$$
 (7.31)

That is such modes are non-propagating, and half of them are unstable with a growth rate independent of the wave-vector. Next, we consider wave-numbers such that $\sigma_{ij}k_ik_j = 0$, noting that such modes will always exist. In the shear-adapted basis they are characterized by $k^1 = k^2 = k^3$ if the determinant is not vanishing (that is $s_1 \neq s_2$), and $k_1 = k_2$ when it does. It follows that for such modes

$$-\frac{1}{2}\text{Tr}(\sigma^2) + \sigma_{ij}^2 \hat{k}^i \hat{k}^j = -\frac{1}{6}\text{Tr}(\sigma^2) , \qquad (7.32)$$

⁴We note that we obtain the same dispersion relation also in the opposite limit where the speed of sound is tiny.

where $\hat{k} = \mathbf{k}/|\mathbf{k}|$. We then obtain

$$\omega^{2} = -\frac{1}{6} \operatorname{Tr}(\sigma^{2}) \Longrightarrow \omega = \pm i \sqrt{\frac{1}{6} \operatorname{Tr}(\sigma^{2})} .$$
(7.33)

These modes are also non-propagating, and half of them are unstable with a (constant) growth rate about a factor of 2 smaller. As the dispersion relation is quadratic (in the sound-proof limit), we can explicitly solve it and confirm the expectation (and well known fact) that shearing flows are generically unstable.

Let us now build on this and discuss how vorticity and shear viscosity impact on the generic instability of sheared flows. First we consider the case where the background has negligible vorticity but non-vanishing shear viscosity. As sanity check, we observe that if we also set the background shear to be negligible, and take the sound-proof limit⁵ we obtain

$$\omega^{2} + i\eta k^{2}\omega - \frac{1}{4}\eta^{2}(k^{2})^{2} = \left(\omega + \frac{i}{2}\eta k^{2}\right)^{2} = 0, \qquad (7.34)$$

with stable roots provided $\eta > 0$. We recall that when viscosity is of microphysical origin, $\eta > 0$ follows from the second law of thermodynamics (see [134, 21]). If viscosity is instead due to filtering, a positive value of η corresponds to an eddy-type model where energy is cascading to smaller/unresolved scales (see [193, 136, 143]). With this observation in mind, let us go back to the case with both both shear and viscosity, in which case the dispersion relation is

$$\omega^4 + a_3\omega^3 + a_2\omega^2 + a_1\omega + a_0 = 0 , \qquad (7.35)$$

with

$$a_3 = \frac{5}{3}i\eta k^2 , (7.36a)$$

$$a_{2} = -c_{s}^{2}k^{2} + \frac{1}{2}\operatorname{Tr}(\sigma^{2}) - \frac{1}{12}\eta \left[11\eta (k^{2})^{2} - 2\sigma_{ij}k_{i}k_{j}\right] , \qquad (7.36b)$$

$$a_0 = c_s^2 \left[\sigma_{ij}^2 k_i k_j - \frac{1}{2} \operatorname{Tr}(\sigma^2) k^2 - \frac{1}{2} \eta k^2 \sigma_{ij} k_i k_j + \frac{1}{4} \eta^2 (k^2)^3 \right] , \qquad (7.36c)$$

and

$$a_{1} = ic_{s}^{2} \left[\sigma_{ij}k_{i}k_{j} - \eta(k^{2})^{2}\right] + i\left\{-\frac{1}{6}\left[\sigma_{ij}^{2}k_{i}k_{j} - 2\operatorname{Tr}(\sigma^{2})k^{2}\right] + \frac{1}{12}\eta^{2}k^{2}(\sigma_{ij}k_{i}k_{j}) - \frac{1}{6}\eta^{3}(k^{2})^{3} - \det(\sigma)\right\}.$$
 (7.36d)

⁵We have also verified, using the Routh-Hurwitz criterion (see [131] and appendix E) that the same result holds true in general, not only in the sound-proof limit.

As before, we first consider modes such that $\sigma_{ij}\hat{k}_j = 0$, whose dispersion relation is

$$\omega^{2} + i\eta k^{2}\omega - \frac{1}{4} \left(\eta^{2} (k^{2})^{2} - 2\text{Tr}(\sigma^{2}) \right) = 0.$$
(7.37)

Assuming $\eta > 0$, stability corresponds to

$$\eta^2 (k^2)^2 - 2 \operatorname{Tr}(\sigma^2) > 0.$$
 (7.38)

In essence, comparing this to eq. (7.31) we see that viscosity tends to stabilize shearunstable modes, with a larger impact at smaller scales. This makes intuitive sense. Next, consider modes such that $\sigma_{ij}\hat{k}_i\hat{k}_j = 0$, whose dispersion relation is

$$\omega^{2} + i\eta k^{2}\omega - \frac{1}{12} \left(3\eta^{2} (k^{2})^{2} - 2\operatorname{Tr}(\sigma^{2}) \right) = 0 , \qquad (7.39)$$

and we made use of eq. (7.32). As before, these modes—to be compared with their counterparts in eq. (7.33)—are also stable (assuming $\eta > 0$) provided the last term in the previous equation is negative. That is, provided the wave-number is sufficiently large. We have verified that the same trend is true for generic wavevectors. In essence, we learn (as one may have expected) that shear viscosity generically slows the growth rate of unstable shear modes, and stabilises modes with small enough wavelengths.

Turning to the case where the background has non-negligible vorticity and shear, the dispersion relation is

$$\omega^4 + a_2 \omega^2 + a_1 \omega + a_0 = 0 , \qquad (7.40)$$

with

$$a_2 = -c_s^2 k^2 - \mathbf{W}^2 + \frac{1}{2} \operatorname{Tr}(\sigma^2)$$
, (7.41a)

$$a_1 = i \left[c_s^2 \sigma_{ij} k_i k_j - \det(\sigma) - \sigma_{ij} W_i W_j \right] , \qquad (7.41b)$$

$$a_0 = c_s^2 \left[\sigma_{ij}^2 k_i k_j - \frac{1}{2} \operatorname{Tr}(\sigma^2) k^2 + (\mathbf{k} \cdot \mathbf{W})^2 \right] .$$
 (7.41c)

Taking the sound-proof limit, we first observe that the fastest growing modes encountered before, namely those characterized by $\sigma_{ij}\hat{k}_j = 0$ are not guaranteed to exist anymore, as the determinant of the shear matrix cannot be assumed to be negligible in general (see section 7.2.1). Should these modes exist, though, their dispersion relation would be

$$\omega^{2} = -\frac{1}{2}\operatorname{Tr}(\sigma^{2}) + (\hat{k} \cdot \mathbf{W})^{2} \Longrightarrow \omega = \pm i\sqrt{-\frac{1}{2}\operatorname{Tr}(\sigma^{2}) + (\hat{k} \cdot \mathbf{W})^{2}}, \qquad (7.42)$$

and we see, comparing this to eq. (7.31), that vorticity tends to stabilize them. We also observe that—in contrast to shear viscosity—vorticity affects all such modes by reducing their growth rate in a way that does not depend on their wave-number (although the direction of propagation is important). Next—and also because the modes we just

looked at may not exist—we consider modes such that $\sigma_{ij}\hat{k}_i\hat{k}_j = 0$ obtaining

$$\omega^{2} = -\frac{1}{6} \operatorname{Tr}(\sigma^{2}) + (\hat{k} \cdot \mathbf{W})^{2} .$$
 (7.43)

Comparing this to eq. (7.33), we observe again that vorticity tends to stabilize such modes in a way that does not depend on their wave-number. We have verified that the same trend is also true for generic wave-vectors. As a final point, it is easy to verify that the case with only background vorticity is generally stable (not only in the sound-proof limit).

In summary, a sheared background flow is generically unstable already at the hydrodynamic level, which is a well-known fact. However, we have considered the impact that shear viscosity and/or vorticity have on the instability of the possible hydrodynamic modes. The results show that shear viscosity tends to weaken the instability in general, with larger effects for larger wave-numbers. Meanwhile, vorticity has a stabilizing effect which does not depend on the wave-number. Finally, let us also point to appendix D.2 where we show that the general dispersion relation derived here is shown to encompass the classic Rayleigh stability criterion.

7.5 Magneto-shear instability in the local frame

Having explored the hydrodynamic case, let us perturb the corresponding MHD equations and study the impact of the magnetic field on the generic shear instabilities we encountered. We consider a barotropic equation of state and retain gradients in the background velocity only, as we want to focus on the magneto-shear nature of the instability (cf. [108, 200]). The continuity equation is obviously unchanged, while the perturbed Euler equation becomes

$$\partial_t \delta v_i + \delta v^j \nabla_j v_i + \frac{1}{\rho} \nabla_i \delta P + \frac{1}{\mu_0 \rho} \left[B_j \nabla_i \delta B^j - B^j \nabla_j \delta B_i \right] = 0$$

$$\implies -i\omega \delta v_i + i \frac{c_s^2}{\rho} k_i \delta \rho + \frac{i}{\mu_0 \rho} \left[(B_j \delta B^j) k_i - (B^j k_j) \delta B_i \right] + \sigma_{ij} \delta v^j + \epsilon_{ijk} W^j \delta v^k = 0.$$
(7.44)

Finally, the perturbed induction equation is

$$\partial_t \delta B^i + B^i \nabla_j \delta v^j - B^j \nabla_j \delta v^i - \delta B^j \nabla_j v^i + \delta B^i \nabla_j v^j = 0$$

$$\implies -i\omega \delta B^i + iB^i (k_j \delta v^j) - i(B^j k_j) \delta v^i - \sigma^{ij} \delta B_j - \epsilon^{ijk} W_k \delta B_j + \frac{2}{3} \theta \delta B^i = 0. \quad (7.45)$$

We will now discuss the linearized system that follows from these equations. In order to keep the discussion tidy, we will first recap the mode analysis for the homogeneous case and then move on to consider a background with non negligible shear and vorticity (separately).

7.5.1 Homogeneous background: a recap

In order to derive the fully compressible dispersion relation for the homogeneous case, we first re-scale the magnetic field as

$$\mathbf{v}_A \doteq \frac{\mathbf{B}}{\sqrt{\mu_0 \rho}}, \quad \delta \mathbf{v}_A \doteq \frac{\delta \mathbf{B}}{\sqrt{\mu_0 \rho}},$$
 (7.46)

and introduce a convenient basis { \hat{v}_A , \hat{q} , \hat{s} } where $\hat{v}_A = \mathbf{v}_A / |\mathbf{v}_A|$ while \hat{q} , \hat{s} complete it to an orthonormal basis. For instance, assuming \mathbf{v}_A is not aligned with \mathbf{k} we can construct it as

$$\mathbf{q} = \mathbf{k} - (\mathbf{k} \cdot \hat{v}_A) \,\hat{v}_A \,, \qquad \hat{q} = \frac{\mathbf{q}}{|\mathbf{q}|} \,, \qquad \hat{s} = \hat{v}_A \times \hat{q} \,, \tag{7.47}$$

so that⁶

$$\mathbf{k} = k^{v_A} \hat{v_A} + k^q \hat{q} \ . \tag{7.48}$$

The coefficient matrix of the linearized system can then be written as (cf. eqs. (7.23), (7.44) and (7.45) and ignore background vorticity and shear)

$$\mathbf{M} = \begin{pmatrix} \mathbf{A} & \mathbf{C} \\ \mathbf{C}^{\top} & \mathbf{D} \end{pmatrix} , \qquad (7.49)$$

with

$$\mathbf{A} = \begin{pmatrix} -\omega & \rho k_{v_A} & \rho k_q & 0\\ \frac{c_s^2}{\rho} k_{v_A} & -\omega & 0 & 0\\ \frac{c_s^2}{\rho} k_q & 0 & -\omega & 0\\ 0 & 0 & 0 & -\omega \end{pmatrix}, \quad \mathbf{C} = \begin{pmatrix} 0 & 0 & 0\\ 0 & 0 & 0\\ bk_q & -bk_{v_A} & 0\\ 0 & 0 & -bk_{v_A} \end{pmatrix}, \quad (7.50a)$$

and

$$\mathbf{D} = \begin{pmatrix} -\omega & 0 & 0\\ 0 & -\omega & 0\\ 0 & 0 & -\omega \end{pmatrix} \,. \tag{7.50b}$$

As D is clearly invertible, we can reduce M into factors via the Schur complement

$$\begin{pmatrix} \mathbf{A} & \mathbf{C} \\ \mathbf{C}^{\top} & \mathbf{D} \end{pmatrix} = \begin{pmatrix} \mathbf{I}_4 & \mathbf{C}\mathbf{D}^{-1} \\ \mathbf{0}_{4\times3} & \mathbf{I}_3 \end{pmatrix} \begin{pmatrix} \mathbf{A} - \mathbf{C}\mathbf{D}^{-1}\mathbf{C}^{\top} & \mathbf{0}_{4\times3} \\ \mathbf{0}_{3\times4} & \mathbf{D} \end{pmatrix} \begin{pmatrix} \mathbf{I}_4 & \mathbf{0}_{4\times3} \\ \mathbf{D}^{-1}\mathbf{C}^{\top} & \mathbf{I}_3 \end{pmatrix} , \quad (7.51)$$

and then compute the determinant as

$$det(\mathbf{M}) = det(\mathbf{D}) det(\mathbf{A} - \mathbf{C}\mathbf{D}^{-1}\mathbf{C}^{\top}) .$$
 (7.52)

⁶If the wave-vector is along the background magnetic field we just have to set $k^q = 0$ in the following.

The resulting dispersion relation is

$$-\omega\left(\omega^{2}-\left(\mathbf{v}_{A}\cdot\mathbf{k}\right)^{2}\right)\left[\omega^{4}-\left(v_{A}^{2}+c_{s}^{2}\right)k^{2}\omega^{2}+c_{s}^{2}k^{2}\left(\mathbf{v}_{A}\cdot\mathbf{k}\right)^{2}\right]=0,\qquad(7.53)$$

where the roots of the quadratic polynomial correspond to Alfvén waves, while those of the quartic one in square brackets describe (fast and slow) magneto-sonic waves [86].

Before moving on to discuss the impact of shear and vorticity, let us briefly note what happens to the modes when we take the sound-proof limit—where the speed of sound is large. From eq. (7.53) we see that fast magneto-sonic waves are filtered out, while the slow ones reduce to Alfvén waves. In the opposite limit—when disturbances are much faster than the sound waves—the dispersion relation describes Alfvén waves and the low- c_s limit of fast magneto-sonic waves. This limit corresponds to ignoring fluid pressure perturbations while retaining variations in the magnetic pressure.

7.5.2 Sheared Background

Let us now consider the case where the background vorticity is negligible while shear terms are not. Re-scaling the magnetic field as in eq. (7.46) and decomposing eqs. (7.23), (7.44) and (7.45) (ignoring vorticity terms) as well as $\delta \mathbf{v}$ and $\delta \mathbf{v}_A$ in the shear-adapted basis, the coefficients' matrix of the linearized system of equations reads

$$\begin{pmatrix} -\omega & \rho k_1 & \rho k_2 & \rho k_3 & 0 & 0 & 0 \\ \frac{c_s^2}{\rho} k_1 & -\omega - i\sigma_1 & 0 & 0 & I_1 & v_A^2 k_1 & v_A^3 k_1 \\ \frac{c_s^2}{\rho} k_2 & 0 & -\omega - i\sigma_2 & 0 & v_A^1 k_2 & I_2 & v_A^3 k_2 \\ \frac{c_s^2}{\rho} k_3 & 0 & 0 & -\omega - i\sigma_3 & v_A^1 k_3 & v_A^2 k_3 & I_3 \\ 0 & I_1 & v_A^1 k_2 & v_A^1 k_3 & -\omega + i\sigma_1 & 0 & 0 \\ 0 & v_A^2 k_1 & I_2 & v_A^2 k_3 & 0 & -\omega + i\sigma_2 & 0 \\ 0 & v_A^3 k_1 & v_A^3 k_2 & I_3 & 0 & 0 & -\omega + i\sigma_3 \end{pmatrix} ,$$
 (7.54)

where

$$I_1 = v_A^1 k_1 - (\mathbf{v}_A \cdot \mathbf{k}) , \qquad \sigma_3 = -(\sigma_1 + \sigma_2) , \qquad (7.55)$$

and I_2 , I_3 are defined similarly.

In a similar fashion as for the hydrodynamic case considered above, we will decompose the coefficients of the characteristic polynomial in terms of scalars built from background quantities. As we might have expected, the resulting dispersion relation is a complicated seventh-degree polynomial (and we sanity-checked it reduces to the homogeneous case when we set to vanish the shear terms). In order to learn something useful out of it, we then consider the sound-proof limit and retain only terms proportional to the speed of sound. We end up with the following dispersion relation

$$a_5\omega^5 + a_4\omega^4 + a_3\omega^3 + a_2\omega^2 + a_1\omega + a_0 = 0 , \qquad (7.56)$$

with

$$a_{0} = -i \left\{ \det(\boldsymbol{\sigma}) \left[\sigma_{ij}^{2} k^{i} k^{j} - \frac{1}{2} \operatorname{Tr}(\boldsymbol{\sigma}^{2}) \right] + (\mathbf{v}_{A} \cdot \mathbf{k})^{2} \left[\det(\boldsymbol{\sigma}) k^{2} - \frac{1}{2} (\sigma_{ij} k^{i} k^{j}) \operatorname{Tr}(\boldsymbol{\sigma}^{2}) \right] + (\mathbf{v}_{A} \cdot \mathbf{k})^{4} \sigma_{ij} k^{i} k^{j} \right\}, \quad (7.57a)$$

$$a_{1} = \left\{ \left(\mathbf{v}_{A} \cdot \mathbf{k} \right)^{4} k^{2} + \left(\mathbf{v}_{A} \cdot \mathbf{k} \right)^{2} \left[\sigma_{ij}^{2} k^{i} k^{j} - \operatorname{Tr}(\sigma^{2}) k^{2} \right] + \operatorname{det}(\sigma) \left(\sigma_{ij} k^{i} k^{j} \right) + \frac{1}{2} \operatorname{Tr}(\sigma^{2}) \left[\frac{1}{2} \operatorname{Tr}(\sigma^{2}) k^{2} - \sigma_{ij}^{2} k^{i} k^{j} \right] \right\}, \quad (7.57b)$$

$$a_{2} = i \left\{ -\frac{1}{2} (\sigma_{ij} k^{i} k^{j}) \operatorname{Tr}(\sigma^{2}) + \det(\sigma) k^{2} + 2(\mathbf{v}_{A} \cdot \mathbf{k})^{2} (\sigma_{ij} k^{i} k^{j}) \right\},$$
(7.57c)

$$a_3 = \left[\operatorname{Tr}(\boldsymbol{\sigma}^2) k^2 - 2(\mathbf{v}_A \cdot \mathbf{k})^2 k^2 - \sigma_{ij}^2 k^i k^j \right] , \qquad (7.57d)$$

$$a_4 = -i(\sigma_{ij}k^ik^j) , \qquad (7.57e)$$

$$a_5 = k^2$$
. (7.57f)

As in the hydrodynamic case considered earlier, we first consider the case det(σ) = 0, and look for modes such that $\sigma^{ij}k_j = 0$. It is then easy to see that the general dispersion relation in eq. (7.56) simplifies to (ignoring a trivial root)

$$\left[\omega^2 - \left(\frac{1}{2}\mathrm{Tr}(\boldsymbol{\sigma}^2) - (\mathbf{v}_A \cdot \mathbf{k})^2\right)\right]^2 = 0.$$
(7.58)

Comparing to the corresponding hydrodynamic modes in eq. (7.31), we immediately see that the magnetic field tends to have a stabilizing effect (provided it is not orthogonal to the wave-vector, in which case it has no effect whatsoever).

Next, we take (again, as before) det(σ) = 0 and consider modes such that $\sigma^{ij}k_ik_j = 0$ (but $\sigma^{ij}k_j \neq 0$). The relevant dispersion relation can then be written (making use of eq. (7.32))

$$\omega^4 + b_2 \omega^2 + b_4 = 0 , \qquad (7.59)$$

with

$$b_2 = \frac{2}{3} \operatorname{Tr}(\sigma^2) - 2(\mathbf{v}_A \cdot \mathbf{k})^2 , \qquad (7.60a)$$



FIGURE 7.1: Real and imaginary part of the solutions of eq. (7.59),with both the frequency and $|\mathbf{v}_A \cdot \mathbf{k}|$ in units of $\sqrt{\text{Tr}(\sigma^2)}$. The solutions plotted correspond to the fastest growing modes evolving on top of an MHD sheared background. We see that the magnetic field has a stabilizing effect, as the growth rates are reduced with respect to those of the corresponding hydrodynamic modes. The stabilizing effect is all the more pronounced the more the wave-vector is aligned with the magnetic field lines, and is switched off for modes propagating in the directions perpendicular to the magnetic field lines. In particular, modes corresponding to sufficiently large values of $|\mathbf{v}_A \cdot \mathbf{k}|$ are turned stable.

and

$$b_4 = \frac{1}{12} \operatorname{Tr}(\sigma^2)^2 - \frac{2}{3} \operatorname{Tr}(\sigma^2) (\mathbf{v}_A \cdot \mathbf{k})^2 + (\mathbf{v}_A \cdot \mathbf{k})^4 .$$
(7.60b)

The same stabilizing effect of the magnetic field is evident from fig. 7.1, where both the frequency and $|\mathbf{v}_A \cdot \mathbf{k}|$ are plotted in units of $\sqrt{\text{Tr}(\sigma^2)}$.

The key point here is that, while the background shear is required for the instability (the vanishing-shear modes are stable Alfvèn waves in the sound-proof limit), the magnetic field is not the main driver. This is evident from the results as the imaginary part of the unstable modes remains finite in the limit $\mathbf{v}_A \rightarrow 0$, and the limiting value coincides with the hydrodynamic result (from the previous section). This observation, possibly unexpected at first sight, deserves a thorough discussion, and we will return to this issue in section 7.6.1. Before we expand on this, let us stress that the results make intuitive sense. The magnetic field impacts on the instability in that it breaks the hydrodynamic isotropy and dampens the growth of unstable modes propagating along magnetic field lines. This also suggests that shear-instability driven turbulence is isotropic in the hydrodynamic case but inherently anisotropic for magnetized flows, consistently with the overall picture discussed in section 5.1. Before moving on, it is also worth noting that the background velocity profile considered by Balbus & Hawley [31, 33] is characterized by having a shear matrix with vanishing determinant (and expansion rate), and also that for axisymmetric modes $\sigma^{ij}k_ik_i = 0$, while for the fastest growing MRI modes (propagating vertically) $\sigma^{ij}k_j = 0$.

7.5.3 Background with vorticity

Before we make contact with the usual MRI and the Rayleigh criterion, let us also consider the case with non-negligible background vorticity only. We re-scale the magnetic field as in eq. (7.46) and introduce a convenient basis $\{\hat{W}, \hat{q}, \hat{s}\}$, where $\hat{W} = \mathbf{W}/|\mathbf{W}|$ while \hat{q}, \hat{s} complete it to an orthonormal basis. For instance, assuming \mathbf{v}_A is not aligned with \mathbf{W} we can construct it as

$$\mathbf{q} = \mathbf{v}_A - (\mathbf{v}_A \cdot \hat{W}) \hat{W}, \qquad \hat{q} = \frac{\mathbf{q}}{|\mathbf{q}|}, \qquad \hat{s} = \hat{W} \times \hat{q}, \qquad (7.61)$$

and the magnetic field⁷

$$\mathbf{v}_A = v_A^W \hat{W} + v_A^q \hat{q} \ . \tag{7.62}$$

The coefficient matrix of the linearized system then is (cf. eqs. (7.23), (7.44) and (7.45) and ignore shear terms)

$$\begin{pmatrix} -\omega & \rho k^{W} & \rho k^{q} & \rho k^{s} & 0 & 0 & 0 \\ \frac{c_{s}^{2}}{\rho} k^{W} & -\omega & 0 & 0 & -v_{A}^{q} k^{q} & v_{A}^{q} k^{W} & 0 \\ \frac{c_{s}^{2}}{\rho} k^{q} & 0 & -\omega & +iW & v_{A}^{W} k^{q} & -v_{A}^{W} k^{W} & 0 \\ \frac{c_{s}^{2}}{\rho} k^{s} & 0 & -iW & -\omega & v_{A}^{W} k^{s} & v_{A}^{q} k^{s} & -(v_{A}^{W} k^{W} + v_{A}^{q} k^{q}) \\ 0 & -v_{A}^{q} k^{q} & v_{A}^{W} k^{q} & v_{A}^{W} k^{s} & -\omega & 0 & 0 \\ 0 & v_{A}^{q} k^{W} & -v_{A}^{W} k^{W} & v_{A}^{q} k^{s} & 0 & -\omega & -iW \\ 0 & 0 & 0 & -(v_{A}^{W} k^{W} + v_{A}^{q} k^{q}) & 0 & +iW & -\omega \end{pmatrix}$$

$$(7.63)$$

As for the sheared case, after having sanity-checked the result by contrasting it against the homogeneous background dispersion relation, we take the sound-proof limit. The sound-proof dispersion relation can then be written as

$$\omega^4 + b_2 \omega^2 + b_4 = 0 , \qquad (7.64)$$

with

$$b_2 = -\left[W^2 + (\hat{k} \cdot \mathbf{W})^2 + 2(\mathbf{v}_A \cdot \mathbf{k})^2\right] , \qquad (7.65a)$$

$$b_4 = \left[(\mathbf{v}_A \cdot \mathbf{k})^2 + W^2 \right] \left[(\hat{k} \cdot \mathbf{W})^2 + (\mathbf{v}_A \cdot \mathbf{k})^2 \right] .$$
(7.65b)

As this is a particularly simple quartic polynomial, we can study the stability of its roots analytically. Considering eq. (7.64) as an equation for ω^2 and computing the discriminant we obtain

$$\left[W^2 - (\hat{k} \cdot \mathbf{W})^2\right]^2 \ge 0 \tag{7.66}$$

⁷Note that the definition of \hat{q} changes when the background magnetic field is aligned with the vorticity, even though in what follows we would simply have to set $v_A^q = 0$.

so that ω^2 -roots are real. As complex roots of a real algebraic polynomials occur in pairs of complex conjugates, complex ω^2 -roots would correspond to an instability. In order to have stable roots though, we also need other conditions to be met. We, in fact need $b_2 < 0$ and $b_4 > 0$ to make sure that the ω^2 -roots are real and positive, so that ω roots are real as well. As this is evidently the case, we conclude that magnetized flows are generically stable in this case. This feature is unchanged from the corresponding fluid case, so that it is reasonable to expect that the same trend we discussed for the purely hydrodynamical case will also apply to the magnetized case with both shear and vorticity: Vorticity tends to stabilize shear-unstable modes in a manner independent of the wave number.

7.6 Concluding remarks: The MRI in perspective

We set out with the intention of discussing the magneto-rotational instability in a general background, relaxing the symmetry constraints associated with the standard analysis and possibly deriving an instability "criterion" relevant for (highly) dynamical environments and nonlinear simulations. However, having set up the analysis (and the required tools) in an arguably sensible way, we arrived at results which were not in line with the "naïve" expectations. Given this, it makes sense to comment on the implications. Moreover, we need to highlight an important "missing ingredient" in the discussion; the need to involve some suitable filtering operation to make the discussion sensible in the first place. We will deal with each of these questions in turn, starting with the implications of our results for the MRI.

7.6.1 The MRI vs the Rayleigh criterion

A key aspect of the MRI is that adding a weak magnetic field on top of a hydrodynamically stable shearing flow changes the nature of the problem and makes it unstable. In discussing this problem, however, it is often "forgotten" that the relevant hydrodynamic stability criterion [180] guarantees stability <u>only for</u> axisymmetric modes (cf. the discussion in appendix D.2). Adding a magnetic field renders such modes unstable technically, the non-axisymmetric ones are not [32]. Thus it is clear that the MRI is relevant only in situations where we can think of axisymmetric modes being "preferred" in some sense. An immediate example of this is an accretion disk, which involves a globally axisymmetric background for the perturbations. This then immediately tells us that applying the results to the dynamical context of neutron star mergers is a much more subtle endeavour. In fact, this exercise is problematic from the outset.

To back up this claim, we show in appendix D that we can reproduce the MRI perturbation equations and dispersion relation through the local frame construction. However, for the specific MRI calculation there exists a preferred local frame. This local frame is associated with an observer that is co-rotating with the fluid on some orbit, and the coordinate axes rotate in such a way that one of them always points in the radial direction of the global cylindrical coordinate system. Another coordinate axis always points in the azimuthal direction. This local frame is "preferred" as the axes are (by construction) tied to those of the most natural global coordinate system. In a sense, we could set up different local co-rotating observers and construct the global axes by stitching together the local ones. In the case of a general and truly local analysis, however, this additional piece of information is not available.

Moreover, we show in appendix D.2 how one may set up (for the circular and axisymmetric background flow) a local frame that is "co-moving but not co-rotating" with the fluid. In doing so, we derived the corresponding dispersion relation, confirmed that the result is consistent with the general formulae, and showed how we can recover the usual Rayleigh criterion (and hence also the MRI criterion) as long as we perform the conversion to the relevant co-rotating frame frequency.

These arguments clarify the sense in which the MRI (and similarly the Rayleigh stability) is a "global instability analyzed with local tools". The local analysis needs to be "augmented" by pieces of information that cannot be truly local. The upshot of this is that, in a merger-like scenario (where assumptions regarding the global properties of the flow are debatable) we should probably not expect the standard instability criteria to provide a faithful indication/diagnostic of what is actually going on. The standard argument will apply, but only if there is a meaningful sense of (Rayleigh stable) flow on a scale larger than that at which the plane-wave analysis is carried out. This complicates the discussion for any given numerical simulation, but so be it.

7.6.2 The missing ingredient: Filtering

Throughout the discussion we have focussed on the analytical development, sweeping issues associated with actual numerical data "under the carpet". The key issue here is that we ignored the question of how one would, in practice, construct the background suitable for the perturbation analysis given nonlinear simulation dynamics. In words, the answer is easy: We need to apply some suitable filtering operation to remove small scale fluctuations from a gradually varying "background". In a nonlinear setting this split is (obviously) not guaranteed to make sense. Suppose that the instability we are trying to uncover acts on some characteristic scale *L*, say. Then we need a background that varies on a larger scale than this, otherwise the notion of a shear flow that becomes unstable due to smaller scale waves makes no sense. This argument relies on an explicit filtering step, separating the instability scale *L* from the variation of the background. The construction of such a filter should be possible, at least in principle, in

many situations (see, for example, Celora et al. [58]). Of course, the scale separation may not apply in actual problems of interest.

Further complicating the discussion is the unavoidable implicit filtering associated with the finite numerical resolution. We know from the large body of work on turbulence simulations that sub-grid dynamics may play an important role in a robust description of the dynamics. This typically involves a suitable large-eddy scheme to represent the subgrid dynamics. Hence, the analysis involves elements of choice (effectively, the closure relations). Crucially, the effective field theory that is/should be simulated is not that of the ideal theory. All current models—both the ones discussed in [49, 177] as well as the covariant scheme of chapter 4—modify the principal part of the equations of motion. Therefore the analysis of the model "that is actually solved" is fundamentally changed, even when the closure terms are small. In essence, an instability analysis of numerical simulation data needs to consider the impact of an effective viscosity/resistivity. Given the presently available tools, we do not have a particularly good handle on this issue. We are forced to conclude that we need to make progress on the development of robust large-eddy models before we can make a sensible attempt to demonstrate the presence of the MRI in a highly dynamical environment.

Part IV

Conclusions

Hydrodynamics is an incredibly useful framework with myriads of applications at all scales, so that even after centuries of research it continues to be extremely fascinating and valuable. In this thesis, we have studied different and interconnected problems in the modelling of relativistic fluids, from dissipation to turbulence. The motivation for this work lies in the (extremely thrilling) promises of gravitational wave astronomy, and the range of exciting physics we can explore with binary neutron star mergers.

After a brief introduction/review in chapter 2 of the different modelling strategies for dissipative fluids currently on the market, we explored in chapter 3 the close to equilibrium regime of the action-based dissipative multi-fluid model of Andersson and Comer [20]. A first motivation for this lies in the fact that the equations of motion are derived from an action principle, and as such are valid (in principle) in the non-linear regime with no reference to some equilibrium state. However, as the close to equilibrium regime is likely to be relevant for much of the applications we have in mind, it is worth exploring how the model behaves in such a limit. This also has the additional advantage of facilitating a direct comparison with alternative existing models, all of which are based on an expansion around such a notion of equilibrium.

In developing complicated dissipative models, however, we need to be pragmatic and keep in mind the extreme computational cost of simulating them. Hence, we continued in chapter 4 focusing on the foundational aspects of performing a "spatial filtering" in relativity. We do so as this is a common strategy for dealing with computationally demanding turbulent flows, for which direct numerical simulations are often impractical. The discussion we provided focuses on the formal underpinnings, as the strategy is complicated by the covariance principle of General Relativity. We then argued that it is natural to set up the framework in the fibration associated with fluid elements, and showed how one can perform filtering ensuring consistency with the tenets of relativity. The framework we put forward has the additional advantage that it allows for a direct link with the underlying thermodynamics, which is ultimately what we aim to constraint with binary neutron star merger observations. In the process we also demonstrated how the filtered equations of motion are effectively equivalent to those describing a dissipative fluid, thus leading back to similar (although, as we discuss in chapter 4, not equivalent) issues faced when modelling dissipation in relativity.

As we argued already in this work, for accurate neutron star modelling we are tasked with even more complicated settings involving multiple interpenetrating flows, should this be in the form of a two-fluid plasma and/or superfluid/superconducting mixtures. Given this, we continued in chapter 5 by discussing the first steps towards extending the framework of chapter 4 to magneto-hydrodynamics. We do so as this is the first step towards a multi-fluid LES framework in that it adds the electromagnetic degrees of freedom to the picture, while remaining in the realm of single fluid models. In the last part of this work we focused on applications to problems of relevance for binary neutron star mergers. In chapter 6 we focused on modelling (fast) reactions for neutron star simulations, and the associated bulk viscosity. The reason being that reactions are thought to source the dominant dissipative mechanism at play in mergers. Our discussion, in particular, focuses on the impact that inevitable numerical limitations have on the way we should frame the modelling. We then considered the magneto-rotational instability, which is thought to be a key mechanism for sustaining the development of turbulence in the outer layers of binary neutron star merger remnants. Our aim has been to provide an analysis of this mechanism that is well suited for highly-dynamical environments such as mergers, where usual criteria based on rather restricting assumptions may not hold.

Whilst we have presented our different contributions in a relatively independent fashion, it should have become rather clear by now that they are not. This is true from a physics perspective, as all the different aspect we touched upon, from dissipation to turbulence, play a role "at the same time" in mergers. It is even more true if we consider that similar modelling strategies developed in one "area" can find useful applications in another. This is demonstrated, for example, by the specific closure scheme we put forward in chapter 4, where we adapted some of the ideas developed to address the stability and causality issues of traditional dissipative models to fix the issues encountered with the natural relativistic generalization of the model originally put forward by Smagorinsky [205] in a Newtonian setting.

In terms of future work, there are a number of possible avenues worthwhile pursuing that originate from the analysis presented in this thesis. They involve, not surprisingly, both aspects of framework developments as well as more specific applications.

In terms of framework developments, for example, it would be worthwhile working towards extensions of the framework of chapter 4 to the case of multiple interpenetrating flows/mixtures. Moreover, we also mentioned briefly at the end of chapter 5 how the framework discussed in chapter 4 suggests—due to the covariance of the resulting model—that we may enhance the role of the "large-eddy strategy" to a tool for linking models valid at different scales. Clearly, there are several applications we can envisage, from dynamo theory to try and unify the somewhat arbitrarily separated small scale and large scale models, to superfluids to link mesoscopic models where each single vortex line is resolved to coarse-grained descriptions of the kind discussed in section 2.3—for which the two-fluid models were originally developed [21].

Furthermore, as we discussed in section 4.7, the scale-gap and fluctuations within a fluid box implies that the equation of state used for large-scale merger simulations may not be trivially linked to the underlying microphysics. Notably, as any numerical simulation is performed on a finite grid, there is always at least an implicit filtering associated with it. As the best grid resolution in large-scale merger simulations is of

order tens of meters, the impact of this is potentially significant. It would be worth exploring this potential disconnect and try to quantify the impact this may have on neutron star parameters extracted from observations.

Further developments of the large-eddy strategy in both these directions would obviously be rather incomplete if not provided with suitable closure schemes. Future developments will also have to involve aspects of developing, testing and calibrating novel/different closure schemes. As we may also envisage rather different schemes to be better suited for diverse applications, there is plenty of work to be done.

Appendix A

Transporting a tetrad and Fermi Coordinates

In this appendix we introduce the notion of Fermi coordinates, as these have been used but not derived/discussed in the main body of the thesis. We start by discussing the transport of a tetrad along a curve/worldline since the notion of Fermi coordinates builds on this [101, 155]. In our discussion we will, in particular, make explicit contact with the notion of spin coefficients that have to be introduced whenever one wants to work with an orthonormal basis or tetrad (see notation section at the beginning of this work).

A.1 Transporting a tetrad along a curve

We start our analysis by considering a curve/worldline in spacetime, which we here take as time-like with tangent vector U^a , and assuming a set of four orthonormal vectors is given along the worldline. Namely, we have a set of four vectors $e_{\hat{a}}$, with $\hat{a} = 0, 1, 2, 3$ such that

$$g(e_{\hat{a}}, e_{\hat{b}}) = \eta_{\hat{a}\hat{b}} , \qquad (A.1)$$

where we are considering the metric *g* as a bi-linear form on the tangent space, and $\eta_{\hat{a}\hat{b}}$ is the Minkowski metric. We further assume that we can take the time-like unit vector of the tetrad as the one tangent to the curve, that is $U = e_{\hat{0}}$. We are interested in the rate-of-change of the tetrad basis vectors as we move along the curve. Since *U* is the tangent to the curve, we can start from

$$\nabla_U e_{\hat{a}} = \Omega^b_{\ \hat{a}} \, e_{\hat{b}} \,, \tag{A.2}$$

where we have introduced the bi-linear form $\Omega_{\hat{a}}^{\hat{b}}$ to re-write the rate-of-change as a linear combination of the tetrad basis vector themselves. Next, we observe that $\Omega_{\hat{b}\hat{a}}$ must be an anti-symmetric bi-linear form¹, as this follows from the orthonormal character of the basis, that is

$$\nabla_U \left(g(e_{\hat{a}}, e_{\hat{b}}) \right) = 0 \Longrightarrow \Omega_{\hat{a}\hat{b}} = \Omega_{\hat{b}\hat{a}} . \tag{A.3}$$

As such, we can decompose Ω as any skew-symmetric bi-linear form [101]. In our case this gives

$$\Omega_{\hat{a}\hat{b}} = U_{\hat{a}}a_{\hat{b}} - a_{\hat{a}}U_{\hat{b}} - \varepsilon_{\hat{a}\hat{b}\hat{c}\hat{d}}U^{\hat{c}}W^{d} , \qquad (A.4)$$

where $a_{\hat{a}}$ is the four-acceleration of the worldline, while $W^{\hat{a}}$ is a generic vector orthogonal to $U^{\hat{a}}$. To better understand the physical meaning of such decomposition, we introduce the spatial index $\hat{i} = 1, 2, 3$ and use it also to denote the three unit vectors orthogonal to U. We then get

$$\left(\nabla_{U}U\right)^{\hat{a}} = a^{\hat{a}} , \qquad (A.5)$$

$$\left(\nabla_{U}e_{\hat{i}}\right)_{\hat{a}} = a_{\hat{i}}U_{\hat{a}} + \varepsilon^{U}_{\hat{a}\hat{b}\hat{i}}W^{\hat{b}} .$$
(A.6)

Hence, when the worldline is a geodesic we have $a^{\hat{a}} = 0$ and the time-like unit vector is unchanged as we move along the curve, while the spatial ones $e_{\hat{i}}$ change due to some (spatial) rotation in the subspace orthogonal to U. In particular, we stress that whilst the four acceleration is an intrinsic property of the worldline, the vector W represents the angular velocity associated with the rotation of the observer frame. In essence, given a time-like worldline/curve, there exist an infinite number of compatible tetrads such that $e_{\hat{0}} = U$, and these are all related by some rotation in the plane orthogonal to the curve. We can then further split the bilinear form Ω as the sum of two terms

$$\Omega_{\hat{a}\hat{b}} = \Omega_{\hat{a}\hat{b}}^{FW} + \Omega_{\hat{a}\hat{b}}^{rot} , \qquad (A.7a)$$

with

$$\Omega_{\hat{a}\hat{b}}^{FW} = U_{\hat{a}}a_{\hat{b}} - a_{\hat{a}}U_{\hat{b}} , \quad \text{and} \ \Omega_{\hat{a}\hat{b}}^{rot} = \varepsilon_{\hat{a}\hat{b}\hat{c}}^{U}W^{\hat{c}} .$$
(A.7b)

This splitting is meaningful because it separates the terms in Ω we have control over (the rotation part) from those that are given once the curve is specified (the Fermi Walker part). In particular, a vector is said to be Fermi-transported (or Fermi-Walker transported) along a worldline if its components change with Ω^{FW} .

Before moving on to discuss Fermi coordinates, we take the opportunity to make contact with the spin connection coefficients, which are normally introduced when working with a tetrad basis. Using the spin coefficients [50] we would get

$$\nabla_{U}e_{\hat{a}} = U^{b}\nabla_{b}e_{\hat{a}} = U^{b}\omega_{b\ \hat{a}}^{\ \hat{b}}e_{\hat{b}} = \omega_{\hat{0}\ \hat{a}}^{\ \hat{b}}e_{\hat{b}}^{\ }, \qquad (A.8)$$

¹For clarity, let us point out that the bi-linear form is obtained lowering the contravariant index via the metric, namely $\Omega_{\hat{a}\hat{b}} = g_{\hat{a}\hat{c}}\Omega_{\hat{f}}^{\hat{c}}$.

so that, contrasting this with eq. (A.2) we see

$$\omega_{\hat{0}\,\hat{a}}^{\,\hat{b}} = \Omega_{\,\hat{a}}^{\hat{b}} \,. \tag{A.9}$$

In essence, the bilinear form Ω determines the "time-part" of the spin connection coefficients. This is intuitive given that the spin connection coefficients have to be used when a tetrad is assigned throughout (a region of) the spacetime, and not only along a worldline.

A.2 Fermi coordinates

We now discuss how to introduce a (local) set of coordinates that is suitable for describing the physics as measured by some local observer [155, 101], that is in the vicinity of the observer worldline. These go under the name of Fermi coordinates, as it was Fermi who originally introduced them in 1922 [78, 79] (the work by Walker [221] is also relevant). The concept was then further developed by, in particular Manasse and Misner [148], where the authors focus on the particular case where the central curve is a geodesic. The same idea was then generalized to the more general case of an accelerated and rotating observer by Misner et al. [155].

Given some observer's worldline, the associated Fermi coordinates can be constructed via the following procedure—which is perhaps best described in fig. A.1. First consider the tangent vector to the (time-like) curve U, and a tetrad defined over the entire worldline. In the case of Fermi coordinates this is given at a point P_0 and then Fermi-transported along the curve. We can then assign coordinates to spacetime points (or viceversa) in the vicinity of the worldline as follows. We start with four real numbers $\{x^0, x^{\hat{1}}, x^{\hat{2}}, x^{\hat{3}}\}$, and first move along the central curve by x^0 (in a parameterization compatible with U). We then consider the numbers $\alpha^{\hat{i}} = x^{\hat{i}}/s$ with $s^2 = \sum_i (x^{\hat{i}})^2$, and the connecting geodesic starting perpendicular from the central curve with unit tangent vector $v = \alpha^{\hat{i}}e_{\hat{i}}$. As geodesics satisfy a second order differential equation, they are uniquely defined given a point and a tangent vector (at the point). Next, move along the connecting geodesic by s (in terms of the connecting geodesic's proper length), and assign to such point the coordinates $(x^0, x^{\hat{1}}, x^{\hat{2}}, x^{\hat{3}})$.

Let us first of all note that along the central curve the basis vector of the coordinates grid are, by construction, identical to the orthonormal tetrad

$$\frac{\partial}{\partial x^{\hat{a}}} = e_{\hat{a}} . \tag{A.10}$$

This is important as we can immediately draw two important conclusions. First, the metric in the given coordinates takes the Minkowski form along the entire central



FIGURE A.1: Left: Tetrad transported along the observer worldline. Right: Connecting geodesics starting from and perpendicular to the central curve. Figure adapted from Misner et al. [155].

curve. Second, we can use this simple fact to show that such coordinates are welldefined in some neighbourhood of the central curve [148, 155]. This is intuitively clear as points close enough to the central curve are uniquely connected to the central curve by one (and only one) geodesic. On larger distances (in terms of the connecting geodesic proper length) the procedure fails because different geodesics can mix and touch. This can happen due to both the acceleration of the central curve and the curvature of the spacetime itself—and also due to the tetrad rotation if the basis is not Fermi transported along the curve.

We can now use such coordinates to write down an expansion for the metric in the neighbourhood of the central curve. As we know already that the metric along the central curve takes the Minkowski form, we only need the first derivatives of the metric evaluated on the central curve. These can be obtained from the Christoffel symbols which are related to the spin coefficients and hence to the bilinear form Ω introduced above. In particular, by inspecting eq. (A.10) one can easily see that

$$\Gamma^{\hat{a}}_{\hat{b}\hat{0}} = \Omega^{\hat{a}}_{\hat{b}} , \qquad (A.11)$$

so that, by means of eq. (A.4), we readily obtain

$$\Gamma_{\hat{0}\hat{0}\hat{0}} = \Gamma_{\hat{0}\hat{0}}^{\hat{0}} = 0 , \qquad (A.12)$$

$$\Gamma_{\hat{0}\hat{i}\hat{0}} = -\Gamma_{\hat{i}\hat{0}}^{\hat{0}} = -a_{\hat{i}} , \qquad (A.13)$$

$$\Gamma_{\hat{0}\hat{j}\hat{0}} = -\Gamma^{0}_{\hat{j}\hat{0}} = -a_{\hat{j}} , \qquad (A.13)$$

$$\Gamma_{\hat{i}\hat{j}\hat{0}} = \Gamma^{\hat{i}}_{\hat{j}\hat{0}} = -\varepsilon_{\hat{0}\hat{i}\hat{j}\hat{k}}W^{\hat{k}} . \tag{A.14}$$

The remaining Christoffel symbols can be found using the geodesic equation satisfied by the connecting geodesics

$$\frac{d^2 x^{\hat{a}}}{ds^2} + \Gamma^{\hat{a}}_{\hat{b}\hat{c}} \frac{dx^{\hat{b}}}{ds} \frac{dx^{\hat{c}}}{ds} = 0.$$
 (A.15)

As the connecting geodesics are given by $x^{\hat{0}} = \text{const}$ and $x^{\hat{i}} = \alpha^{\hat{i}}s$, we readily obtain

$$\Gamma_{\hat{a}\hat{i}\hat{j}} = \Gamma^{\hat{a}}_{\ \hat{i}\hat{j}} = 0 .$$
 (A.16)

Next, note that by means of the metric compatibility condition (with the connection given by the Christoffel symbols) we have

$$g_{\hat{a}\hat{b},\hat{c}} = 2\Gamma_{(\hat{a}\hat{b})c}$$
, (A.17)

and using eqs. (A.12) and (A.16) we obtain

$$g_{\hat{0}\hat{0}} = g_{\hat{0}\hat{0}}|_{G} + g_{\hat{0}\hat{0},\hat{a}}x^{\hat{a}} = -\left(1 + 2a_{\hat{j}}x^{\hat{j}}\right) + \mathcal{O}(x^{\hat{j}})^{2}, \qquad (A.18)$$

$$g_{\hat{0}\hat{i}} = g_{\hat{0}\hat{i}}|_{G} + g_{\hat{0}\hat{i},\hat{a}}x^{\hat{a}} = -\varepsilon_{\hat{0}\hat{i}\hat{j}\hat{k}}W^{\hat{k}} + \mathcal{O}(x^{\hat{j}})^{2}, \qquad (A.19)$$

$$g_{\hat{0}\hat{i}} = g_{\hat{0}\hat{i}}|_{G} + g_{\hat{0}\hat{i},\hat{a}}x^{\hat{a}} = -\varepsilon_{\hat{0}\hat{i}\hat{j}\hat{k}}W^{k} + \mathcal{O}(x^{j})^{2}, \qquad (A.19)$$

$$g_{\hat{i}\hat{j}} = g_{\hat{i}\hat{j}}\big|_{G} + g_{\hat{i}\hat{j},\hat{a}}x^{\hat{a}} = \eta_{\hat{i}\hat{j}} + \mathcal{O}(x^{j})^{2} .$$
(A.20)

In essence, we obtained an expansion for the metric away from the central worldline. The first order expansion depends on the worldline acceleration, and also on the (arbitrary) vector describing the angular rotation of the basis vectors as we move along the worldline. Notably, no information about the space-time curvature enters at first order in the expansion.

We conclude with some observations/comments. First of all, let us observe that we can always choose to work with a non-rotating tetrad, so that the associated expansion for the metric simplifies accordingly (noting that this is precisely the choice made in chapter 4, for example). In this case, the coordinates are called Fermi-coordinates [78, 79, 208]. If the observer is also freely-falling, the central curve is a geodesic and the (non-rotating) coordinates are called Fermi normal coordinates [148]—in which case the metric expansion contains no first order terms. The second order corrections to the Minkowski metric have been computed explicitly by Manasse and Misner [148] (for Fermi normal coordinates), showing in particular that the second order corrections are uniquely determined by the Riemann tensor (evaluated on the central curve). The work by Rakhmanov [179], where the Fermi coordinates expansion of the metric is computed to all orders given a spacetime describing a plane gravitational wave, is also relevant. Finally, we also note that when a similar scheme is used with light-like connecting geodesics, the resulting coordinates are known as optical coordinates [179].

Appendix **B**

Multi-scale arguments and the invariant manifold method

Multi-scale methods are useful whenever the (system of) equations to solve contains different scales, so that is physically (and numerically) useful/convenient to solve an approximate system instead. In this appendix we briefly summarize key results from [170, 223] that are used in the main body of the thesis, specifically in chapter 6.

B.1 Invariant manifold approach

Assume we have a system of ordinary differential equations written as

$$\dot{x} = f(x, y), \tag{B.1a}$$

$$\dot{y} = \epsilon^{-1} g(x, y). \tag{B.1b}$$

The variables *x* are called *slow*, and the variables *y* are called *fast*, whilst $\epsilon \ll 1$ is a parameter.

In the *invariant manifold* approach we assume that there exists an *equilibrium* (*fast*) *state* $\varphi(x)$ such that

$$g(x,\varphi(x)) \equiv 0. \tag{B.2}$$

We can then write the fast variables y as an expansion in the small parameter ϵ about the equilibrium state as

$$y = \varphi(x) + \epsilon y_1 + \mathcal{O}(\epsilon^2).$$
 (B.3)

Using the equations of motion eq. (B.1) we find that the behaviour of the slow variables x is approximated, to second order in ϵ , by the solution to

$$\dot{X} = F_0(X) + \epsilon F_1(X) \tag{B.4a}$$

$$= f(X,\varphi(X)) + \epsilon \nabla_y f(X,\varphi(X)) \left(\nabla_y g(X,\varphi(X)) \right)^{-1} \nabla_x \varphi(X) f(X,\varphi(X)).$$
(B.4b)

The solution to the simplified system approximates the solution to the full system eq. (B.1) to $\mathcal{O}(\epsilon^2)$ up to times $\mathcal{O}(1)$. The first order correction term needs to be applied consistently to all variables in the reduced system for this accuracy result to hold.

B.2 Two timescale approach

Strictly, the invariant manifold approach is only valid for ordinary differential equations. A more general approach that applies to partial differential equations is the two-scale approach. Using the two timescale approach as an example, this introduces the *fast time* $\tau = t/\epsilon$ which is then treated as an independent variable. Applied to eq. (B.1) this leads to

$$\partial_t x + \epsilon^{-1} \partial_\tau x = f(x, y),$$
 (B.5a)

$$\partial_t y + \epsilon^{-1} \partial_\tau y = \epsilon^{-1} g(x, y).$$
 (B.5b)

By gathering terms in powers of ϵ , the fast behaviour can be integrated out by taking the integral average in τ .

The result of the mathematical calculation is identical to the invariant manifold approach, when applied to ordinary differential equations. The calculation is general enough to include partial differential equations, and illustrates a different interpretation and potential problems. The interpretation is that the reduced system is valid for the *integral average* of the slow variables: the fast scales have been integrated out. The potential problem is the requirement that the integral average of the fast behaviour is assumed to not contribute at leading order in ϵ . This "resonance" behaviour cannot be captured by these approaches.

B.3 Linear fast dynamics

A particularly relevant example is where the fast behaviour is linear, or can be linearised. In this case we write the full system eq. (B.1) as

$$\dot{x} = f(x, y), \tag{B.6a}$$

$$\dot{y} = \epsilon^{-1}(-Ay + B), \tag{B.6b}$$

with A = A(x) and B = B(x) are constants in the fast variables *y*. The equilibrium solution is therefore $\varphi(x) = B/A$, and the simplified system is

$$\dot{X} = f(X, \varphi(X)) \left[1 + \epsilon A^{-1} \nabla_y f(X, \varphi(x)) \nabla_x \varphi(X) \right].$$
(B.7)

B.4 Constructing the fast terms

The construction in this section relies on the ratio of scales ϵ being explicit in the equations of motion. Usually a non-dimensionalisation of the system is needed to make the scales explicit. However, with complex nonlinear terms (such as tabulated net reaction rates which include many reaction channels) the precise form of the terms may not be obvious.

Here we need only the leading order terms and so can proceed as follows. We start from a system of equations which we expect to have fast behaviour

$$\dot{z} = h(x, z), \tag{B.8}$$

where x are any variables we expect to be slow. We assume that we know how h scales asymptotically with the ratio of scales. That assumption means we can explicitly compute

$$h^{\text{fast}} = \lim_{\epsilon \to 0} (\epsilon h) .$$
 (B.9)

This defines the source term for the fast behaviour as the piece that diverges linearly with the ratio of scales in the limit of infinitely fast speeds. We then split the source into fast and slow pieces using

$$h^{\rm slow} = h - \epsilon^{-1} h^{\rm fast},\tag{B.10}$$

and perform the equivalent split on the variables z as

$$\dot{z}^{\text{fast}} = \epsilon^{-1} h^{\text{fast}},$$
 (B.11a)

$$\dot{z}^{\text{slow}} = h^{\text{slow}}.$$
 (B.11b)

We can then identify the fast variables y with z^{fast} and augment the slow variables x with z^{slow} .

Appendix C

Working with the CompOSE database

Our analysis of the reactive problem in chapter 6 is—ultimately—aimed at numerical implementations. Given this target, it makes sense to consider how the discussion impacts on the matter model that needs to provide the input physics. In this appendix we will spell out the connection with the compOSE database, which provides a useful collection of state-of-the-art equation of state models. As a result, the arguments draw heavily on the compOSE manual [210], in particular section 4.1.2 ("Thermodynamic consistency") of version 2.0. The aim here is to explain how the various thermodynamical coefficients introduced in the main text can be worked out from an actual equation of state table. This is obviously a necessary step in the process. It also helps highlight to what extent existing tabulated data needs to be augmented in the future.

All equations of state relevant for our work in the compOSE database are provided as tables of (T, n, Y_q) , where Y_q is the fraction of charged strongly interacting particles, which for a system without muons corresponds to the electron fraction $Y_q = Y_e$ (as local charge neutrality is assumed to hold). The central thermodynamical potential is the Helmholtz free energy density $f = \varepsilon - Ts$, and some key quantities in the construction of the tables are

$$\left\{\frac{p}{n}, \frac{s}{n}, \frac{\mu_{\rm b}}{m_{\rm n}} - 1, \frac{\mu_{\rm q}}{m_{\rm n}}, \frac{\mu_{\rm l}}{m_{\rm n}}, \frac{f}{nm_{\rm n}} - 1, \frac{\varepsilon}{nm_{\rm n}} - 1\right\}, \qquad (C.1)$$

where m_n is the neutron mass—also provided in the tables and specific to each model while μ_b , μ_q , μ_l are the baryon, charge and lepton "chemical potentials" (respectively). The energy cost of adding a neutron, proton or electron to the system is then

$$\mu_{\rm n} = \mu_{\rm b}$$
, $\mu_{\rm p} = \mu_{\rm b} + \mu_{\rm q}$, $\mu_{\rm e} = -\mu_{\rm q} + \mu_{\rm l}$ (C.2)

as follows straightaway from the respective baryon, charge¹ and lepton number. The baryon, charge and lepton chemical potentials are (in general) used to build the free energy, even though in the charge-neutral case with leptons this reduces to

$$f = (\mu_{\rm b} + \mu_{\rm l} Y_{\rm e})n - p$$
. (C.3)

From this we see that there are only two independent chemical potentials at the thermodynamical level—consistent with a three-parameter equation of state—that can be written as derivatives of the Helmholtz free energy, namely μ_b and μ_l .

Now, we need to connect with the quantities used in the main text. We have (following from the Gibbs relation eq. (6.12))

$$f = \varepsilon - Ts$$
, $df = -sdT + \mu_n dn - \beta dn_e$, (C.4)

where $\beta = \mu_n - \mu_p - \mu_e$ in the cold equilibrium assumed throughout this paper. From this differential we see that, by definition

$$\mu_{n} = \left(\frac{\partial f}{\partial n}\right)_{T,n_{e}} = \left(\frac{\partial f}{\partial n}\right)_{T,Y_{e}} - Y_{e}\left(\frac{\partial f}{\partial n_{e}}\right)_{T,n}, \qquad (C.5a)$$

$$-\beta = \left(\frac{\partial f}{\partial n_{\rm e}}\right)_{T,n} = \frac{1}{n} \left(\frac{\partial f}{\partial Y_{\rm e}}\right)_{T,n} \,. \tag{C.5b}$$

Contrasting this with the results in section 4.1.2 of the compOSE manual, it is easy to see that² $\mu_n = \mu_b$ and $\beta = -\mu_l$. In practice, the affinity β can be either read off from the tables directly, or computed as above. Because the equation of state table is three-dimensional, the quantity extracted is inevitably a function of $\beta = \beta(n, T, Y_q)$. This is consistent with the results of section 6.2, where we accounted for the fact that β depends on either the temperature or the energy density.

Now, the coefficients A and B introduced in section 6.2 can be obtained as combinations of derivatives of β considered as a function of (n, ε, Y_e) . We need to link these expressions to derivatives that can be computed from the available tables (or, which may be more practical for the future, enhance the table with the required information).

¹This is the total charge, not the charge of the strongly interacting particles corresponding to $n_q = nY_q$. ²Note that μ_q potential is not an independent thermodynamic quantity as the system cannot create protons alone because of the charge neutrality assumption.

That is, we have to change variables to arrive at

$$\left(\frac{\partial\beta}{\partial n}\right)_{\varepsilon,Y_{e}} = \left(\frac{\partial\beta}{\partial n}\right)_{T,Y_{e}} - \left(\frac{\partial\varepsilon}{\partial n}\right)_{T,Y_{e}} \left(\frac{\partial\varepsilon}{\partial T}\right)_{n,Y_{e}}^{-1} \left(\frac{\partial\beta}{\partial T}\right)_{n,Y_{e}}, \quad (C.6a)$$

$$\left(\frac{\partial\beta}{\partial\varepsilon}\right)_{n,Y_{\rm e}} = \left(\frac{\partial\varepsilon}{\partial T}\right)_{n,Y_{\rm e}}^{-1} \left(\frac{\partial\beta}{\partial T}\right)_{n,Y_{\rm e}}, \qquad (C.6b)$$

$$\left(\frac{\partial\beta}{\partial Y_{\rm e}}\right)_{\varepsilon,n} = \left(\frac{\partial\beta}{\partial Y_{\rm e}}\right)_{T,n} - \left(\frac{\partial\varepsilon}{\partial Y_{\rm e}}\right)_{T,n} \left(\frac{\partial\varepsilon}{\partial T}\right)_{n,Y_{\rm e}}^{-1} \left(\frac{\partial\beta}{\partial T}\right)_{n,Y_{\rm e}}.$$
 (C.6c)

Recalling that these quantities should be evaluated at equilibrium, we see that we also need to construct the corresponding equilibrium table. Operationally, this can be done as follows. We fix n, T and vary Y_e until we find a value for which $\beta = 0$. The corresponding value of Y_e is then what we call Y_e^{eq} and the equilibrium composition will automatically only be a function of (n, T). Evaluating the original three-parameter model at $Y_e = Y_e^{eq}$ gives the corresponding equilibrium energy density and pressure etc. Using expressions analogous to eq. (C.6), we can rewrite derivatives of Y_e^{eq} with respect to (n, ε) in terms of derivatives with respect to (n, T), that can be extracted from the tables.

With the results in eq. (C.6) and evaluating the relevant quantities at equilibrium, we can work out (for a given equation of state) the value of \mathcal{B} , as required for fig. 6.2. In order to compute \mathcal{A} though, we also need to evaluate the restoring term γ (effectively, a measure of the reaction timescale). In fig. C.1 we show \mathcal{A} as obtained from the modified Urca rates for the APR equation of state [210, 195] used in [104]. For this figure we have calculated γ assuming the Fermi surface approximation, which allows us to use the analytic formulae from [9].

Let us stress that the result should be valid for low temperatures ($T \leq 1$ MeV), however the timescales relevant for our purposes occur in the range of (according to A as calculated above) 2 MeV $\leq T \leq 20$ MeV for the densities relevant to the neutron star core, begging the question of how accurate the approximation is. Instead of calculating out-of-equilibrium rates without the Fermi surface approximation, we can take a different approach, and estimate the equilibration timescale using neutrino opacities [106]. Notably, we find some broad similarities between the two estimates—despite expected qualitative differences—particularly in the regions of interest at relevant temperatures and densities.



FIGURE C.1: Plot of \mathcal{A} for the APR equation of state used in [104]. The restoring term γ is calculated assuming the Fermi surface approximation remains valid. Contours are at $\mathcal{A} = \{10^3, 10^4, 10^7, 10^9\}s^{-1}$ (solid, dash, dot-dash, dot).

Appendix D

Formulating the MRI in the local frame

In this appendix we show how to formulate the magneto-rotational instability using the local frame construction of chapter 7. In particular, we will show that using a corotating local frame we can derive the same equations as in [31]. Next in appendix D.2 we focus on the Rayleigh criterion. The discussion we provide here has the advantage that it makes explicit an important underlying assumption that is key to the usual Rayleigh and MRI criteria [180, 33].

Let us start by considering the circular velocity profile assumed in [31], $\mathbf{v} = v^{\hat{\varphi}} \hat{\varphi}$ with $v^{\hat{\varphi}} = \Omega(R)R$ where we use cylindrical coordinates and an orthonormal basis on the "tangent space" (as usual). Consistently with the notational conventions adopted in the rest of the thesis, we then distinguish between indices with a "hat" corresponding to the orthonormal basis, and those without that correspond to the coordinate basis. We then pick an orbit at some radial distance R_0 and choose an observer that is corotating with angular frequency identical to that of the background flow at R_0 , that is $\mathbf{v}_{obs} = \Omega_0 R \hat{\varphi}$ where $\Omega_0 = \Omega(R_0)$. The observer is then accelerated with acceleration $\mathbf{a} = -\Omega_0^2 R \hat{R}$, and the velocity of the fluid with respect to such an observer then is $\mathbf{v}' = (\Omega - \Omega_0) R \hat{\varphi}$. We then set up the axes of the observer local frame so that one is pointing in the radial direction (\hat{e}_1) , one is pointing in the azimuthal direction (\hat{e}_2) and the third one is aligned with the rotation axis (\hat{e}_3) . Introducing coordinates associated to this observer, we can then write the background fluid velocity as

$$\mathbf{v}' = \frac{\mathrm{d}\Omega}{\mathrm{d}\ln R} \bigg|_{R_0} x' \hat{e}_2 + \mathcal{O}(x'^2) . \tag{D.1}$$

We have neglected terms of order $O(x'^2)$ as we will only need the velocity and its gradients evaluated at the origin of the frame—so that such terms will not enter the
perturbation equations anyway. Computing the gradients we then obtain

$$\partial'_i v'_j = \begin{pmatrix} 0 & s_0 & 0\\ 0 & 0 & 0\\ 0 & 0 & 0 \end{pmatrix} , \qquad s_0 = \frac{d\Omega}{d\ln R} \Big|_{R_0} . \tag{D.2}$$

As the local frame of the observer is rotating with angular velocity $\Omega_0 \hat{e}_3$, we need to include the Coriolis force into the perturbation equations. We then write the perturbed Euler and continuity equations (dropping the primes for clarity, and retaining only fast-gradients in the background velocity)

$$\partial_t \delta v_i + 2\Omega_0 \epsilon_{i3k} \delta v^k + \delta v^j \partial_j v_i + \frac{c_s^2}{\rho} \partial_i \delta \rho + \frac{1}{\mu_0 \rho} \left[B_j \partial_i \delta B^j - B^j \partial_j \delta B_i \right] = 0 , \qquad (D.3a)$$

$$\partial_t \delta \rho + \rho \partial_i \delta v^i = 0$$
, (D.3b)

and, introducing a WKB plane-wave expansion (as detailed in section 7.1),

$$-i\omega\delta v_i + 2\Omega_0\epsilon_{i3k}\delta v^k + s_0\delta_{i2}\delta v^1 + i\frac{c_s^2}{\rho}k_i\delta\rho + \frac{i}{\mu_0\rho}\left[B_jk_i\delta B^j - B^jk_j\delta B_i\right] = 0, \quad (D.4a)$$

$$-i\omega\delta\rho + i\rho k_i\delta v^i = 0. (D.4b)$$

Next, focus on the induction equation. As we have discussed section 7.3, the induction equation in the co-rotating frame retains the inertial form. We then have

$$\delta\partial_j \left(v^j B^i - v^i b^j \right) = \delta v^j \partial_j B^i + v^j \partial_j \delta B^i - B^j \partial_j \delta v^i - \delta B^j \partial_j v^i , \qquad (D.5)$$

where we made use of i)the no-monopoles constraint ii)the vanishing expansion of the background flow iii)the Boussinesq approximation to get rid of the divergence of the perturbed velocity. Introducing the WKB plane-wave expansion and evaluating the background quantities at the origin of the local frame we then end up with

$$-i\omega\delta B^{i} - iB^{j}k_{j}\delta v^{i} - \delta B^{1}s_{0}\delta^{i2} = 0.$$
(D.6)

In eqs. (D.4) and (D.6) we recognize the terms entering the perturbation equations in [31] (with the exception of background gradients in the pressure that we are here neglecting). We also note that we here do not need to formally neglect terms of the form B/R as these terms do not appear in the explicit local frame construction. We conclude by noting that, at the special relativistic level, a uniformly rotating observer and the corotating one are not the same as the latter is also accelerated (see Gourgoulhon [101], ch. 13). However, such a difference is irrelevant at the level of the Newtonian perturbation equations since i) pseudo-acceleration terms drop out of the perturbed Euler equation ii) non-inertial terms in the induction equation involving the four-acceleration are negligible in the Newtonian limit (cf. discussion in section 7.3).

D.1 Another look at the non-inertial MHD equations.

Before we move on to take a closer look at the Rayleigh criterion, let us here show how the terms involving the local frame rotation drop out of the induction equation. Even though we have already argued this happens in general (cf. eq. (7.21)), we here prove this for the specific case of a co-rotating observer. We do so as this allows us to appreciate better why the cancellation comes about.

We show this using the notion of spin coefficients associated with a uniformly rotating observer. That this is going to lead us where we want can be anticipated by recalling the result in eq. (A.9) and that we made use of the bi-linear form Ω in the argument in section 7.3. We then introduce (Born) coordinates associated with a uniformly rotating observer (axes suitably oriented so that the angular velocity is $\Omega_0 \hat{z}$)

$$t' = t$$
, $z' = z$, $x' = R\cos(\Omega_0 t + \varphi)$, $y' = R\sin(\Omega_0 t + \varphi)$, (D.7)

where primed coordinates are Cartesian (i.e. non-rotating). Computing the spin coefficients (starting from a flat metric) we then obtain

$$\omega_{\varphi \, \hat{\varphi}}^{\,\hat{R}} = -1 \,, \quad \omega_{\varphi \, \hat{R}}^{\,\hat{\varphi}} = +1 \,, \quad \omega_{t \, \hat{\varphi}}^{\,\hat{R}} = -\Omega_0 \,, \quad \omega_{t \, \hat{R}}^{\,\hat{\varphi}} = \Omega_0 \,, \tag{D.8}$$

showing that, as the coordinates "mix space and time" we need to introduce a covariant derivative in the time-direction as well. We then write the non-inertial induction equation as (cf. eq. (7.19))

$$\nabla_t B^{\hat{i}} + \nabla_{\hat{j}} \left(v^{\hat{j}} B^{\hat{i}} - v^{\hat{i}} B^{\hat{j}} \right) + \epsilon^{\hat{i}\hat{j}\hat{k}} B_{\hat{j}} \Omega^0_{\hat{k}} = 0 .$$
 (D.9)

It is then easy to verify, by means of eq. (D.8) that

$$\nabla_t B^{\hat{i}} + \epsilon^{\hat{i}\hat{j}\hat{k}} B_{\hat{j}} \Omega^0_{\hat{k}} = \partial_t B^{\hat{i}} , \qquad (D.10)$$

thus confirming the result in eq. (7.21) and the use of the inertial induction equations. The co-rotating frame rotation vector is $\Omega_0 \hat{e}_3$, which happens to be the same (by construction) as the vorticity of the observer. This is why we see the same cancellation as in eq. (7.21) where we assumed the two are equal.

D.2 A closer look at the Rayleigh stability criterion

The key point of the magneto-rotational instability is that the circular velocity background is stable against axisymmetric hydrodynamic perturbations, while adding a (however weak) magnetic field changes completely the nature of the system and makes it unstable to such perturbations. We now revisit the Rayleigh criterion with the aim of bringing to the fore important aspects to be kept in mind when looking at the general results derived in the main text, specifically in sections 7.4 and 7.5.

Starting from eq. (D.4), and ignoring terms associated with the magnetic field, we write the coefficients matrix (ordering the perturbed quantities as $\{\delta \rho / \rho, \delta v^{\hat{R}}, \delta v^{\hat{\varphi}}, \delta v^{\hat{z}}\}$)

$$\begin{pmatrix} -\omega & k_1 & k_2 & k_3 \\ c_s^2 k_1 & -\omega & 2i\Omega_0 & 0 \\ c_s^2 k_2 & -i\frac{\kappa^2}{2\Omega_0} & -\omega & 0 \\ c_s^2 k_3 & 0 & 0 & -\omega \end{pmatrix}, \qquad \frac{\kappa^2}{2\Omega_0} = 2\Omega_0 + s_0 = 2\Omega_0 + \frac{\mathrm{d}\Omega}{\mathrm{d}\ln R}\Big|_{R_0}, \qquad (D.11)$$

and the dispersion relation reads

$$\omega^4 - (c_s^2 \mathbf{k}^2 - \kappa^2) \,\omega^2 + i c_s^2 s_0 k_1 k_2 \omega + c_s^2 \kappa^2 (k_3)^2 = 0 \,. \tag{D.12}$$

If we now consider the sound-proof limit, namely $c_s^2 \gg 1$, we then end up with

$$\mathbf{k}^2 \omega^2 - i s_0 k_1 k_2 \omega - \kappa^2 k_3^2 = 0 .$$
 (D.13)

From this we easily see that, if we assume axisymmetric perturbations, namely $k_2 = 0$, we obtain the usual Rayleigh stability criterion, that is $\kappa^2 > 0$. We stress that, as is well-known, the criterion does not guarantee that non-axisymmetric modes are stable. In fact, rewriting the dispersion relation in terms of $\Delta = -i\omega$ and using the Routh-Hurwitz criterion (cf. appendix E) we find that, on top of the Rayleigh criterion we would also need

$$s_0 k_1 k_2 \le 0$$
. (D.14)

We also note that, the story changes if we take the opposite limit instead, namely $c_s^2 \ll 1$, in which case the Rayleigh criterion is sufficient to guarantee stability of also non-axisymmetric perturbations. This would also be the case had we assumed incompressibility from the start.

Having discussed the usual Rayleigh criterion using the co-rotating observer, we now re-work through it using an observer that is orbiting with the fluid at a given orbital distance R_0 but whose (local frame) axes are non rotating. We do this for two reasons. First, it will allow for a direct comparison with the general results discussed in the main text, specifically in section 7.4. Second, we have argued that choosing to work with a rotating or non-rotating observer is, in general, just a matter of taste. We then pick up an orbit R_0 as before, choose the observer to be co-orbiting with the background flow at the specific orbit

$$\mathbf{v}_{obs} = -\Omega_0 y_0 \hat{x} + \Omega_0 x_0 \hat{y} , \qquad (D.15)$$

where we used global Cartesian coordinates and $x_0(t) = R_0 \cos(\Omega_0 t)$, $y_0(t) = R_0 \sin(\Omega_0 t)$ describe the worldline of the observer (the origin of the axes is suitably chosen so that

 $z_0(t) = 0$). The background fluid velocity is then

$$\mathbf{v} = -\Omega x \hat{y} + \Omega y \hat{x}$$
, $\Omega = \Omega(\sqrt{x^2 + y^2})$ (D.16)

so that, considering the relative velocity $\mathbf{v}' = \mathbf{v} - \mathbf{v}_{obs}$ and expanding around (x_0, y_0) we obtain

$$\mathbf{v}' = -\left[s_0 \frac{x_0 y_0}{x_0^2 + y_0^2} x' + \left(\Omega_0 + s_0 \frac{y_0^2}{x_0^2 + y_0^2}\right) y'\right] \hat{x} + \left[\left(\Omega_0 + s_0 \frac{x_0^2}{x_0^2 + y_0^2}\right) x' + s_0 \frac{x_0 y_0}{x_0^2 + y_0^2} x'\right] \hat{y} , \quad (D.17)$$

where $x' = x - x_0$, $y' = y - y_0$. We can now choose a local region around a specific point (x_0, y_0, z_0) on the orbit and choose to re-orient the axes by a constant rotation so that the observer velocity is moving only in the *y*-direction. We then set up the local frame in such a way that the local axis are non-rotating and oriented like the global cartesian ones. We can therefore write the gradients as

$$\partial'_i v'_j = \begin{pmatrix} 0 & \Omega_0 + s_0 & 0\\ -\Omega_0 & 0 & 0\\ 0 & 0 & 0 \end{pmatrix} , \qquad (D.18)$$

and the coefficients matrix of the linearized Euler plus continuity system is (cf. eq. (D.4) and ignore both magnetic field terms and the Coriolis force as the axis are non-rotating)

$$\begin{pmatrix} -\omega & k_1 & k_2 & k_3 \\ c_s^2 k_1 & -\omega & i\Omega_0 & 0 \\ c_s^2 k_2 & -i(\Omega_0 + s_0) & -\omega & 0 \\ c_s^2 k_3 & 0 & 0 & -\omega \end{pmatrix}.$$
 (D.19)

We can then compute the dispersion relation to find

$$\omega^4 - \left[c_s^2 \mathbf{k}^2 + \Omega_0(\Omega_0 + s_0)\right] \omega^2 + i c_s^2 k_1 k_2 s_0 \,\omega + c_s^2 \Omega_0(\Omega_0 + s_0) (k_3)^2 = 0 , \qquad (D.20)$$

and observe this is consistent with the general dispersion relation in eq. (7.40) when restricted to the shear and vorticity associated with eq. (D.18). However, this is not quite the dispersion relation we obtained above (cf. eq. (D.12)). The reason for this is that the two local observers we have considered measure different frequencies, as the axes of the co-rotating observer rotate with angular velocity $\Omega_0 \hat{e}_3$ with respect to the other. To show why this is the resolution to the apparent conflict, let us consider once again the Born coordinates (cf. eqs. (D.7) and (D.8)). Given any vector $a^{\hat{i}}$ we have

$$\nabla_t a^{\hat{i}} = \partial_t \hat{a}^i + \Omega_0 \epsilon^{\hat{i}\hat{3}\hat{k}} a_{\hat{k}} . \tag{D.21}$$

This relation, when we introduce a plane-wave WKB expansion translates to

$$-i\omega_{rot}\delta a^{\hat{i}} = -i\omega_{nr}\delta a^{\hat{i}} + \Omega_0 \epsilon^{\hat{i}\hat{3}\hat{k}}\delta a_{\hat{k}} , \qquad (D.22)$$

where ω_{rot} is the frequency measured by the co-rotating observer, while ω_{nr} is the frequency measured by an observer that has the same worldline but uses non-rotating axes. Specifying eq. (D.22) to the perturbed velocity (noting that it would not apply to the continuity equation as the density is a scalar), and noting that the frequency in eq. (D.19) correspons to ω_{nr} , we can reconcile the results obtained from eq. (D.19) with those from eq. (D.11). We also note here that the same logic applies when we consider magnetized flows. That is, if we work with the inertial induction equation and compute the background velocity gradients as in eq. (D.18), we also need to take into account of the relation in eq. (D.22) for magnetic field disturbances to get back to eq. (D.6) and the MRI dispersion relation.

Appendix E

The Routh-Hurwitz criterion

In this appendix we review the Routh-Hurwitz criterion (see [131]), which gives valuable information about the roots of a polynomial with real coefficients. The criterion is often useful for studies of the linear stability of a system of equations, and can also be used prior to a numerical investigation (to inform the numerical study).

Given a real algebraic equation

$$x^{n} + \tilde{a}_{1}x^{n-1} + \dots + \tilde{a}_{n-1}x + \tilde{a}_{n} = 0$$
, (E.1)

the Routh-Hurwitz criterion states that the number of roots with positive real part corresponds to the number of sign changes—disregarding vanishing terms—in the following sequence

$$T_0$$
, T_1 , T_1T_2 , T_2T_3 , ..., $T_{n-1}T_{n-2}$, \tilde{a}_n , (E.2)

where

$$T_{0} = 1, \quad T_{1} = \tilde{a}_{1}, \quad T_{2} = \det \begin{pmatrix} \tilde{a}_{1} & 1 \\ \tilde{a}_{3} & \tilde{a}_{2} \end{pmatrix}, \quad T_{3} = \det \begin{pmatrix} \tilde{a}_{1} & 1 & 0 \\ \tilde{a}_{3} & \tilde{a}_{2} & \tilde{a}_{1} \\ \tilde{a}_{5} & \tilde{a}_{4} & \tilde{a}_{3} \end{pmatrix}, \quad (E.3)$$

and so on. Throughout this work we have derived a number of dispersion relations in terms of the frequency ω (as a function of the wave-vector **k**), whereby linear stability corresponds to its roots having negative imaginary part. Rewriting the dispersion relation in terms of $\Delta = -i\omega$, stability corresponds to Δ -roots having negative real part. As such, using of the Routh-Hurwitz criterion we can directly obtain information about the stability of a system without having to explicitly find the solutions to the dispersion relation—which is always possible numerically (but often requires an expensive parameter study) while is viable analytically only in special/simple cases.

We conclude this appendix by noting a caveat that is not explicitly mentioned in [131], namely that Routh-Hurwitz criterion can be used *only* when all the coefficients in a

polynomial are non-vanishing. We show this with a trivial trivial example:

$$(x^2 - 2)(x^2 - 3) = x^4 - 5x^2 + 6 = 0 \Longrightarrow x = \pm\sqrt{2}, \pm\sqrt{3}.$$
 (E.4)

Using the Routh-Hurwitz criterion, the sequence we obtain (neglecting vanishing terms) is 1, 6 and we would deduce there are no roots with positive real part. This is obviously wrong.

References

- [1] B. P. Abbott et al. Observation of gravitational waves from a binary black hole merger. *Phys. Rev. Lett.*, 116:061102, Feb. 2016.
- [2] B. P. Abbott et al. GW170817: Observation of Gravitational Waves from a Binary Neutron Star Inspiral. *Phys. Rev. Lett.*, 119(16):161101, 2017.
- [3] B. P. Abbott et al. Multi-messenger Observations of a Binary Neutron Star Merger. Astrophys. J., Lett., 848(2):L12, 2017.
- [4] R. Aguilera-Miret, D. Viganò, F. Carrasco, B. Miñano, and C. Palenzuela. Turbulent magnetic-field amplification in the first 10 milliseconds after a binary neutron star merger: Comparing high-resolution and large-eddy simulations. *Phys. Rev. D*, 102:103006, Nov. 2020.
- [5] R. Aguilera-Miret, D. Viganò, and C. Palenzuela. Universality of the Turbulent Magnetic Field in Hypermassive Neutron Stars Produced by Binary Mergers. *Astrophys. J., Lett.*, 926(2):L31, 2022.
- [6] M. Alford, A. Harutyunyan, and A. Sedrakian. Bulk viscosity of baryonic matter with trapped neutrinos. *Phys. Rev. D*, 100(10):103021, 2019.
- [7] M. Alford, A. Harutyunyan, and A. Sedrakian. Bulk viscous damping of density oscillations in neutron star mergers. *Particles*, 3(2):500–517, 2020.
- [8] M. Alford, A. Harutyunyan, and A. Sedrakian. Bulk viscosity from Urca processes: npeµ matter in the neutrino-trapped regime. *Phys. Rev. D*, 104(10):103027, 2021.
- [9] M. G. Alford and S. P. Harris. β equilibrium in neutron-star mergers. *Phys. Rev. C*, 98:065806, Dec. 2018.
- [10] M. G. Alford and S. P. Harris. Damping of density oscillations in neutrinotransparent nuclear matter. *Phys. Rev. C*, 100:035803, Sept. 2019.
- [11] M. G. Alford, S. Mahmoodifar, and K. Schwenzer. Large amplitude behavior of the bulk viscosity of dense matter. J. Phys. G, 37:125202, 2010.

- [12] M. G. Alford, L. Bovard, M. Hanauske, L. Rezzolla, and K. Schwenzer. Viscous Dissipation and Heat Conduction in Binary Neutron-Star Mergers. *Phys. Rev. Lett.*, 120(4):041101, Jan. 2018.
- [13] M. G. Alford, L. Bovard, M. Hanauske, L. Rezzolla, and K. Schwenzer. Viscous dissipation and heat conduction in binary neutron-star mergers. *Phys. Rev. Lett.*, 120:041101, Jan. 2018.
- [14] M. G. Alford, A. Haber, S. P. Harris, and Z. Zhang. Beta equilibrium under neutron star merger conditions. *Universe*, 7(11), 2021.
- [15] M. Anderson, E. W. Hirschmann, L. Lehner, S. L. Liebling, P. M. Motl, D. Neilsen, C. Palenzuela, and J. E. Tohline. Magnetized neutron-star mergers and gravitational-wave signals. *Phys. Rev. Lett.*, 100:191101, May 2008.
- [16] N. Andersson. Resistive relativistic magnetohydrodynamics from a charged multifluids perspective. *Phys. Rev. D*, 86:043002, Aug. 2012.
- [17] N. Andersson. Gravitational-wave Astronomy: Exploring the Dark Side of the Universe. Oxford University Press, 2020.
- [18] N. Andersson and G. L. Comer. A flux-conservative formalism for convective and dissipative multi-fluid systems, with application to newtonian superfluid neutron stars. *Class. Quant. Grav.*, 23:5505–5530, 2006.
- [19] N. Andersson and G. L. Comer. Variational multi-fluid dynamics and causal heat conductivity. Proc. R. Soc. Lond. A, 466:1373–1387, May 2010.
- [20] N. Andersson and G. L. Comer. A covariant action principle for dissipative fluid dynamics: from formalism to fundamental physics. *Class. Quant. Grav.*, 32(7): 075008, Apr. 2015.
- [21] N. Andersson and G. L. Comer. Relativistic fluid dynamics: physics for many different scales. *Living Rev. Rel.*, 24(1):3, 2021.
- [22] N. Andersson and C. S. Lopez-Monsalvo. A consistent first-order model for relativistic heat flow. *Class. Quant. Grav.*, 28(19):195023, Oct. 2011.
- [23] N. Andersson, G. L. Comer, and I. Hawke. A variational approach to resistive relativistic plasmas. *Class. Quant. Grav.*, 34(12):125001, May 2017.
- [24] N. Andersson, K. Dionysopoulou, I. Hawke, and G. L. Comer. Beyond ideal magnetohydrodynamics: resistive, reactive and relativistic plasmas. *Class. Quant. Grav.*, 34(12):125002, June 2017.
- [25] N. Andersson, I. Hawke, K. Dionysopoulou, and G. L. Comer. Beyond ideal magnetohydrodynamics: from fibration to 3 + 1 foliation. *Class. Quant. Grav.*, 34 (12):125003, June 2017.

- [26] N. Andersson, I. Hawke, T. Celora, and G. L. Comer. The physics of non-ideal general relativistic magnetohydrodynamics. *Mon. Not. R. Astron. Soc.*, 509(3): 3737–3750, 2021.
- [27] A. F. Andreev and E. P. Bashkin. Three-velocity hydrodynamics of superfluid solutions. J. Exp. Theor. Phys., 42:164, Jan. 1976.
- [28] A. M. Anile. *Relativistic fluids and magneto-fluids*. Cambridge University Press, 1989.
- [29] M. Antonelli, A. Montoli, and P. M. Pizzochero. Insights Into the Physics of Neutron Star Interiors from Pulsar Glitches. In Astrophysics in the XXI Century with Compact Stars. Edited by C.A.Z. Vasconcellos. eISBN 978-981-12-2094-4. Singapore: World Scientific, pages 219–281. 2022.
- [30] L. Baiotti and L. Rezzolla. Binary neutron star mergers: a review of einstein's richest laboratory. *Rep. Prog. Phys.*, 80(9):096901, July 2017.
- [31] S. A. Balbus and J. F. Hawley. A Powerful Local Shear Instability in Weakly Magnetized Disks. I. Linear Analysis. *Astrophys. J.*, 376:214, July 1991.
- [32] S. A. Balbus and J. F. Hawley. A Powerful Local Shear Instability in Weakly Magnetized Disks. IV. Nonaxisymmetric Perturbations. *Astrophys. J.*, 400:610– 621, Dec. 1992.
- [33] S. A. Balbus and J. F. Hawley. Instability, turbulence, and enhanced transport in accretion disks. *Rev. Mod. Phys.*, 70:1–53, Jan. 1998.
- [34] A. Barletta. The boussinesq approximation for buoyant flows. *Mech. Res. Commun.*, 124:103939, 2022.
- [35] J. D. Barrow, R. Maartens, and C. G. Tsagas. Cosmology with inhomogeneous magnetic fields. *Phys. Rept.*, 449:131–171, 2007.
- [36] A. Bauswein, O. Just, H.-T. Janka, and N. Stergioulas. Neutron-star radius constraints from GW170817 and future detections. *Astrophys. J.*, 850(2):L34, Nov. 2017.
- [37] F. Becattini. Thermodynamic Equilibrium in Relativity: Four-temperature, Killing Vectors and Lie Derivatives. *Acta Phys. Pol. B*, 47(7):1819, Jan. 2016.
- [38] J. D. Bekenstein and E. Oron. New conservation laws in general-relativistic magnetohydrodynamics. *Phys. Rev. D*, 18(6):1809–1819, Sept. 1978.
- [39] P. Bellan. Fundamentals of Plasma Physics. Cambridge University Press, 2008.
- [40] F. S. Bemfica, M. M. Disconzi, and J. Noronha. Causality and existence of solutions of relativistic viscous fluid dynamics with gravity. *Phys. Rev. D*, 98(10): 104064, 2018.

- [41] F. S. Bemfica, M. M. Disconzi, and J. Noronha. Causality of the Einstein-Israel-Stewart Theory with Bulk Viscosity. *Phys. Rev. Lett.*, 122(22):221602, 2019.
- [42] F. S. Bemfica, M. M. Disconzi, and J. Noronha. First-Order General-Relativistic Viscous Fluid Dynamics. *Phys. Rev. X*, 12(2):021044, 2022.
- [43] A. Beresnyak. Spectral slope and kolmogorov constant of mhd turbulence. *Phys. Rev. Lett.*, 106:075001, Feb. 2011.
- [44] A. Beresnyak. MHD turbulence. Living Rev. Comput. Astrophys., 5(1), Sept. 2019.
- [45] S. Bernuzzi. Neutron star merger remnants. Gen. Relativ. Gravit., 52:108, 2020.
- [46] J. Bhattacharya, S. Bhattacharyya, and M. Rangamani. Non-dissipative hydrodynamics: Effective actions versus entropy current. *J. High Energy Phys.*, 02:153, 2013.
- [47] A. Brandenburg and K. Subramanian. Astrophysical magnetic fields and nonlinear dynamo theory. *Physics Reports*, 417(1):1 – 209, 2005.
- [48] N. Bucciantini and L. Del Zanna. A fully covariant mean-field dynamo closure for numerical 3+1 resistive GRMHD. *Mon. Not. R. Astron. Soc.*, 428:71, 2013.
- [49] F. Carrasco, D. Viganò, and C. Palenzuela. Gradient subgrid-scale model for relativistic mhd large-eddy simulations. *Phys. Rev. D*, 101:063003, Mar. 2020.
- [50] S. Carroll. Spacetime and Geometry. Cambridge University Press, 2019.
- [51] B. Carter. Conductivity with causality in relativistic hydrodynamics: the regular solution to Eckart's problem. In *Gravitation and Cosmology*, pages 58–65, Jan. 1988.
- [52] B. Carter. Covariant theory of conductivity in ideal fluid or solid media. *Relativistic fluid dynamics*, 1385:1–64, 1989.
- [53] B. Carter. Convective variational approach to relativistic thermodynamics of dissipative fluids. *Proc. R. Soc. A*, 433(1887):45–62, 1991.
- [54] B. Carter and D. Langlois. Relativistic models for superconducting superfluid mixtures. *Nucl. Phys. B*, 531:478–504, 1998.
- [55] B. Carter and H. Quintana. Foundations of General Relativistic High-Pressure Elasticity Theory. *Proc. R. Soc. Lond. A*, 331(1584):57–83, Nov. 1972.
- [56] H. B. G. Casimir. On onsager's principle of microscopic reversibility. *Rev. Mod. Phys.*, 17:343–350, Apr 1945.
- [57] T. Celora, N. Andersson, and G. L. Comer. Linearizing a non-linear formulation for general relativistic dissipative fluids. *Class. Quant. Grav.*, 38(6), Mar. 2021.

- [58] T. Celora, N. Andersson, I. Hawke, and G. L. Comer. Covariant approach to relativistic large-eddy simulations: The fibration picture. *Phys. Rev. D*, 104(8): 084090, 2021.
- [59] T. Celora, I. Hawke, P. C. Hammond, N. Andersson, and G. L. Comer. Formulating bulk viscosity for neutron star simulations. *Phys. Rev. D*, 105(10):103016, 2022.
- [60] T. Celora, I. Hawke, N. Andersson, and G. L. Comer. Local magneto-shear instability in Newtonian gravity. *arXiv e-prints*, 4 2023.
- [61] C. Cercignani and G. M. Kremer. *The Relativistic Boltzmann Equation*. Springer-Verlag New York, 1998.
- [62] N. Chamel and P. Haensel. Physics of Neutron Star Crusts. *Living Rev. Rel.*, 11: 10, 2008.
- [63] S. Chandrasekhar. The Stability of Non-Dissipative Couette Flow in Hydromagnetics. Proc. Natl. Acad. Sci., 46(2):253–257, Feb. 1960.
- [64] R. Ciolfi. Binary neutron star mergers after gw170817. Frontiers in Astronomy and Space Sciences, 7:27, 2020.
- [65] F. Cipolletta, J. V. Kalinani, B. Giacomazzo, and R. Ciolfi. Spritz: a new fully general-relativistic magnetohydrodynamic code. *Class. Quant. Grav.*, 37(13): 135010, 2020.
- [66] G. L. Comer and R. Joynt. Relativistic mean field model for entrainment in general relativistic superfluid neutron stars. *Phys. Rev. D*, 68:023002, July 2003.
- [67] G. L. Comer and D. Langlois. Hamiltonian formulation for multi-constituent relativistic perfect fluids . *Class. Quant. Grav.*, 10:2317–2327, Nov. 1993.
- [68] G. L. Comer and D. Langlois. Hamiltonian formulation for relativistic superfluids. *Class. Quant. Grav.*, 11:709–721, Mar. 1994.
- [69] T. Damour, A. Nagar, and S. Bernuzzi. Improved effective-one-body description of coalescing nonspinning black-hole binaries and its numerical-relativity completion. *Phys. Rev. D*, 87:084035, Apr. 2013.
- [70] S. Dartevelle. Comprehensive Approaches to Multiphase Flows in Geophysics. United States. National Nuclear Security Administration, 2005.
- [71] M. D. Duez, Y. T. Liu, S. L. Shapiro, M. Shibata, and B. C. Stephens. Collapse of magnetized hypermassive neutron stars in general relativity. *Phys. Rev. Lett.*, 96: 031101, Jan. 2006.

- [72] M. D. Duez, A. Knight, F. Foucart, M. Haddadi, J. Jesse, F. Hebert, L. E. Kidder, H. P. Pfeiffer, and M. A. Scheel. Comparison of momentum transport models for numerical relativity. *Phys. Rev. D*, 102(10):104050, 2020.
- [73] C. Eckart. The Thermodynamics of Irreversible Processes. III. Relativistic Theory of the Simple Fluid. *Phys. Rev.*, 58:919–924, Nov. 1940.
- [74] G. F. R. Ellis. Relativistic cosmology. Cargese Lect. Phys., 6:1–60, 1973.
- [75] G. F. R. Ellis. Inhomogeneity effects in Cosmology. Class. Quant. Grav., 28:164001, 2011.
- [76] M. Evans et al. A Horizon Study for Cosmic Explorer: Science, Observatories, and Community. arXiv e-prints, Sept. 2021.
- [77] G. L. Eyink and T. D. Drivas. Cascades and dissipative anomalies in relativistic fluid turbulence. *Phys. Rev. X*, 8:011023, Feb. 2018.
- [78] E. Fermi. On the phenomena that occur in the neighborhood of a world line. *Atti R. Accad. Lincei Rend. Cl. Sci. Fis. Mat. Nat.*, 31(1):21, 1922.
- [79] E. Fermi. On the phenomena that occur in the neighborhood of a world line. *Atti R. Accad. Lincei Rend. Cl. Sci. Fis. Mat. Nat.*, 31(1):51, 1922.
- [80] G. Fiorella Burgio and A. F. Fantina. Nuclear Equation of state for Compact Stars and Supernovae. Astrophys. Space Sci. Libr., 457:255–335, 2018.
- [81] J. Friedman and N. Stergioulas. *Rotating Relativistic Stars*. Cambridge Monographs on Mathematical Physics. Cambridge University Press, 2013.
- [82] J. L. Friedman and B. F. Schutz. On the stability of relativistic systems. Astrophys. J., 200:204–220, Aug. 1975.
- [83] J. L. Friedman and B. F. Schutz. Lagrangian perturbation theory of nonrelativistic fluids. *Astrophys. J.*, 221:937–957, May 1978.
- [84] U. Frisch. Turbulence. The legacy of A.N. Kolmogorov. Cambridge University Press, 1995.
- [85] M. Fukuma and Y. Sakatani. Entropic formulation of relativistic continuum mechanics. *Phys. Rev. E*, 84(2):026315, Aug. 2011.
- [86] S. Galtier. *Introduction to Modern Magnetohydrodynamics*. Cambridge University Press, 2016.
- [87] L. Gavassino and M. Antonelli. Thermodynamics of uncharged relativistic multifluids. *Class. Quant. Grav.*, 37(2):025014, Jan. 2020.

- [88] L. Gavassino and M. Antonelli. Unified Extended Irreversible Thermodynamics and the stability of relativistic theories for dissipation. *Front. Astron. Space Sci.*, 8: 686344, 2021.
- [89] L. Gavassino, M. Antonelli, and B. Haskell. Bulk viscosity in relativistic fluids: from thermodynamics to hydrodynamics. *Class. Quant. Grav.*, 38(7):075001, 2021.
- [90] R. P. Geroch. Relativistic theories of dissipative fluids. *J. Math. Phys.*, 36:4226, 1995.
- [91] R. P. Geroch and L. Lindblom. Dissipative relativistic fluid theories of divergence type. *Phys. Rev. D*, 41:1855, 1990.
- [92] R. P. Geroch and L. Lindblom. Causal theories of dissipative relativistic fluids. *Annals Phys.*, 207(2):394–416, May 1991.
- [93] B. Giacomazzo and L. Rezzolla. WhiskyMHD: A New numerical code for general relativistic magnetohydrodynamics. *Class. Quant. Grav.*, 24:S235–S258, 2007.
- [94] B. Giacomazzo, J. Zrake, P. Duffell, A. I. MacFadyen, and R. Perna. Producing Magnetar Magnetic Fields in the Merger of Binary Neutron Stars. *Astrophys. J.*, 809(1):39, 2015.
- [95] P. Goldreich and D. Lynden-Bell. II. Spiral arms as sheared gravitational instabilities. Mon. Not. R. Astron. Soc., 130:125, Jan. 1965.
- [96] P. Goldreich and S. Sridhar. Toward a theory of interstellar turbulence. 2. Strong Alfvenic turbulence. *Astrophys. J.*, 438:763–775, 1995.
- [97] P. Goldreich and S. Sridhar. Magnetohydrodynamic turbulence revisited. Astrophys. J., 485(2):680, Aug. 1997.
- [98] J. Goodman and G. Xu. Parasitic Instabilities in Magnetized, Differentially Rotating Disks. Astrophys. J., 432:213, Sept. 1994.
- [99] E. Gourgoulhon. An Introduction to the theory of rotating relativistic stars. In CompStar 2010: School and Workshop on Computational Tools for Compact Star Astrophysics, Mar. 2010.
- [100] E. Gourgoulhon. 3+1 Formalism in General Relativity. Springer-Verlag Berlin Heidelberg, 2012.
- [101] E. Gourgoulhon. *Special Relativity in General Frames*. Springer-Verlag Berlin Heidelberg, 2013.
- [102] J. Guilet, A. Bauswein, O. Just, and H.-T. Janka. Magnetorotational instability in neutron star mergers: impact of neutrinos. *Mon. Not. R. Astron. Soc.*, 471(2): 1879–1887, July 2017.

- [103] P. Haensel, A. Potekhin, and D. Yakovlev. Neutron Stars 1: Equation of State and Structure. Astrophysics and Space Science Library. Springer New York, 2007.
- [104] P. Hammond, I. Hawke, and N. Andersson. Thermal aspects of neutron star mergers. *Phys. Rev. D*, 104(10):103006, 2021.
- [105] P. Hammond, I. Hawke, and N. Andersson. Impact of nuclear reactions on gravitational waves from neutron star mergers. *Phys. Rev. D*, 107(4):043023, 2023.
- [106] P. C. Hammond. *Numerical evolution of binary neutron star mergers*. PhD thesis, University of Southampton, 2022.
- [107] B. Haskell and A. Melatos. Models of pulsar glitches. Int. J. Mod. Phys. D, 24(3): 1530008, Jan. 2015.
- [108] J. F. Hawley and S. A. Balbus. A Powerful Local Shear Instability in Weakly Magnetized Disks. II. Nonlinear Evolution. *Astrophys. J.*, 376:223, July 1991.
- [109] J. F. Hawley, C. F. Gammie, and S. A. Balbus. Local Three-dimensional Magnetohydrodynamic Simulations of Accretion Disks. *Astrophys. J.*, 440:742, Feb. 1995.
- [110] J. F. Hawley, X. Guan, and J. H. Krolik. Assessing Quantitative Results in Accretion Simulations: From Local to Global. *Astrophys. J.*, 738:84, 2011.
- [111] J. F. Hawley, S. A. Richers, X. Guan, and J. H. Krolik. Testing Convergence for Global Accretion Disks. *Astrophys. J.*, 772:102, 2013.
- [112] K. Hayashi, K. Kiuchi, K. Kyutoku, Y. Sekiguchi, and M. Shibata. Generalrelativistic neutrino-radiation magnetohydrodynamics simulation of secondslong black hole-neutron star mergers: Dependence on initial magnetic field strength, configuration, and neutron-star equation of state. *preprint*, Nov. 2022.
- [113] G. W. Hill. Researches in the lunar theory. Am. J. Math., 1(1):5–26, 1878.
- [114] W. A. Hiscock and L. Lindblom. Stability and causality in dissipative relativistic fluids. *Annals Phys.*, 151(2):466–496, Jan. 1983.
- [115] W. A. Hiscock and L. Lindblom. Generic instabilities in first-order dissipative relativistic fluid theories. *Phys. Rev. D*, 31:725–733, 1985.
- [116] W. A. Hiscock and L. Lindblom. Nonlinear pathologies in relativistic heatconducting fluid theories. *Phys. Lett. A*, 131(9):509–513, Sept. 1988.
- [117] R. E. Hoult and P. Kovtun. Stable and causal relativistic Navier-Stokes equations. J. High Energy Phys., 06:67, 2020.
- [118] W. Israel. Nonstationary irreversible thermodynamics: A causal relativistic theory. Annals Phys., 100(1):310 – 331, 1976.

- [119] W. Israel and J. M. Stewart. Transient relativistic thermodynamics and kinetic theory. *Annals Phys.*, 118(2):341–372, Apr. 1979.
- [120] W. Israel and J. M. Stewart. On transient relativistic thermodynamics and kinetic theory. II. Proc. R. Soc. Lond. A, 365(1720):43–52, Feb. 1979.
- [121] D. Jou, G. Lebon, and J. Casas-Vázquez. Extended Irreversible Thermodynamics. Springer Netherlands, 3 edition, 2010.
- [122] M. Karlovini and L. Samuelsson. Elastic stars in general relativity: IV. Axial perturbations. *Class. Quant. Grav.*, 24(13):3171–3189, July 2007.
- [123] D. Kasen, B. Metzger, J. Barnes, E. Quataert, and E. Ramirez-Ruiz. Origin of the heavy elements in binary neutron-star mergers from a gravitational wave event. *Nature*, 551:80, 2017.
- [124] K. Kiuchi, K. Kyutoku, Y. Sekiguchi, M. Shibata, and T. Wada. High resolution numerical relativity simulations for the merger of binary magnetized neutron stars. *Phys. Rev. D*, 90:041502, Aug. 2014.
- [125] K. Kiuchi, P. Cerdá-Durán, K. Kyutoku, Y. Sekiguchi, and M. Shibata. Efficient magnetic-field amplification due to the kelvin-helmholtz instability in binary neutron star mergers. *Phys. Rev. D*, 92:124034, Dec. 2015.
- [126] K. Kiuchi, K. Kyutoku, Y. Sekiguchi, and M. Shibata. Global simulations of strongly magnetized remnant massive neutron stars formed in binary neutron star mergers. *Phys. Rev. D*, 97(12), June 2018.
- [127] K. Kiuchi, S. Fujibayashi, K. Hayashi, K. Kyutoku, Y. Sekiguchi, and M. Shibata. Self-consistent picture of the mass ejection from a one second-long binary neutron star merger leaving a short-lived remnant in general-relativistic neutrinoradiation magnetohydrodynamic simulation. *preprint*, Nov. 2022.
- [128] A. N. Kolmogorov. The Local Structure of Turbulence in Incompressible. Viscous Fluid for Very Large Reynolds' Numbers. *Dokl. Akad. Nauk SSSR*, 30:301–305, Jan. 1941a.
- [129] A. N. Kolmogorov. On the degeneration of isotropic turbulence in an incompressible viscous fluid. *Dokl. Akad. Nauk SSSR*, 31:319–323, 1941b.
- [130] A. N. Kolmogorov. Dissipation of Energy in Locally Isotropic Turbulence. Dokl. Akad. Nauk SSSR, 32:16, Apr. 1941c.
- [131] G. Korn and T. Korn. *Mathematical Handbook for Scientists and Engineers*. Dover Civil and Mechanical Engineering. Dover Publications, 2013.
- [132] P. Kovtun. Lectures on hydrodynamic fluctuations in relativistic theories. *J. Phys. A*, 45:473001, 2012.

- [133] P. Kovtun. First-order relativistic hydrodynamics is stable. *J. High Energy Phys.*, 10:034, 2019.
- [134] L. D. Landau and E. M. Lifshitz. Fluid mechanics. Pergamon Press, 1959.
- [135] J. Lattimer. Neutron stars and the nuclear matter equation of state. Annu. Rev. Nucl. Part. Sci., 71(1):433–464, 2021.
- [136] T. I. L.C. Berselli and W. Layton. *Mathematics of Large Eddy Simulation of Turbulent Flows*. Springer, Heidelberg, 2006.
- [137] A. Leonard. Energy cascade in large-eddy simulations of turbulent fluid flows. *Adv. in Geophysics A 18*, 1974.
- [138] M. Lesieur. Turbulence in Fluids. Springer Science & Business Media, 2008.
- [139] D. K. Lilly. The representation of small-scale turbulence in numerical simulation experiments. *Proc. IBM Scientific Computing Symposium on Environmental Sciences*, 1967.
- [140] L. Lindblom. The Relaxation effect in dissipative relativistic fluid theories. *Annals Phys.*, 247:1, 1996.
- [141] R. W. Lindquist. Relativistic transport theory. Annals Phys., 37(3):487–518, May 1966.
- [142] C. S. Lopez-Monsalvo and N. Andersson. Thermal dynamics in general relativity. *Proc. R. Soc. Lond. A*, 467(2127):738–759, Mar. 2011.
- [143] P. C. M. Lesieur, O. Métais. Large-Eddy Simulations of Turbulence. Cambridge University Press, 2005.
- [144] M. Maggiore. *A Modern Introduction to Quantum Field Theory*. Oxford University Press, 2005.
- [145] M. Maggiore. *Gravitational Waves: Theory and Experiments,* volume 1 of *Gravitational Waves.* OUP Oxford, 2008.
- [146] M. Maggiore et al. Science Case for the Einstein Telescope. J. Cosmol. Astropart. Phys., 03:050, 2020.
- [147] S. M. Mahajan and V. Krishan. Existence of the Magnetorotational Instability. *Astrophys. J.*, 682(1):602–607, July 2008.
- [148] F. K. Manasse and C. W. Misner. Fermi normal coordinates and some basic concepts in differential geometry. J. Math. Phys., 4(6):735–745, 1963.
- [149] B. Margalit and B. D. Metzger. Constraining the Maximum Mass of Neutron Stars From Multi-Messenger Observations of GW170817. *Astrophys. J., Lett.*, 850 (2):L19, 2017.

- [150] B. Margalit, A. S. Jermyn, B. D. Metzger, L. F. Roberts, and E. Quataert. Angularmomentum Transport in Proto-neutron Stars and the Fate of Neutron Star Merger Remnants. *Astrophys. J.*, 939(1):51, 2022.
- [151] J. McDonough. *Introductory Lectures on Turbulence*. CreateSpace Independent Publishing Platform, 2014.
- [152] B. Metzger. Kilonovae. Living Rev. Relativ., 23(1), Jan. 2020.
- [153] D. Mihalas and B. Weibel-Mihalas. Foundations of Radiation Hydrodynamics. Dover Books on Physics. Dover Publications, 1999. URL https://books.google.co. uk/books?id=f75C_GN9KZwC.
- [154] M. Miravet-Tenés, P. Cerdá-Durán, M. Obergaulinger, and J. A. Font. Assessment of a new sub-grid model for magnetohydrodynamical turbulence. I. Magnetorotational instability. *Mon. Not. R. Astron. Soc.*, 517(3):3505–3524, 2022.
- [155] C. Misner, K. Thorne, J. Wheeler, and D. Kaiser. *Gravitation*. Princeton University Press, 2017.
- [156] D. Montenegro and G. Torrieri. Causality and dissipation in relativistic polarizable fluids. *Phys. Rev. D*, 100(5):056011, Sept. 2019.
- [157] D. Montenegro and G. Torrieri. Linear response theory and effective action of relativistic hydrodynamics with spin. *Phys. Rev. D*, 102(3):036007, 2020.
- [158] E. R. Most, S. P. Harris, C. Plumberg, M. G. Alford, J. Noronha, J. Noronha-Hostler, F. Pretorius, H. Witek, and N. Yunes. Projecting the likely importance of weak-interaction-driven bulk viscosity in neutron star mergers. *Mon. Not. R. Astron. Soc.*, 509(1):1096–1108, Oct. 2021.
- [159] E. R. Most, A. Haber, S. P. Harris, Z. Zhang, M. G. Alford, and J. Noronha. Emergence of microphysical viscosity in binary neutron star post-merger dynamics. *preprint*, July 2022.
- [160] I. Müller. Zum Paradoxon der Warmeleitungstheorie. Z. Phys., 198:329–344, 1967.
- [161] I. Müller and T. Ruggeri. Rational extended thermodynamics. Springer-Verlag New York, 1998.
- [162] C. Musolino and L. Rezzolla. A practical guide to a moment approach for neutrino transport in numerical relativity. *arXiv e-prints*, 4 2023.
- [163] A. I. Nesterov. Riemann normal coordinates, Fermi reference system and the geodesic deviation equation. *Class. Quant. Grav.*, 16:465–477, 1999.
- [164] M. Oertel, M. Hempel, T. Klähn, and S. Typel. Equations of state for supernovae and compact stars. *Rev. Mod. Phys.*, 89:015007, Mar. 2017.

- [165] T. S. Olson and W. A. Hiscock. Stability, Causality and Hyperbolicity in Carter's 'Regular' Theory of Relativistic Heat Conducting Fluids. *Phys. Rev. D*, 41:3687, 1990.
- [166] L. Onsager. Reciprocal relations in irreversible processes. i. Phys. Rev., 37:405–426, Feb. 1931.
- [167] B. Paczynski. Gamma-ray bursters at cosmological distances. Astrophys. J., Lett., 308:L43–L46, Sept. 1986.
- [168] K. Palapanidis. *Relativistic fluid dynamics and electromagnetic media*. PhD thesis, University of Southampton, Oct. 2018.
- [169] C. Palenzuela, R. Aguilera-Miret, F. Carrasco, R. Ciolfi, J. V. Kalinani, W. Kastaun, B. Miñano, and D. Viganò. Turbulent magnetic field amplification in binary neutron star mergers. *Phys. Rev. D*, 106(2):023013, 2022.
- [170] G. Pavliotis and A. Stuart. *Multiscale Methods*. Springer Science+Business Media, LLC, 2008.
- [171] E. Pian et al. Spectroscopic identification of r-process nucleosynthesis in a double neutron star merger. *Nature*, 551:67–70, 2017.
- [172] E. Poisson and C. Will. *Gravity: Newtonian, Post-Newtonian, Relativistic*. Cambridge University Press, 2014.
- [173] D. J. Price and S. Rosswog. Producing ultrastrong magnetic fields in neutron star mergers. *Science*, 312(5774):719–722, 2006.
- [174] D. Priou. Comparison between variational and traditional approaches to relativistic thermodynamics of dissipative fluids. *Phys. Rev. D*, 43:1223–1234, Feb. 1991.
- [175] R. Prix. Variational description of multifluid hydrodynamics: Uncharged fluids. *Phys. Rev. D*, 69:043001, Feb. 2004.
- [176] D. Radice. General-relativistic large-eddy simulations of binary neutron star mergers. *Astrophys. J.*, 838(1):L2, Mar. 2017.
- [177] D. Radice. Binary Neutron Star Merger Simulations with a Calibrated Turbulence Model. Symmetry, 12(8):1249, 2020.
- [178] D. Radice, S. Bernuzzi, A. Perego, and R. Haas. A new moment-based generalrelativistic neutrino-radiation transport code: Methods and first applications to neutron star mergers. *Mon. Not. R. Astron. Soc.*, 512(1):1499–1521, 2022.
- [179] M. Rakhmanov. Fermi-normal, optical, and wave-synchronous coordinates for spacetime with a plane gravitational wave. *Class. Quant. Grav.*, 31:085006, 2014.

- [180] L. Rayleigh. On the dynamics of revolving fluids. Proc. R. Soc. Lond. A, 93(648): 148–154, 1917.
- [181] L. E. Reichl. A Modern Course in Statistical Physics. Wiley-VCH, Weinheim, Germany, 2016.
- [182] D. Reitze et al. Cosmic Explorer: The U.S. Contribution to Gravitational-Wave Astronomy beyond LIGO. *Bull. Am. Astron. Soc.*, 51(7):035, 2019.
- [183] L. Rezzolla and J. C. Miller. Relativistic radiative transfer for spherical flows. *Class. Quant. Grav.*, 11:1815–1832, 1994.
- [184] L. Rezzolla and O. Zanotti. *Relativistic Hydrodynamics*. Oxford University Press, 2018.
- [185] L. Rezzolla, E. R. Most, and L. R. Weih. Using gravitational-wave observations and quasi-universal relations to constrain the maximum mass of neutron stars. *Astrophys. J.*, 852(2):L25, Jan. 2018.
- [186] L. Rezzolla, P. Pizzochero, D. Jones, N. Rea, and I. Vidaña. *The Physics and Astro-physics of Neutron Stars*. Astrophysics and Space Science Library. Springer International Publishing, 2019.
- [187] F. Rincon. Dynamo theories. J. Plasma Phys., 85(4):205850401, 2019.
- [188] P. Romatschke and U. Romatschke. *Relativistic Fluid Dynamics In and Out of Equilibrium*. Cambridge Monographs on Mathematical Physics. Cambridge University Press, May 2019.
- [189] M. Ruiz, S. L. Shapiro, and A. Tsokaros. Gw170817, general relativistic magnetohydrodynamic simulations, and the neutron star maximum mass. *Phys. Rev. D*, 97:021501, Jan. 2018.
- [190] J. F. Salazar and T. Zannias. On Extended Thermodynamics: From Classical to the Relativistic Regime. *Int. J. Mod. Phys. D*, 29:2030010, 2020.
- [191] R. F. Sawyer. Bulk viscosity of hot neutron-star matter and the maximum rotation rates of neutron stars. *Phys. Rev. D*, 39:3804–3806, June 1989.
- [192] A. A. Schekochihin. MHD turbulence: a biased review. J. Plasma Phys., 88(5): 155880501, 2022.
- [193] W. Schmidt. Large Eddy Simulations in Astrophysics. *Living rev. comput. astro-phys.*, 1(1):2, Oct. 2015.
- [194] A. Schmitt and P. Shternin. Reaction rates and transport in neutron stars. Astrophys. Space Sci. Libr., 457:455–574, 2018.

- [195] A. S. Schneider, C. Constantinou, B. Muccioli, and M. Prakash. Akmalpandharipande-ravenhall equation of state for simulations of supernovae, neutron stars, and binary mergers. *Phys. Rev. C*, 100:025803, Aug. 2019.
- [196] B. F. Schutz and R. Sorkin. Variational aspects of relativistic field theories, with application to perfect fluids. *Annals Phys.*, 107:1–43, Jan. 1977.
- [197] N. Shakura, K. Postnov, D. Kolesnikov, and G. Lipunova. On the appearance of non-local MRI in Keplerian accretion discs. *preprint*, Oct. 2022.
- [198] N. I. Shakura and R. A. Sunyaev. Black holes in binary systems. observational appearance. Astron. Astrophys., 24:337–355, 1973.
- [199] S. Shapiro and S. Teukolsky. Black Holes, White Dwarfs and Neutron Stars: The Physics of Compact Objects. A Wiley-interscience publication. Wiley, 1983.
- [200] M. Shibata. Numerical Relativity. World Scientific, 2015.
- [201] M. Shibata and K. Hotokezaka. Merger and mass ejection of neutron star binaries. *Annu. Rev. Nucl. Part. Sci.*, 69(1):41–64, 2019.
- [202] M. Shibata, K. Kiuchi, and Y.-i. Sekiguchi. General relativistic viscous hydrodynamics of differentially rotating neutron stars. *Phys. Rev. D*, 95:083005, Apr. 2017.
- [203] M. Shibata, E. Zhou, K. Kiuchi, and S. Fujibayashi. Constraint on the maximum mass of neutron stars using gw170817 event. *Phys. Rev. D*, 100:023015, July 2019.
- [204] D. M. Siegel, R. Ciolfi, A. I. Harte, and L. Rezzolla. Magnetorotational instability in relativistic hypermassive neutron stars. *Phys. Rev. D*, 87(12):121302, 2013.
- [205] J. Smagorinsky. General Circulation Experiments with the Primitive Equations. Monthly Weather Review, 91(3):99, Jan. 1963.
- [206] J. M. Stewart. On transient relativistic thermodynamics and kinetic theory. Proc. R. Soc. Lond. A, 357(1688):59–75, Oct. 1977.
- [207] N. Straumann. General Relativity. Springer International Edition, 2004.
- [208] J. L. Synge. Relativity: The General Theory. North-Holland, Amsterdam, 1960.
- [209] A. Taub. General Relativistic Variational Principle for Perfect Fluids. *Phys. Rev.*, 94:1468–1470, 1954.
- [210] CompOSE Core Team. CompOSE. https://compose.obspm.fr/home.
- [211] K. S. Thorne. Multipole Expansions of Gravitational Radiation. *Rev. Mod. Phys.*, 52:299–339, 1980.
- [212] K. S. Thorne. Relativistic radiative transfer: moment formalisms. Mon. Not. R. Astron. Soc., 194(2):439–473, 02 1981.

- [213] K. S. Thorne and R. D. Blandford. *Modern Classical Physics*. Princeton University Press, 2017.
- [214] R. C. Tolman. Thermodynamics and relativity. Science, 77(1995):291–298, 1933.
- [215] R. C. Tolman. Thermodynamics and relativity. ii. Science, 77(1996):313–317, 1933.
- [216] G. M. Vasil, D. Lecoanet, B. P. Brown, T. S. Wood, and E. G. Zweibel. Energy conservation and gravity waves in sound-proof treatments of stellar interiors. II. lagrangian constrained analysis. *Astrophys. J.*, 773(2):169, Aug. 2013.
- [217] E. Velikhov. Stability of an ideally conducting liquid flowing between cylinders rotating in a magnetic field. *J. Exp. Theor. Phys.*, 36(9):995–998, 1959.
- [218] I. Vidaña. Hyperons: the strange ingredients of the nuclear equation of state. *Proc. Roy. Soc. Lond. A*, 474:0145, 2018.
- [219] D. Viganò, R. Aguilera-Miret, and C. Palenzuela. Extension of the subgrid-scale gradient model for compressible magnetohydrodynamics turbulent instabilities. *Phys. Fluids*, 31(10):105102, Oct. 2019.
- [220] D. Viganò, R. Aguilera-Miret, F. Carrasco, B. Miñano, and C. Palenzuela. General relativistic mhd large eddy simulations with gradient subgrid-scale model. *Phys. Rev. D*, 101:123019, June 2020.
- [221] A. G. Walker. Relative coordinates. Proc. R. Soc. Edinb. A, 52:345–353, 1933.
- [222] L. R. Weih, H. Olivares, and L. Rezzolla. Two-moment scheme for generalrelativistic radiation hydrodynamics: a systematic description and new applications. *Mon. Not. R. Astron. Soc.*, 495(2):2285–2304, 2020.
- [223] E. Weinan. *Principles of Multiscale Modeling*. Cambridge University Press, 2011.
- [224] S. Weinberg. *Gravitation and Cosmology: Principles and Applications of the General Theory of Relativity.* Wiley, 1972.
- [225] S. Weinberg. The Quantum Theory of Fields, volume 1, Foundations. Cambridge University Press, 1995.
- [226] C. M. Will. The Confrontation between General Relativity and Experiment. *Living Rev. Rel.*, 17:4, 2014.
- [227] F. Zappa, S. Bernuzzi, D. Radice, and A. Perego. Binary neutron star merger simulations with neutrino transport and turbulent viscosity: impact of different schemes and grid resolution. *Mon. Not. R. Astron. Soc.*, 520(1), Jan. 2023.
- [228] O. Zier and V. Springel. Simulating the magnetorotational instability on a moving mesh with the shearing box approximation. *Mon. Not. R. Astron. Soc.*, 517(2): 2639–2658, 2022.

[229] J. Zrake and A. I. MacFadyen. Magnetic energy production by turbulence in binary neutron star mergers. *Astrophys. J.*, 769(2):L29, May 2013.